

HEARING ON THE REPORT OF THE CHIEF JUSTICE OF GIBRALTAR

REFERRAL UNDER SECTION 4 OF THE JUDICIAL COMMITTEE ACT 1833

before

Lord Phillips
Lord Hope
Lord Rodger
Lady Hale
Lord Brown
Lord Judge
Lord Clarke

ADVICE DELIVERED

ON

12 November 2009

Heard on 15,16, 17, and 18 June 2009

Chief Justice of Gibraltar
Michael Beloff QC
Paul Stanley
(Instructed by Charles
Gomez & Co and Carter
Ruck)

Governor of Gibraltar Timothy Otty QC (Instructed by Clifford Chance LLP)

Government of Gibraltar
James Eadie QC
(Instructed by R J M
Garcia)

LORD PHILLIPS:

1. The task of the Committee is to advise Her Majesty whether The Hon. Mr Justice Schofield, Chief Justice of Gibraltar, should be removed from office by reason of inability to discharge the functions of his office or for misbehaviour. The independence of the judiciary requires that a judge should never be removed without good cause and that the question of removal be determined by an appropriate independent and impartial tribunal. This principle applies with particular force where the judge in question is a Chief Justice. In this case the latter requirement has been abundantly satisfied both by the composition of the Tribunal that conducted the initial enquiry into the relevant facts and by the composition of this Committee. This is the advice of the majority of the Committee, namely, Lord Phillips, Lord Brown, Lord Judge and Lord Clarke.

Security of tenure of judicial office under the Constitution

2. Gibraltar has two senior judges, the Chief Justice and a second Puisne Judge. The Gibraltar Constitution Order 2006 ("the 2006 Order") provides:

"Tenure of office of judges

- 64 (1) Subject to the provisions of this section, a person holding the office of Chief Justice or of Puisne Judge shall vacate that office when he attains the age of 67...
- (2) The Chief Justice ... may be removed from office only for inability to discharge the functions of his office (whether arising from infirmity of body or mind or any other cause) or for misbehaviour...
- (3) The Chief Justice... shall be removed from office by the Governor if the question of the removal of that judge from office has, at the request of the Governor made in pursuance of subsection (4),been referred by Her Majesty to the Judicial Committee

of Her Majesty's Privy Council under section 4 of the Judicial Committee Act 1833 or any other enactment enabling Her Majesty in that behalf, and the Judicial Committee has advised Her Majesty that the judge ought to be removed from office for inability as aforesaid or misbehaviour.

- (4) If the Governor considers that the question of removing the Chief Justice...from office for inability as aforesaid or for misbehaviour ought to be investigated, then
 - (a) the Governor shall appoint a tribunal...;
 - (b) the tribunal shall inquire into the matter and report on the facts thereof to the Governor and advise the Governor whether he should request that the question of the removal of that judge should be referred by Her Majesty to the Judicial Committee [of Her Majesty's Privy Council]; and
 - (c) if the tribunal so advises, the Governor shall request that the question should be referred accordingly.

. .

(6) If the question of removing the Chief Justice...has been referred to a tribunal under subsection (4), the Governor may suspend him from performing the functions of his office...

. . .

- (8) The powers of the Governor under this section shall be exercised by him in accordance with the advice of the Judicial Service Commission."
- 3. These provisions, other than those in subsection (8), reproduced provisions to the same effect in section 60 of the Gibraltar Constitution Order 1969 ("the 1969 Order").

- 4. On 17 April 2007 all the Queen's Counsel in Gibraltar, with the exception of the Speaker in the House of Assembly, were among the signatories ("the Signatories") to a Memorandum to the Governor that expressed on behalf of themselves and their respective firms "their deep concern at a state of affairs which has developed seriously affecting the administration of justice and the reputational image of Gibraltar". They stated that they had lost confidence in the ability of the Chief Justice to discharge the functions of his office. At the request of the Governor they submitted a Supplementary Memorandum ("the Second Memorandum") dated 21 May which set out in detail the reasons for their dissatisfaction with the Chief Justice. One firm, Hassans & Co, wrote dissociating themselves from those parts of the Second Memorandum that related to matters in respect of which they had professionally represented the Chief Justice or his wife.
- 5. Copies of the two Memoranda were supplied to the Chief Justice and his solicitors sent a preliminary response to the Governor. All of these documents were considered by the Judicial Service Commission, which advised the Governor to appoint a Tribunal under section 64(4) of the 2006 Order. The Governor did so on 14 September 2007 and on 17 September he suspended the Chief Justice, pursuant to section 64(6) of the Order.
- 6. One member of the Tribunal recused himself and was replaced, whereupon the Tribunal consisted of the Rt Hon Lord Cullen of Whitekirk, the Rt Hon Sir Peter Gibson and the Rt Hon Sir Jonathan Parker.
- 7. There were two directions hearings. These were attended by representatives of the Chief Justice and of the Government of Gibraltar, the Governor of Gibraltar and the signatories to the Memoranda ("the Signatories") as interested parties. At the second hearing all agreed

"that in making its report to the Governor the Tribunal should express its own view on whether the facts found by the Tribunal showed any inability on the part of the Chief Justice to discharge the functions of his office or any misbehaviour on his part and, if so, to reach its own conclusion as to whether such inability or misbehaviour should warrant his removal from office."

The parties agreed a Statement of Issues that set out 23 episodes between 1999 and 2007 where the conduct of the Chief Justice had been the subject of criticism.

8. The Tribunal sat from 7 to 28 July to hear evidence of fact. 18 witnesses gave oral evidence. The Tribunal also had regard to written statements by a further 11 witnesses who

did not give oral evidence. These included Mrs Schofield, the wife of the Chief Justice, who had been expected to give evidence. She declined to give evidence on the ground that she did not consider satisfactory the terms on which funding had been provided to her to enable her to obtain legal advice in relation to certain matters relating to her appearance before the Tribunal. The Tribunal indicated that it did not consider that this justified her failure to appear to give evidence. They were right to do so. Had Mrs Schofield wished to give evidence there was no reason why she should not have done so.

The Report of the Tribunal

- 9. The Report of the Tribunal to the Governor is dated 12 November 2008. It was provided to the Chief Justice on 3 December and published on the following day. The Report makes findings of primary fact in relation to each of the 23 episodes identified in the Statement of Issues. In relation to all but one of these it criticises the conduct of the Chief Justice. Some of these criticisms are based on inferences drawn from the primary facts as to the knowledge, perception or motive of the Chief Justice. In making findings of fact the Tribunal has applied the civil standard of proof of balance of probability. The Tribunal commented that in a number of instances the conduct of the Chief Justice amounted to impropriety, giving four examples of this, but added that no single instance of misbehaviour showed that the Chief Justice was unfit to hold office.
- 10. The conclusions of the Tribunal were set out in the following final paragraphs of its Report:
 - "7.36 The conduct of the Chief Justice which we have summarised earlier in this chapter directly affected the way in which he discharged part of the responsibilities of his office, such as his relations with the Governor and the Government (covering such matters as funding, proposed legislation and appointments), and his relations with the representatives of the Bar. The conduct stemmed, in our view, from a number of characteristics of his personality and attitude, as follows.
 - 7.37 First, the Chief Justice did not seem to be alive to the boundary between what was and what was not proper for someone in his position to do or say. He repeatedly showed a lack of judgment in this respect. He also showed the lack of a sense of proportion, and tended to over-react to perceived slights. He did not observe appropriate restraint or respect for accuracy in his public pronouncements.
 - 7.38 Secondly, he showed a pre-occupation, bordering on an obsession, with judicial independence. He claimed that it was under threat when this was not the case. This led to his responding in an improper or excessive manner to executive

action of which he disapproved. Allied to this was his preoccupation with the status of his office and his continuance in office. This showed itself in a number of ways ranging from petty discourtesy to the Chief Minister to the unfounded accusation that the Government had long sought to have him removed from office.

7.39 Fourthly, he showed himself to be unable to restrain himself from supporting his wife in her attack of the members of the Bar Council or her libel action against its Chairman. Although he affected a lack of interest in her communications with the Bar Council he was more than content that his silence should be interpreted as support for her communications. He knew that it would have been improper for him to have sent them. He was unable to grasp that his association with them would have been seen by a fair minded and well informed observer as improper.

7.40 Fifthly, the perceptions arising from the conduct of the Chief Justice inevitably rendered it impossible for the Chief Justice to sit in a significant number of cases.

7.41 At the same time the Chief Justice showed himself to be indifferent as to the effect, or the perceived effect, of his conduct on his relations with the Government and the Bar, the standing of the judiciary and the administration of justice in Gibraltar. This would inevitably affect the reputation of his office. In the particular context of Gibraltar, which is a small jurisdiction as a number of witnesses reminded us, the significance of public perception is inevitably magnified. In his witness statement Mr Neish observed: 'The public in Gibraltar is much closer to public figures than in the case of say, England. Their scrutiny is more intense and their actions more directly felt'. While it is true that public opinion in Gibraltar is not unanimous in its disapproval of the conduct of the Chief Justice we are in no doubt that its effect has been to polarise public opinion in a way which is damaging to the reputation of the office, and hence to the interests of good governance of Gibraltar. By his conduct he has antagonised a large number of those who practise before him.

7.42 In these circumstances we conclude that the Chief Justice is unable to discharge the functions of his office. We are satisfied that this inability warrants the removal of the Chief Justice from office.

Our advice

7.43 Accordingly, in terms of section 64(4)(c) we advise the Governor that he should request that the question of the

removal of the Chief Justice should be referred by Her Majesty to the Judicial Committee of the Privy Council."

The parties

11. The Chief Justice has been represented before the Committee by Mr Michael Beloff QC and Mr Paul Stanley, neither of whom appeared for him before the Tribunal. The other parties have been the Government of Gibraltar, represented by Mr James Eadie QC, the Governor of Gibraltar, represented by Mr Timothy Otty QC, and the Signatories, who submitted a written case but were not represented at the hearing. These parties have supported the conclusions of the Tribunal.

The significance of the Tribunal's Report

12. The Constitution Order does not expressly provide for the Tribunal to express a view as to whether a judge's conduct justifies removal. All parties sensibly agreed, nonetheless, that it should do so for the Tribunal's view, if not expressed, was likely to be implicit in its advice to the Governor. On 17 May 2008 an Application Notice filed on behalf of the Chief Justice submitted:

"The Tribunal has been entrusted with a duty to determine and report on the facts and there will be no rehearing in relation to the facts if the matter progresses to the Privy Council. The Tribunal therefore has a crucial determinative role in the removal of a Chief Justice."

Before the Committee Mr Beloff QC submitted that the Committee was exercising its own original jurisdiction under section 64 of the Constitution and section 4 of the Judicial Committee Act 1833 in advising Her Majesty whether the Chief Justice should be removed and could, if it chose, conduct its own hearing of the evidence. He accepted, however, that as master of its own procedure, the Committee was unlikely to do so, unless it accepted his submission that the Tribunal had erred in its approach to fact-finding by applying the civil rather than the criminal standard of proof.

13. Section 64(4) of the 2006 Order provides for an inquiry into the facts by a Tribunal of three present or past holders of high judicial office. This Committee is not bound by the findings of fact of that Tribunal but can properly act upon them. In the present case the primary findings of fact are not challenged. In some cases secondary findings, or inferences from the primary findings, have been challenged. In those cases the Committee has reviewed the secondary findings in the light of the primary findings upon which they have been based.

14. In the event it is not surprising that the hearing before us was treated by the parties as if it were an appeal against the conclusions of the Tribunal. While the jurisdiction of the Committee is indeed original, it is convenient to follow the course taken by the parties and to consider the respects in which the conclusions of the Tribunal have been challenged by Mr Beloff, taking the issues of fact in chronological order. For convenience we annexe the Report of the Tribunal to this advice. This advice will not refer to all the evidence that is set out in detail in the Tribunal's Report, and the two should be read together. Before turning to the facts there are, however, a number of preliminary matters to consider.

Standard of proof

- 15. The Tribunal applied the civil standard of proof when resolving issues of fact. It gave its reasons for applying this standard in the Fifth Schedule to its Report. It held that the proceedings before the Tribunal were not to be equated with disciplinary proceedings where the criminal burden of proof was applicable. They were concerned with the public interest that called, on the one hand, for the protection of a judge against unfounded or illegitimate interference with his tenure of office and, on the other, for his removal should he show unfitness for office that need not necessarily be based on misbehaviour. In these circumstances it was appropriate to apply the civil standard of proof.
- 16. We have not found the issue of standard of proof an easy one. Judicial independence is of cardinal importance. There is a case for saying that a judge should not be removed for misconduct unless this is proved to the criminal standard – see the comments of Lord Mustill at paragraphs 81 to 83 of the Report of the Tribunal of 14 December 2007 into the question of removing from office the Chief Justice of Trinidad and Tobago pursuant to provisions of the Constitution of the Republic of Trinidad and Tobago that are similar to section 64 of the 2006 Order. The present proceedings are not, however, concerned with disciplining the Chief Justice for misconduct but with deciding whether he is fit to perform his office and the Committee has decided, as did the Tribunal, that issues of fact that bear on that question should be determined according to the civil standard of proof. That said, this is not a case where our advice is going to turn on the standard of proof applied to fact finding. Most of the primary facts are a matter of record and not disputed. The challenge made by Mr Beloff is to the inferences drawn by the Tribunal from the primary facts and to the Tribunal's conclusion that the Chief Justice's conduct amounted to impropriety and that it justified his removal on the ground of his inability to perform the functions of his office. The latter is not a question of fact subject to a standard of proof but a matter for judicial assessment.
- 17. The Tribunal applied the civil standard of proof according to what it described as the "flexible approach" that "the more improbable the event, the stronger must be the evidence that it did occur" see *In re D* (Secretary of State for Northern Ireland intervening) [2008] 1 WLR 1499, per Lord Carswell at paras 23 and 25. That approach is no more than the rational way of determining facts on a balance of probabilities. The more improbable the event the greater the weight of the evidence that must exist before the scales tilt in favour of a finding

that the event occurred. There is one area in relation to which we consider that this approach is particularly important. In their written case the signatories have drawn attention to 21 instances (after allowing for the fact that two appear twice in their list) where the Tribunal made findings adverse to the Chief Justice's credibility. The signatories rely on these as demonstrating "mendacity" on his part. A firm basis is required for any finding that the Chief Justice was deliberately untruthful when giving evidence, albeit that in assessing any individual aspect of the Chief Justice's evidence the Tribunal could properly have regard to the content of his evidence overall and the manner in which he gave that evidence.

Procedural fairness

- 18. Mr Beloff submitted that the procedure before the Tribunal had not been fair in as much as the case against the Chief Justice had not been clearly spelt out at the start of the proceedings and some of the adverse findings made by the Tribunal were in respect of matters that formed no part of the Statement of Issues. He also submitted that there was a lack of clarity in the reasons given by the Tribunal for concluding that the Chief Justice's behaviour showed that he was unfit to remain in office. In general we reject this submission. The Tribunal took great care to ensure that the Chief Justice had proper notice of the allegations that were made against him. As Mr Eadie QC for the Government pointed out in his printed case:
 - i) The procedure was inquisitorial, not adversarial;
 - ii) The Terms of Reference referred to the Memorandum and the Supplementary Memorandum of the signatories that contained substantial and specific allegations against him;
 - iii) A list of issues was prepared and amended that identified the case against the Chief Justice in some detail:
 - iv) There were opening statements by all concerned, including Counsel to the Tribunal:
 - v) Witness statements were prepared and circulated before witnesses gave evidence.
- 19. Mr Beloff referred to two specific episodes in relation to which findings adverse to the Chief Justice were made that had not been foreshadowed in the Statement of Issues. We will deal with Mr Beloff's submissions when considering the episodes in question.

- 20. The Tribunal recorded at the outset (1.21) that no criticism was made of the Chief Justice's ability as a lawyer to decide the cases before him. The criticisms levelled against the Chief Justice related, with one important exception, to his behaviour outside court and to the manner in which he discharged other responsibilities of the office of Chief Justice of Gibraltar. What did that office involve?
- 21. Gibraltar is a self-governing British overseas territory of a modest size, covering a little less than 7 square kilometres. It has a population of approximately 29,000. It has two major political parties, the Gibraltar Social Democrats, headed by Mr Peter Caruana QC, which was at all material times in power, and the Gibraltar Socialist Labour Party, headed by Mr Joe Bossano. The unicameral Parliament consists of seventeen elected members.
- 22. The permanent professional judiciary of Gibraltar is very small. Apart from the Chief Justice and one Puisne Judge, who is the only other member of the Supreme Court, there is one Stipendiary Magistrate. The court is staffed by civil servants, who are headed by the Registrar. There is a Deputy Registrar and a small body of court staff, including three bailiffs.
- 23. The fused legal profession, as at January 2008, consisted of 26 firms comprising 166 lawyers. Firms that undertook a substantial amount of litigation included Hassans (54 lawyers), Triay & Triay (20 lawyers), Triay Stagnetto Neish (12 lawyers) and Attias & Levy (9 lawyers). The Gibraltar Bar is regulated by the Gibraltar Bar Council.
- 24. There was when the Chief Justice was appointed no formal position of head of the judiciary of Gibraltar, but we consider that, as the senior resident judge, the Chief Justice could properly regard himself as occupying that position. He was responsible for the day to day administration of justice by the local judiciary. He presided over the formal ceremony of the Opening of the Legal Year and spoke at it. He was responsible for negotiating with the Gibraltar Government the provision of an adequate infrastructure for the administration of justice.
- 25. In a jurisdiction as small as Gibraltar there is bound to be interrelation between those in the different arms of State and, indeed, in every aspect of life. In his first witness statement the Chief Justice commented:

"By way of context, it is important to remember how closely connected are lawyers and members of the government. The Chief Minister is the son in law of J E Triay QC whose cousin is Louis Triay. Freddie Vasquez is a cousin of the Chief Minister as is Robert Vasquez. The Chief Minister was a partner in Triay and Triay until he went into politics. Guy Stagnetto is also a cousin of the Chief Minister and James Neish is a close friend of the Chief Minister. Daniel Feetham, Minister for Justice in this Government was until recently a partner in Hassans. His brother and sister in law are still partners in Hassans. Keith Azopardi, a partner in Attias and Levy was Deputy Chief Minister to Peter Caruana until his resignation and was a member of the select committee on the draft constitution."

26. In his written representation Mr Neish QC, the Chairman of the Bar, remarked:

"The seriousness of the matters complained of must be judged from the standpoint of Gibraltar and not from that of a larger jurisdiction. In, say, London with its large number of judges the conduct of individual judges would not have the same impact on the judiciary, or on the operation of the principle of the separation of powers or on the justice system generally as would the conduct of a Chief Justice in a two judge jurisdiction like Gibraltar. Office holders in Gibraltar have to be particularly sensitive to the need to maintain the respect and confidence of the public, which are as necessary, if not more so than institutional safeguards, for the proper discharge of their functions. The public in Gibraltar is much closer to public figures than in the case of say, England. Their scrutiny is more intense and their actions more directly felt."

Similar comments were made by a number of other witnesses.

27. Gibraltar has a particularly lively press, anxious for copy and, in common with the media in other jurisdictions, eager to identify any actual or supposed conflict between the judiciary and the executive. As the senior first instance judge the Chief Justice would normally preside over the trial of any dispute, whether of public or private law, that involved the Government. While it is entirely appropriate for a Chief Justice to take a firm stand in any matter that affects the independence of the judiciary or the due administration of justice, to the extent of making robust public pronouncement on these matters, both as de facto head of the judiciary and as the judge who would be dealing with actions involving the Government, in this small jurisdiction, the Chief Justice needed to exercise particular sensitivity and discretion in his dealings with Government.

28. Mr Beloff did not challenge submissions made by other parties that the Bangalore Principles of Judicial Conduct 2002 and the Guide to Judicial Conduct, published by the Judges' Council of England and Wales in October 2004 provided guidance as to the standard of conduct to be expected of a judge. The following provisions of the Bangalore Principles are of particular relevance.

"IMPARTIALITY

Principle:

Impartiality is essential to the proper discharge of the judicial office. It applies not only to the decision itself but also to the process by which the decision is made.

Application:

- 2.1 A judge shall perform his or her judicial duties without favour, bias or prejudice.
- 2.2 A judge shall ensure that his or her conduct, both in and out of court, maintains and enhances the confidence of the public, the legal profession and litigants in the impartiality of the judge and of the judiciary.
- 2.3 A judge shall, so far as is reasonable, so conduct himself or herself as to minimise the occasions on which it will be necessary for the judge to be disqualified from hearing or deciding cases.

. .

PROPRIETY

Principle:

Propriety, and the appearance of propriety, are essential to the performance of all of the activities of a judge.

Application:

- 4.1 A judge shall avoid impropriety and the appearance of impropriety in all of the judge's activities.
- 4.2 As a subject of constant public scrutiny, a judge must accept personal restrictions that might be viewed as burdensome by the ordinary citizen and should do so freely and willingly. In particular, a judge shall conduct himself or herself in a way that is consistent with the dignity of the judicial office.

4.3 A judge shall, in his or her personal relations with individual members of the legal profession who practise regularly in the judge's court, avoid situations which might reasonably give rise to the suspicion or appearance of favouritism or partiality.

. . .

4.6 A judge, like any other citizen, is entitled to freedom of expression, belief, association and assembly, but in exercising such rights, a judge shall always conduct himself or herself in such a matter as to preserve the dignity of the judicial office and the impartiality and independence of the judiciary.

. . .

- 4.8 A judge shall not allow the judge's family, social or other relationships improperly to influence the judge's judicial conduct and judgment as a judge."
- 29. To like effect, the Guide to Judicial Conduct advises:

"JUDICIAL INDEPENDENCE

2.1 . . . The judiciary, whether viewed as an entity or by its individual membership, is and must be seen to be, independent of the legislative and executive arms of government. The relationship between the judiciary and the other arms should be one of mutual respect, each recognising the proper role of the others.

. .

IMPARTIALITY

- 3.1 A judge should strive to ensure that his or her conduct, both in and out of court, maintains and enhances the confidence of the public, the legal profession and litigants, in the impartiality or the judge and of the judiciary.
- 3.2 Because the judge's primary task and responsibility is to discharge the duties of office, it follows that a judge should, so far as is reasonable, avoid extra-judicial activities that are likely to cause the judge to have to refrain from sitting on a case because of a reasonable apprehension of bias or because of a conflict of interest that would arise from the activity."

Later the Guide deals with "Personal relationships and perceived bias" and warns that "Personal friendship with, or personal animosity towards, a party is ...a compelling reason for disqualification" and warns under the heading "Activities outside the court" that judges should exercise their freedom to talk to the media "with the greatest circumspection".

30. A summary of the standard of behaviour to be expected from a judge was given by Gonthier J when delivering the judgment of the Supreme Court of Canada in *Therrien v Canada (Ministry of Justice) and another* [2001] 2 SCR 3:

"The public will therefore demand virtually irreproachable conduct from anyone performing a judicial function. It will at least demand that they give the appearance of that kind of conduct. They must be and must give the appearance of being an example of impartiality, independence and integrity. What is demanded of them is something far above what is demanded of their fellow citizens."

- 31. While the highest standards are expected of a judge, failure to meet those standards will not of itself be enough to justify removal of a judge. So important is judicial independence that removal of a judge can only be justified where the shortcomings of the judge are so serious as to destroy confidence in the judge's ability properly to perform the judicial function. As Gonthier J put it at paragraph 147 of the same case:
 - "...before making a recommendation that a judge be removed, the question to be asked is whether the conduct for which he or she is blamed is so manifestly and totally contrary to the impartiality, integrity and independence of the judiciary that the confidence of individuals appearing before the judge, or of the public in its justice system, would be undermined, rendering the judge incapable of performing the duties of his office."

We will revert to this topic when considering the statutory criteria for removal of a judge under section 64 of the 2006 Order in the concluding section of this advice. There are, however, two matters that call for further comment at this stage.

Freedom of expression

32. Gibraltar is subject to the European Convention on Human Rights. Mr Beloff submitted that the Tribunal had wrongly held the Chief Justice at fault for public statements made by him or his wife that were no more than the exercise by them of the right of freedom of expression that is recognised by article 10 of the Convention. He argued that this freedom of expression could only lawfully be restricted by clear provisions of law that satisfied the requirements of article 10(2) and that none such had been demonstrated. We do not accept this argument. So far as any judge is concerned, the proper exercise of judicial office

necessarily circumscribes the freedom of expression open to those who do not have to ensure that they are seen to be acting without fear or favour, affection or ill will. There is an abundance of jurisprudence, to some of which we have already referred, that defines the requirements of judicial office in this respect. The position of the wife of the Chief Justice calls for further comment.

Statements by the wife of the Chief Justice

- 33. The wife of the Chief Justice made a large number of public statements or allegations, including allegations in a libel action that she started against the Chairman of the Bar. The Tribunal found that the majority of these were made with the knowledge and approval of the Chief Justice, sometimes rejecting evidence that he gave to the contrary on the ground that it was not credible. These allegations included attacks on the Bar Council which would have been improper if made by the Chief Justice himself. The Tribunal held that the Chief Justice's implicit support of those attacks constituted misbehaviour. The Tribunal further found that on a number of occasions the Chief Justice should have publicly dissociated himself from his wife's comments and criticised him for his failure to do so. Mr Beloff submitted that these findings on the part of the Tribunal were unsound. He submitted that there was no basis for doubting the Chief Justice when he said that he was not aware of his wife's activities. The Chief Justice's wife was and should have been treated as being independent of the Chief Justice. He could not properly be associated with her actions and was under no obligation to dissociate himself from them.
- 34. The Chief Justice married his wife, Anne Kariuki when he was serving as an acting judge of the High Court of Kenya. She was a practising member of the Kenyan Bar. They have two children. It appears plainly from the evidence that the Chief Justice and his wife were and are devoted to one another. It is equally plain that Mrs Schofield has a strong, indeed headstrong, personality. When she believed that her husband was being attacked she made her feelings plain in coming to his defence. On a number of occasions she said in her witness statements that she was acting independently of her husband and that he was unaware of her actions. In general the Tribunal rejected such evidence as incredible.
- 35. When reviewing the Tribunal's conclusions we have borne the following matters in mind. Neither the Chief Justice nor his wife suggested that they had a deliberate policy of not discussing certain matters. They did not seek to build a "Chinese wall" in their home. In these circumstances the inevitable inference was that they would discuss matters that were of serious concern to either, or both. Indeed, on occasion the Chief Justice gave evidence that such discussions had occurred. It might have been the case that Mrs Schofield had a deliberate policy of concealing from her husband actions that she proposed to take of which she knew her husband would disapprove, but neither of them suggested that this was so. Such a policy would have been almost certain to result in marital discord, but there is no hint of this in the evidence. Nor did the Chief Justice suggest that he ever sought to dissuade his wife from action that she proposed to take.

- 36. We will review the findings made by the Tribunal that associated the Chief Justice with the public statements made by his wife. What is clear on the evidence is that others associated the Chief Justice with those statements. In so far as it was reasonable for them to do so, this was likely to impact on the standing of the Chief Justice and on his performance of his judicial duties, as indeed it did when applications were made that he should recuse himself from certain cases. Dissociating himself from his wife's actions and statements would have been an appropriate step to take in an attempt to avoid the implication that they had his approval, but the question would have remained, and does remain, as to how it was that his wife came to make public statements that were bound professionally to embarrass her husband if he was opposed to her doing so.
- 37. When giving evidence on 23 July 2008 the Chief Justice was asked by Lord Cullen whether his wife considered that his prosecution for the MOT offence was politically motivated. He did not answer that question because he said that he might be breaching confidences between himself and his wife. The failure of Mrs Schofield to give evidence left a number of questions unanswered. It may be that the answers to those questions would not have assisted her husband's case.

The conclusions of the Tribunal

- 38. The Tribunal found that the Chief Justice's public behaviour "fell far short of what befitted the dignity of his office" (7.7). That office required him to have and to be seen to have "the detached, unbiased, unprejudiced, impartial, open-minded and even-handed approach which is the hallmark of a judge" (7.9). The Chief Justice failed so to conduct himself in his relationship with the Governor, with the Government and with the legal profession. He addressed what he claimed to be threats to judicial independence in a manner that was "confrontational", "improper...inappropriate and disproportionate" (7.8 to 7.11). He showed "hostility" towards the Government, the Chief Minister and members of the legal profession including making "serious and unfounded allegations against the Government" (7.12 to 7.15, 7.35 (iv)). He associated himself with, or failed to dissociate himself from, conduct by his wife that he could not himself have committed without impropriety [7.16 to 7.27].
- 39. Mr Beloff challenged the conclusion of the Tribunal that the conduct of the Chief Justice in the individual episodes justified these descriptions. He also challenged the Tribunal's conclusion that the conduct of the Chief Justice, when taken as a whole, amounted to inability to discharge the functions of his office or misbehaviour within section 64(2) of the 2006 Order justifying his removal.
- 40. We propose first to examine the individual episodes in the light of the attacks made by Mr Beloff on the Tribunal's findings in relation to these. In doing so it will consider, in

particular, the findings relied upon by the signatories as demonstrating mendacity on the part of the Chief Justice. In the light of its conclusions it will then consider Mr Beloff's challenge to the Tribunal's finding that the Chief Justice has shown himself unable to perform the functions of his office.

- 41. The episodes considered by the Tribunal stretch back as far as 1999. If any of these episodes constituted serious impropriety on the part of the Chief Justice they might be capable, for this reason, of demonstrating his unfitness to remain in office despite their antiquity. The Tribunal included two of the early episodes in its four examples of misbehaviour, while not suggesting that they fell into this category (7.35). The early episodes have, however, this additional significance:
 - i) An important part of the case against the Chief Justice is that he publicly accused the government, and in particular the Chief Minister, of improperly attempting to drive him from office, when that accusation was unfounded and should not have been made. By the end of the hearing before the Tribunal the Chief Justice accepted that the accusation was unfounded, but contended that he had had reasonable grounds for believing in its justification. He himself relied upon the episodes stretching back to 1999 in support of this contention. Insofar as the reasonableness of the Chief Justice's belief that there were attempts improperly to oust him from office is relevant to the assessment of his behaviour it is necessary to look at the earlier episodes.
 - ii) The Tribunal's conclusion that the Chief Justice was unfit to remain in office was based in part on the "characteristics of his personality and attitude" manifested by his conduct (7.36). Insofar as these characteristics are significant, the early episodes have relevance to their assessment.

The address by the Chief Justice at the Opening of the Legal Year in October 1999.

42. This is dealt with in paragraphs 2.1 to 2.24 of the Tribunal's Report. The Tribunal criticised a passage at the end of the Chief Justice's address. He referred to attending the Commonwealth Law Conference in Malaysia and there discussing the draft Latimer House Guidelines. He drew attention to principles set out in that draft including:

"Sufficient funding to enable the judiciary to perfume its functions to the highest standards should be provided' and

'The administration of monies allocated to the judiciary should be under the control of the judiciary'.

He then continued:

'The judiciary has encountered one or two instances in the past year where the denial or delay of the release of funds by the Government has had the potential to affect adversely the administration of justice. The matter is one of practical importance, but there is also a fundamental principle involved. The Chief Justices of the Commonwealth were all agreed that those who control the judiciary's purse strings exercise enormous influence and have the capacity to undermine the judiciary's independence. That is why the Latimer House Guidelines are expressed as they are. A system should be in place which on the one hand enables the judiciary to administer justice properly whilst on the other subjects the judiciary's budget to proper controls, but which enables the judiciary to function independently and without improper restraint. It is a matter which I shall be addressing in the coming year."

- 43. In making this statement the Chief Justice was subsequently to explain that he had two instances in mind. One related to the provision of funding for a proposal made by the Chief Justice that a number of members of the Bar should be permitted to sit as part time Stipendiary Magistrates for a period of three weeks in the coming year. The other related to the provision of funding to enable the Chief Justice to travel to Warwick to attend a Judicial Studies Board Conference on Lord Woolf's reforms to civil procedure at the beginning of February 2000. The Chief Justice had told the Government that he considered that it was "vital" that he should attend this.
- 44. The Tribunal expressed a number of "serious concerns" about this part of the Chief Justice's address. The first was that it was "seriously inaccurate and misleading". As to the funding of the Stipendiary Magistrates' scheme, the Government had agreed to fund the scheme. The reason why the scheme was not put into operation was not lack of funding but disagreement between the Government and the Chief Justice as to how the candidates should be selected. The Committee agrees that this incident could not properly be relied on by the Chief Justice in support of his comment that there had been "one or two instances."
- 45. As to the failure to fund the Chief Justice's travel to the Conference, the Tribunal commented that it was questionable whether it was "vital" for him to attend this, that he did not disclose the fact that funding had subsequently been provided by the UK Government to enable him to attend the Conference and that the request had been for funding in excess of the judicial budget which the Government had met in full.
- 46. It seems to us that there was probably more to the refusal of the Government to provide funding for the Chief Justice's travel than met the eye. The facts are as follows. On

10 November Mrs Dawson, the Registrar of the Supreme Court, made a request to the Chief Secretary for funding to enable the Chief Justice to attend the Conference. She sent a chaser on the 20 November, which crossed with a request from the Assistant to the Chief Secretary for further details of the Conference. Mrs Dawson replied on 23 November:

"Lord Woolf (Master of the Rolls) has put forward proposals for a complete overhaul of the system of procedures in the Civil Courts and this could impact heavily on our Supreme Court. I understand that the changes are imminent.

The English Judicial Studies Board is holding a series of seminars to familiarise Judges with the proposed new procedures and the Chief Justice has managed to secure a place for himself on one such seminar.

He considers it vital that he attend "

On December 1 Mrs Dawson wrote stating that the Chief Justice was anxious to know whether approval had been given for the funding. The Assistant to the Chief Secretary replied:

"The Government does not consider that the subject matter of this conference is of sufficient value to Gibraltar at this stage. Fund cannot be approved for this purpose. Please not that funds under the Sub-head for conferences are now fully committed for the remainder of the financial year."

- 47. In a parliamentary debate on judicial independence on 20 November 2000 the Chief Minister stated that with hindsight it would have been a much better decision to have allowed the Chief Justice to go to the Conference. He explained that towards the end of the year requests for supplementary funding are turned down except in the case of "enormous necessity" and that is what happened on this occasion.
- 48. In paragraph 13 of his witness statement Mr Garcia, who took over the office of Chief Secretary in March 2007, gave the following explanation of the Government's refusal to provide funding for the Chief Justice's visit to Warwick.

"It transpired that indeed the Government had refused to provide the requested funds for the Chief Justice's attendance at the Warwick University course in 1998. This was because the Chief Justice had that year taken 66 working days leave of absence, at a time when his contractual annual leave entitlement was 30 days. Although there was an element of absence relating to official business (6 working days at seminars and conferences) and possibly some sick leave, the Government felt that there was a significant excess over the annual leave entitlement."

- 49. When he gave oral evidence, Mr Garcia was not asked about this statement but we consider it possible that it explains what, on the face of it, appears a surprising unwillingness to make available the small sum needed to cover the expenses of the Chief Justice's visit to Warwick. Mrs Dawson in her witness statement, while not doubting the Chief Justice's professional ability as a judge, was critical of various aspects of his behaviour out of court. In particular she criticised the Chief Justice for taking too much leave and attending too many conferences. She stated that she had a good relationship with Mr Montado, Mr Garcia's predecessor as Chief Secretary, and would often call or write to him to tell him things which the Chief Justice had done which she thought might potentially cause embarrassment. In oral evidence she agreed that she knew that there were people who were unhappy with the Chief Justice, including the Chief Minister. We consider it at least possible that the refusal to fund the Chief Justice's visit to Warwick was influenced by a perception that he had been spending too much of his time outside Gibraltar.
- 50. Whether or not this is correct, it was a serious rebuff to the Chief Justice, after he had stated that he considered it "vital" that he should go to the Warwick conference, to have his request for funding turned down on the ground that the Government did not consider that the conference was of sufficient value to Gibraltar. In the circumstances it would not have been surprising if the Chief Justice had made a passing reference to this in his speech at the opening of the legal year. While it did not fully justify the statement that he made about funding, we do not consider that the Tribunal was justified in describing as "seriously inaccurate and misleading" the statement that there had been "one or two instances in the past year where the denial or delay of the release of funds by the Government had had the potential effect to affect adversely the administration of justice". The wording implied that it had not, in fact, had that effect, and we do not see that the fact that the U.K. Government subsequently provided the funds for the visit to Warwick had, of itself, any bearing on the accuracy of the Chief Justice's statement. The Tribunal also described the Chief Justice's statement as "highly critical of the Government", said that it demonstrated "a complete lack of judgment in his choice of language" and that the "language, tone and manner of his remarks were inappropriate for the holder of his office". Once again we do not consider the severity of these criticisms to be justified.
- 51. The remarks were made in the context of an account of the draft Latimer House Guidelines on the Supremacy of Parliament and the Independence of the Judiciary. These included a principle that the administration of monies allocated to the judiciary for the running of the courts should be under the control of the judiciary. This somewhat ambitious aspiration did not ultimately form part of the Latimer House Guidelines. We consider, however, that it was legitimate for the Chief Justice to make reference to this in his address. It was also legitimate for him to remark, that the Chief Justices of the Commonwealth were all agreed that those who control the judiciary's purse strings exercise enormous influence and have the capacity to undermine the judiciary's independence.
- 52. What can be said about the Chief Justice's remarks is that they were injudicious in that they were unspecific and open to misinterpretation. This was particularly the case having regard to the juxtaposition of his reference to the one or two instances of denial or

delay of release of funds and the reference to undermining the judiciary's independence. He said that he did not anticipate the reaction that greeted his remarks. The Tribunal did not express any reservations about that assertion. They did, however, criticise the Chief Justice for doing nothing to assuage the furore that his remarks created. Furore it certainly was. On 15 October the *Gibraltar Chronicle* reported as the major news item on the front page that Dr Garcia, the leader of the Liberal Party, was calling for a public inquiry into "claims of potential or perceived interference by the Government with the judiciary" made by the Chief Justice. It reported that when asked to comment on the exact meaning of his comments, a spokesman had replied that if the Chief Justice had wished to elaborate, he would have done so in his speech. The Chief Justice confirmed that he instructed the response to be given in these terms. The *New People* published on the same day had a banner headline "Chief Justice warns of Government interference in Judiciary".

- 53. The Chief Justice said that he was surprised, albeit not horrified, at the manner in which his remarks had been distorted. The Tribunal criticised him for not taking steps to put the record straight. We consider that this criticism was merited. Having been responsible for an unjustified political attack on the Government the obvious course would have been to do his best to defuse the situation. The Government issued a press release contending that Dr Garcia had distorted the Chief Justice's remarks and identifying what it claimed to have been the only two occasions on which it had refused to provide the judiciary with funding, one being funds for social entertaining and the other "international conferences including one by the Chief Justice in Malaysia".
- 54. The Chief Justice said that he formed the view that the Government was "trivialising" his complaint and indulging in "spin" and that this called for a public response from himself. This was not a rational reaction. Nor was the press release that the Chief Justice issued on 18 October, his subsequent public exchanges with the Government, his consultation of Mr Causer in relation to constraints that he feared that the law might impose on disclosing details of the two incidents to which he had referred in his address, and his public and unwarranted challenge to the Government's statement in relation to the practice of consultation with the Chief Minister before making judicial appointments. This unhappy chapter demonstrated lack of judgment and inappropriate behaviour of a high order that brought the Chief Justice into unnecessary public confrontation with the Government that was of the Chief Justice's own making and that had no justification.
- 55. The Tribunal held that the Chief Justice was "determined to embarrass the Government publicly" (2.21) and that his concern about confidentiality and the Official Secrets Act was "an irrelevant diversion from answering the substance of the point against him, namely that he had not told the Government about his concerns privately, so that it could be seen whether they truly had the potential to affect adversely the administration of justice." (2.24) We do not consider that these findings of bad faith are justified. We see no ground for concluding that the Chief Justice's consultation of Mr Causer was a charade. It was, however, a further demonstration of bad judgment. There was no reason to think that the Government would object to the Chief Justice clarifying the concerns to which he had

referred and, had he had any doubt, he could simply have explained the position and asked them if they minded.

56. The conduct of the Chief Justice in fanning the flames caused by his comments at the 1999 opening of the legal year has relevance to the question of whether defects of character and personality have resulted in a current inability to perform the functions of his office. We do not consider, however, that the Tribunal should have treated this matter as an incident of misbehaviour that formed part of the justification for removing him from office ten years later.

The Maids Issue

- 57. This is dealt with in paragraphs 3.1 to 3.27 of the Tribunal's Report.
- 58. In 1996 and 1997 the Chief Justice was guilty of a number of breaches of the criminal law in relation to the employment of a Jamaican woman, Ms Jackie Williams as a domestic. The Tribunal criticised the Chief Justice (i) for these breaches of the law and (ii) for failing to give the Governor, David Durie CMG, a true account of the position in 2002.
- 59. Ms Williams began to work for the Chief Justice in or about September 1996 and her employment was terminated in October 1997. In breach of Regulation 7 of the Employment Regulations 1994 he failed to obtain a work permit for her. What he did was to sign in blank three forms, one of which was an application for a work permit, leaving it to Ms Williams to fill these in and submit them. Ms Williams filled them in as his agent, but she deliberately inserted false information as to her terms of employment in order to obtain the work permit that was subsequently issued. Ms Williams stated that she was employed for 39 hours per week at £3.15 per hour. In fact her working hours were significantly less than this the Chief Justice's evidence was that she worked less than 15 hours a week and earned £150 month.
- 60. The following account of the Chief Justice's obligations in relation to tax and PAYE is based on a note provided to the Committee by Mr Beloff. Although Ms Williams' pay was too low to attract liability to income tax she failed to obtain a certificate to that effect. In these circumstances the Chief Justice was under a liability under PAYE regulations to deduct tax at a "default" rate called Code X. The Chief Justice made no such deductions. While under the Social Security Regulations Ms Williams was, because of the short hours that she worked, exempt from some aspects of the social security scheme, the Chief Justice was under an obligation to make social security payments to cover insurance against accident. The Chief Justice made no such payments. It was the Chief Justice's evidence that he had assumed that PAYE was not payable and was unaware of the social security obligation. He

only became aware of his errors as a result of evidence given to the Tribunal. He would have expected the authorities to make demands for payments if any payments were due.

- 61. By a written contract dated 9 October 1997 the Chief Justice employed Ms Danvers as a housekeeper/cook at a salary of £450 a month. He obtained a work permit in respect of her. Her employment terminated in April 2000. At that point she lodged a complaint with the Ombudsman and with the TGWU that she had been paid less than the minimum wage. That complaint does not appear to have been well-founded. At this point the Chief Justice contacted both the tax and the social security authorities. He was informed that there were outstanding social security obligations and he paid these. He was also informed of outstanding PAYE obligations and entered into negotiations to determine the amount of these. On 28 June he paid £990.30 in relation to these obligations.
- 62. On 26 May 2000 under a headline "M'Lord you have broken the law" *Vox* devoted its front page to the story of Ms Danvers' employment by the Chief Justice and his failure to pay PAYE and Social Insurance contributions. Correspondence ensued with the Governor, which related initially only to Ms Danvers. The Governor made a public statement that he was personally looking into the matter.
- 63. On 9 June *Vox* published an article under the headline "The Chief Justice must now resign" in which it was suggested that the Governor should widen his enquiry to cover Ms Williams. The article raised the question of whether her social insurance and income tax was paid up. On 16 June the Chief Justice wrote a lengthy letter to the Governor which largely related to Ms Danvers but also referred to Ms Williams. This included the statement that his recent researches had disclosed that he might be liable to pay social security contributions in respect of Ms Williams.
- 64. On 3 July the Chief Justice wrote to the Governor:

"I have placed the matter of Ms Williams (now Mrs Moreno) before leading counsel. Having perused the legislation with some care counsel is satisfied that by reason of its terms and the hours she worked she does not come within the legislation. I am thus advised that there is no liability upon me.

Mrs Moreno did not earn sufficient wages to come within the P.A.Y.E legislation."

65. One of the duties of the Attorney General, Mr Rhoda QC, was to give legal advice to the Governor. By 12 July 2000, after discussions between the Attorney General and Mr Desmond de Silva QC, who was advising the Chief Justice an exchange of letters between the

Governor and the Chief Justice had been agreed. That from the Governor was to read as follows:

"Thank you for the information you have provided in response to my letter to you of 1 June. I have also received information from the relevant public authorities.

I note that you registered your employment of Ms Danvers with the Employment Training Board and that you have recently settled your outstanding obligations for PAYE and for social security in her regard. I find it regrettable that you did not make more effort to regularise this matter earlier, but I accept that you did not deliberately seek to avoid your obligations.

You will appreciate, I know, that someone in your position should be particularly careful to fulfil your legal obligations in good time and I would be grateful for your assurance that you will do so in future.

As far as your earlier employment of Ms Williams is concerned, I take your letter of 3 July to mean that your considered view is that no liability falls to you in respect of either PAYE or social security.

As you must be aware, I am not in a position to take an independent view of that matter. You will appreciate therefore that if in the event it turned out that there was an outstanding liability this could have the effect of calling the whole issue to be revisited."

66. The letter to be written by the Chief Justice in reply read:

"Thank you for your letter of 12 July. I do regret, that through oversight, I did not make greater efforts to meet my obligations in time in respect of social security and PAYE in my employment of Ms Danvers. I, of course, accept that I should be particularly careful to fulfil my obligations in good time and I can assure you that this situation will not arise again. I can also confirm that I have no reason to believe that I have any outstanding liabilities in respect of Ms Williams."

67. These letters were not in fact released. Delay in bringing the matter to a conclusion was attributable, at least in part, to the MOT prosecution, to which we will shortly turn. On 16 August 2000 the Chief Justice wrote a letter to the Governor which asked, among other things, whether he had taken advantage of the delay that had occurred to reach a conclusion in relation to the Jacqueline Williams matter. On 23 August the Chief Justice wrote a further letter in the course of which he observed that the Governor had determined the issue in

relation to Ms Danvers and invited him to do the same in relation to Ms Williams. On 31 August the Governor replied in relation to this:

"In your letter of 23 August you raised the matter of Ms Williams. Since at an earlier date you chose not to provide me the extra information for which I asked in relation to Ms Williams, I continue to rely on the assurance which you gave me on 12 July in respect of her. I have conducted no separate investigation in her regard."

It is not clear from the evidence what extra information was sought or in what circumstances.

68. Ultimately on 4 October the Governor made an announcement of his decision. It stated that he had received full cooperation from all concerned, including from the relevant authorities. It continued:

"The information I have received shows that Ms Danvers' employment was registered with the Employment Training Board and that all outstanding PAYE payments and social security contributions have now been met. It is regrettable that matters were not regularised at an earlier stage but I have accepted that the Chief Justice did not deliberately seek to avoid his obligations.

The Chief Justice has also assured me, in relation to another former employee of his, Ms Williams, that he has no outstanding liabilities.

I have concluded, in view of the information and assurances which I have received, that it would not be appropriate for me to take any formal action in exercise of my constitutional powers."

69. On 9 October the Government issued a press release that stated that as the decision was constitutionally for the Governor the Government did not consider it necessary or appropriate to comment on his conclusions. It then said that since the Governor's statement only stated part of the facts in relation to the Chief Justice's two maids the Government considered that the public interest required the facts of the two cases to be put into the public domain. The facts then set out in relation to Ms Williams included the following:

"Miss Williams herself registered her employment with the Department of Social Services for the purposes of Social Insurance. The Department of Social Security has no record of social insurance contributions having been paid in respect of Ms Williams' employment. The Commissioner of Income Tax has no record of tax having been paid in respect of Ms Williams' employment and has no record of her employment.

It is clear from the Governor's statement that the Chief Justice does not, in these circumstances, consider that he has any outstanding liability."

- 70. The Maids Issue was the second of the four examples of misbehaviour itemised by the Tribunal at the end of its report. There the Tribunal summarised the misbehaviour as persisting "in a reckless disregard for compliance with the law" and making "a less than frank disclosure to the Governor" (7.35).
- 71. As the Chief Justice accepted, his position required that he should show a scrupulous regard for the law. In these circumstances leaving Ms Williams, without supervision, to fill in forms for the accuracy of which he was responsible and his failure to make any or adequate enquiries in relation to his obligations in respect of PAYE and social security payments were reprehensible, particularly in the case of Ms Danvers. The false assumption that no obligations arose in relation to Ms Williams was perhaps more understandable. These failings were, however, water long under the bridge. The Governor had decided to draw a line under the Chief Justice's failings in relation to Ms Danvers. Although he had not been informed of the true position in relation to Ms Williams we do not consider that it would or should have affected his decision. In these circumstances we do not consider that these breaches of the Chief Justice's legal obligations should, in 2009, have weighed against him in the scale as significant misbehaviour. Potentially more significant were the following matters relied upon by the signatories as part of their case that the Chief Justice was mendacious:
 - "a) the Tribunal's rejection of the Chief Justice's assertion in oral evidence that he must have mentioned Ms Williams to the Commissioner of Income Tax;"
 - b) the Tribunal's finding that the statement in the Chief Justice's fourth witness statement that he had given "a full explanation" to the Governor in relation to Ms Williams was "plainly untrue";
- 72. In the course of cross-examination the Chief Justice said:

"I discussed these issues with the Collector of Income Tax in relation to Ms Danvers, and it is inconceivable I did not mention Ms Williams at the time, and there was no comment by him to my recollection that I had fallen foul of the law".

73. This statement was made in the context of discussion of a statement made by the Chief Justice when giving evidence in chief that there was a relaxed attitude to matters such as PAYE in Gibraltar. The statement was not challenged. The Chief Justice was reconstructing what he believed must have been said in 2000 rather than asserting a positive recollection of what was said. It may well be that there was an element of wishful thinking in

the reconstruction. It would not be right, however, to treat this as an example of deliberate mendacity.

- 74. The question of whether the Chief Justice had misinformed the Governor as to the advice that he had received from counsel was not included in the Statement of Issues. That he had done so was put to him, vigorously, by Mr Eadie in cross-examination. Mr Beloff submitted, with some justification, that it was not fair to expect the Chief Justice to meet this allegation without notice of it. He placed before the Committee an 18 page analysis of the evidence in relation to the maids' issue. This demonstrated that the Chief Justice professed no longer to remember the details of the advice that he sought or received. He identified the leading counsel to whom he had referred as either Mr Stagnetto or Mr de Silva. His evidence was that he had been sure that the letter that he had sent to the Governor was accurate.
- 75. Mr Stagnetto sent an e-mail dated 21 July 2008 to the Chief Justice's solicitors, which was put in evidence before the Tribunal. It stated:

"The issue of the employment of Jacqueline Williams arose in the context of a press release by the Governor the terms of which had been negotiated by Mr Desmond de Silva QC on behalf of the Chief Justice and the Attorney General/Governor in relation to the employment of Mrs Danvers. Before the agreed date of publication it emerged that the Chief Justice had also employed Jacqueline Williams and the Governor wanted to be satisfied that the Chief Justice had complied with all his obligations in relation to her employment before publishing his report. I was informed of this by the Chief Justice and he told me that her pay did not reach the threshold to bring her within the PAYE legislation. I replied to the effect that if that was so he had nothing to worry about and he should inform the Governor accordingly. He may have taken advice from Mr De Silva as indeed he did in the case of Mrs Danvers. I was never asked by the Chief Justice nor indeed did I give my opinion in writing on the matter."

76. In his fifth witness statement the Chief Justice said that the Government's press release of 9 October:

"whilst being strictly accurate, seems to suggest that contributions were due in respect of Ms. Williams. This is a misleading innuendo since no contributions were due and unpaid in respect of her. Furthermore, on my understanding the tax situation of Ms. Williams was unlawfully released to the public. If it was being suggested at the time, or if it is

suggested now, that I had any liability in respect of Ms. Williams it would be astonishing indeed, given that at no time either before or since her departure from our employment in October 1997 has there been any communication to me of any existing claim from any Government Department."

Mr de Silva had no recollection of advising on this matter.

77. The submission made by Mr Eadie on behalf of the Government is that the Chief Justice in 2000 instructed his lawyers to ascertain his position in 2000 rather than the position in 1996 to 1997 when Ms Williams was working for him and then gave a misleading description of the position so as to lead the Governor to believe that he had been advised that he had committed no breach of the law during the earlier period. Implicit in this submission is that the Chief Justice believed that there was a difference between his legal obligations in 1996 to 1997 and in 2000 and that he was or might be in breach of an obligation to pay PAYE in the earlier period that did not persist to 2000. We can see no basis for drawing such a conclusion. Right up to the time that he made his 5th witness statement the Chief Justice gave every indication of believing that no PAYE obligation could have arisen in relation to Ms Williams. That was not an unreasonable belief, albeit erroneous. In fact, as a matter of strict law, the obligation that he had been under to make PAYE deductions while Ms Williams had worked for him had persisted to 2000. We do not consider that there is any basis for holding that the Chief Justice set out in 2000 deliberately to deceive the Governor. Because of the passage of time there is a question mark over the advice sought and given by leading counsel, but no case of mendacity on the part of the Chief Justice has been made out.

Dissatisfaction with the Chief Justice

78. On 28 June 2000 five Queen's Counsel sent a memorandum to the Governor expressing regret at his delay in announcing whether he had found any grounds for establishing a Tribunal of Inquiry which was the limit of his powers in relation to the conduct of members of the judiciary under the constitution. They stated that:

"the delay has resulted in division between those whose views on the quality and integrity of the administration of justice permit them to regard the allegations as a storm in a teacup and those who cannot so regard them."

It is clear that there was in Gibraltar, at least by this stage, a significant body of opinion, both at the Bar and outside it, critical of the conduct of the Chief Justice. Whilst the Chief Justice was not justified in alleging that the Government's Press Release of 9 October made a "misleading innuendo", the fact and terms of the release suggested that the dissatisfaction was felt within the Government. Lord Luce, who was Governor between February 1997 and March 2000, spoke of noticing a growing strain in the relationship between the Chief Justice and the Chief Minister and said that when he renewed the Chief Justice's contract in 1999 he was "not entirely happy with the lack of leadership of the Chief Justice in that capacity". At

the ceremony of the opening of the legal year Mr Robert Vasquez, the Chairman of the Bar, who had been dissuaded from making a speech critical of the Chief Justice, instead, in breach of tradition, refrained from making any speech at all. The Chief Justice responded, angrily, by saying to him "Robert, don't rape the Constitution". The Tribunal commented that this language was not consistent with the dignity and status of the office of Chief Justice. We do not attach significance to this sentence, spoken no doubt in a moment of tension.

Instructions to the Registrar of the Supreme Court in respect of expenditure in May 2000.

79. This issue is dealt with in paragraphs 3.28 to 3.31 of the Tribunal's Report. It relates to a request by the Chief Justice to the Registrar for the use of funds earmarked "for general and office expenses" to cover the costs of a party that the Chief Justice had organised at his residence on 9 May 2000 to introduce the new Governor to the Bench and Bar. The Tribunal concluded that this issue was of "only limited significance" to the general issues falling for its consideration. We agree.

Allegations of interception of the Chief Justice's telephone in 1999 and 2000

- 80. In 1999 the Chief Justice and his wife became concerned that there might be interference with their telephones. The Chief Justice reported this concern to the police. On 30 May 2000 the Gibraltar newspaper *Panorama* carried an article under a front page headline that stated "Police asked to investigate phone tapping, claim at home of Chief Justice. 'Campaign to hound my husband out of office' Mrs Schofield". The article referred to the Maids Issue. It reported an interview with Mrs Schofield in which she had said "I am not the Chief Justice and I am going to fight for what I think is an unfairness...They are trying to discredit my husband. They are trying to hound him out of office." On the same day an article in *Sunday Business*, which is not published in Gibraltar, reported that the Chief Justice had told friends that he believed that he was under surveillance because of a clash with the Chief Minister over claims of political interference in the judiciary's independence.
- 81. The Chief Justice denied the truth of the article in *Sunday Business*. He said that he and his wife had never regarded the telephone tapping to be in any way the responsibility of the Government of Gibraltar. His wife, in her second statement, accepted her responsibility for the article in *Panorama*. She said that her intention was to raise awareness in a matter of public interest.
- 82. The Chief Justice was asked whether he considered that the Maids Issue was part of an attempt to discredit him or drive him from office and he said that he did not, either at the time or in 2007. The Chief Justice was asked whether he discussed the article with his wife

at about the time that it was published and he said that he had. He was not asked about the nature of the discussion. He was asked whether he considered making a public statement that it was not his view that he was being hounded from office and he said that he did not. The Tribunal commented:

"His unwillingness to correct what had been attributed to him and to dissociate himself from his wife's comments on this issue was not satisfactorily explained by him. He must have been aware of the damage thereby caused to Gibraltar's financial and legal reputation."

83. The Chief Justice at other points of his evidence made plain that it was not his policy to join issue with the press on misleading press reports as once one started such a dialogue there was uncertainty as to where it would lead. We accept that there are dangers in engaging in dialogue with the press, and would not on this occasion criticise the Chief Justice for not publicly dissociating himself from his wife's comments to the press. Those comments were made, however, after statements made by his wife that she would be prepared to fight for her husband "in the gutter" – (3.20 - 3.22). We consider that the Chief Justice should have been concerned by his wife's conduct. He should have made plain such concern to his wife. He should have explained to her that her conduct was damaging both to Gibraltar and to his position as Chief Justice.

The MOT prosecution

- 84. The Tribunal's Report 3.39 3.52 describes the series of events relating to the prosecution of the Chief Justice for failing to have a valid MOT certificate for his car. We shall not repeat that exercise. The Report criticises the Chief Justice for behaviour which was not "consonant with the proper conduct of a Chief Justice in a small jurisdiction and with the dignity of his office". While we agree with that criticism the Report does not place this incident in its context nor set out the full story of this matter.
- 85. The Chief Justice's MOT certificate expired on 11 January 2000. He overlooked the need to renew it. In May 2000 his road tax licence was about to expire. Mr Mendez, the Deputy Registrar, drew this to his attention. He could not re-tax his car until he had obtained a new MOT certificate. The car's log book had been lost, and a duplicate had to be obtained before he could seek a new MOT certificate. A booking was made for an MOT inspection at the end of August. Mr Mendez received an assurance on the telephone from a woman police officer that, if the Chief Justice had an appointment for an MOT he could continue to drive pending that appointment. At the same time the police had announced a general amnesty for anyone driving a car that was untaxed that would expire on 31 July. Mrs Dawson, the Registrar, advised the Chief Justice that he should not drive after that date.

- 86. On 28 July the Chief Justice was stopped by police constable Perera, who was carrying out routine checks on vehicle documents. He explained why he did not have a valid MOT certificate or road tax licence. He then went on holiday.
- 87. On 31 July his lawyer, Mr Stagnetto, was contacted by the Attorney General who said that the matter had been drawn to the attention of the Governor who wished to seek an explanation from the Chief Justice as the matter affected the agreed disposal of the maids issue.
- 88. On his return from holiday, on 16 August, the Chief Justice was seen at his residence in the presence of his lawyer, Mr Stagnetto, by PC Perera and PS Vinales. He was asked to produce evidence of the appointment that had been made for the MOT test, which he did. He was then told that he would be reported for not being in possession of an MOT certificate and a road tax licence.
- 89. On 21 August the Chief Justice received a letter from the Governor, which ended as follows:

"Whilst the allegations relate to motoring matters, they also relate to a failure to comply in a proper and timely fashion with obligations of a public nature. I am mindful of the fact that the allegations in respect of the Danvers affair related to a failure on your part to make proper and timely contributions to the public revenue.

Two issues in particular require to be clarified, first whether the allegations relating to 28 July are true, and in the event that they are true, how it was that in the light of those matters you were prepared to give me assurances which were contained in your letter of 12 July. Unless, and until, I am satisfied by your explanations in relation to these matters, I cannot make a public statement in the terms which were initially envisaged, namely my continuing support for your position. It is for this reason that I want to discuss matters with you and would welcome the opportunity to do so later today."

90. The Chief Justice, through his lawyer, challenged the suggestion that the MOT matter had any bearing on the resolution of the Maids Issue. On 23 August the Chief Justice received a formal written caution signed on behalf of the Commissioner of Police indicating that it had been decided to issue him with a caution. The letter ended "This caution has been recorded for further reference".

91. On 25 August the Attorney General, Mr Rhoda, wrote to Mr Stagnetto in the following terms:

"I have now had the benefit of seeing both the Governor's letter of 21 August 2000 to the Chief Justice, and the Chief Justice's reply of 23 August 2000.

The matter has progressed and that my understanding is that a Formal Caution has now been issued by the Commissioner of Police to the Chief Justice.

Until such time as I know whether or not the Chief Justice is prepared to accept the decision to caution him, thereby acknowledging the truth of the allegations, it would not be right for me to comment upon the causal connection between the events of 28 July 2000 and the assurances given by the Chief Justice in his letter of 12 July 2000.

As you firm is now acting for the Chief Justice in respect of events of 28 July 2000, perhaps you would let me know whether or not your client is prepared to accept the caution."

92. The Attorney General explained the subsequent course of events in his witness statement:

"I was not involved in issuing the first caution notice to the Chief Justice. The first caution notice that was issued was defective in that it did not require him to accept his guilt. Had the Chief Justice accepted the caution, matters might have ended there. Instead the Chief Justice's lawyers became involved and quibbled about the wording of the caution. At that point, the Commissioner of Police sought legal advice and the defect in the first caution was identified. A new caution was therefore issued which did require the Chief Justice to accept his guilt. The Chief Justice did not respond to this caution notice until the eleventh hour and effectively did not accept his guilt. In the circumstances, I made the decision as Director of Public Prosecutions to prosecute him."

- 93. Ultimately, on October 4th, the Governor published his decision on the Maids Issue without reference to the MOT matter, as this was *sub judice*.
- 94. The Chief Justice has always considered that his prosecution was "grossly unfair". We can understand that reaction. The evidence indicates that the first written caution that he received was the informal way that a trivial motor vehicle offence was normally dealt with at the time in Gibraltar. The subsequent action taken by the Attorney General was, so it seems

to us, taken by the Attorney General because he wished to be in a position to advise the Governor on the Chief Justice's position with an unequivocal acceptance by the Chief Justice of the offences that he had committed.

- 95. The Chief Justice said in his fifth witness statement that he felt that the MOT matter could be used to try to pressurise him out of office or at least seriously embarrass him. His wife in her first statement said that it was her view that the matter was a "set up" intended to provide the Governor with further evidence to seek the removal of the Chief Justice.
- 96. The Chief Justice told the Tribunal that he did not consider that his prosecution was politically motivated. When asked whether his wife had thought the prosecution politically motivated, he replied that he did not wish to breach any confidences between himself and his wife.
- 97. Mr Beloff submitted that the Tribunal erred in making any criticism of the Chief Justice in relation to the MOT affair. He could not be criticised for permitting international observers to be present at his trial, nor for raising defences that his lawyers had advised were arguable.
- 98. We do not agree. The conduct of the Chief Justice demonstrated a serious lack of judgment that was calculated to bring his office into disrepute, to put a heavy and unnecessary burden on the Stipendiary Magistrate of having to try his own Chief Justice in the full glare of publicity and to cast aspersions over the fairness of that trial, which was, in fact, meticulously conducted.
- 99. Given that the Chief Justice did not consider it fair that the Commissioner of Police sought an express admission that he had committed the MOT offence and given that the Governor had indicated that this might have relevance to the disposal of the maids affair, it still made no sense at all not to admit that he had committed what was, in the circumstances, a trivial offence; there was, after all, no doubt that he had done so. Fighting the prosecution on technical defences was patently an absurd thing to do and one that, accordingly, brought his office into disrepute.
- 100. When being questioned by the Tribunal the Chief Justice said that he had no knowledge of any briefing that his wife made to the international observers. That answer was not challenged. The Tribunal none the less said that they did not accept that he did not know the content of the observers' briefing. We do not consider that adverse finding to be justified. That said it can properly be inferred that the Chief Justice knew that his wife had arranged the attendance of international observers because she thought that his prosecution was politically motivated. He said that he did not share that view. Whether he did or not, there was no

possible justification for observers at the trial unless there was a suggestion that the trial itself would be unfair. Their presence necessarily carried an inference adverse to the Stipendiary Magistrate that was unwarranted. The Chief Justice could and should have made it plain to the international observers that he did not consider that there was any justification for their presence.

- 101. The Chief Justice was asked about his view of the motivation for the MOT prosecution:
 - "Q. Do you suggest that this prosecution was politically motivated?
 - A. No, but it was instituted against a background of what was going on between Governor Durie and the Attorney General and myself vis-à-vis the Maids Issue. I did consider that it was strange indeed that the Governor knew that a formal caution had been issued before me before ever I did.
 - Q. Did you –
 - A. And I put it no higher than that. I can put it no higher than that.
 - Q. Did you discuss that strangeness with your wife in 2000?
 - A. Undoubtedly."

In the course of the discussion Mrs Schofield must have expressed her view that the prosecution was politically motivated. This passage suggests that the Chief Justice was not convinced that she was wrong.

102. In August 2000 a conversation took place between the Attorney General and Mr De Silva, who had acted for the Chief Justice in connection with the Maids Issue, at Sotogrande where they were both on holiday. There was a degree of conflict of evidence about what was said at that meeting. It is common ground, however, that the Attorney General agreed that it "would be a good idea if the Chief Justice could be found something elsewhere". The Chief Justice did not suggest that this view in any way affected the action taken by the Attorney General in relation to the MOT prosecution. Had he had any doubts as to this they should have been assuaged by (i) the fact that the Attorney General offered to drop the prosecution and permit the Chief Justice to make a statement if he accepted that he had committed the offence; (ii) the statement by the Attorney General after his conviction that if he dropped an appeal this would, in his opinion, have no effect on the tenure of his office and (iii) the fact that his conviction had no adverse consequences on his tenure of office.

103. In paragraphs 4.7 to 4.26 of the Tribunal's Report the Tribunal deals with the reaction of the Chief Justice to the issue by the Governor on 7 February 2002 of a warrant renewing his appointment for a period of only one year in place of the customary three year warrant. The Chief Justice was given no warning of this, nor explanation for it. Nor was any explanation put before the Tribunal or the Committee. The Tribunal speculated that it may be that the Governor was dissatisfied with the Chief Justice's conduct and wished to mark his disapproval in some way. This seems a likely explanation. The Tribunal described the Chief Justice's letter of response of 11 February as "Notable for its hostility and threat". We consider that it is not surprising that the Chief Justice responded in strong terms. The letter accused the Governor of an "intention to purport to limit my tenure of office".

104. The Governor replied on 13 February, giving no explanation for the one year warrant and stating somewhat equivocally:

"You raise the question of security of tenure and your independence as a judge. I know your views on the effect of the Constitution which are that, in accordance with the terms of Section 60 of the Gibraltar Constitution Order 1969, the Chief Justice of Gibraltar has security of tenure until he attains the age of 67 years.

The long-standing practice of issuing time-limited Warrants is not intended to, nor could it, affect the provisions of the Constitution.

I hope that this is clear, and that it will set your mind as rest."

105. Meanwhile, on 12 February the Chief Justice had raised the question of the effect of the one year warrant in a chambers hearing in a criminal trial in which Mr David Hughes was appearing for the defendants. A note of the hearing taken by Mr Mendez, the Deputy Registrar, includes the following statement by the Chief Justice:

"Chief Justice: I have told the Governor that he may not offend the Constitution or violate my independence. If the Governor has not withdrawn his purported action by close of business today, I shall want the Attorney General to be here, to address me on the validity of the Governor's purported action. I may feel that I must abandon this case and indeed suspend all sittings of the Supreme Court.

Hughes: You are right. An opposite view to yours would not even be arguable.

. . .

This is a grave matter. We may have to run arguments on this. You are correct in your views.

Chief Justice: Hughes, how did you know so much about this? Is it out?

Hughes: It was discussed at the Bar Council. Maybe you should ask the Chairman of the Bar to be present tomorrow as amicus curiae."

On the following day in open court he raised the question of whether his independence might be challenged, referring to the Scottish case of *Starrs v Ruxton* [2000] SLT 42. Counsel reassured him that no such suggestion would be made.

106. What had happened in chambers was reported in the *Gibraltar Chronicle* on the following day. On 15 February the *New People* carried an article headed "Chief Justice in constitutional row with UK." It quoted verbatim from an angry letter that Mrs Schofield had written to the Foreign Secretary alleging attempts to hound her children out of their home and harassment aimed at inducing the Chief Justice to depart of his own accord.

107. The Chief Justice challenged the accuracy of Mr Mendez's note. We agree with the Tribunal that it provides the best evidence of how the Chief Justice reacted in court to the receipt of the one year warrant save that, having considered the oral evidence given to the Tribunal, it finds it more likely that the Chief Justice threatened to suspend all his own sittings rather than all the sittings of the court. We consider that the Chief Justice's behaviour in Court once again showed lack of judgment or a sense of proportionality. If the terms of the warrants issued by the Government in fact impacted on the Chief Justice's term of office, the *Starrs v Ruxton* issue of whether the Chief Justice had the necessary independence already existed. The Chief Justice did not, however, believe that the terms of the warrants affected his security of tenure, nor did anyone else suggest that they did.

108. The Chief Justice said that he had no knowledge of how the *Gibraltar Chronicle* obtained its information about the matter. He said that his wife told him about the letter that she had written to the Foreign & Commonwealth Office after she had written it. He did not think to dissociate himself from her letter as she was ploughing an independent furrow, talking about family rights. He said that he was not concerned about these because he did not feel that the warrant of appointment had any legal effect.

- 109. This was a further example of the Chief Justice's "laissez faire" approach to actions of his wife that had implications for his own position. The Tribunal pointed out, with justification, that there was an appearance of collusion between his wife and himself.
- 110. Mr Beloff attacked, in particular, a finding of the Tribunal that the Chief Justice had colluded with Mr Hughes in raising the possible implication of the one year warrant in court. He submitted first that it was procedurally unfair for this finding to have been made when the Chief Justice had had no notice that such an allegation was to be made. Secondly he submitted that the finding was not justified on the evidence.
- 111. We agree with both points. This was a serious allegation of bad faith and one that required notice if it were to be properly addressed. On a full analysis of the evidence the finding of the Tribunal is not justified. That finding was that Mr Hughes called on the Chief Justice in his chambers and between them they agreed that Mr Hughes would raise the matter. It necessarily follows from this that the Chief Justice's question "Hughes, how did you know so much about this? Is it out?" And his reply "It was discussed at the Bar Council..." was a dishonest charade, designed to conceal their collusion. We have considered with care the detailed analysis of the evidence prepared by those acting for the Chief Justice and concluded that it does not support the finding of collusion made by the Tribunal.

Summary thus far

112. These matters, some of them significant, form the background to the more immediate events that led to the appointment of the Tribunal. They reflect badly on the judgment of the Chief Justice, leading to inappropriate behaviour on a number of occasions. They also establish a propensity on the part of Mrs Schofield to take ill-advised action in support of her husband that was capable of carrying the implication that he associated himself with it and that was damaging to his standing as Chief Justice. There followed, however, a period of some four years without relevant incident. During this period, as the Chief Minister pointed out, nothing occurred capable of suggesting dissatisfaction on the part of the Government with the Chief Justice. On the contrary, the Government treated the Chief Justice generously by the provision of some financial assistance beyond that to which he was entitled under his terms of service. Equally during this period there were no events that, on the findings of the Tribunal, were cause for criticism of the Chief Justice.

The swearing in of the Deputy Governor in July 2006

113. This matter is dealt with in paragraphs 4.32 to 4.37 of the Tribunal's Report. It was a single incident of thoroughly unattractive behaviour by the Chief Justice. He lost his temper because, in accordance with the wishes of the outgoing Governor, Sir Francis Richards, the

Chief Minister had been given precedence over him in the grouping of those bidding him farewell at the Naval Dockyard. He took this out on the Deputy Governor whom he considered responsible for the arrangements. The Tribunal was justified in describing this as disgraceful behaviour, governed by pique that was inconsistent with the dignity of his office. It could properly be described as misbehaviour, albeit not of major consequence.

The debate over the 2006 Constitution

- This matter is covered by paragraphs 4.38 to 4.63 of the Tribunal's Report. The facts can be summarised as follows: In 1999 a Select Committee had been established by the Gibraltar House of Assembly to consider reform of the Constitution. There was a consultation period which closed in February 2001. The first reaction of the Gibraltar judiciary was a paper submitted by the Chief Justice to the Governor, with a copy to the Chief Minister in March 2005, making recommendations for amendments to the Constitution. Although the Chief Justice described these as "made after extensive consultation with my colleagues and are those of the whole judiciary" this overstated the position. recommendations included the institution of a Judicial Service Commission ("JSC") to make judicial appointments. Modified recommendations were made by a letter of 21 February 2006, inspired by the provisions of the United Kingdom Constitutional Reform Act 2005. These recommended a Judicial Appointments Commission made up of the Chief Justice, the President of the Court of Appeal, the Chairman of the Justices of the Peace, a member of the Bar proposed by the Bar, one appointee of the Chief Minister and one appointee of the Governor.
- 115. On 17 March a UK delegation and a Gibraltar cross-party delegation reached agreement on a new draft Constitution. The Chief Justice did not press for a copy of this. On 5 July it was placed on the Government website. The Chief Justice did not obtain a copy until 4 August. On 11 August, following an earlier short letter of concern, he wrote to the Chief Minister enclosing "the submissions of the Chief Justice, the Puisne Judge, the Stipendiary Magistrate and the Registrar on the draft Constitution" ("the submission"). Article 57(1) of the draft Constitution provided for the creation of a JSC consisting of the President of the Court of Appeal as Chairman, the Chief Justice, the Stipendiary Magistrate, two members appointed by the Governor in accordance with the advice of the Chief Minister and two members appointed by the Governor acting in his discretion. The submission objected that the majority of the JSC would be appointed by the executive and suggested that it should include a member of the legal profession and a lay member. It also suggested that the Chief Justice should be the chair of the JSC, or should not be a member at all.
- 116. The submission took particular objection to Article 57(3) which provided:
 - "The Governor, with the prior approval of the Secretary of State, may disregard the advice of the Judicial Service

Commission in any case where he judges that compliance with that advice would prejudice Her Majesty's Service."

This, it was said, was in conflict with the requirements of judicial independence. A similar objection to Article 57(3) was to be made by the Bar Council.

- 117. The submission concluded with a statement that the judiciary would consider all possible steps that it might take to prevent these errors of principle being promulgated including "reluctantly but if necessary, a petition to Her Majesty through Her Privy Council".
- 118. These objections to the draft Constitution received media coverage. This included an article in Panorama on 15 August headed "Chief Justice in campaign against new Constitution."
- 119. On 25 August Hassans wrote on behalf of the Chief Justice to the Acting Governor and the Chief Minister stating that some of the provisions of the draft Constitution would adversely affect judicial independence and the rule of law in Gibraltar and public confidence in due administration of justice by the courts. The letter complained of lack of consultation and called for consultation before any referendum on the Constitution, failing which it would be necessary to have recourse to the Judicial Committee of the Privy Council. This was followed by a letter on 5 September threatening that the Chief Justice would petition the Privy Council directly and unilaterally unless by 8 September the Gibraltar and UK Governments confirmed that they would refer the matter to the Judicial Committee.
- 120. The Chief Secretary replied to the Chief Justice on 7 September in a lengthy letter which refuted the points made by Hassans. So far as Article 57(3) was concerned it gave the following explanation:

"Section 57(3) was included in the draft Constitution at the behest of the UK side. It was said by HMG to form an important element of the UK position in the negotiating process, as part of the UK's agreement to the establishment of the new Judicial Service Commission (which significantly reduces the Governor's powers in this area in comparison to the present Constitution and those of other Overseas Territories). According to HMG it requires the provision to reflect the continuing constitutional relationship between the United Kingdom and Gibraltar and the UK's interest in the good administration of justice. The UK Government does not, therefore, accept that the provision is objectionable. Even though the Gibraltar Government does not consider the provision to be necessary, it does not consider it objectionable on the grounds that you allege.

The UK Government has informed the Gibraltar Government, and the Bar Council that it would only envisage the power being used in extremely rare and exceptional circumstances. These might include a case where the UK Government had information on a recommended candidate that could not be shared with the Judicial Service Commission for reasons of confidentiality, or where a wholly unqualified or unsuitable candidate were recommended for appointment to a particular office;

The section is not intended to, and does not, give the Governor an enabling power. It is deliberately drafted as a veto power only. This was clearly agreed between the delegations during the negotiations on the text when the words 'and act in his own discretion' (which were in a previous draft) were removed at the request of the Gibraltar side. The effect in practice is that, if the veto power were exercised, the Judicial Service Commission would have to reconsider the matter and tender such further advice to the Governor as it thought appropriate;"

A similar undertaking had been given in a letter sent on the previous day by the Chief Minister and the Acting Governor to the Bar Council.

- 121. The Assistant Deputy Governor also wrote on the 7 September refuting Hassans' contentions and offering the Chief Justice a meeting with officials in London who would explain in more detail the UK Government's position on the judicial aspects of the draft Constitution.
- 122. The Chief Justice's speech at the opening of the legal year on 6 October was devoted almost entirely to the draft Constitution. The Tribunal's criticisms of the Chief Justice focus on this. In essence the criticism is that the Chief Justice improperly used this speech to enter the political arena by making in public an attack on parts of the draft constitution that had been agreed by the Governments of the United Kingdom and of Gibraltar and on which a referendum was shortly to be held. The consequence was that the press had a field day and the opposition seized on his remarks to justify opposing the proposed constitution, so that "consensus turned to dissent". Six specific criticisms were made of his speech. Underlying those criticisms were express or implicit criticisms of the earlier conduct of the Chief Justice in relation to the draft Constitution. We will deal with each criticism in turn.
- 123. The first criticism is that in his speech the Chief Justice purported to be speaking on behalf of the locally based judiciary when he had not properly consulted them in relation to the various steps that he had taken, or in relation to the text of his address. He was acting on his own initiative.

- 124. The judiciary in question consisted of Mr Justice Dudley, the puisne judge, Mr Pitto, the stipendiary magistrate and Miss Desoiza, the Registrar. Mr Justice Dudley did not give evidence or make a witness statement, so there is no direct evidence on his views. Miss Desoiza's attitude was that she did not want to get involved. She had expressed concern at the possibility that she might get conflicting instructions from the Chief Secretary and the Chief Justice, but she had not expressed concern about being under the disciplinary control of the executive, as suggested by the Chief Justice in his speech. Mr Pitto said that he did not consider that he had been consulted before the Chief Justice made the written submission of 7 March 2005. He said that discussion "tended to be more about the Chief Justice giving me his views". He did not dissent from those views, but thought that ultimately they were a matter for those drafting the Constitution, not to be imposed by the judiciary. He was not happy with the proposal of petitioning the Privy Council, and the Chief Justice was aware of this. Mr Pitto spoke of one short meeting at which the Chief Justice's views were discussed with Mr Dudley and he did not dissent from them. He also, however, was not in favour of a petition to the Privy Council.
- 125. The position can be summarised as follows. There had been no formal consultation of the local judiciary. While its members would not have dissented from the changes that the Chief Justice wished to procure to the draft Constitution, they would not have approved of proactive attempts by the judiciary to achieve those changes, such as the proposed petition to the Privy Council. Nor would they have approved the aggressive terms of the Chief Justice's speech, which he had not discussed with them. The Chief Justice was, in reality, conducting a one man campaign in the name of the judiciary as a whole.
- 126. Also open to criticism is the letter by the Chief Justice to the Chief Minister of 11 August which purported to contain the submissions of all four members of the judiciary. The threat, if necessary, to petition Her Majesty through the Privy Council against errors of principle in the draft Constitution, was one which had no support from the other members of the judiciary and Mr Pitto had made it plain that he was opposed to it.
- 127. The second criticism is that the Chief Justice acted as if the judiciary had equal standing to the Governments of the United Kingdom and Gibraltar in negotiating the draft constitution. The tone of the Chief Justice's communications in relation to the draft Constitution was indeed high handed and did not reflect the reality that the new Constitution was primarily a matter for the Governments of the United Kingdom and Gibraltar.
- 128. The third criticism is that the speech was deliberately worded so as to give rise to an implication of bad faith on the part of the two Governments. There is force in this criticism. Anyone listening to the Chief Justice's address would have formed the impression that he was alleging that the Government had deliberately withheld from him details of the three provisions of the draft Constitution that gave him particular cause for concern. We would have been inclined to conclude that this was the subjective view of the Chief Justice at the time, but he did not, when giving evidence to the Tribunal, seek to defend his words on that

basis. He claimed to have been misunderstood. We are satisfied that the impression that his words created was deliberate.

- 129. The fourth criticism is that the address did not present a balanced view of the proposed Constitution or of the process by which it had been agreed. The Chief Justice himself accepted in evidence the force of this criticism. His speech was indeed thoroughly unbalanced. The draft Constitution was, as we will shortly explain, beneficial overall to Gibraltar. The Chief Justice represented it as if it had at its heart an attack on judicial independence.
- 130. Fifthly the Tribunal criticises the Chief Justice for failing to make plain in his address that he had been consulted about the composition of the Judicial Service Commission. That criticism is not valid. The Chief Justice stated that the Chief Minister had provided him with details of the proposed composition of the Judicial Service Commission and that the judiciary had made proposals in relation to this. The Tribunal also criticises the Chief Justice for not referring to the reservations of Mr Pitto and Mr Justice Dudley. These reservations related, however, to the proposal to petition the Privy Council, to which the Chief Justice did not refer in his address. This criticism is more appropriately directed to the letter of 11 August.
- 131. The sixth, last and most important criticism is that the Chief Justice's remarks were polemic in tone. This is coupled with the general criticism that the Chief Justice acted improperly in undertaking a public campaign with regard to part of the subject of the referendum. This calls for more detailed consideration.

132. The Chief Justice ended his speech as follows:

"The issues involved are fundamental and will affect future generations of Gibraltarians. It is for the Judiciary to protect its independence so that it may in turn protect the rule of law. As Head of the Judiciary I have a duty to ensure that the Constitution together with the assurances do indeed provide the necessary safeguards. I am therefore in the process of putting together a team of constitutional experts who will give an independent opinion on whether the Constitution does provide for an independent Judiciary given the recent communication from the UK and Gibraltar Governments. I shall make that opinion available to both Governments, to all members of the Gibraltar Bar and will also make it public. If there are still issues of concern I have advised the UK Government that I shall take up their offer of a meeting. If we do not agree on a way forward I shall seek the further advice of the expert team.

I am already some way towards putting the team together and I hope to have a detailed, joint, opinion in my hands by mid-November at the latest "

- 133. The Chief Justice consulted Sir Sydney Kentridge QC, Keir Starmer QC and Richard Tur. Their Opinion, provided in November, expressed "considerable reservations" about the provisions of Article 57, "both as to general constitutional principle and European Convention compliance". These related to the composition of the JSC and to the power given to the Governor under section 57(3). The advice none the less pointed out that the institution of a Judicial Service Commission was "a significant step towards ensuring judicial independence". Furthermore, the advice opened with the statement that the authors had "left out of account that the Draft Constitution is undoubtedly an advance on the existing Gibraltar Constitution and colonial constitutions generally."
- 134. The Chief Justice and Sir Sydney Kentridge attended a meeting at the FCO where the only concession that they achieved was an addition to the explanatory note to the Constitution stating that the executive powers of the JSC were "subject only to an exceptional power of veto by the Governor". This did no more than repeat the undertaking already given to the Bar Council. The Chief Justice took no further action and the new Constitution was approved in a referendum on 30 November.
- 135. The new Constitution, laboriously negotiated between the United Kingdom and Gibraltar, represented a significant advance for the independence of Gibraltar. The Chief Justice's criticisms of aspects of the Constitution that advanced judicial independence but did not, in his view, go far enough were criticisms that he could fairly advance. They were, however, relatively insignificant in the context of the Constitution as a whole. In his speech the Chief Justice presented the draft Constitution as if it were a threat to the independence of the judiciary rather than a significant advance of this. The fact that the advance did not go as far as he wished could not justify action that might derail, or assist in derailing, the approval of the new Constitution on the forthcoming referendum. The Chief Justice's address carried that risk. When giving evidence the Chief Justice confirmed that he was conscious that a referendum was in the offing. He said that he was not trying to influence votes but to influence the Constitution. Later he added "I did not seriously consider that this would derail the referendum".
- 136. We accept that the Chief Justice's actions were motivated by the praiseworthy aim of promoting judicial independence. But to attempt to achieve this by a public and unbalanced attack on the draft Constitution shortly before the referendum showed a serious lack of judgment. The Chief Justice accepted that, with hindsight, his actions might be open to criticism. The problem with his conduct was that he did not exercise the foresight required of someone in his position. It may be open to question whether his headstrong action amounted to misbehaviour. It certainly demonstrated an inability to pay due regard to the likely political consequences of his actions.

- 137. This is dealt with in paragraphs 5.1 to 5.35 of the Tribunal's Report. It marks the start of the most critical part of the case against the Chief Justice. At the heart of this case is the relationship between the Chief Justice and his wife, the extent to which he was party to her actions and the extent to which he was reasonably perceived, and should have appreciated that he would be reasonably perceived, as being party to her actions.
- 138. We have already commented on the fact that the Chief Justice and his wife were and are clearly devoted to one another. Her actions, throughout the story, misguided as they frequently were, demonstrate a passionate concern for her husband and his position. Passages in his evidence disclose devotion to and support for his wife. The Chief Justice did not have a policy of keeping any matters confidential from his wife. He shared with her an email address. He also shared with her his own concerns. We would not suggest that a man in his position should not share matters that are confidential with his wife. If he does so, however, he implicitly undertakes responsibility for the maintenance by her of that confidence.
- 139. In the course of introducing the 2006 Order the Government announced its intention of creating a Minister of Justice and passing a Judicial Service Act. On 20 February the Chief Minister wrote to the Chief Justice recording that at a dinner that had been given for judges of the Court of Appeal the previous week the Chief Justice had told him that he would be going on holiday in Argentina with his daughter at the end of the following week (24 February) and that the Chief Minister had promised to try to get the draft Judicial Service Bill to him before he left. The Chief Minister enclosed the draft Bill. He stated that the Government intended to publish the Bill the following week, which would be the beginning of a six week consultation period, so that the Chief Justice would have more time to respond on his return from holiday before the Bill went to Parliament. The letter ended:

"Finally, since no other consultee will yet have seen the draft Bill, I must ask that you treat it confidentially and 'for your eyes only' until the consultation paper issues next week".

- 140. In evidence the Chief Justice confirmed that the letter also enclosed the Consultation Paper, although it is possible that he received this a little later. It was addressed to "the President and Justices of the Court of Appeal, the Chief Justice and other judges of the Supreme Court, the Stipendiary Magistrate, the Chairman of the Justices, the Chairman of the Bar Council (for consultation with him and the Bar Council) and the Attorney General."
- 141. The draft Bill very largely reflected, as the Consultation Paper pointed out, provisions of the British Constitutional Reform Act 2005. This was true of Clause 5, which provided a mechanism under which the President of the Court of Appeal and the Chief Justice could

have, what the Consultation Paper described as "an appropriate and dignified method of making [the views of the judiciary] known to the Parliament, and through the Parliament, to the public at large. Clause 6 provided:

"President of the Courts of Gibraltar

- (1) The President of the Court of Appeal holds the office President of the Courts of Gibraltar.
- (2) As President of the Courts of Gibraltar he has overall responsibility—
- (a) for representing the views of the judiciary of Gibraltar to Parliament, to the Minister and to the Government generally;
- (b) for the maintenance of appropriate arrangements for the welfare, training and guidance of the judiciary of Gibraltar within the resources made available by the Government;
- (c) for the maintenance of appropriate arrangements for the allocation of work within courts.
- (3) The President of the Courts of Gibraltar is president of the courts listed in subsection (4) and is entitled to sit in any of these courts.
- (4) The courts are –
- (a) The Court of Appeal
- (b) The Supreme Court
- (c) The Magistrates' Court
- (d) The Coroner Court"

As to these provisions, the Consultation Paper stated:

"Although the Court of Appeal is an itinerant court, it is nevertheless a Gibraltar Court and its President is therefore the most senior member of Gibraltar's own judiciary. The Government no longer considers it appropriate, in the context of the new constitution, for the Court of Appeal to 'look like' an external court (even though it has never actually been that). Accordingly, Section 6 establishes the President of the Court of Appeal as the President of the Courts of Gibraltar."

- 142. The Chief Justice said in evidence that he was horrified at the implication of Clause 6 for the administration of justice. It is plain that he was also horrified at the implication of the Clause for his own position. This is not surprising. The Chief Justice had regarded himself as head of the judiciary of Gibraltar, although that position had received no formal recognition. We consider that he was reasonable to do so, for the leadership role of that position was one more naturally performed by the senior member of the local judiciary rather than the President of a Court of Appeal that only visited Gibraltar from time to time.
- 143. The Chief Justice and his wife formed the view that Clause 6 of the Bill was directed at him personally and was intended to drive him from office by forcing his resignation. By the end of the hearing before the Tribunal this was no longer his position, nor was it before the Committee. We consider that the Chief Justice and his wife could reasonably have concluded that Clause 6 was motivated, in part, by dissatisfaction with aspects of his own conduct in the office of Chief Justice but not that it was deliberately aimed at him, or at forcing his resignation.
- 144. The Chief Justice wrote to the Chief Minister on 21 February stating that the provisions which put the present functions of the Chief Justice in the hands of a visiting President of the Court of Appeal were of particular concern. He stated that he felt that he must discuss the Draft with the members of the Court of Appeal while they were in Gibraltar that week. This was a breach of confidence that he should not have committed. The proper course would have been for the Chief Justice to ask the Chief Minister to agree to his doing this. In the event his action forced the Chief Minister's hand and he immediately distributed the consultation paper to all the consultees.
- 145. On 23 February the Chief Justice got the Registrar to send a copy of the draft Bill to every member of the Bar, together with a statement that he felt that it might have wider constitutional implications and inviting the views of the recipients. The Tribunal inferred that this was a further breach of confidence. We are not of that view. The Chief Minister had asked the Chief Justice to treat his copy as confidential until the start of the consultation. His initial breach of confidence had caused the Chief Minister to advance the start of consultation, thereby releasing the Chief Justice from the duty of confidentiality. The Tribunal found that the Chief Justice's conduct in by-passing the Bar Council was not merely high-handed but deliberately provocative. We would agree that it was high handed and provocative, but does not feel justified in finding that the latter was deliberate. It was inappropriate and ill-judged, as was demonstrated by the fact that only two members of the Bar responded to his request for their views. The Chief Justice had, however, been indirectly responsible for a much more reprehensible publication of the Government's proposals.
- 146. On 22 February Mrs Schofield went to see Mr Charles Gomez, a barrister with a sole practice, taking with her a copy of the draft bill. According to his witness statement she asked him to advise on its implications so far as she and her family were concerned. On the following day *Vox* published an article under the heading "Now Caruana Attempts to Twist

the Law". It purported to be based on a "leaked draft" of the Judicial Service Bill. It set out clause 6 in the context of a description of the Chief Justice's receipt of the "leaked draft" and his reaction to it. It reported that Charles Gomez had confirmed that he was "looking to initiate proceedings to declare the act invalid after being consulted by a concerned citizen". The article alleged that the draft Bill had been deliberately disclosed to the Chief Justice just when he was about to go on holiday.

- 147. The Tribunal found that the "leaked draft" that Mrs Schofield had taken to Mr Gomez was the draft that had been sent to the Chief Justice and that Mrs Schofield and Mr Gomez were together responsible for the article in *Vox*. The Chief Justice did not challenge these conclusions. He said, however, that he had only showed his wife clause 6 of the Bill, that he did not know how she had come by a copy, that, although he had become aware, perhaps before he left for Argentina, that she had been to see Mr Gomez, he had not expected her to go public, that he might have "skim read" the article in *Vox* before he left for Argentina but that he did not discuss with his wife whether she had provided *Vox* with a copy of the draft Bill.
- 148. All of this evidence the Tribunal rejected. It found that the Chief Justice had shown the draft Bill and the consultation paper to his wife as soon as he received them, that he left the draft Bill in their house where it was available to her, that he expected that she would "go public" and was at the very least content that she should do so and supportive of her in that respect and that he discussed the article in Vox with her before he left for Argentina.
- 149. The Tribunal based this last finding on the "overwhelming likelihood" of the Chief Justice having behaved in this way and stated that it found the Chief Justice's evidence on each of these matters "deliberately evasive". We have read the relevant part of the transcript of the evidence of the Chief Justice, and share the impression that his evidence made on the Tribunal. We also consider that the Tribunal's findings accord with the overwhelming likelihood of the situation. Both the Chief Justice and his wife were understandably deeply concerned at the provisions of clause 6. The Chief Justice accepts that they discussed this. It was only to be expected that they would also discuss what, if any, action to take in the face of it. No reason has been advanced as to why Mrs Schofield should conceal from her husband the actions that she proposed to take. Her witness statements are significantly silent as to what transpired at this stage of the story.
- 150. This was a discreditable episode. The copy of the Bill that the Chief Minister had disclosed to the Chief Justice in confidence in advance of the consultation period had been used to provide copy for an attack by *Vox* on the Government's proposals. The Chief Justice bore responsibility for permitting this to occur.

151. While the Chief Justice was on holiday in Argentina his wife entered into e-mail correspondence with Mr James Neish, QC, the Chairman of the Bar. On 25 February she sent him a lengthy e-mail making recommendations "as a member of the public and a person who may be affected by any constitutional and contractual implications of the draft Bill". They included suggestions as to what the Bar Council should do before meeting to discuss the draft Bill. These included that members disclose "personal, political and business relationship[s] with the Government of Gibraltar, Minister for Justice or any other member of the Gibraltar Government" and whether they had discussed the proposed Bill prior to it being referred to the Bar Council for consultation. The matters that she suggested the Council should consider when it did meet to discuss the draft Bill ended with:

"Whether the Chief Minister in including these provisions is demoting, demeaning, harassing, the justice for statements or decisions that he may have made in the performance in his role as Chief Justice about the Chief Minister and whether this is an abuse of office and or interference with the Chief Justice."

152. The e-mail ended with a demand for a reply by the end of the following day, failing which Mrs Schofield would go public. She ended by alleging that the section [implicitly 6]:

"to my mind is an 'attempted rape' of the Gibraltar Constitution and of the Chief Justice's office and contract. In my view it is intended to force a resignation of the CJ unless he accepts a demotion or to force him to sue in which case we shall hear calls for him to resign."

- 153. Mr Neish made a short and restrained reply on 27 February. At about this time Mrs Schofield went to visit her family in Kenya. While she was away *Vox* carried on 2 March an article that reported the Chief Justice's personal consultation of members of the Bar under a headline "Schofield bypasses Bar in legal clash". It also reported Mrs Schofield's letter to the Bar Council acting "in my capacity as a private individual". The Tribunal accepted that Mrs Schofield was not responsible for providing *Vox* with this information.
- 154. The Chief Justice returned from holiday on 15 March, by which time his wife had returned from Kenya. He accepted that on the day of his return his wife, probably in the late afternoon, told him the gist of the e-mails that she had sent to Mr Neish. He said that she told him that Mr Neish had sent a bland reply. When asked for details of the conversation he said repeatedly that he could not remember. He said that he took very little interest in the matter and expected it to blow over. The Tribunal found the Chief Justice's evidence on this "wholly unconvincing" and "frankly incredible". It found that he acquainted himself fully with the terms of her correspondence, that he did not expect the matter to blow over but that he was fully aware that his wife intended to pursue the matter as vigorously as she could, which indeed she did.

- 155. On the evening of 15 March Mrs Schofield sent a further e-mail to Mr Neish in aggressive terms alleging that he and/or members of his firm had been involved in the drafting of the Bill and alleging that this gave rise to a conflict of interest. The Chief Justice said that he recalled his wife telling him that she had received this information from a local lawyer, but that he could not remember when this was. He said that he was surprised that she should have been typing e-mails on the evening of his return from Argentina and could not remember her sending this e-mail.
- 156. We consider it incredible that, after discussing with her husband her e-mail correspondence with the Chairman of the Bar, Mrs Schofield would have sent this e-mail without discussing her intention to do so with her husband. We share the Tribunal's conclusion that the likelihood is that he would have read the earlier e-mail correspondence at this point. What is more significant, however, is that he did not seek to dissuade her from sending the e-mail that she sent that evening.
- 157. The Bar Council met to consider the e-mails that had been sent by Mrs Schofield. As a result of their deliberations Mr Neish sent the three letters of 3 April that are quoted in detail in the Tribunal's Report. The first letter was to Mrs Schofield protesting in strong terms at her conduct and including the following paragraph

"This is unprecedented conduct in Gibraltar by the wife of a Chief Justice. Notwithstanding the ludicrous fiction under which you have sought to interfere, i.e. 'as a member of the public', the inescapable fact is that you are the wife of the Chief Justice acting in your common interests. Your actions cannot be dissociated from the Chief Justice and, I regret to say, impact upon his position. This is an issue which will be addressed separately."

158. The second letter was to the Attorney General, inclosing the previous correspondence with Mrs Schofield, including his letter to her of 3 April. This stated:

"As the Chief Justice's wife whatever Mrs Schofield has written or done is liable to be construed as having the express or implied approval or knowledge of the Chief Justice or as expressing their common views. The Chief Justice has not distanced himself from Mrs Schofield's e-mails or their contents."

159. The third letter was to the Chief Justice, informing him of the letter that had been sent to the Attorney General. It ended

"Mrs Schofield is not an ordinary member of the public – she is the Chief Justice's wife. Her e-mails were therefore liable to be construed – as in fact they have been – as enjoying your express or implied approval or knowledge or as reflecting your and Mrs Schofield's common views. We note that you have not distanced yourself from those e-mails."

It was at this point that the Chief Justice said that he first gave detailed consideration to the earlier e-mail correspondence.

- 160. It was put to the Chief Justice that the last sentence in this letter was an implicit invitation to distance himself from the statements in his wife's e-mails to the Bar Council. He accepted that these and subsequent e-mails sent by his wife could not properly or appropriately have been sent by him, but denied that he would be associated with them. He said that his wife was known to the Bar and well known in Gibraltar as an independent-minded person. He did not consider that public opinion would necessarily think that her views were his. He said that he did not consider that the last sentence in Mr Neish's letter to him was an implicit invitation to dissociate himself from his wife's e-mails. It was simply a statement of fact.
- 161. The Tribunal was highly critical of this passage of the Chief Justice's evidence, rejecting it as untrue. It found that it must have been clear to any reader, and most especially to the Chief Justice, that the Bar Council considered that he should have distanced himself from his wife's e-mails, that by not doing so he had provided tacit support for the statements made in them and that, by the final sentence of the letter the Bar Council was inviting him to remedy the situation by making a statement distancing himself from the statements. The Tribunal found it "extraordinary" that he did not do so and concluded that he knew that his silence would almost certainly be interpreted as support for his wife's views and was more than content that should be so.
- 162. We do not agree that the last sentence in Mr Neish's letter was an implicit invitation to the Chief Justice to distance himself from his wife's conduct. Rather it implicitly accused him, in the light of his failure to do so, of approving her conduct and sharing her views. It expressly spelt out to the Chief Justice that these conclusions were a consequence of his silence. The reality was that the Chief Justice did largely share his wife's views. He believed that the Chief Minister was trying to get rid of him as Chief Justice. He also believed that some members of the Bar Council, and in particular Mr Neish "as a good friend and ally" of the Chief Minister, were "partisan". At no stage did the Chief Justice suggest that he attempted to dissuade his wife from making the statements that she did. In the absence of any

statement that he should not be associated with them the natural conclusion that was drawn was that he shared her views.

163. The same is true of his wife's subsequent communications at this time. These included her e-mail to Mr Peter Schirmer of *Vox* of 4 April, which she gave him express permission to publish. This made the following allegations, which were among those quoted by *Vox* in an article published on 13 April.

"I consider the action by the Bar Council an attempt to harass me into silence. In a matter where I assert that the Chief Minister now clothed as Minister for Justice has been trying to get rid of the Chief Justice, the use of consultation and legislation to legitimise this long held ambition is an attempted rape of the Constitution and the Chief Justice."

164. We consider that the Tribunal was right not to accept the Chief Justice's evidence that he was unaware of the e-mail sent by his wife to Mr Neish on 5 April in answer to his letter of 3 April. On his own evidence the Chief Justice asked for copies of his wife's e-mail correspondence when he received Mr Neish's letter of 3 April. His wife was plainly incensed by the letter that she received from Mr Neish on the same day. How could they possibly have failed to discuss how she was going to respond? In the event her letter included the allegation that Mr Neish and other members of the Bar Council were guilty of "conduct unbecoming" in failing to recuse themselves from taking a part in making decisions in relation to the Judicial Service Bill. It also alleged that no Chief Justice had been put through what the Chief Justice had been put through since the Chief Minister came to power.

165. On 25 April Mrs Schofield sent a fax to the International Commission of Jurists in Kenya, with copies to the British Foreign Secretary, the Chief Minister and the Leader of the Opposition. The letter set out a list of events dating back to 1998 as evidencing the intention of the Chief Minister to get rid of the Chief Justice. These included the Maids Issue and the MOT prosecution. This letter immediately entered the public domain, for it was the subject of an article in the *Gibraltar Chronicle* on the following day under the headline "Why the Chief Minister wants to get rid of the Chief Justice" Mrs Schofield tells Kenya Jurists Committee'. We are satisfied that the Chief Justice was content that his wife should make these allegations. Indeed it is significant that in his evidence to the Tribunal he never suggested that he at any stage disapproved of the allegations made by his wife, remonstrated with her for making them or sought to dissuade her from making similar allegations in the future.

- 166. This action and its repercussions are covered by paragraphs 5.37 to 5.93 of the Tribunal's Report.
- 167. On 10 April, on Mrs Schofield's instructions, Mr Gomez wrote a letter addressed to "the General Council of the Bar in Gibraltar" alleging that Mr Neish's letter of 3 April was "a very serious defamation" in as much as it accused her of "interfering in the affairs of the Bar with the intention of inhibiting the Bar's freedom of expression and undermining the independence of the Bar". The claim called for an undertaking on behalf of the Council members not to repeat the allegations, failing which he had "firm instructions to apply for an injunction". Mr Neish replied on 12 April, alleging that it was Mrs Schofield who had made defamatory statements about members of the Bar Council and refusing to give an undertaking.
- 168. On 13 April Mrs Schofield issued defamation proceedings, naming Mr Neish as sole defendant. The claim alleged that Mr Neish's letter of 3 April meant that her communications with the Bar Council were a dishonest and cynical manoeuvre and that she was motivated exclusively by a desire to protect her personal interests by inhibiting the Bar's freedom of expression and undermining its independence.
- The Chief Justice stated in evidence that he could not recall any conversation with his wife about her intention to issue libel proceedings though she might have told him that she had sent a letter before action. He said that it would not have been appropriate for him to have dissociated himself from the libel action or to have interfered with the exercise of her legal rights. The Tribunal did not accept that the Chief Justice was unaware of his wife's intention to issue defamation proceedings nor do we. The Tribunal was prepared to accept that the Chief Justice did not know the detailed nature of those proceedings until after they had been issued. We consider that had the Chief Justice exercised reasonable prudence he would have found out precisely what action his wife was minded to take, for it should have been obvious to him that if she commenced a libel action against the Chairman of the Bar in relation to his letter it would have the potential to make his position very difficult, as indeed it did. Had he discovered the details of the claim his wife proposed to make he should have sought to dissuade her, both out of consideration for her position and for his own. The accusations that she had made in her e-mails had been intemperate and inappropriate and the Chief Justice had, not unreasonably, been associated with them. Mr Neish's response had been robust, but to base a claim for libel on it was asking for trouble. And trouble there was.
- 170. The Bar Council had agreed that if Mrs Schofield commenced the threatened libel action, Mr Neish would ask the Chief Justice to recuse himself from hearing any matters involving members of the Bar Council or their firms. One of those firms was Hassans. This

was on the basis that they could not properly appear before the Chief Justice when his wife had issued proceedings against them with the approval of her husband. Although, in the event, Mr Neish was named in the action as sole defendant, the Bar Council considered that, in accordance with Mr Gomez's letter before action, the action was aimed against them as a whole. They concluded that clients on whose behalf they were acting could not be sure of having a fair hearing before the Chief Justice.

- 171. On the morning of 16 April 2007 a case management conference was listed before the Chief Justice in the case of *Sonia Bossino v Attorney General*. Hassans were acting for the claimant. At their request Mr Neish appeared at the hearing. He informed the Chief Justice of the decision of the Bar Council and asked him to recuse himself for the reasons set out by the Tribunal in its Report. The Chief Justice adjourned the application.
- 172. What then transpired is set out in detail in the Tribunal's Report. It is a confused and confusing picture. The background is (i) information received by Mrs Schofield from Hassans that on the following day, 17 April, a vote of no confidence in the Chief Justice was to be moved at the AGM of the Gibraltar Bar; (ii) the conclusion reached by Mrs Schofield and Mr Gomez that both the recusal application and the proposed resolution at the AGM were intended to undermine Mrs Schofield's defamation claim. The latter conclusion caused Mrs Schofield and Mr Gomez to request a hearing in chambers before the Chief Justice on the afternoon of 16 April. The Tribunal found that the Chief Justice deliberately omitted from his first witness statement the fact that his wife had telephoned him to inform him of the possibility of this request. Mr Beloff drew attention to the fact that this conversation featured in his wife's second statement and comments that this indicates either that he and his wife had no intention of concealing this information or is powerful evidence that they act independently. We see the force of the former point and is not satisfied that the Chief Justice deliberately concealed the fact of the telephone conversation with his wife.
- The Chief Justice sat on the afternoon of 16 April on a chambers hearing, convened at the request of Mr Gomez on behalf of Mrs Schofield at which Mr Gomez and his wife were present. The basic facts in relation to that hearing set out in paragraph 5.73 of the Tribunal's report are not challenged. The Tribunal concluded that the Chief Justice's conduct demonstrated "a reckless disregard for the requirements and reputation of the office of Chief Justice" and constituted "judicial misconduct of the most serious kind". We consider that these findings are over severe. The Chief Justice should not have agreed to an ex parte hearing when the party in question was his wife. The capacities in which she and Mr Gomez appeared were as claimant and counsel in her libel action. Having wrongly entertained the hearing, the Chief Justice should have ascertained this at the outset and, having done so, permitted the hearing to go no further. That said, he granted no relief and made no order. His actions showed lack of judgment of a high order and amounted to judicial misconduct. It was not, however, judicial misconduct of the most serious kind. Nor would we conclude that the Chief Justice was motivated by "the need (as he saw it) for Mrs Schofield's position as a claimant in her libel action to be protected" (5.64). His conduct was bred of muddle, not design.

174. On 3 May Triay & Triay, who acted for Mr Neish in the libel case, wrote to the Deputy Registrar, a letter "Re: Schofield v Neish" asking for a copy of the Judge's note of the application in that case that had taken place on 16 April. On instructions from the judge Mr Mendez wrote back on 4 May saying that no notes were taken and that the matter which was brought before him was not *Schofield v Neish* but the *Bossino* recusal application. The Chief Justice's assertion that the hearing had been in the *Bossino* matter appears to us to have been a piece of *ex post facto* rationalisation in an attempt at damage limitation. We do not, however, share the Tribunal's view that his reply that no note was taken was highly misleading or that his statement to the Tribunal that he thought that Triay & Triay's request related to notes taken by him was an "afterthought". His understanding of the situation was, in fact, correct, as demonstrated by Triay & Triay's letter of 9 May expressing surprise that the judge should not have taken notes and asking for a copy of the note taken by the court clerk.

175. In the event a vote of no confidence was not passed on 17 April. Instead the Signatories first Memorandum was sent to the Governor.

176. The general recusal application that had been made in *Bossino* was restored on 24 April and then withdrawn in the circumstances described in the Tribunal's report. The Chief Justice permitted Mr Gomez to be present to represent his wife. Mr Gomez explained that he was there because of concern that the "flurry of applications" was meant to put pressure on Mrs Schofield in her libel action. The Tribunal found that in treating Mrs Schofield as an interested party the Chief Justice was, "to put it no higher", giving a degree of judicial credence to Mr Gomez's submission that the purpose of the recusal application was to stifle the libel action and that he should not have done this. Rather he should have recognised that his wife's continuing involvement in the application was yet another reason for his recusal and that his failure to recuse himself could only have inflicted further damage on his office of Chief Justice.

177. We do not draw the same conclusions from this episode. The hearing was in open court and it was on that basis that the Chief Justice said that it was open to Mr Gomez to remain, albeit that he recognised the possibility that Mr Gomez might seek to play some part in the proceedings. We cannot see that Mr Gomez had any legitimate part to play in the recusal proceedings and it would have been better had the Chief Justice so held. It did not follow, however, that he accepted Mr Gomez assertion that the recusal proceedings were designed to put pressure on his wife. Nor do we think that the Chief Justice should have recused himself in the *Bossino* case. No application was made that he should do so. He came prepared to hear the general recusal application, which was not pursued.

178. On 21 May the Signatories submitted to the Governor their Second Memorandum, setting out in detail their reasons for asserting that they had lost confidence in the ability of the Chief Justice to discharge his functions and that his continuance in office would cause prejudice to the administration of justice and to the reputation and image of Gibraltar. A copy of this was sent to the Chief Justice. This document ran to 20 pages. Most of the individual matters relied upon subsequently featured in the Statement of Issues, were considered by the Tribunal and have been considered by the Committee in its turn. There was one recurrent theme. Mrs Schofield had repeatedly acted in a reprehensible manner in support of the Chief Justice and the Chief Justice had done nothing to dissociate himself from her conduct, leading to the inference that it had his approval.

The recusal application of 22 May

179. The Tribunal deal with this at 5.94 to 5.102 of their Report. The application was made by counsel for the claimant company, which was wholly owned by the Government. It was that the Chief Justice should recuse himself on the ground of apparent bias resulting from certain public statements by his wife from which he had not dissociated himself. The Chief Justice adjourned the application so that he could seek the assistance of an *amicus curiae*, but in doing so stated that there was an interested party who had a right to be heard and who should be notified. The Tribunal observed that there was no possible ground on which Mrs Schofield could be treated as an interested party in this application and that the Chief Justice's failure to appreciate this was another example of his inability to comprehend the constraints and responsibilities of his office, in particular where Mrs Schofield was concerned. We consider that that observation was well founded.

180. Before the date for the hearing of the adjourned application the Chief Justice was suspended from office.

The cancellation of the opening of the legal year 2007 to 2008.

181. This episode is dealt with in paragraphs 6.1 to 6.8 of the Tribunal's Report. On 11 July Mr Mendez, then the Acting Registrar, received instructions from the Chief Justice that he understood to mean that he would no longer be holding a Ceremonial Opening of the Legal Year. He made an announcement to that effect. There was an issue before the Tribunal as to whether the Chief Justice had intended his instructions to relate solely to the forthcoming ceremony or to indicate the abolition of the ceremony for all time. The Tribunal resolved this issue against the Chief Justice. We do not consider that there was a firm basis for doing so, nor that the issue is of significance. Nor do we consider that there was a firm

basis for the Tribunal's conclusion that the Chief Justice's decision was motivated by concern not to afford those whom he perceived as hostile to him the chance to criticise his conduct or challenge his views. Our conclusion is that the Chief Justice's conduct was probably motivated by a desire to mark his objection to being deprived of the position of head of the judiciary. The ceremony was a well established part of Gibraltar's traditions and the Chief Justice must have anticipated that his action, even if restricted only to the forthcoming ceremony, would cause concern and dismay in Gibraltar. If he had any concern as to whether it would be appropriate for him to conduct the ceremony, this was a matter that he could and should have discussed with the President, Sir Murray Stuart-Smith, who in the event took over responsibility for the ceremony. The cancellation occupied the front page of the Gibraltar Chronicle on 12 July and provoked a leader headed "Justice is not one man's property". Once again the Chief Justice showed a lack of judgment in relation to his public conduct in a respect that was damaging to his office. The Tribunal was justified in describing it as a "deliberately provocative act which only served to exacerbate existing tensions".

Mrs Schofield's complaint against Freddie Vasquez QC.

182. The Tribunal deals with this in paragraphs 6.9 to 6.14 of their Report. Mrs Schofield made an unfounded disciplinary complaint against Mr Vasquez in relation to a letter that he wrote to the *Gibraltar Chronicle*, which was published on 14 August 2007. She complained among other matters that this letter sought to undermine the Chief Justice in the eyes of the public. This is but one example of an ill-advised attempt by Mrs Schofield to fight her husband's battles. The Tribunal did not find that the Chief Justice was complicit in his wife's action, albeit that there would have been a natural inference that he approved it. We agree with Mr Beloff that this was not an incident of overall significance.

The Chief Justice's proceedings for judicial review

- 183. The Tribunal deal with this at pages 6.15 to 6.36 of their Report. The background to this part of the story is as follows. The Chief Justice was not the only person who had had criticisms to make of clause 6 of the Judicial Service Bill. Thus:
 - i) On 5 March 2007 Sir Paul Kennedy wrote to the Legal Secretary to the Chief Minister giving the view of the members of the Court of Appeal. These were that, while there was no objection to the President of the Court of Appeal being given the office of President of the Courts of Gibraltar and overall responsibility for those courts, as he was not permanently resident in Gibraltar the Bill should make it clear that "the *direct* responsibility for the day to day discharge of the duties set out in clause 6(2)(b) and (c) of the Bill (as opposed to the *overall* responsibility) lies with the Chief Justice".
 - ii) On 23 March Mr Justice Dudley wrote making similar points, but also questioning whether it was practical or desirable for the President rather than

the Chief Justice to be the titular head of the courts. He suggested that the powers in 6(2)(a)(b) and (c) should be vested in the Chief Justice. On April 2 Sir Murray Stuart-Smith wrote on behalf of the Court of Appeal endorsing his comments in relation to 6(2)(a).

- iii) On 23 March the President of the Gibraltar Magistrates Association wrote saying that on occasion the Association required prompt access to the Head of the Judiciary and expressing concern at the practicability of accessing the President at short notice.
- iv) On 26 March the Governor wrote pointing out that it was likely to prove necessary for certain administrative functions to be carried out by the resident judiciary and suggesting that the Act should make express provision for this.
- v) On 30 March the Bar Council wrote:
- "(i) Whilst recognising that the present incumbent of the Office of President of the Court of Appeal is the most senior judge in our 'judicial system' and notwithstanding the advances in telecommunications systems the Bar Council is not persuaded that it is practicable for the President of the Court of Appeal to be de facto President of the Courts of Gibraltar. Accordingly the Council does not support Clause 6 (2) of the draft Judicial Service Act.
- (ii) Section 6 (3) may contravene or alternatively be inconsistent with the provisions of section 60 (2), 62 and 64 (1) of the Constitution."
- 184. The Government subsequently responded to these comments by amending 6(3) of the draft Bill to read:
 - "(3) As President of the Courts of Gibraltar he has overall responsibility—
 - (a) for representing the views of the judiciary of Gibraltar to Parliament, to the Minister and to the Government generally;
 - (b) for the maintenance of appropriate arrangements for the welfare, training and guidance of the judiciary of Gibraltar within the resources made available by the Government;
 - (c) for the maintenance of appropriate arrangements for the allocation of work within courts.

(4) Subject to subsection (3), for the Supreme Court and all lower courts the Chief Justice shall have direct day to day responsibility for the mattes set out in subparagraphs (b) and (c) of subsection (3)."

The Act containing this wording came into force on July 5.

- 185. Meanwhile the Chief Justice had sought the advice of Rabinder Singh QC and Alex Bailin on whether the proposed Act would be *ultra vires*, whether action challenging it could be brought in the United Kingdom (as he preferred) and whether he had a claim for constructive dismissal, on the basis that, as stated in the Instructions, 'Since 1998, there have been major disagreements between Derek and the Chief Minster who has been trying to get rid of him, so far without success'. On 28 March they provided a written opinion on the draft Bill. They advised that section 6 would be unconstitutional since 'it undermines the core constitutional principle of the independence of the judiciary, which itself forms part of the rule of law'. An amendment to provide that for the Supreme Court and lower courts the Chief Justice should have direct responsibility for the matters specified in subsection (2)(b) and (c) would in their opinion create 'an artificial and unworkable division of functions'. They considered that the independence of the judiciary, enshrined in the Constitution, required that the President of the Courts should have a degree of security of tenure which the Court of Appeal judges did not have, and the Chief Justice did.
- 186. On 30 August 2007 the Chief Justice filed an application for judicial review as a claimant in his own court seeking a declaration that sections 6 and 37(3) of the Judicial Service Act were ultra vires the Constitution. The grounds for the claim attached to the claim form ended:
 - "4.9 It is submitted that to demote the office of Chief Justice, and to appoint the President of the Court of Appeal as President of the Courts of Gibraltar, is to undermine the principle of Judicial Independence in Gibraltar, and to imperil the rights guaranteed in section 8 (1) and 8 (8) of the Constitution. This is because appointments to the Court of Appeal are short term and renewable. Renewal is on the recommendation of the Judicial Service Commission, whose members are appointed in such a way as not to offer any guarantee of their independence.
 - 4.10 It is not uncommon for the head of the judiciary not to sit habitually in the highest court of the jurisdiction in question. See the position of the Lord Chief Justice in England and Wales, the Lord President in Scotland, the Lord Chief Justice of Northern Ireland. See also the position of the Chief Justice of New Zealand before the abolition of appeal to Her Majesty in Council from that Jurisdiction.
 - 4.11 In contrast, the claiming is unaware of any jurisdiction in which full-time professional judge resides, but in which the

head of the judiciary is a short-term, part-time judge residing outside the jurisdiction."

The claim was withdrawn on 13 December on the ground that the Chief Justice could not cope simultaneously with judicial review proceedings and proceedings in relation to his own suspension.

187. The Chief Justice filed a statement in support of his application for Judicial Review which included the following allegations:

"Personal Attempts to remove me from office

- In the course of late 1998 and early 1999 I was called to a number of meetings with the then Governor Sir Richard Luce. Sir Richard asked me to consider my position as Chief Justice, and ultimately suggested that I accept a six month warrant of appointment. When I pressed him as to the reason for his suggestions he told me that the Chief Minister had made representations that my 'contract' be not renewed. I made it clear to the Governor that I had security of tenure under the Constitution and would stay in post as per the Constitution.
- In 1999, at the ceremonial opening of the legal year, I spoke publicly of instances in which the delay or denial of funds by the Government had had the potential adversely to affect the administration of justice. A copy of my address is exhibited hereto as exhibit 7.
- I referred to two particular incidents. Firstly, I had sought funding to attend a Judicial Studies Board seminar on the Woolf reforms, which were to be introduced into Gibraltar. Gibraltar has no equivalent of the Judicial Studies Board. Funding for this was refused by the Government, although the UK government eventually provided funding.
- 17 Secondly, I had initiated discussions with the Government to institute a system of part-time acting stipendiary magistrates, similar to recorders or deputy district judges in England & Wales. The benefit of this would have been twofold. It would have assisted in reducing backlog in the magistrates' court. It would also have provided a pool of local practitioners with judicial experience, which would have been valuable when making future judicial appointments. The Government were willing to agree to this, provided that they had some say in who was appointed. I considered this to be unacceptable.
- In February 2002, whilst was sitting in a criminal trial, I was presented by the Governor with a warrant purporting to

appoint me Chief Justice for 1 year. I asked counsel in the criminal trial to consider the position. My view was that, if the warrant did limit my appointment to 1 year, I would not be an independent tribunal as required by the 1969 Constitution then in force.

- Counsel for 2 of the defendants (who is also one of my counsel in this matter) made representations that, pursuant to the 1969 Constitution, a Chief Justice appointed held office until the age of 67, or until removed by the Constitutionally established procedure. Counsel for the other defendants (a member of the same firm) adopted these representations. Counsel for the prosecution (a member of the Attorney-General's chambers) did not address the Court in any meaningful way.
- I received no explanation from the Governor as to the powers he considered he had to circumvent my tenure of office.
- I had expressly asked the Attorney General to attend the Court and address me on the warrant. He did not do so, nor did he provide any explanation for why he did not do so.
- On 04.09.2002, Sir Desmond de Silva QC wrote to me, to record that he had received an approach from the Attorney General, indicating that I would be assisted in finding judicial employment elsewhere if I were to leave Gibraltar. A copy of Sir Desmond's letter is exhibited hereto as exhibit 8.
- I considered and consider that I would be untrue to my oath if I were to accept such an inducement or bow to such pressure."
- 188. The Tribunal were critical of the Chief Justice in relation to a number of aspects of his application for judicial review. First they found that, so far as this stage of the story was concerned, there was evidence that the Chief Justice and his wife were working together in relation to the legal proceedings that each had initiated. Secondly they criticised the Chief Justice for the allegations against the Chief Minister and the Attorney that he chose to make in his statement in support of his application in that the allegations were neither well founded nor relevant to the application. Thirdly they questioned the justification for bringing judicial review proceedings at all.
- 189. So far as the first point is concerned, there could be no objection to the Chief Justice and his wife assisting one another if the endeavours to which their mutual assistance related were not open to criticism. Indeed it would be strange if they were not helping one another. Nevertheless his wife stated in her witness statement that her husband had no involvement in her defamation proceedings and the Chief Justice confirmed this. Equally he said that she

had no hand in his judicial review application. So far as the latter is concerned the Tribunal pointed out that papers disclosed by the Chief Justice's solicitors included an e-mail from his wife to them which expressed her views on some legal questions with which the action was concerned. Faced with this the Chief Justice accepted that she was "putting her fingers in the pie" but said that this was to be discouraged. So far as the former is concerned, the Tribunal drew attention to the strong similarity between the application by the Chief Justice and orders for which Mrs Schofield applied on 17 July in her libel action. The Chief Justice sought to explain this by saying that he and his wife, both lawyers, were likely to articulate matters in the same way.

- 190. The Tribunal did not find this explanation acceptable and nor do we. Mrs Schofield gave notice that she intended to apply for the following orders in her libel action:
 - "3. A declaration that the current composition of the Judicial Service Commission as the Governor's advisory body, on inter alia, appointments for acting judges undermines the fundamental right to a fair trial under the Gibraltar Constitution ('the Constitution') and European Convention of Human Rights ('ECHR').
 - 4. A declaration that Section 6 of the Judicial Service Act is ultra vires the Constitution and therefore null and void.
 - 5. A declaration that Section 26 (e) of the Judicial Service Act is ultra vires the Constitution and undermines independence of the judiciary.
 - 6. Declarations that Section 33(3) and Section 33(4) of the Judicial Service Act undermine independence of the judiciary in Gibraltar and therefore are null and void.
 - 7. Section 37(1)(d) and 37(3) of the Judicial Service Act are ultra vires the Constitution and therefore are null and void.
 - 8. A declaration that Section 42(1) (a) (b) (c) (d) and 42(2) of the Judicial Service Act is ultra vires the Constitution and undermines the independence of the judiciary in Gibraltar.
 - 9. A declaration that the devolved powers to the Gibraltar Parliament do not empower Parliament to change any constitutional provisions or structures."

These were not appropriate orders to seek in her libel action. They appear to have been an attempt to duplicate the battle that the Chief Justice was about to launch by his judicial review proceedings. Their source would seem to be documents prepared in relation to those proceedings, kept in a file that was common to both the Chief Justice and his wife, together with conversations that she must have had with the Chief Justice about those proceedings.

- 191. What is more significant is that in his judicial review proceedings the Chief Justice for the first time himself advanced allegations that his wife had previously made as to attempts by the Chief Minister to get rid of him stretching back to 1999. This lends strong support to the conclusion that the Chief Justice was in sympathy with the allegations made publicly by his wife at the earlier stages of the story.
- 192. In his statement the Chief Justice accused the Chief Minister of having made personal attempts to oust him from office and alleged that the provisions of the Constitutional Reform Act and the Judicial Service Act that related to the role of the Chief Justice were designed to achieve the same end. This remained the Chief Justice's case at the start of the hearing before the Tribunal. The Chief Justice's stance received some support from the Leader of the Opposition. He told the Tribunal that he believed that there had been "an orchestrated campaign against him politically driven". Later he said:

"It seems to me that the only reason we have a head of the judiciary based in London is because Mr Caruana got a bee in his bonnet about it and wanted to do it to spite the current Chief Justice. It is a pure value judgment, but it is consistent with his normal reactions when he deals with other people that cross him."

- 193. The Chief Minister gave evidence to the Tribunal. He was asked about the Chief Justice's contention that Lord Luce had told him, in relation to the expiry of his warrant in 1999 that the Chief Minister had made representations that he should vacate office. He said that he had no recollection of making any such representation. He added in relation to that period:
 - "...it has never been my position that Derek Schofield should be removed from the bench and certainly not for any of the reasons at the time that the Government was unhappy with him about. But there have been occasions where we have believed that his behaviour was not what the Government would have expected of him."
- 194. The Chief Minister said that he knew nothing about the decision in 2002 to offer the Chief Justice a one year warrant of appointment in place of the usual three year warrant. He said that the Government exercised restraint in its public handling of "pretty provocative events, politically damaging, electorally damaging to the Government". The beginning of the "terminal process" that led the Government to form the view that Mr Schofield could no longer remain as Chief Justice of Gibraltar were his wife's statements to the Bar Council and the Kenyan Jurists that the Chief Minister was trying to hound the Chief Justice out of office. He added that the process was concluded by:

"Derek Schofield's witness statement in which for the first time, although the Government of course had believed it from the beginning, and we said so in our recusal application, that we thought it was not possible to distinguish so clinically the position of Mr and Mrs Schofield. It was actually confirmed in his statement when he actually swore a statement in his own court, openly accusing the Government of using executive and legislative means to remove him from office. At that point I think the Government decided that this is it. One of us has got to go."

- 195. By the end of the hearing before the Tribunal the Chief Justice had ceased to contend that the Chief Minister had tried personally and by legislative means to drive him from office. Instead it was submitted, as it has been before the Committee, that the Chief Justice had had reasonable grounds for forming that view. Our conclusions are as follows. At the time of the maids incident and the MOT prosecution the Chief Justice had reasonable grounds to believe, and there are still reasonable ground to believe, that the Government would have been happy to see the Chief Justice replaced, having particular regard to the furore that had followed his statement at the opening of the legal year in 1999. The Government may well have hoped that there would prove to be some means by which this could properly be achieved. The Chief Minister may well have made that view clear to Lord Luce in his regular meetings with him. It may well be that he shared the view, agreed to by the Attorney General in August 2000, that "it would be a good idea if the Chief Justice could be found something elsewhere". The Chief Justice had, however, security of tenure, and no attempt was made to deprive him of this.
- 196. There was no justification for concluding that the Chief Minister had been involved in attempts to remove the Chief Justice from office in 1999 or 2000 or that he was responsible for the decision of the Governor to issue a warrant for only one year in 2002. Nor in the period from 2002 to 2006 was there any basis for concluding that the Chief Minister wished to engineer the removal of the Chief Justice from office.
- 197. The Chief Justice, and an objective observer, might have concluded that there was a possibility that the diminution in the standing of the Chief Justice brought about by the provisions in the 2006 Order and the Judicial Service Act reflected at least in part a conclusion, influenced by the Chief Justice's own performance in office, that it would be better for the position of head of the judiciary of Gibraltar to be filled by a person who had achieved judicial office carrying higher standing than that of Chief Justice of Gibraltar. It was not reasonable to conclude that the motive for the legislative changes was to get rid of the Chief Justice or to reduce his personal standing.
- 198. Had the Chief Justice's application for judicial review been supported by argument that focussed solely on the question of whether section 6 of the Judicial Service Act was

constitutional, we consider that his action in making that application would have shown a grave lack of judgment. Bringing legal proceedings in the court over which he presided would have been bound to raise severe practical problems to the administration of justice in Gibraltar. They would be likely to be seen to be motivated not simply, or perhaps not at all, by the Chief Justice's concern for the Constitution of Gibraltar, but by concern for his personal position and as part of the battle that his wife was fighting on his behalf.

199. In the event the Chief Justice ensured that such a conclusion would be drawn by supporting his application with a statement that alleged that the legislation was motivated by personal animosity against him on the part of the Chief Minister. This was based not on hard evidence but on conjecture. Even if that conjecture had been reasonable, which it was not, the Chief Justice should not have founded allegations upon it in his judicial review application. In the first place the motives of the Chief Minister had no relevance to the constitutionality of the Judicial Service Act. In the second place the allegations were calculated to damage the standing of his office and to render it at least questionable whether he could continue to hold that office. The Tribunal was correct to hold that the action of the Chief Justice in bringing the judicial review proceedings constituted misbehaviour.

Inability and misbehaviour

200. As Mr Beloff pointed out, there is a lack of clarity in 7.35 to 7.42 of their Report as to the precise basis upon which the Tribunal held that the Chief Justice's conduct justified his removal from office under section 64(2) of the 2006 Order. They gave four instances of misbehaviour, commenting that no single instance demonstrated that the Chief Justice was unfit to hold office, but did not hold that his cumulative misbehaviour demonstrated that he was unfit for office. Rather the Tribunal appears to have held that, by reason of a combination of defects of character, misbehaviour and the effects of his behaviour on how he and his office were perceived, his removal was justified because of his *inability* to discharge the functions of his office.

201. There is considerable jurisprudence on the test of both "misbehaviour" and "inability" in the context of the removal from office of a judge or public official. This demonstrates that there is a degree of overlap between the two. The most recent authoritative guidance on the meaning of misbehaviour is to be found in the advice of this Committee in the case of Lawrence v Attorney General of Grenada [2007] UKPC 18; [2007] 1 WLR 1474. That appeal was against the removal from office of the Director of Audit of Grenada under a provision in identical wording to that of section 64(2). Her misbehaviour had consisted of the single act of sending to the Minister of Finance, who was also the Prime Minister of Grenada, with copies to the Clerk of Parliament and the Speaker of the House of Representatives of an intemperate and abusive letter that accused him of tampering with her report prior to its being laid before Parliament.

202. Giving the advice of the majority of the Committee Lord Scott of Foscote remarked at paragraph 23 that "misbehaviour" was a word which drew its meaning from the context in which it was used. He went on at paragraph 25 to quote with approval paragraph 85 of the decision of Gray J in *Clark v Vanstone* [2004] FCA 1105, (2004) 81 ALD 21:

"It is clear from these expressions of opinion that, in order to constitute misbehaviour by the holder of an office, the conduct concerned need not be criminal conduct and need not occur in the course of the performance of the duties of the office. For present purposes, the important proposition to be drawn from these expressions of opinion is that, in a case in which the term 'misbehaviour' is used with reference to the holder of an office. the content of its meaning is to be determined by reference to the effect of the conduct on the capacity of the person to continue to hold the office. In turn, the capacity to continue to hold an office has two aspects. The conduct of the person concerned might be such that it affects directly the person's ability to carry out the office. Alternatively, or in addition, it may affect the perceptions of others in relation to the office, so that any purported performance of the duties of the office will be perceived widely as corrupt, improper or inimical to the interests of the persons, or the organisation, for whose benefit the functions of the office are performed. In either case, the danger is that the office itself will be brought into disrepute as a result of the conduct of its holder. If that is likely to be the then the conduct is properly characterised misbehaviour for the purposes of the relevant legislation."

- 203. Lord Scott derived from this passage four ingredients that will normally need to be present before conduct can be characterised as "misbehaviour" for the purposes of removal from office. We consider that Lord Scott's analysis can helpfully be applied in the present case. The search for the four ingredients raises the following four questions:
 - i) Has the Chief Justice's conduct affected directly his ability to carry out the duties and discharge the functions of his office?
 - ii) Has that conduct adversely affected the perception of others as to his ability to carry out those duties and discharge those functions?
 - iii) Would it be perceived to be inimical to the due administration of justice in Gibraltar if the Chief Justice remains in office?
 - iv) Has the office of Chief Justice been brought into disrepute by the Chief Justice's conduct?

204. So far as "inability" is concerned, assistance can be derived from the decision of the House of Lords in Stewart v Secretary of State for Scotland 1998 SC (HL) 81. That appeal arose out of a report by the Lord President and the Lord Justice-Clerk to the Secretary of State for Scotland pursuant to section 12(1) of the Sheriff Courts (Scotland) Act 1971 that the appellant was "unfit for office by reason of inability, neglect of duty or misbehaviour". The conduct that was the subject of investigation consisted of a consistent pattern of bizarre behaviour both on and off the Bench. The finding was that this did not constitute "misbehaviour" but was attributable to a character flaw which amounted to "inability". The House upheld the finding of the Court of Session that "inability" was not to be restricted to unfitness through illness but extended to unfitness through a defect in character. In the leading speech Lord Jauncey of Tullichettle rejected the submission that the importance of judicial independence required that "inability" be accorded a narrow meaning. He held that the fact that the decision as to a sheriff's unfitness lay with two senior judges was the bulwark standing between the sheriff and any undue interference by the executive.

205. A similar point can be made in the present case. There is good reason to give "inability" in section 64(2) the wide meaning that the word naturally bears. If for whatever reason a judge becomes unable properly to perform his judicial function it is desirable in the public interest that there should be power to remove him, provided always that the decision is taken by an appropriate and impartial tribunal.

206. We consider that it was open to the Tribunal to proceed on the basis that defect of character and the effects of conduct reflecting that defect, including incidents of misbehaviour, were cumulatively capable of amounting to "inability to discharge the functions of his office" within section 64(2). It remains to consider their finding that, on the facts of this case, such inability had been demonstrated in the case of the Chief Justice.

An overview

207. The Tribunal heard the Chief Justice give evidence over the best part of two days. They took an adverse view of him as a witness. Repeatedly they found his evidence incredible and rejected it. We have reviewed the Tribunal's findings and, on a number of occasions, concluded that there was no firm basis for rejecting the Chief Justice's evidence. However on the central issue of the extent to which the Chief Justice was aware of his wife's actions and intentions we have concluded that the Tribunal had good reason to reject the Chief Justice's evidence that he neither knew nor was particularly interested in what his wife was doing. The transcript fully supports the Tribunal's finding that the Chief Justice was evasive. Repeatedly he said that he could not remember matters that one would expect to have been etched on his memory. It is simply incredible that he and his wife would not have discussed matters that were of great mutual concern and how they were going to react to them.

- 208. The evidence demonstrates that Mrs Schofield has a strong personality and we have concluded that, in her relationship with her husband, it was the dominant personality. From the time of the public exchanges between the Government and her husband that followed his comments at the opening of the legal year in 1999 she formed the view that the Government, and in particular the Chief Minister, was bent on hounding her husband from office. She undoubtedly communicated her views to her husband. If he was not persuaded that she was right, he was not persuaded that she was wrong. Over time he also came to believe that the Chief Minister was trying to hound him from office.
- 209. The Chief Justice initially kept his thoughts to himself. His wife did not. She went public with them and she acted on them. There was never any indication that he was not content that she should do so, nor has he given any such indication to this day. His attitude has been that his wife is an independent person, that she is entitled to do and say what she chooses and that no one would reasonably associate him with his wife's conduct. In that he has been wrong. He has been associated, and reasonably associated, with his wife's conduct.
- 210. Finally, in his judicial review proceedings, the Chief Justice publicly adopted his wife's contention that the Chief Minister had, from 1999, been bent on driving him from office. Through his counsel he persisted in so contending at the opening of his case before the Tribunal.
- 211. The Chief Minister gave evidence. He vigorously refuted the suggestion that he and his Government had been attempting to drive the Chief Justice from office. He stated that, on the contrary, his Government had acted with considerable restraint in the face of provocation from the Chief Justice, and illustrated that contention at some length.
- 212. The transcript of the Chief Minister's evidence is compelling. So, it seems, was his evidence itself. It was not challenged by Mr Fitzgerald QC, who was acting for the Chief Justice. Rather it was put to him that the Chief Justice had reasonable grounds for believing that he was bent on driving him from office. That also he refuted. "Reasonable grounds for belief" was the way in which the Chief Justice's case was advanced in Mr Fitzgerald's closing submissions. Neither then nor since has the Chief Justice apologised for the allegations made against the Chief Minister that he has accepted were not justified.
- 213. This case has aspects of a Greek tragedy. The Chief Justice's wife's conduct in reaction to what she mistakenly believed were attempts to drive him from office threatened to make that office untenable by him unless he publicly dissociated himself from her conduct. This he was not prepared to do. He was not persuaded that her allegations were unfounded and in the end he publicly associated himself with them.

- 214. We turn to the four questions based on the approach of Lord Scott in *Lawrence*. The first requires consideration of the practical effect of the Chief Justice's conduct on his ability to discharge the functions of his office. So far as that is concerned the debate before us has focussed on the extent to which the Chief Justice might be obliged to recuse himself from trying cases involving those whom he himself has attacked or is seen as having attacked by reason of association with attacks made by his wife. As this depends in part on perception it is convenient to address first the second question, which relates to the perception of others as to the Chief Justice's ability to perform his functions.
- 215. The Tribunal had before it an abundance of evidence of the perception of others as to the consequences of the Chief Justice's conduct. The signatories to the Memoranda were Louis Triay QC, signing on his own behalf and on behalf of his firm Triay Stagnetto & Nash, JE Triay QC, signing on behalf of himself and of his firm Triay & Triay, James Levy QC on his own behalf and on behalf of his firm Hassans, A Levy on his own behalf and on behalf of his firm Attias & Levy, AV Stagneto QC, David Dumas QC and AA Vasquez QC. A common theme of the Second Memorandum to which the Signatories subscribed and of individual witness statements made by Signatories or member of their firms was that, having failed to disassociate himself from his wife's public utterances, the Chief Justice was seen as supporting these. In a witness statement which impressed the Committee as balanced and moderate, Mr Francis Triay put his viewpoint as follows:
 - The size of our community is small, and because most land is owned by the Government of Gibraltar and most business activities require licensing by the Government in some way or another, the Government is necessarily involved in a large number of legal and business matters. The legal profession is also small in number and accordingly, it is very important for the proper functioning of the community and the administration of justice that Government representatives and those involved in the administration of justices keep to their proper roles within the small community. When the Chief Justice began to involve himself in political matters, I felt that he was exceeding his role as Chief Justice and this would undermine the administration of justice. This situation was made worse when unfortunately his wife began to crusade to protect him from what she perceived to be a conspiracy since her actions have only served to further undermine his position as Chief Justice, especially as the Chief Justice has invariably taken up the same crusades.
 - 5. In my view, Mrs Schofield has attempted to protect her husband but, in the process, she has created the very situation which she says she has tried to avoid, namely undermining and

further polarising the position of the Chief Justice, especially as the Chief Justice has fought the same battles.

- 6. I cannot know the extent to which the Chief Justice operated in a joint enterprise with his wife. My overall impression is that they must have acted in concert to some large extent in view of the fact that the points that she made were all ultimately to protect or related to the Chief Justice and that the Chief Justice did nothing to intervene or disassociate himself from what she did or her allegations. It would also be very unusual in a marriage, for a husband or wife to take such high profile action on a matter which necessarily affects the other, without the other being privy to and being part of the action taken, especially where the views of both are in the main the same."
- 216. We quote this because it reflects our own conclusions as to the association of the Chief Justice with his wife's public utterances. As to the consequences of this perception, the Signatories put these as follows in their Second Memorandum:

"We feel it is impossible for the Hon Derek Schofield CJ to sit in cases involving judicial reviews of Government or Ministerial decisions when his wife has very publicly accused the Chief Minister of wanting to get rid of her husband . . . The Hon Derek Schofield has not dissociated himself from those comments. Because he has not done so, a fair-minded and informed observer is bound to conclude that the Chief Justice either shares those views and is therefore likely to be biased against the Government or any Minister concerned or that, at the very least, there is a real possibility of subconscious bias on his part. Administrative Law is an expanding area of law as much here in Gibraltar as in England and Wales. There are increasing numbers of court challenges to the decisions of public authorities. A judge who is prevented from dealing with such cases because of perceived prejudice or bias is a judge who cannot properly discharge his functions."

217. The Second Memorandum was polemic in tone and it would not be right to treat its contents as reflecting a universal viewpoint. Mr Triay observed that the Chief Justice's conduct has polarised the legal profession in Gibraltar with some lawyers at times supportive of the Chief Justice whilst others have been against him. It would not be right, however, to treat the large body of lawyers who were prepared to subscribe to the Memoranda as partisan. We accept that the picture painted in the Second Memorandum portrayed the fairly held views of a significant proportion of the legal practitioners in Gibraltar.

- 218. The Tribunal made findings as to the effect that perceptions of bias caused by the Chief Justice's conduct would have on his ability to try cases. He himself had accepted that he would have problems in sitting on any case in which the Chief Minister was involved either as a witness or because a policy of Government for which he was responsible was in issue. The Tribunal rightly observed that these were likely to be among the most important so far as concerned their impact on the public. The Tribunal queried, however the practicability of identifying cases involving the Government which did not involve the Chief Minister. We endorse these findings.
- 219. The Tribunal further found that the Chief Justice would face a similar problem in sitting on cases involving Mr Rhoda, the Attorney General because of his allegation that the Attorney General had been involved in an attempt to remove him in 1999. We do not consider that this could give rise to any appearance of bias ten years later. More pertinent is the statement by Mr Rhoda that he would feel a great deal of unease in appearing before the Chief Justice having regard to the fact that "the Schofields have taken issue and asked me to recuse myself from the Judicial Service Commission and as Chairman of the Admissions and Disciplinary Committee in respect of Mrs Schofield's complaint against Freddie Vasquez QC " and the e-mail that she sent on 5 April 2007 stating that she would be instructing her lawyers to ask the Attorney General to recuse himself from dealing with any issue relating to herself or her husband. Overall we consider that there would be no more than a possibility of a recusal application and that only in relation to a case in which the Attorney General was personally involved
- 220. The Tribunal also found that the Chief Justice's known antipathy towards certain members of the Bar Council, in particular Mr Neish the leader of the Bar, would be likely to affect the ability of the Chief Justice to be seen to do justice in some individual cases. This raises the wider question of whether there would be a perception of bias against the lawyers who were, or who were in the firms of, the Signatories to the two Memoranda. Mr Catania, a partner in the firm Attias & Levy, suggested in his witness statement that the Chief Justice would be perceived as favouring those lawyers who had supported him and his wife as opposed to those who had been party to the Memoranda. While there would be a risk of this we do not consider that it would be right in principle to consider as a ground for removal of a judge, an appearance of bias based on resentment that the judge might be thought to feel towards advocates who had sought his removal. Were this not so, an application to a judge to recuse himself might be self-fulfilling.
- 221. While we have not accepted all the allegations of apparent bias that would arise if the Chief Justice were to continue to sit, those that they have accepted are significant. A Chief Justice of Gibraltar who was unable to sit on cases involving the Government would be substantially disabled from performing his judicial function.

- 222. We turn to the third and fourth questions based on *Lawrence*, which are linked. Essentially the question is whether the behaviour of the Chief Justice has brought himself and his office into such disrepute that it would damage the image of the administration of justice in Gibraltar if he continues to serve as Chief Justice. No question has ever been raised as to the Chief Justice's judicial ability to resolve issues of fact and law. His conduct has, however, shown repeated and serious shortcomings and misjudgements' in his public behaviour. The office of Chief Justice, it must be recognised, carries demands well beyond those placed upon ordinary sitting judges, however senior. We have concluded that the Tribunal was right to criticise his conduct in that office in relation to the following episodes
- The content of his address at the opening of the legal year in 1999 and his interchanges with the Government thereafter
- Failure to have due regard to his legal obligations in relation to the maids that he employed in 1996 and 1997.
- His defence to the MOT prosecution in 2000.
- His behaviour in court after the issue of the one year warrant.
- The swearing in of the Deputy Governor in July 2006.
- His reaction to the draft 2006 Constitution culminating in his speech at the opening of the legal year.
- His initial reaction to the Judicial Service Bill.
- Allowing his wife to appear before him in court on 16 April 2007
- His reference to his wife's interest at the recusal application of 22 May 2007.
- The cancellation of the 2007 opening of the legal year ceremony.
- His implicit support of repeated accusations made by his wife against the Government and the Chief Minister and against members of the Bar Council.
- His judicial review proceedings and the allegations made against the Chief Minister in support of them.
- 223. These episodes demonstrated the defects of personality and attitude identified by the Tribunal at 7.37 to 7.39 of their Report. That indeed is the principal relevance of the episodes that predated 2006. The last two items also resulted in an inability to preside over hearings involving the Chief Minister because of the appearance of bias. There would also be a risk of applications to recuse himself in hearings involving the Attorney General.
- 224. This conduct infringed almost every one of the principles in the two Guides to Judicial Conduct that set out in paragraphs 28 and 29, namely 2.2, 2.3, 4.1, 4.2, 4.3, 4.6 and 4.8 of the Bangalore Principles and 3.1 and 3.2 of the Guide to Judicial Conduct and the

warning in this guide in relation to disqualification and to talking to the media with the greatest circumspection.

- 225. The conduct of the Chief Justice has brought him and his office into disrepute. The Tribunal had evidence and submissions from a large proportion of those who practise in the courts of Gibraltar that in their perception his conduct had adversely affected his ability to carry out his duties and functions and that it would be inimical to the due administration of justice in Gibraltar if he remained in office. Indeed they went further and submitted that his occupation of his office had been rendered untenable.
- 226. Mr Mendez, who had always loyally served the Chief Justice as Deputy Registrar and who was praised by the Tribunal as a reliable witness spoke in his witness statement of his perception of the loss of respect sustained by the Chief Justice among the people of Gibraltar and added that if the Chief Justice returned to his post he would not continue working with him.
- 227. We have reached the conclusion that the actions of the Chief Justice and his wife have rendered his position as Chief Justice of Gibraltar untenable. The Chief Minister was realistic in saying that the terminal process began with Mrs Schofield's publicised statements to the Bar Council and to the Kenyan Jurists that the Chief Minister was trying to hound her husband out of office and ended when he brought judicial review proceedings in which he publicly adopted that allegation.
- 228. Although there were a number of incidents that qualified as misbehaviour we would follow the example of the Tribunal in finding that these were incidents in a course of conduct that has resulted in an inability on the part of the Chief Justice to discharge the functions of his office.
- 229. Accordingly we humbly advise her Majesty that the Hon Mr Justice Schofield be removed from the office of Chief Justice of Gibraltar.

LORD HOPE, (with whom Lord Rodger and Lady Hale agree)

230. I regret that I am unable to agree with the opinion of the majority in this extremely important and very troublesome case. While we share their view that in numerous respects the Tribunal's criticisms of the Chief Justice of Gibraltar cannot be accepted (see paras 50-52, 55-56, 71-73, 78, 82-83, 110-111, 113, 145, 172-174, 181 and 219-220), I find myself in the same position as Lord Rodger and Lady Hale. We do not regard the remaining grounds

of criticism as justifying the conclusion which the majority have reached that he ought to be removed from his office of Chief Justice on the ground of inability to discharge the functions of that office. The following are my reasons for reaching this view.

Procedural Backgound

- 231. Section 64 of the Gibraltar Constitution Order 2006 ("the Order") provides that the Chief Justice shall be removed from office by the Governor if the question of his removal has, at the request of the Governor made in pursuance of section 64(4), been referred by Her Majesty to the Judicial Committee of the Privy Council and the Judicial Committee has advised Her Majesty that the judge ought to be removed from office "for inability to discharge the functions of his office or for misbehaviour": see section 64(2). Section 64(4) lays down the procedure that the Governor must follow before he makes such a request. He must appoint a tribunal whose task is to inquire into the matter, to report on the facts to the Governor and to advise him whether he should request that the question of the judge's removal should be referred by Her Majesty to the Judicial Committee. If the tribunal so advises, the Governor is required by the subsection to request that the question be referred accordingly.
- The steps laid down by section 64(4) of the Order have all been fulfilled. A tribunal was duly appointed. There were two hearings for directions, which were followed by a full hearing at which evidence was led. Having considered the evidence and a substantial number of documents, the Tribunal concluded that the Chief Justice is unable to discharge the functions of his office and that this inability warrants his removal from office: para 7.42. Accordingly it advised the Governor that he should request that the question of his removal should be referred by Her Majesty to the Judicial Committee. It is now for the Board to consider whether the Chief Justice should be removed from office. It is a heavy responsibility. We must look at the facts with an appropriate measure of detachment from the cut and thrust that characterised the events that gave rise to this procedure. Our task is to set them into a wider context in the light of our own knowledge and experience of the relationships that may develop between the judiciary and the executive. We must be sensitive to the fact that it takes two to create an argument, and to the risk that one side or the other may be over-reacting to perceived insults. Above all, we must be careful not to undermine the principle of independence by judging too harshly one man's judgment of what it required of him.
- 233. The Board is not, it must be appreciated, sitting as a court of appeal. The jurisdiction which it is required to exercise in a case of this kind is an original jurisdiction. It must make up its own mind as to whether or not the allegation that the Chief Justice is unable to discharge the functions of his office has been made out. The raw material that it needs to address this issue is to be found in the report and the supporting documents. But the Tribunal's findings are not binding on the Board, nor is the conclusion that the Tribunal drew from those findings. The weight to be given to them must depend on whether the Board is

satisfied that the Tribunal addressed the issues that were before them fully and fairly, and that it directed its mind to the right questions when it reached the conclusion that the Chief Justice should be removed from office.

- 234. The report is, as one would expect of such a distinguished Tribunal, clear and comprehensive. I see no reason to doubt the fairness of the procedure that was adopted. Nevertheless it seems to me that the report suffers from a number of significant defects and that, taken overall, it requires to be approached with a great deal of caution. Although the Tribunal says in para 1.21 that it took full account of complimentary remarks made by a number of witnesses, the impression that is created by the remorseless criticism of him that follows is that there was a marked lack of balance in its approach to the issues that were before it. I have the distinct impression that it failed to give proper weight to the context in which the various events which it was considering had arisen and to analyse them with a due sense of perspective and detachment. It seems to me that, in the result, it has presented us with a one-sided version of these events. It did not pay adequate regard to the situation in which the Chief Justice found himself as those events unfolded, to the room for differences of opinion as to what upholding the principle of judicial independence required of him and to the nature of the jurisdiction in which he was operating.
- 235. I accept with gratitude the careful way in which, for the majority, Lord Phillips has reviewed the facts and analysed the Tribunal's report. But, for the reasons I shall explain, I would draw a different conclusion than he does from the product of this analysis. Here too I think that the approach lacks the necessary sense of perspective and detachment. It fails to give proper weight to the crucial importance of protecting senior judges against attacks by the executive upon their efforts to uphold judicial independence in their jurisdiction. Errors of judgment there may well have been. But to say that this amounts to an inability to discharge the functions of the office within the meaning of section 64(2) of the Order which justifies his removal from office seems to me to go too far. It risks setting a dangerous precedent.

Judicial independence

236. It is a striking fact that the first and last of the episodes listed by the majority as those for which they consider that the Tribunal was right to criticise the Chief Justice's conduct (para 222) were instances where he was addressing himself to what he saw as risks to the independence of the judiciary. This was the theme of the relevant part of his address at the opening of the legal year in 1999. It was also the theme of his statement that was filed in support of his proceedings for judicial review on 29 August 2007. Leaving aside some of the earlier episodes which fall outside this pattern, it is a theme that runs right through the problems that his relationship with the executive gave rise to. One of the central questions in this case is how far it was open to the Chief Justice to exercise his own judgment as to what the need to protect the independence of the judiciary required of him.

237. In his address to the Commonwealth Law Conference in Nairobi on 12 September 2007 Lord Phillips of Worth Matravers CJ said:

"A judge should value his independence above gold. Not for his or her own benefit, but because it is of the essence of the rule of law."

Nine years earlier, in June 1998, the Chief Justice was among the group of parliamentarians, judges, lawyers and legal academics who joined together at Latimer House at a Colloquium on Parliamentary Sovereignty and Judicial Independence within the Commonwealth. He participated in these discussions as a representative of the Commonwealth Magistrates' and Judges' Association and presented a paper on maintaining judicial independence in a small He brought to these discussions his own experience of the risks to the independence of the judiciary in Commonwealth Africa. There were other examples that would have had a part to play in these discussions, notably the way in which Chief Justice Gubbay of Zimbabwe had been forced out of office in that jurisdiction by the executive. The product of the Colloquium was the Latimer House Guidelines in which various principles were set out for preserving judicial independence. There is no doubt that the Chief Justice saw it as part of his mission in life to promote these principles and, by doing so, to set an example for others in small jurisdictions such as his own. He may be criticised for the way in which he pursued this aim. He was perhaps too ready to perceive threats to his independence when in reality there were none. Perhaps he took insufficient care in his choice of language. But the importance of the principle is not in doubt. I do not detect any reasons for doubting his good faith in seeking to do all that he could to uphold it.

The Chief Justice is by no means alone among the holders of this office in having said or done things in pursuit of judicial independence or the rule of law that have given rise to controversy. In his address at the opening of the legal year in Trinidad and Tobago on 16 September 1999 Chief Justice de la Bastide said that efforts were being made to make the judiciary's access to funds voted to it by Parliament subject to the approval or disapproval of the Attorney General and that if this were to happen he would have a stranglehold on the judiciary. He went on to express fears that the independence of the Bar was being compromised and that the President of the Law Association had joined with the Attorney General to subvert the independence of the judiciary. These and other remarks gave rise to a lengthy and acrimonious statement by the Attorney General and a caustic press statement by the President of the Law Association: Sir Fred Phillips, Commonwealth Caribbean Constitutional Law (Cavendish Publishing Ltd, 2002), p 280. Sir Telford Georges was asked to intervene in the dispute. He drew attention to flaws in a statement to Parliament by the Attorney General in which he claimed that he had responsibility for overseeing the judiciary, but he ended his report with a warning that highly placed judges should not lose their nerve and descend, or threaten to descend, into the arena when provoked: Phillips, op cit, pp 281-A Commission of Inquiry was appointed by the Government to investigate the administration of justice in Trinidad and Tobago presided over by Lord Mackay of Clashfern. It found that the Chief Justice's allegations that the executive was endeavouring to undermine the independence of the judiciary were not well founded. But the Chief Justice's ability to continue to perform the duties of his office was not called into question.

- Bruising exchanges between the senior judiciary and the executive are not unknown in England and Wales, and the behaviour of the senior judiciary is not always above reproach. In December 1934 Lord Chief Justice Hewart's attack on the Lord Chancellor and permanent officials in the Lord Chancellor's Department for their supposed attempt, in the Supreme Court of Judicature Bill, to interfere with the independence of the judiciary was so scathing, so ill-judged and so extreme that only the intervention of the Marquess of Reading saved the day: Hansard Debates, HL, vol 95, cols 224-237. Lord Lane was Chief Justice when Mrs Thatcher was Prime Minister and it fell to Lord Mackay of Clashfern as Lord Chancellor to promote the reforms to the legal profession which were introduced by the Courts and Legal Services Act 1990. In a debate in the House of Lords on 7 April 1989 he was subjected to a sustained and bitter attack by the senior judiciary. In the course of the debate Lord Lane accused him of discourtesy in failing to consult him before he announced the proposed reforms and, in a passage which was widely regarded as intemperate, of conduct that was reminiscent of that perpetrated by Nazi Germany: Hansard, HL Debates, vol 505, cols 1329-1331. In a debate in the House of Lords on 7 April 1989 Lord Mackay was able to set the record straight by referring to an exchange of letters between himself and Lord Lane before the Green Papers were published. He told the House that Lord Lane had authorised him to say he withdrew the word "discourteous": col 1475. No-one, of course, suggested that any steps should be taken to remove either Lord Hewart or Lord Lane from office because of their remarks.
- 240. It is not unknown too for a member of the executive to perceive a slight by a member of the senior judiciary in asserting his independence when in reality there is none. In the course of his address to the Commonwealth Law Conference in Nairobi on 12 September 2007 Lord Phillips recalled Charles Clarke's furious public reaction, when he was Home Secretary, to the refusal by Lord Bingham of Cornhill, the Senior Law Lord, to meet him to discuss issues of principle that might be raised by the measures that he was proposing to meet the challenge of terrorism. He failed to appreciate, of course, that it was not open to Lord Bingham to risk compromising his judicial independence by entering into discussions of that kind with the executive.
- The ability of the press too to stir up trouble must not be underestimated. As Lord Phillips has mentioned (para 27), Gibraltar has a particularly lively press, anxious for copy and, in common with the media in other jurisdictions, eager to identify any actual or supposed conflict between the judiciary and the executive. I encountered this phenomenon during my last year as Lord President of the Court of Session when Michael Forsyth was the Secretary of State for Scotland. He decided to follow the example of the Home Secretary, Michael Howard, in his approach to criminal justice which had been criticised vigorously and in public by the Chief Justice, Lord Taylor of Gosforth. In HM Advocate v McKay, 1996 JC 110, in which the Crown sought to abandon an appeal against a sentence on the ground that it was unduly lenient, I took the opportunity at p 115, in view of the concern that was being expressed on this issue south of the border, to assert the court's right to exercise leniency in the matter of sentencing whenever this was appropriate. This attracted the headline in *The* Scotsman's report the next day: "Warning over threat to justice". From then on the press sought to exploit what they saw as a rift between me and the Secretary of State over issues of policy. As report fed upon report the number of occasions on which I had intervened were said to have been much greater than they actually were. So much so that when it was

announced that I was to resign as Lord President on my appointment as a Lord of Appeal in Ordinary the headlines were "Was he pushed or did he jump?" and "Forsyth tightens grip on crime as Lord Hope quits". I was able to correct this impression in an interview after my resignation. But it was obvious to me that any attempt to do so earlier would only have provided the press with further copy and made matters worse.

I refer to these examples – and others could no doubt be cited (see, for example, the matters referred to in para 5 of Lord Mustill's Report of the Tribunal in the question of removing Chief Justice Sharma from office as Chief Justice of Trinidad and Tobago) - in order to emphasise the latitude that must be given to senior judges when they engage with the executive on matters which may be thought to affect the independence of the judiciary. Upholding that principle is one of their most important responsibilities. The rule of law depends upon it. Of course, they must be careful, as Sir Telford Georges remarked, not to descend into the arena when they are provoked. To do so risks bringing their office into disrepute. But sometimes a venture in that direction is unavoidable in order to make the point that there is a boundary beyond which the executive may not go in its dealings with the judiciary. From time to time mistakes will be made, as tends to happen in the heat of battle. But they should not be characterised as instances of misbehaviour or indicative of inability. That would only be appropriate in the most extreme case. Ideally, it would only be after a warning has been given by a higher judicial authority that conduct of a kind which has given offence out of all proportion to the circumstances must not be repeated. In the case of this Chief Justice no such warning was given. It is unclear who, if anyone, was in a position to do this. He was, in effect, on his own. All the more reason for according him a measure of latitude, so long as his actions were not malicious or in bad faith.

The episodes in general

243. The majority have concluded that the Tribunal was right to criticise the Chief Justice in relation to the twelve episodes that they list in para 222 of their judgment. I wish to concentrate on only four of them: the first, the seventh and the last two. As for the remainder, I would make these brief points. First, as the majority have observed (para 112), there was an interval of four years between the fourth episode, which relates to his behaviour after the issue of the one year warrant on 7 February 2002 in respect of which the majority reject the Tribunal's findings of bad faith and collusion (para 111), and the fifth episode in July 2006. No criticism is made of the behaviour of the Chief Justice during this period, either on or off the bench. This is a strong indication that, whatever one is to make of these earlier episodes, they are not indicative of an inability to discharge the functions of his office. So I would discount the second, third and fourth episodes entirely from the overall assessment. The first episode, as it forms part of a pattern, does deserve further scrutiny.

244. As for the fifth episode, the swearing in of the Deputy Governor, the majority say that the Tribunal was justified in describing this as disgraceful behaviour and that it could properly be described as misbehaviour although not of major consequence. It seems to me

however that if, as they say, it was not of major consequence – a view with which I agree – it must follow that it was not misbehaviour of the kind that would justify removal from office. This was an isolated incident, and I accord it a low priority in the overall assessment of inability. The sixth episode relates to the debate over the 2006 Constitution. The majority conclude that the Chief Justice's actions on this occasion were motivated by the praiseworthy aim of promoting judicial independence but that this showed a serious lack of judgment. They say that it may be open to question whether this amounted to misbehaviour but that it showed an inability to pay due regard to the consequences of his actions. In my opinion the appropriate finding, in the light of these remarks, is that it has not been shown that this was an example of misbehaviour. I agree that his lack of judgment falls to be taken into account in the overall assessment of inability. But I would accord him greater latitude for mistakes of judgment in pursuit of that aim than he has been given by the majority.

I leave aside for the moment the seventh episode. As for the eighth which relates to the Chief Justice's action in allowing his wife to appear before him in court on 16 April 2007, I think that the proper conclusion in the light of the majority's comments (para 177), with which I agree, is that it cannot be characterised as an example of misbehaviour. In regard to the ninth episode as to his wife's interest at the recusal application of 22 May 2007, the majority accept the Tribunal's conclusion (para 179) that he had no possible ground for treating his wife as an interested party who should be notified and that this was an example of his inability to comprehend the constraints and responsibilities of his office, particularly where his wife was concerned. While his conduct on this occasion is open to some criticism, it is not said to have led to an injustice and had no lasting effect. I would regard it as insignificant in the overall assessment. In regard to the tenth episode, the cancellation of the opening of the legal year 2007 to 2008, the majority reject the Tribunal's main conclusions about it which were against the Chief Justice. But they accept its description of it as a deliberately provocative act which only served to exacerbate existing tensions (para 181). I agree that he showed a lack of judgment, but I also agree that his conduct was probably motivated by a desire to mark his objection to being deprived by the legislative reform of the position as head of the judiciary. I would not leave this episode out of account in the overall assessment of inability. But it needs to be put carefully into the overall context.

246. As for the overall context, two other factors need to be taken into account in the assessment. The first is the absence of any suggestion that the Chief Justice was incapable of performing his judicial duties. As the majority recognise, no question has ever been raised about his judicial ability to resolve issues of fact and law: para 222. This is an important factor in his favour. To a substantial degree his performance of the duties of the office was beyond reproach. The second is to be found in the evidence of Mr Peter Caruana QC, the Chief Minister of Gibraltar. He gave evidence on the sixth day of the hearing. The Tribunal preferred his evidence to that of the Chief Justice where they were in disagreement. He insisted that it had never been his view that the Chief Justice's contract should not be renewed or that he should be removed from office until the Government had adopted that position in the light of the events of 2006 and 2007. He said that it had never been his position that he should be removed from the bench, although there had been occasions when the Government believed that his behaviour was not what it would have expected of him. In cross-examination, when asked to pinpoint the stage at which, in his view, the public interest required him to be no longer Chief Justice of Gibraltar he said that the beginning of the terminal process was probably his wife's statements about his being hounded out of office and the things she was saying to Kenyan jurists and the Bar Council, and that it was concluded by his witness statement in support of his claim for judicial review in which he openly accused the Government of using executive and legislative means to remove him from office. Looking at the overall history, he accepted that, while there were periods of tension, there were quite long periods of complete normality of relations.

247. The conclusion that I would draw from this evidence is that the key to the case against the Chief Justice lies in these two episodes: his wife's behaviour and the extent to which, if at all, it is right to hold what she said or did against him; and the remarks that he made in his witness statement. As for the rest, there is an obvious danger in highlighting in a single inquiry a series of relatively isolated instances of misguided, tactless or irritable behaviour which extend over many years. A sense of perspective is lost unless one takes account of the whole picture, including the long periods when relations with the executive were perfectly normal. Had it not been for these two episodes, which the Chief Minister himself identified as marking the start and the end of the terminal process, there would have been no case for his removal from office. It is with that background that I turn to the first, seventh and last two episodes.

The first episode: the opening of the legal year 1999

248. As I have said, the interval of time between the group of episodes of which this forms part and those that follow is a strong indication that, whatever one is to make of them, they are not indicative of the Chief Justice's inability to discharge the functions of his office. The Chief Minister's evidence supports this assessment. Furthermore the majority, I think rightly, reject the Tribunal's findings of bad faith and its decision to treat this matter as an incident of misbehaviour forming part of the justification for removing him from office ten years later (paras 55 and 56). But they say that his remarks, which are quoted in para 2.4 of the Tribunal's report, were a demonstration of bad judgment which has relevance to the question whether there was a defect of character and personality resulting in a current inability to perform the functions of his office.

249. In my opinion, against the background of what had been discussed and agreed at Latimer House in June 1998 and was published as the Latimer House Principles in the Commonwealth Judicial Journal in December 1998, this is an unduly harsh judgment. The Chief Justice's remarks were comparatively restrained in comparison with those which had been made on the same theme by Chief Justice de la Bastide in Trinidad and Tobago a month earlier. It is true that the principle in the Guidelines that the administration of monies allocated to the judiciary should be under the control of the judiciary did not meet with the approval of the Commonwealth Heads of Government in December 2003. But that was long after the date of this episode. A fair interpretation of the Chief Justice's remarks is that he was putting down a marker for further discussion with the executive during the coming year. This was in accordance with the agreed guidelines as they stood at the time, although the

unspoken examples of interference that he had in mind did not really stand up to detailed scrutiny. The political storm which ensued was due to a substantial extent to the efforts of opposition politicians and the press which provoked what might be thought to have been a surprising over-reaction by the executive. Wisely, the Chief Justice did not pursue this line in his future addresses.

The last episode: the statement in the judicial review proceedings

- 250. I take this episode next, as it is linked to the first by the Chief Justice's concern to uphold and protect the independence of the judiciary. The majority say that there are grounds for thinking that, even if the application for judicial review had been supported by argument that focussed solely on whether section 6 of the Judicial Service Act was unconstitutional, his action in making the application in his own name would have shown a grave lack of judgment in view of the practical problems that this would give rise to (para 198). But the real burden of their criticism of him is directed to what he said in his statement, passages from which are quoted in paras 6.19-6.20 of the Tribunal's report.
- 251. The statement should, of course, be read as a whole. It is true that there are passages in it which contain serious allegations against the executive under the headings "Personal attempts to remove me from office" in paras 14-23, "Legislative attempts to remove me from office/diminish the office of Chief Justice" in para 24-32 and "Reduction of Powers of the Office of Chief Justice" in para 33-48. But the Chief Justice set out his reasons for making the statement in paras 3-4, where he said:
 - "3. I have given anxious thought before bringing this action, but consider that I am obliged to do so. I am gravely concerned that the [Judicial Service Act] undermines the rule of law in Gibraltar, in that it undermines the principle of judicial independence. The JSA diminishes the office of Chief Justice and gives its powers to the President of the Court of Appeal who resides outside Gibraltar and necessarily will be less familiar with domestic issues.
 - 4. In order to demonstrate that my concerns are not of a purely theoretical nature, I will give examples of instances in which those whom the JSA entrusts with safeguarding judicial independence failed to act with proper regard for the independence of the judiciary."

In para 32, although he said that he believed that the executive's determination to press ahead with the Bill was born out of a desire to diminish his powers as Chief Justice because he would not leave, he referred to his concern that this would have on the powers of future Chief Justices. In para 42 he said that he was taking the proceedings not for the purpose of

protecting his own position but to safeguard the office of Chief Justice and the principle of judicial independence.

- As the Tribunal records in para 6.22 of its report, the Chief Justice said in evidence 252. that he had signed the statement after taking legal advice and in good faith and that he considered that there was an evidential basis for everything that he had stated. He repeated his belief that he had been the victim of a series of personal and legislative attempts to remove him from office and that the Attorney General was knowingly involved in an improper attempt to remove him. He said that he had been reluctant to include these claims because he knew that this would be criticised but that he had accepted legal advice that he ought to do so. The Tribunal was shown evidence that contradicted the Chief Justice's evidence about his reluctance and it did not accept his evidence on this point: para 6.23. It also rejected his claim that his allegations were well founded. But it did not go so far as to hold that he was acting in bad faith because he knew that his allegations were baseless. The majority do not make such a finding against him either. Their criticism of him is that his allegations would be seen as motivated not simply by concern for the Constitution of Gibraltar but by concern for his personal position and as part of the battle that his wife was fighting on his behalf (para 198). But it seems to me that it is open to question whether this is a fair reading of the statement as a whole, having regard to what he said in paras 4, 32 and 42.
- 253. That having been said, I would accept that the Chief Justice's conduct in bringing these proceedings and supporting it by a statement in these terms raises a serious question as to whether he had shown that he was unable to discharge the functions of his office or was guilty of misbehaviour of a kind which justified his removal.

The seventh and second last episodes: Mrs Schofield

- 254. Common to these episodes is the allegation that the Chief Justice was aligning himself with his wife's allegations, which she started making after the opening ceremony in 1999, that the executive were trying to hound him out office, and that his actions give rise at least to the perception that they were helping each other in the respective legal proceedings which they were taking.
- 255. This gives rise to an important issue of principle. To what extent, and in what circumstances, is a judge to be held accountable for the actions of his or her spouse or other close relatives? It should be said at once that the circumstances of this case are, in this respect, highly unusual. Mrs Schofield is, as the majority have noted, a practising member of the Kenyan Bar and she possesses a strong, indeed headstrong, personality (para 34). Some indication of her highly unconventional behaviour can be obtained from the terms of her written application for disclosure which the Tribunal received in the course of its inquiry and which, at the close of the proceedings on the sixth day of the hearing, it rejected: see the

opening pages of the transcript of the proceedings on that day. At first sight, of course, she alone is answerable for her own actions and anything that she said or did that might be regarded as misbehaviour cannot be attributed to the Chief Justice. But the case is not as easy as that, because she was conducting a highly articulate and provocative campaign in his defence in the belief that he was being hounded out of office. The question is whether it is a fair reading of the situation that he had associated himself with that campaign and that it was in reality being conducted by them both in concert with each other.

256. The majority, following observations to a similar effect by the Tribunal, say that dissociating himself from her conduct would have been an appropriate step for the Chief Justice to take in an attempt to avoid the implication that what she was doing had his approval (para 36). They also say, more generally, that his conduct infringed almost every one of the principles in Bangalore Principles and the Guide to Judicial Conduct in England and Wales which was published in October 2004 (paras 29, 30 and 224). But it is difficult to find anything in these Guides which tells the judge what to do in the unusual circumstances exemplified by this case. Para 3.3 of the Guide to Conduct contains a warning that care must be exercised where a close member of a judge's family is politically active. But that is not really this case. As for the suggestion that the Chief Justice should have taken steps to dissociate himself publicly from what she was doing, it seems to me to be open to a series of objections.

As the Chief Justice himself said in his first witness statement, his position was that there was no joint enterprise between him and his wife. His approach was and had always been that she was absolutely entitled to take her own view and her own course of action, and he would never seek to interfere or endorse or dissociate himself from her independent opinions. I, for my part, would accept and respect the position which he said he adopted. There can be no doubt that she was entitled to the freedom of her own opinions, her own way of expressing them and her own beliefs. The days are long gone when a husband and wife were treated as one person in law and the husband was that person. It is not unknown for senior figures in public life to have spouses or partners who pursue their own careers and interests, in the course of which they may say or do things that are controversial and Any difficulties that this may give rise to should be resolved between embarrassing. themselves, if they can be resolved at all, in private. Judges are not to be taken as supporting or endorsing their spouse's or partner's conduct if they do not publicly dissociate themselves from it. The law should recognise that they are independent actors and that the deeds of the one are not to be visited on the other.

258. Furthermore, no findings have been made about the nature and efficacy of any steps the Chief Justice might reasonably have taken to dissociate himself publicly from her course of conduct. Such information as the Board has about her activities suggests that she would strongly resist any such interference, claiming with some justification in public that it was her right to do so. Public dissociation in such circumstances would be likely to lead to even more public controversy. The whole question as to how the Chief Justice was to go about this exercise and its likely consequences remains unexplored. I cannot accept that this provides a sound basis for alleging that he was guilty of misbehaviour by association with what his wife

was doing. As for the question whether he indicated by the remarks that he made in his statement in the judicial review proceedings that he associated himself with his wife's allegations, as the Chief Minister suggested, I would for the same reasons reject that conclusion. His remarks can be taken, on their own terms, to be an expression by him of his own beliefs. I do not think that there is a sound basis for holding that he was, by making these remarks, endorsing his wife's behaviour.

259. It is not difficult to understand that, in such a small and highly charged community to which she did not belong either by birth or upbringing, Mrs Schofield's activities would be likely to give offence to those who were the target of her allegations. But she and the Chief Justice were distinct individuals, leading separate lives. That this was how, up until the end, their activities were perceived is borne out by the Chief Minister's evidence that the beginning of the terminal process was Mrs Schofield's statements to the Kenyan jurists and the Bar Council which, he said, was beginning to cross the bounds of what society at large in Gibraltar and the Government at large ought to be expected to tolerate. Up until then relations between the Chief Justice and the executive had been normal for long periods. It is true that the Chief Minister then said that it was not possible to distinguish clinically between them, and that the Government's suspicions were confirmed when the Chief Justice swore his statement. But the important point is that it was not until the end that, according to this evidence, the executive formed the view that the Chief Justice could be associated with what she had been saying for many years.

260. For these reasons I cannot agree with the conclusion of majority that his wife's behaviour was one of the circumstances that rendered the position of the Chief Justice untenable (para 227). In my opinion it is by reference to his own actions alone, not those of his wife, that his ability to perform the functions of his office must be judged.

Misbehaviour and inability

261. Section 64(2) of the 2003 Order provides:

"The Chief Justice, a Puisne Judge, the President of the Court of Appeal or a Justice of Appeal may be removed from office *only* for inability to discharge the functions of his office (whether arsing from infirmity of body or mind or any other cause) or for misbehaviour, and shall not be removed except in accordance with subsection (3)" [my emphasis]

262. I have emphasised the word "only" in this quotation because it serves as a reminder that the grounds for removal of these judicial officers are limited to those expressly mentioned in the subsection. This means that careful attention needs to be given to the meaning of the expressions it uses and to the application of that meaning to the facts of the

case. This is not to say that those expressions must be read narrowly. But one must understand what they mean, having regard to what must follow from a finding that the facts show that the concepts described by one or other of them are satisfied. The principle of judicial independence has a part to play in this assessment. It is protected by the nature of the procedure that the section lays down. But the procedure, which places the responsibility for advising Her Majesty on the Judicial Committee, is not the only protection. It lies too in the principle that judicial officers should not be removed from office save in circumstances where the integrity of the judicial function itself has been compromised.

263. There is not much guidance in the authorities as to how the circumstances of this case should be approached. *Therrien v Minister of Justice* [2001] 2 SCR 3 was a case of misbehaviour. The judge was faced with a complaint that he had failed to disclose information that was prejudicial to him to the members of the committee to select persons qualified for appointment as judges. In para 147 Gonthier J said that the public's confidence in its justice system was at the very heart of the case:

"Thus, before making a recommendation that a judge be removed, the question to be asked is whether the conduct for which he or she is blamed is so manifestly and totally contrary to the impartiality, integrity and independence of the judiciary that the confidence of individuals appearing before the judge, or of the public in its justice system, would be undermined, rendering the judge incapable of performing the duties of his office."

In Lawrence v Attorney General of Grenada [2007] 1 WLR 1474, the allegation was that the Director of Audit had written an intemperate and abusive letter to the Minister containing allegations of improperly tampering with her report prior to its being laid before Parliament. Lord Scott of Foscote quoted with approval a passage in the judgment of Gray J in Clark v Vanstone [2004] FCA 1105 which is quoted by the majority (para 202). As Gray J said, the word "misbehaviour" takes its meaning from the statutory context. If the conduct is such that it is liable to bring the office itself into disrepute it can be properly characterised as misbehaviour. But even if the conduct can be so characterised the question remains whether it is conduct of such gravity, applying the standard referred to in Therrien v Minister of Justice, that the recommendation must be that the judge must be removed from office.

264. This case has been treated however by both the Tribunal and the majority as one of inability. The Tribunal said in para 7.35 of its report that, while there were a number of instances of misbehaviour of the Chief Justice, it would not go so far as to say that any single instance amounted to such misbehaviour as to show that he was thereby unfit for office. Its conclusion was that he had shown by his conduct overall that he was unable to discharge the functions of his office: para 7.42. The majority adopt the same approach (para 228). So would I. I too would look at his conduct overall.

265. The only case which offers guidance as to the meaning in this context of the word "inability" is *Stewart v Secretary of State for Scotland*, 1996 SC 271, aff'd 1998 SC (HL) 81. In the Inner House at p 283 Lord McCluskey said that the word should be given its ordinary understood meaning, reading it in the context of the question whether the person was unfit for office. At p 285 Lord Clyde said that the intention of the statutory provision was that sheriffs may be removed from office if they are found to be unfit for office because they have failed to do things which it was their duty to do, or they have done things which they should not have done, or because there are things that they are unable to do. In the House of Lords Lord Jauncey of Tullichettle, rejecting the appellant's argument for a narrow construction, said at p 86 that it would be absurd if the senior judges were required to report that a sheriff was fit for office although they were entirely satisfied that due to some defect in character or quirk of behaviour not amounting to mental illness he was wholly unfitted to perform judicial functions.

266. I think that the phrase "wholly unfitted to perform judicial functions" captures the essence of what the word means in this context. It sets a high standard. But it seems to me right, if the rule of law is to be preserved, that it should do so. The rule of law is in the hands of the judges, and it is of crucial importance that their independence should be protected against allegations which do not achieve that standard. Section 64(2) is in harmony with this approach, as it refers to infirmity of body or mind or any other "cause". This was indeed what the senior judges found established in the case of *Stewart*, as he had persisted in his unacceptable behaviour notwithstanding two previous warnings that his conduct could not be tolerated in perpetuity. They held that this demonstrated an underlying defect in character which manifested itself in various ways to the severe prejudice of the sheriff's function as a judge. It was this defect in character which was the cause of his inability.

The assessment

267. I would apply that approach to this case. The question, as I see it, is whether the conduct of the Chief Justice in the respects that, for the reasons already given, I would take into account demonstrated that he was wholly unfitted to perform the functions of Chief Justice in the jurisdiction to which he had been appointed. The Tribunal said, in support of its conclusion, that his conduct stemmed from a number of characteristics of his personality and attitude: para 7.36. It set out these characteristics in paras 7.37 to 7.41 of its report. The majority, in making their assessment, say that the defects of personality demonstrated by the episodes on which they rely are those identified by the Tribunal at paras 7.37 to 7.39: in short, a lack of judgment as to what was and was not proper for someone in his position to do or say; showing a pre-occupation, bordering on an obsession, with judicial independence; and his inability to grasp that his association with his wife's communications would be perceived by a fair minded and well informed observer as improper.

268. For the reasons already given, I do not accept that the Chief Justice's attitude to his wife's behaviour can be regarded as a defect in his character or personality. The suggestion

that his pre-occupation with the principle of judicial independence was such a defect seems to me to be venturing into very dangerous territory. I refer to what I have already said about the importance of the principle, the duty of a Chief Justice to preserve and uphold it and the latitude that must be given to him in the performance of this function. It must also be borne in mind that the Chief Justice was not without some justification for his suspicions in his dealings with the Government. The reasons for the one year warrant, referred to by the majority in para 103, have never been explained. Nor have the reasons for publicly depriving him of his de facto position as head of the Gibraltar judiciary.

269. I would agree that there are instances where his conduct showed a lack of judgment. But it must not be forgotten, that in contrast to the cases referred to above, no criticism is made of the Chief Justice's ability to perform his judicial functions, and the fact is that for most of the time he was fulfilling his other functions as Chief Justice in a way that did not attract criticism. The events of the concluding period must be seen in that light. As the Tribunal was not prepared to say that those events in themselves amounted to misbehaviour of such gravity as to justify his removal from office, the case must stand or fall on the issue of inability. Taking the whole progress of events in the round, including the absence of criticism of his conduct on the bench and the Chief Minister's acceptance that there were long periods when the relationship between the Chief Justice and the executive were harmonious, I do not think that the proposition that his conduct demonstrated inability to fulfil the functions of his office, in the sense that he was wholly unfitted to perform them, has been made out.

Conclusion

270. The Chief Justice has now been suspended from office for more than two years. He has been exposed to a long and bruising inquiry, the effect of which has been to harden attitudes on either side. In these circumstances it is probably unrealistic to think that he could now resume the functions of his office. I would not hold this consequence against him. It was not his choice that he should be suspended. But we are where we are, and it seems to me that the proper course for him now to take would be for him simply to resign. I would hold that he should be given that opportunity and that, if he were to do so, no adverse inferences of any kind should be drawn against him. The case that was made against him has not been made out. We would have humbly advised Her Majesty that the Chief Justice ought not to be removed from his office as Chief Justice of Gibraltar.

IN THE MATTER OF A TRIBUNAL OF INQUIRY APPOINTED UNDER S.64 OF THE CONSTITUTION OF GIBRALTAR

Before:

The Rt Hon Lord Cullen of Whitekirk KT The Rt Hon Sir Peter Gibson The Rt Hon Sir Jonathan Parker

Report of the Tribunal

to His Excellency Lieutenant General Sir Robert Fulton, KBE Governor and Commander-in-Chief

CONTENTS

INTRODUCTION	2
Chapter 1: BACKGROUND	6
Chapter 2: 1999	14
Chapter 3: 2000 – 2001	30
Chapter 4: 2002 – 2006	55
Chapter 5: FEBRUARY – MAY 2007	88
Chapter 6: AUGUST – DECEMBER 2007	141
Chapter 7: CONCLUSIONS	158
First Schedule	175
Second Schedule	181
Third Schedule	193
Fourth Schedule	196
Fifth Schedule	205

INTRODUCTION

1. The members of the Tribunal were appointed by His Excellency the Governor by letters dated 14 September 2007, and, in the case of its Chairman, dated 11 December 2007, under section 64 (4) (a) of the Gibraltar Constitution Order 2006 and in accordance with the advice of the Judicial Service Commission (JSC), to advise him on the matter of the Chief Justice. The letters of appointment stated:

"The Tribunal should inquire into and report on whether the Chief Justice of Gibraltar is unable to discharge the functions of his office by reason of inability or for misbehaviour having regard to the Memorandum and Supplementary Memorandum with two Appendices submitted by 13 senior representatives of the legal profession on 17 April 2007 and any other submissions and evidence which may be placed before it and report on the facts thereof to me and advise me whether I should request that the question of the removal of that judge should be referred by Her Majesty to the Judicial Committee."

2. Subsection (2) of section 64 provides that the Chief Justice "may be removed from office only for inability to discharge the functions of his office (whether arising from infirmity of body or mind or any other cause) or for misbehaviour, and shall not be removed except in accordance with subsection (3)" (which requires a reference of the matter to the Judicial Committee of the Privy Council). Subsection (4) provides that if the Governor considers that the question of removing the Chief Justice from office for inability as aforesaid or for misbehaviour ought to be investigated, he is to appoint a tribunal, which "shall inquire into the matter and report on the facts thereof to the Governor and advise the

Governor whether he should request that the question of the removal of that judge should be referred by Her Majesty to the Judicial Committee". If the Tribunal so advises, the Governor is to request that the question should be referred accordingly.

- 3. Subsection (8) of section 64 requires that the powers of the Governor are to be exercised by him in accordance with the advice of the JSC. The JSC, which is established by section 57 of the Constitution, is chaired by the President of the Gibraltar Court of Appeal (currently the Rt Hon Sir Murray Stuart-Smith). The other members include the Chief Justice, the Stipendiary Magistrate and the Attorney General.
- 4. The signatories to the Memorandum dated 17 April 2007 referred to in the letters of appointment (to whom we will refer as "the Signatories"), who included all Queen's Counsel in Gibraltar, with the exception of the Speaker in the House of Assembly, expressed, on behalf of themselves and their respective firms, "their deep concern at a state of affairs which has developed seriously affecting the administration of Justice and the reputational image of Gibraltar." They stated that they had lost confidence in the ability of the present incumbent of the office of Chief Justice to discharge the functions of his office. In response to a request by the Governor for details of the concerns expressed in the Memorandum, the Signatories submitted the Supplementary Memorandum which was dated 21 May 2007. It should be noted that in a letter to the Governor dated 22 May Hassans, one of the Signatories, qualified their support for the Supplementary Memorandum.
- 5. Copies of the Memorandum and the Supplementary Memorandum were provided to the Chief Justice. By letter dated 26 July 2007 solicitors acting for him wrote to the Governor enclosing his Preliminary Response to the matters raised in the Supplementary Memorandum.
- 6. Having considered the Memorandum, the Supplementary Memorandum and the Chief Justice's Preliminary Response, the JSC advised the Governor to appoint a Tribunal under section 64(4) of the Constitution.

 On 17 September 2007, on the advice of the JSC, the Governor suspended

- the Chief Justice from performing the functions of his office under section 64 (6).
- 7. Mr Timothy Otty QC was appointed Counsel to the Tribunal and Clifford Chance as Solicitors to the Tribunal.
- 8. It will be noted that our terms of reference are not in exact correspondence with the terms of section 64. We took the view, which was communicated to the Chief Justice by letter from the Solicitors to the Tribunal dated 21 December 2007, that in so far as it might be suggested that there was any discrepancy between the terms of reference and section 64 of the Constitution, the text in the Constitution was to be preferred.
- 9. Prior to the full hearing we held two hearings for directions, arising from which we were satisfied that, in addition to the Chief Justice, the Signatories, the Government of Gibraltar (to which we will refer as "the Government") and Mrs Anne W Schofield, the wife of the Chief Justice, should be entitled to be represented before the Tribunal. In the event Mrs Schofield was not so represented. She did not give evidence before the Tribunal, in the circumstances set forth in the second part of the Fourth Schedule to this Report.
- 10. Mr Michael Llamas was appointed Secretary to the Tribunal. His duties were of an administrative nature.
- 11. The procedural history of this Inquiry is set out in the First Schedule to this Report.
- 12. We would like to express our appreciation for the assistance which counsel and solicitors for the parties provided in enabling the Inquiry to perform its task in an expeditious and fair manner. We are greatly indebted to Counsel to the Tribunal, for the considerable amount of work done by him in preparing and presenting material at the hearings, and to the Solicitors to the Tribunal in organising a large volume of documentation and communications.

13. In considering what facts have been proved by the evidence before us, we have applied the standard of proof on the balance of probabilities, for the reasons set out in the Fifth Schedule to this Report.

CHAPTER 1

BACKGROUND

In this chapter we set out, as a background to the following chapters, details relating to:

- the career of the Chief Justice (paragraphs 1.1 1.5)
- Gibraltar (paragraphs 1.6 1.20)

together with our observations as to the nature and subject-matter of this inquiry(paragraphs 1.21 - 1.25).

The career of the Chief Justice

- 1.1 Derek Schofield is 63 years old (he was born on 20 February 1945). He was called to the English Bar in 1970. In 1974 he was recruited on a short-term contract via the British Overseas Development Agency to the post of Resident Magistrate in Kenya. In the event, he spent thirteen years in Kenya. In 1979 he was appointed Senior Resident Magistrate, Mombasa. In that post, he exercised administrative control over some six courts in the Coast Province of Kenya, coupled with an enhanced civil and criminal jurisdiction. In 1972 he was appointed Acting Judge of the High Court of Kenya. The following year, he was confirmed in that office. In 1983 he was appointed Resident Judge in Kisumu. He served in that office until 1985, when he was transferred to Nairobi.
- 1.2 In 1983 he married Anne Wangeci Kariuki, a Kenyan lawyer. They have two children.
- 1.3 In January 1987 he resigned as a judge in Kenya on the ground that improper pressure had been exerted on him by the Chief Justice of Kenya (who, it is said, had in turn been subjected to pressure from President

- Moi) in connection with an application for committal of the Head of the Kenyan CID for contempt of court. Shortly thereafter, he left Kenya with his wife and their infant children.
- 1.4 In March 1987 he was appointed Judge of the Grand Court of the Cayman Islands.
- 1.5 He continued in that office until February 1996, when he was appointed Chief Justice of Gibraltar following an open competition in the course of which a number of other applicants were interviewed. The interviewing panel consisted of the outgoing Chief Justice of Gibraltar (Sir Alister Kneller), a member of the Gibraltar Court of Appeal who was Sir Alister Kneller's predecessor as Chief Justice, the Deputy Governor of Gibraltar, a legal adviser from the Foreign and Commonwealth Office (FCO) and the Chief Secretary.

Gibraltar

- 1.6 It is convenient at this point for us to give a brief description of the constitutional position of Gibraltar, and of its political and legal systems.
- 1.7 Gibraltar is part of the dominions of Her Majesty the Queen and thus subject to British sovereignty. Her Majesty is Head of State of Gibraltar, and citizens of Gibraltar are British citizens.
- 1.8 Gibraltar is also part of the European Union, by virtue of its being a European territory for whose external relations the United Kingdom is responsible. In consequence European legislation applies to Gibraltar as it does to a Member State, subject to certain special concessions.
- 1.9 The constitution in place in 1996 when Derek Schofield was appointed Chief Justice was established by the Gibraltar Constitution Order 1969 (the 1969 Order). Under that constitution (the 1969 Constitution) the legislative power of the Gibraltar House of Assembly the precursor of the Gibraltar Parliament established by the 2006 Constitution referred to in the Introduction to this Report was strictly limited. Thus:

- Section 35(2) of the 1969 Order provided that save with the consent of the Governor, acting in his discretion, the House of Assembly should not proceed on any bill which in the opinion of the Governor related to or closely concerned a matter which was not "a defined domestic matter", that is to say such a matter as might from time to time be specified by the Governor acting on instructions given by Her Majesty through a Secretary of State (see ibid. section 55).
- Section 34(2) of the 1969 Order provided that should the Governor consider that legislation was necessary or desirable with respect to any such matter "in the interests of maintaining the financial and economic stability of Gibraltar" he could introduce a bill for that purpose, and, notwithstanding that the bill had not been passed by the House of Assembly, he could himself, with the prior approval of a Secretary of State, bring it into law by assenting to it on behalf of Her Majesty.
- Section 37(1) of the 1969 Order provided that any law to which the Governor had given his assent might be disallowed by Her Majesty through a Secretary of State.
- 1.10 So far as the judicature was concerned, section 58(1) of the 1969 Order provided that the Chief Justice and the President and members of the Gibraltar Court of Appeal should be appointed by the Governor in pursuance of instructions given by Her Majesty through a Secretary of State. As to tenure, section 60(1) of the 1969 Order provided that, subject to the provisions of that section relating to removal (provisions which were largely replicated in the 2006 Constitution), a person holding the office of Chief Justice should vacate that office on attaining 67 years. Hence, on his appointment as Chief Justice, Derek Schofield became entitled (subject to the provisions as to removal) to security of tenure until he attained 67 (i.e. until 20 February 2012). Section 60(8) of the 1969 Order provided that the powers of the Governor under that section (i.e. including his powers in relation to the removal of judges) were exercisable by him acting in his discretion. There was no provision in the 1969 Order enabling a judge to be appointed for a fixed term.

- 1.11 In 1996 the political party in power in Gibraltar was the Gibraltar Socialist Labour Party (GSLP) under the leadership of the Hon. Joe Bossano, who was Chief Minister. The largest opposition party was the Gibraltar Social Democrats (GSD) under the leadership of the Hon. Peter Caruana QC, who was leader of the Opposition. However, in the general election held on 17 May 1996 the GSD was returned to power and Mr Caruana QC became Chief Minister, with Mr Bossano becoming the leader of the Opposition. The GSD was re-elected in succeeding general elections held on 10 February 2000, 27 November 2003 and 11 October 2007. Mr Caruana QC contines as Chief Minister, with Mr Bossano continuing as leader of the Opposition.
- 1.12 While he was Chief Minister, Mr Bossano had advocated the reform of the 1969 Constitution to give Gibraltar a greater degree of self-determination. This initiative was in turn taken up and progressed by Mr Caruana when he became Chief Minister. In 2000 a select committee on constitutional reform was set up, with a view to achieving a consensus on proposals for constitutional reform to be submitted to the UK Government with a view to Gibraltar being granted a greater degree of self-determination. In 2004 formal negotiations on constitutional reform began with the UK Government. Thereafter talks were held between the UK Government and a cross-party delegation from Gibraltar led by the Chief Minister and including the leader of the Opposition. In March 2006 the Foreign Secretary (Rt. Hon. Jack Straw MP) announced to the House of Commons that agreement had been reached on the terms of a new constitution for Gibraltar, to be placed before the people of Gibraltar in a referendum. In the referendum, which was held on 30 November 2006, the people of Gibraltar voted in favour of the new constitution. On 2 January 2007 the Gibraltar Constitution Order 2006 (the 2006 Order) came into effect, thereby establishing the 2006 Constitution.
- 1.13 The 2006 Constitution differs from the 1969 Constitution in a number of important respects. For present purposes it suffices to note two of the principal differences.

- In the first place, the 2006 Constitution grants Gibraltar a significantly greater degree of self-determination than it possessed under the 1969 Constitution. Gone are the references to "a defined domestic matter". In their place, section 47(1) of the 2006 Order provides that the Gibraltar Government shall be responsible for all matters save external affairs, defence, internal security and matters falling within the Governor's responsibility for public offices and related matters; and that in respect of external affairs the Governor shall so far as practicable consult with the Chief Minister. There is no equivalent in the 2006 Constitution to the power of veto conferred on the UK by section 37(1) of the 1969 Order (see paragraph 1.9 above).
- 1.15 Secondly, and of greater relevance to this Inquiry, changes were made by the 2006 Constitution in relation to the appointment of judges. Thus:
 - Section 57(1) of the 2006 Order establishes the JSC (referred to in the Introduction to this Report), to be composed of the President of the Court of Appeal (as Chairman), the Chief Justice, the Stipendiary Magistrate, two members appointed by the Governor acting in accordance with the advice of the Chief Minister (i.e. in effect, nominated by the Chief Minister) and two members appointed by the Governor acting in his discretion (i.e. nominated by the Governor).
 - Section 57(2) provides that, subject to the qualification in subsection (3), in appointing judges the Governor shall act *in accordance with the advice of the JSC*. This is in direct contrast to section 58(1) of the 1969 Constitution, which provided (see paragraph 1.10 above) that in appointing judges the Governor was to act "*in pursuance of instructions given by Her Majesty through a Secretary of State.*"
 - Section 57(3) provides that the Governor may, with the prior approval of a Secretary of State, disregard the advice of the JSC "in any case where he judges that compliance with that advice would prejudice Her Majesty's service." However, it is evident from its terms that this power is likely to be exercisable only in extreme circumstances.

- The 2006 Constitution retains the age limit of 67 years for the Chief Justice, but section 64(7) of the 2006 Order introduces a greater degree of flexibility by enabling judges to be appointed for a fixed term.
- 1.16 Hence, consistently with the greater degree of self-determination granted to Gibraltar in relation to the powers of the legislature, the 2006 Constitution grants to Gibraltar a greater degree of self-determination in relation to the judicature by effectively transferring the power of appointing judges from the UK to Gibraltar through the mechanism of the ISC.
- 1.17 The judicial system in Gibraltar is modelled on the English system. The Gibraltar Supreme Court consists of the Chief Justice and one additional Puisne Judge (currently Mr Justice Dudley). The Gibraltar Supreme Court has both civil and criminal jurisdiction, and hears appeals from the Magistrates' Court. The Magistrates' Court is presided over by a Stipendiary Magistrate (currently Mr Charles Pitto) or by lay magistrates. Appeals from the Gibraltar Supreme Court are heard by the Gibraltar Court of Appeal, which currently consists of a President (the Rt Hon Sir Murray Stuart-Smith) and three former members of the English Court of Appeal. Appeals from the Gibraltar Court of Appeal are heard by the Judicial Committee of the Privy Council.
- 1.18 Gibraltar has a fused legal profession, with barristers and solicitors enjoying the same rights of audience in the courts and of direct access by clients. The Gibraltar Bar is regulated by the Gibraltar Bar Council, whose functions are directly comparable with its English counterpart.
- 1.19 In evidence before us is a list of barristers and solicitors of the Supreme Court of Gibraltar as at January 2008. 26 firms are listed, comprising some 166 lawyers. Included in the list are Hassans (54 lawyers), Triay & Triay (20 lawyers), Triay Stagnetto Neish (12 lawyers), and Attias and Levy (9 lawyers). Each of those firms undertakes a considerable amount of litigation.
- 1.20 It should be added that Gibraltar is well served by the media. Among the newspapers published in Gibraltar are the *Gibraltar Chronicle*, which is

published daily (except Sundays), and *The New People, Panorama* and *Vox*, which are published weekly. In addition, the Gibraltar Broadcasting Corporation provides an online news service.

The nature and scope of the subject-matter of the Inquiry

- 1.21 As to the nature of the subject-matter of this Inquiry, it is right to record that at no stage has any doubt been cast on the Chief Justice's ability as a lawyer in deciding cases argued before him. Indeed, his record in terms of successful appeals against his judgments is good. Rather, the matters which this Tribunal is required to investigate concern other aspects of the Chief Justice's conduct, both personal and judicial, during the period from 1999 onwards. In conducting that investigation we take full account of the complimentary remarks about the Chief Justice made by a number of witnesses. Thus, Mr Felix Alvarez, the Chairman of the Equality Rights Group in Gibraltar, complimented the Chief Justice on his enthusiastic and determined support for human rights in Gibraltar; Mr Bossano said that he had every confidence in the Chief Justice and that he believed that a large majority of ordinary people in Gibraltar did so too; and Bishop Caruana (the Bishop of the Diocese of Gibraltar in Europe) said that he had found the Chief Justice to be correct, courteous and polite, and a gentleman.
- 1.22 The scope of this Inquiry was determined by a Statement of Issues. At our first directions hearing we directed that a draft Statement of Issues should be prepared and served. This was in order to identify the criticisms to which the Inquiry would be directed, and ensure that fair notice was given to the Chief Justice. The draft Statement of Issues was based on, and derived from, criticisms of the conduct of the Chief Justice in the Memorandum, the Supplementary Memorandum and the representations received pursuant to our ruling at the first directions hearing. It was revised on 15 April 2008 in the light of our decision at the second directions hearing (see paragraphs 1-3 of the First Schedule to this Report). The headings of the revised Statement of Issues are set out in the Third

- Schedule. It should be noted that the revised Statement of Issues included the response of the Chief Justice to each criticism so far as it was known at 15 April 2008. In addition the Chief Justice set out his position more fully in his witness statements, which ultimately came to be five in number.
- 1.23 As can be seen from the Third Schedule, the Statement of Issues includes in chronological order 23 particular instances or episodes where the conduct of the Chief Justice was the subject of criticism. They formed the factual subject-matter of the full hearing before us. In the chapters of this Report which follow we address them in turn. However, we should point out that we took the view, in the light of the evidence, that episode No 23 (Alleged inappropriate criticism of Registry staff in October 2007) had been given somewhat greater prominence than it merited. There also may have been confusion as to what was said by the Chief Justice. Accordingly we did not consider that the evidence afforded us any material assistance in discharging our functions.
- 1.24 We should add for completeness that we reached a similar conclusion in regard to the matters comprised in the last entry in the Third Schedule (other undated instances of alleged misconduct on the part of the Chief Justice raised by the Deputy Registrar of the Supreme Court). The criticisms of the Chief Justice under this head were either not made out in evidence or were shown to be insignificant.
- 1.25 Again for the sake of completeness we would mention that, although in the event we declined to take up the Government's invitation to inquire into the Chief Justice's relationship with a particular firm of lawyers in Gibraltar, we do not accept the submission of counsel for the Chief Justice that in inviting us to do so the Government was demonstrating improper motivation on its part. Nor do we accept his submission that the fact that the Government invited us to inquire into that matter casts doubt on its representations in relation to those other issues into which we did inquire.

CHAPTER 2

1999

This chapter is concerned with the following:

- the address by the Chief Justice at the Opening of the Legal Year in October 1999 (paragraphs 2.1 2.24)
- the remarks by the Chief Justice about judicial appointments in November 1999 (paragraphs 2.25 2.31)
- the attendance of the Chief Minister at the Supreme Court in December 1999 (paragraphs 2.32 2.35).

The dates in this chapter are in the year 1999, save where otherwise indicated.

The address by the Chief Justice at the Opening of the Legal Year in October 1999

- 2.1 We begin this section of the chapter by referring to two matters which preceded the Opening of the Legal Year in October 1999.
- 2.2 The first related to the proposal by the Chief Justice in May 1998 that a number of members of the Bar should be given the opportunity to sit as Stipendiary Magistrates for 3 weeks in the coming 12 months. On 15 July 1998 the Chief Secretary wrote expressing the agreement of the Chief Minister with this proposal, saying that the procedure previously followed for the selection of the Stipendiary Magistrate should be followed, viz. internal advertisement, and, in the absence of a suitable candidate, external advertisement, followed by selection by a Public Service Commission Board. The Chief Justice was invited to identify suitable candidates. On 27 July 1998 the Registrar wrote to the Chief Secretary, saying that the Chief Justice was pleased that funding was available for this purpose and that he would discuss the way forward

with the Governor. On the same day the Chief Secretary wrote to the Registrar, saying that funding for the proposals would not be made available until the Government had agreed the way forward. The Chief Justice drafted a proposed advertisement inviting candidates to apply to him and on 20 October 1998 sent a copy to the Chief Secretary, saying that a copy was being given to the Deputy Governor to enable him to liaise with the Chief Secretary on the final version. According to the Government's press release of 28 October 1999, to which we will refer later, the long established practice was that the publication of invitations to apply, and the receipt of applications, for appointment as Stipendiary Magistrate were handled by the Government's personnel manager; and in December 1998 the new Deputy Governor confirmed that the usual procedure would be followed. On 18 December the Chief Secretary confirmed that funding was available. However, the press release stated, the Chief Justice decided not to proceed with his proposal.

2.3 The second matter arose out of a letter dated 10 November 1998 which the Registrar wrote on behalf of the Chief Justice to the Chief Secretary, saying that the Chief Justice considered that he should attend what the Registrar called a conference, but which the Chief Justice in his evidence called a seminar, at Warwick University in the following February on the subject of Lord Woolf's proposals for the reform of civil procedure, and seeking additional funding for the purpose. The letter stated: "He considers it vital that he attend". After further details had been supplied at his request the Chief Secretary responded on 1 December that the Government did not consider that the subject of the conference was of sufficient value for Gibraltar at that stage, and that funds for conferences were fully committed for the remainder of the financial year. However, in the event the attendance of the Chief Justice at the seminar was funded with money from the UK Government. Prior to the beginning of the legal year 1999/2000 the (Gibraltar) Government paid for a conference conducted by Lord Woolf in Gibraltar on the same subject and for court staff to go to England to be trained in the Woolf reforms.

2.4 In the course of his address at the Opening of the Legal Year on 8 October 1999 the Chief Justice said that the courts could not fulfil their responsibilities unless they had adequate funding and that once a budget had been approved "the courts should be free to administer that budget according to its own priorities". He referred to the then recently promulgated draft Latimer House Guidelines and their statement of four principles including that:

"Sufficient funding to enable the judiciary to perform its functions to the highest standards should be provided" and

"The administration of monies allocated to the judiciary should be under the control of the judiciary".

He then continued:

"The judiciary has encountered one or two instances in the past year where the denial or delay of the release of funds by the Government has had the potential to affect adversely the administration of justice. The matter is one of practical importance, but there is also a fundamental principle involved. The Chief Justices of the Commonwealth were all agreed that those who control the judiciary's purse strings exercise enormous influence and have the capacity to undermine the judiciary's independence. That is why the Latimer House Guidelines are expressed as they are. A system should be in place which on the one hand enables the judiciary to administer justice properly whilst on the other subjects the judiciary's budget to proper controls, but which enables the judiciary to function independently and without improper restraint. It is a matter which I shall be addressing in the coming year."

2.5 The Chief Justice's comments attracted the attention of politicians and the press. In the *Gibraltar Chronicle* of 15 October Dr Joseph Garcia, the leader of the Liberal Party in Gibraltar, was quoted as expressing concern at "the claims of potential or perceived interference by the ... Government with the judiciary in Gibraltar", and calling for a public inquiry. When asked by the newspaper to comment on the exact meaning of what the Chief Justice had said, a spokesman for him said that if the Chief Justice had wished to elaborate he would have done so in his speech.

- 2.6 In a press release dated 15 October the Government dismissed the call for a public inquiry, and said that the financing of the judiciary was not subject to any political interference, the budget of the judiciary being agreed with the judiciary at the start of the financial year. The Government said that it had only denied additional funds in two instances, namely for international conferences and social entertainment.
- 2.7 In a press release on 18 October 1999 the Chief Justice said that the two instances mentioned by the Government were not those which he had had in mind as having the potential adversely to affect the administration of justice. He repeated verbatim what he had said in his address on the question of funding for the judiciary and said that he was unable to give details of the two instances until he had considered the implications of revealing the contents of confidential files. His comments in his press release were referred to in an article in the *Gibraltar Chronicle* of 19 October 1999.
- 2.8 On the same day the Government responded by a further press release, saying that it had received no representations, in confidential correspondence or otherwise, complaining about insufficiency of financial resources to enable the judiciary to perform its functions, and that it was surprised that this issue had been raised publicly in a manner that was "open to damaging misinterpretation". The Chief Secretary had been instructed to write to the Registrar, asking for details of the two instances. It also pointed out that it had agreed to a budget for the judiciary in excess of that which the judiciary had requested, and that applications for additional funds were considered on a case by case basis.
- 2.9 The then Chief Secretary, Mr Ernest Montado, on the following day wrote to the Registrar, saying that it was regrettable that the concerns raised by the Chief Justice had not been brought to the attention of the Government directly but had instead been made public in a manner which could only result in uninformed speculation, misinterpretation and political manipulation. The Government had been criticised in regard to matters which had not been explained to it, and which were said to be the subject of confidential correspondence, thereby further fuelling damaging public

speculation. He therefore requested full details of the two instances including relevant correspondence.

2.10 On 22 October the Registrar responded, identifying the two instances:

"The first instance the Chief Justice had in mind concerned the delay in the decision to finance agreed proposals for recruitment of members of the Bar to train as Stipendiary Magistrates. The second concerned the refusal of funds for participation in a Judicial Studies Board training seminar on the Access to Justice reforms".

- 2.11 The Government then issued a Press Release on 28 October, making public the information given by the Chief Justice, and stating its position in regard to the two matters mentioned in the above letter. In relation to the first instance it set out the chronology, including the agreement of the Government to implement the Chief Justice's proposal and the subsequent decision of the Chief Justice not to proceed with it. It denied that there had been any delay in the Government agreeing to finance the proposal and denied that the administration of justice could be adversely affected. It set out what it said was the long established practice in Gibraltar in the matter of judicial appointments; that, it said, included the Governor consulting with the Government through the Chief Minister before the Governor made the appointment. In relation to the second instance it accepted that in November 1998 funds were requested but not approved for the Chief Justice's attendance at a conference in England but said that the costs of that attendance were paid by the FCO. It pointed out that in August 1999 the Government had paid for the judiciary and other officials to attend a conference in Gibraltar on the same subject and organised by the Bar and conducted by Lord Woolf, and had provided further funds to enable some of the court's officers to go to UK courts on attachment for training on Access to Justice reforms. The Chief Minister subsequently accepted in a Parliamentary debate on 20 November 2000 that it would have been better if funding had been made available for the Chief Justice to attend the conference in question.
- 2.12 The Chief Justice then instructed English Counsel, John Causer, directly and not through solicitors. On 20 October and 2 November the Chief

Justice had two brief telephone conversations with Mr Causer, who found that neither was very satisfactory as he was speaking on a mobile phone from a public place. He understood from the conversations with the Chief Justice that "the Government had trivialised the funding issue by implying that the Chief Justice was in effect seeking funding for cocktail parties when in fact an entertainment budget was only a tailpiece to the main issues". The Chief Justice then faxed a one page letter of instructions to Mr Causer, with a draft press release which he wished to issue. In that letter the Chief Justice referred to his address at the Opening of the Legal Year and to the two instances where, he said, the delay in or denial of funding had had the potential to affect adversely the administration of justice. He said:

"Political clamour for details of the two instances has led the Government to issue a press release explaining what they are. Unfortunately the Government has put such a "spin" on the facts that I have to put the facts right."

2.13 It should be noted that the Chief Justice did not provide Mr Causer with his address, or its press coverage, or the political comments which it excited, or the press releases of 15, 18 or 19 October, or the letters of 20 and 22 October, or, most importantly, the Government press release of 28 October referred to in that quotation. In his letter of instructions he drew counsel's attention to the references in the draft press release to certain items of correspondence with the Government and to the fact that he had alluded to conversations with the Governor. He did not supply Mr Causer with that correspondence. He asked if the release of the draft would offend the provisions of the Official Secrets Acts. Mr Causer's advice dated 6 November was that both of "the contentious matters" (i.e. the two instances mentioned in the draft press release) would involve revealing the decisions of the Government on funding matters and its motivation and that this would come within the Official Secrets Acts and so the Chief Justice would be exposed to danger. It should also be noted that there appears to have been no recognition in that advice that the Government had already put into the public domain all the relevant facts relating to the two instances by its press release of 28 October.

- 2.14 The Chief Justice did not issue the draft as a press release, but on 11 November wrote a letter to the *Gibraltar Chronicle*, in which he repeated the substance of what he had said in his address, referred to the Government press release of 18 October and his own press release of the same date, and said that he had considered the legal implications of giving details of the two instances and had taken the advice of counsel who confirmed his view that he was constrained by the Official Secrets Act 1911 from divulging information in his possession.
- 2.15 On 16 November the Government issued a press release commenting on the Chief Justice's statement in his letter to the *Gibraltar Chronicle* that he was constrained from divulging information in his possession. The Government said that that suggested that it had information which contradicted what it had said. It said that it was not aware of such information, but that if there were any such documents belonging to it and they were identified by the Chief Justice, it would publish them. The Chief Justice did not respond to that invitation.
- 2.16 The evidence before us highlighted a number of serious concerns in regard to the Chief Justice's address and the events that followed.
- 2.17 First, the remarks of the Chief Justice were seriously inaccurate and misleading. Behind the matter of funding for part-time magistrates lay the question of the method of arriving at their appointment. The proposal for such appointments had not "petered out", as he sought to represent in his oral evidence. It had been dropped by him, despite the merit which it had of clearing a backlog in the Magistrates Court and creating a pool of Gibraltar lawyers with judicial experience. He dropped it because the Government's approval of funding was, in his words, "on condition that it had a say in the appointments" and would have enabled the executive "to put their own appointees in the judiciary". However, the true position was that the Government did not accept his proposal for a new and different procedure under his control, but insisted on the usual procedures for such judicial appointments - which would not have given it control. As regards his attendance at the seminar, we question whether it was "vital" that he attend it, especially when a committee which he

convened postponed the coming into operation of the Woolf reforms for a year. Quite apart from these facts, the Chief Justice did not disclose that both matters to which he had alluded had been resolved: funding for his first proposal had been offered in December 1998; and funding for the second had been met by the UK Government. Moreover, he could have no complaint about the judicial budget, which had been met in full and more. He failed to make it clear that his complaints related to requests for funding in addition to the budget.

2.18 More seriously still, his remarks suggested that the independent functioning of the judiciary had been at risk of being subject to improper restraint by the Government. In his first and third witness statements he said he profoundly disagreed with the view that it was not appropriate for him to make pronouncements on controversial matters, and that he was concerned that the Chief Minister could use funding as a means of controlling judicial training and appointments. He accepted, however, that he had not previously suggested that the conduct of the Government had breached any international standard, or that the instances referred to by him had affected the administration of justice. He did not explain that neither of the two instances would have breached the draft Latimer House Guidelines, which in any event at that time did not set internationally accepted standards and which, when eventually approved in 2003, omitted the principle that the administration of monies allocated to the judiciary should be under the control of the judiciary. If there really was some cause for thinking that these instances had a potential to affect adversely the administration of justice, we do not understand why he did not raise this privately with the Government. In his oral evidence he said that the fundamental principle to which he had referred in his speech was that adequate funding must be made available for things such as judicial training, but he had not asked the Government for a judicial training budget. He gave no prior notice to the Government of his intention to raise any of these points. It had been his practice to ask the Attorney General and the Chairman of the Bar Council for copies of their addresses, but it never occurred to him that it might be courteous or good practice to

provide anyone with a copy of his address. The Chief Minister said in evidence:

"I think what any Government anywhere in the world would reasonably expect of its Chief Justice is that if he wants to take issue with the Government on something, that he should raise it privately before going to war publicly."

- 2.19 Instead he spoke publicly of these instances, in a manner which was highly critical of the Government, and in language which was so unspecific as to create an obvious risk of speculation. As the *Gibraltar Chronicle* in its editorial of 19 October put it, he "[left] the issue in the air, allowing speculation to breed". The Chief Justice demonstrated a complete lack of judgment in his choice of language. According to him, it was "rather a mild mention of the problems with funding". He was not certain that it would have been better to have given some general details of the two instances. In our view the language, tone and manner of his remarks were inappropriate for the holder of his office.
- 2.20 The Chief Justice went on to compound the difficulty created by his remarks. He said that he had not expected the reaction they had received. As regards the references in the press to his address and to Dr Garcia's call for a full public inquiry, he said he was surprised, but not horrified, at the way in which his remarks had been taken up. He said he was concerned that Dr Garcia had gone "somewhat over the top" and that his moderate remarks had been distorted and used inappropriately. However, he did not consider approaching the Government and explaining that his own remarks had been distorted and used inappropriately because "once one enters that arena it would be, I think, entering dangerous territory for me to start making comments that may be published, misinterpreted, and I felt that at that stage it would all die down and it was prudent for me to maintain silen[ce]".
- 2.21 It was, in our view, inappropriate for the Chief Justice, having excited the furore over his speech by his ambiguous and unspecific remarks and knowing that what he said had been misunderstood and distorted, to refuse to clarify the position publicly. We conclude that, having been

- thwarted initially in obtaining the fulfilment of his wishes in relation to the two instances to which he had alluded, he was determined to embarrass the Government publicly.
- 2.22 His suggestion that he might be inhibited from revealing certain matters on the ground that they were confidential was inconsistent with his having alluded to them in the first place. He hinted that there was documentation supporting his concerns but never revealed what was that documentation, even when the Government offered to publish anything he identified. When asked about the reference to confidential files in his press release of 18 October, he said that he could not recall considering going to the Government and asking it to agree to his referring to confidential matters.
- It is also clear, on his own evidence, that it never occurred to him at the 2.23 outset that the contents of his address might have implications under the Official Secrets Acts. That thought was put into his head later by someone, but he could not remember by whom. It did not prevent him telling the Government what were the two instances. Even more unconvincing is his subsequent reliance on Mr Causer. His instructions to counsel were woefully incomplete and tendentious, and the advice was flawed in consequence. It is plain that the Government was not trivialising the funding issue by reference to cocktail parties, yet that is how he chose to characterise what it had done. His draft press release did not have to include any reference to what he counselled the Governor nor to the correspondence with the Government if those were the matters which concerned him. In regard to Mr Causer's advice the Chief Justice said that he was "puzzled as to why the press release of 28 October is not fully addressed", and that he was surprised by the advice, as he had hoped that it would clear him in Official Secrets terms. When asked further questions by the Tribunal on this point two days later the Chief Justice said that it was a misunderstanding that he had expected Mr Causer to deal more fully with the press release. He accepted that he had no answer to the question how Mr Causer could advise without knowing

- what had already been put in the public domain, when his instructions omitted any reference to the Government's press release of 28 October.
- 2.24 We cannot escape the conclusion that the alleged concern about confidentiality and the Official Secrets Acts was an irrelevant diversion from answering the substance of the point against him, namely that he had not told the Government about his concerns privately, so that it could be seen whether they truly had the potential to affect adversely the administration of justice.

Remarks on judicial appointments in November 1999

- 2.25 In the press release of 28 October, to which we referred in paragraph 2.11, the Government gave details of what it said was the long-standing practice in all judicial appointments, which included that the Governor consulted with the Government through the Chief Minister, after which the Governor made the appointment. In his letter to the *Gibraltar Chronicle* of 11 November (see paragraph 2.14) the Chief Justice challenged the correctness of that statement of practice. He said that it was the first time he had heard of such a practice. He repeated his views on the independence of the judiciary, and suggested the appointment of an independent Judicial Service Commission for making or recommending the appointment of magistrates and judges.
- 2.26 The issue of "the long-standing practice" received further discussion in the press. The *Gibraltar Chronicle* of 11 November commented that it was "not good to have two key institutions such as the Government and Chief Justice grating against each other". On the same day the Governor issued a statement noting that the public debate had widened to judicial appointments for which he was responsible. He said that the Governor alone took the final decision on these appointments and that during his term of office he had seen no evidence that any judicial appointment had been subject to political interference. He said:

"I regret that this public discussion is now drawing the judiciary and office of Chief Justice into political debate and causing public confusion and anxiety."

- 2.27 On 16 November Mr Montado, the Chief Secretary, issued a statement setting out his experience since 1986. He said that all judicial appointments were made by the Governor, but that the established practice in the case of the Chief Justice, Additional Judge and Stipendiary Magistrate was for a board to be constituted to interview candidates and to advise the Governor. The Chief Secretary, he explained, was normally a member of the board, representing the Government. The Governor's practice, he said, had been to consult the Chief Minister informally, either directly or through the Deputy Governor, before making his decision. He said that the practice was clearly reflected in Government files going back more than 20 years.
- 2.28 In his first witness statement the Chief Justice denied that he had cast doubt on the veracity of government statements. He said that the subject matter of his comments was properly a matter for his concern and that it was appropriate for him to express his views in the manner he did. The appointments made in his first year of office as Chief Justice by the Governor, Sir Richard Luce, did not follow any process of formal consultation with the Chief Minister. He said in evidence that he was unaware of any past practice of formal consultation with Chief Ministers in relation to judicial appointments, and that his predecessor, Sir Alister Kneller, had told him that he was unaware of such a practice. He continued:

"Genuinely believing that there was no practice of consultation, I felt it would be wrong for me to remain silent in the face of an assertion that there was a practice of consultation lest it be thought that I was accepting a convention which did not exist. So I publicly declared my belief that there was no practice of consultation, to try to hold the line."

- 2.29 When asked in cross-examination why he had not gone privately to the Government and asked it to explain to what consultation practice it was referring, the Chief Justice said that from time to time it became obvious that the Government was "unwilling to engage with me, the judiciary, and when the temperature is such that such engagement will not bring about any results, then in my judgment, knowing the situation in Gibraltar, I have to resort to other methods". He explained that the other method was that of going directly to the press and dealing with the matter through the press and that that method was open to him "because I did not want there to be perceived to be a public acknowledgment by silence that there was a process of consultation".
- 2.30 We think it plain that the Chief Justice did court controversy with the Government by casting doubt on the veracity of government statements on the practice of judicial appointments being made by the Governor after consultation with the Chief Minister. He was wrong to do so because there was such a practice. Contemporaneous documents produced by the Government from the time when Mr Joe Bossano was Chief Minister (1988-1996) amply bear that out. Not for the only time, the basis of an erroneous belief of the Chief Justice is said by him to come from an oral, undocumented and uncorroborated communication with Sir Alister Kneller. The interpretation which Mr Bossano himself placed on the word "consultation" in his evidence does not square with its ordinary meaning. He said that it meant being given an opportunity to change the result, so that for him consultation meant "you actually finish up with a deal that has an input from you".
- 2.31 In cross-examination the Chief Justice described how the issue between him and the Government had died down after he received Mr Causer's advice, but then, he said, "it resurrected itself, not in terms of funding, but in terms of the issue of consultation over judicial appointments". In reality, as was pointed out by Counsel to the Tribunal, it was the Chief Justice himself who resurrected matters by his letter to the newspaper. Instead of attempting to find out privately from the Government the basis of its claim that there was such a practice, he chose by means of a

tendentious letter to the press to raise publicly another disagreement with the Government, as well as continuing to maintain that he was constrained by the Official Secrets Acts from divulging information in his possession. As we noted in paragraph 2.28, the Chief Justice asserted that by silence he might be perceived publicly to acquiesce in a Government statement with which he disagreed. This is in marked contrast with the fact that at no time in relation to his own (or his wife's) statements and their interpretation was he prepared to acknowledge that because of the public perception of acquiescence by silence, an impression might have been created which, if incorrect, fairness would demand that he should correct.

The attendance of the Chief Minister at the Registry of the Supreme Court in December 1999

2.32 It is not in dispute that when Sir Joshua Hassan QC was the Chief Minister, he used to visit staff not only in Government departments but also the Registry of the Supreme Court and Magistrates Court during the Christmas or New Year period, by way of extending Christmas greetings to staff (who were civil servants), and thanking them for their efforts during the year. The present Chief Minister, Mr Peter Caruana QC, has followed the same practice. On 24 December 1999 the Chief Secretary and then Acting Deputy Governor, Mr Montado, sent a note to the Governor in which his first words were:

"I feel duty bound to draw to your attention a matter concerning the Chief Justice which I consider to be deplorable."

He recorded what he had been told by the Registrar, Ms Katherine Dawson, the previous day by telephone. This was that the Chief Justice had instructed her that, were the Chief Minister to visit the Registry, as had been programmed for 30 December, he was not to be met by anyone but should be left to call at the counter and ring the bell for attention; that staff were to remain at their desks; and were those instructions not followed he would consider it an act of disloyalty towards him. Every

- time the Chief Minister called before and after 1999 Ms Dawson and the staff continued to meet and greet him despite the Chief Justice's objections.
- 2.33 In her witness statement Ms Dawson confirmed the accuracy of Mr Montado's note, and described the atmosphere at the Registry as having been "intimidatory". In her oral evidence she accepted that it could well have been her impression (rather than the actual words of the Chief Justice) that he would consider disobedience to his instructions an act of disloyalty towards him. As for the atmosphere at the Registry being "intimidatory", she accepted in cross-examination that it might be better to say that the staff felt uncomfortable because they knew that, when the Chief Minister visited the Registry, if the Chief Justice came in he was going to be extremely unhappy. The Deputy Registrar, Mr Clive Mendez, said in his first witness statement that he knew that the Chief Justice objected to the Chief Minister's visits to the Registry and that this made him and the staff feel uncomfortable. While Ms Dawson was adamant that Mr Bossano did visit the Registry every year and that she remembered him coming, Mr Mendez said that Mr Bossano, not being a lawyer, never did. Mr Bossano said in evidence that it might well be that he popped into the Registry to have a drink with Ms Dawson because they have been friends for a very long time but that he had never made a formal visit to the Registry.
- 2.34 In his first witness statement the Chief Justice said that he thought it not right for the Chief Minister to treat the Supreme Court and its staff as if it were another Government department, but denied that he gave the instructions recorded by Mr Montado. He said that in 1998 he had made a "discreet suggestion" to the Registrar that she should suggest to the Chief Secretary that the visit not take place. He accepted that "very probably" he told the Registrar that "it would be appropriate if the Chief Minister were not met by a fanfare" language disputed by Ms Dawson. He denied that he said that if staff met the Chief Minister as the Chief Minister proposed he would consider it an act of disloyalty. He also

- accepted that he chose to work at home when the Chief Minister called "to avoid any collision in front of staff".
- 2.35 This episode demonstrates the seemingly instinctive hostility felt by the Chief Justice towards the Chief Minister. We find that there was a practice throughout Sir Joshua Hassan's term as Chief Minister to pay an informal goodwill visit to the Registry. Mr Bossano chose not to follow that practice, but the practice was resumed by Mr Caruana. The purpose of the visit was to see not the judiciary but the civil servants who staffed the Registry to thank them for their work. They liked the Chief Minister's visits. A more innocent visit at a festive time of the year it is hard to imagine, and yet the Chief Justice took grave exception to such visits. We do not find that the Chief Justice told the staff that it would be an act of disloyalty for them to disobey his instructions, nor do we find that the atmosphere was intimidatory, but we think it plain that the Chief Justice made known his opposition to such visits and that this was uncomfortable for the staff. We accept Ms Dawson's evidence that the Chief Justice indicated to her that the staff should remain at their desk and let the Chief Minister ring the bell for attention. The petty discourtesy thereby intended to be shown to the Chief Minister was obvious and unwarranted.

CHAPTER 3

2000 - 2001

This chapter deals with:

- the maids issue events up to October 2000 (paragraphs 3.1 3.27)
- instructions to the Registrar of the Supreme Court in respect of expenditure in May 2000 (paragraphs 3.28 3.30)
- the allegations of interception of the Chief Justice's telephone communications in 1999 and 2000 (paragraphs 3.31 3.39)
- the defence advanced on behalf of the Chief Justice on his prosecution in 2001 for an offence under the Motor Vehicles Test Regulations (paragraphs 3.39 3.57).

The maids issue events up to October 2000

- 3.1 From at latest September 1996 until March 1997 the Chief Justice employed a Jamaican national, Ms Jackie Williams, as a domestic in Gibraltar. He did not apply for or obtain a work permit for her, but, according to his evidence, it had been left to her to regularise her own position. On 18 March 1997 the Chief Justice signed three forms for the Employment and Training Board (ETB) in respect of her: (i) a Notification of Vacancy for a domestic working 39 hours a week; (ii) a request for the issue of a work permit for her as a domestic; and (iii) a Notice of Terms of Engagement, which later showed that her terms of employment were for 39 hours per week at £3.15 per hour. On 26 March 1997 a work permit valid for one year was granted to the Chief Justice for her.
- 3.2 On 15 October 1997 the Chief Justice wrote to Inspector Santos of the Visa section of the Government about Ms Williams, stating that, having employed her in the Cayman Islands, he had offered her a new contract when his family moved to Gibraltar. However, that contract had been

terminated on three months' notice on 31 January 1997. She was then reemployed for three hours a day three days a week at £4 per hour. A formal notice terminating her contract, he said, was issued on 1 October 1997. In an undated document, signed by Ms Williams and apparently prepared about 17 October 1997, she acknowledged receipt from the Chief Justice of £786 in settlement of all sums due to her on termination of her employment. They included salary from 1 to 17 October 1997 of £140, a sum which is difficult to reconcile with the figures given to Inspector Santos.

- 3.3 On 22 October 1997 the Chief Justice obtained from the ETB copies of the forms relating to Ms Williams, and he wrote to one of its officials saying that they did not reflect the terms of service under which she was employed. Ms Williams had told Mrs Schofield that the Chief Justice had signed the forms and that the official had completed the details after speaking to him. The Chief Justice also said that he did not recall speaking to the official. He stated that he did realise that Ms Williams needed a work permit but said that his very busy schedule had not allowed him to follow up the matter.
- In his fifth witness statement the Chief Justice said that the Notice of Terms of Engagement had been completed by Ms Williams who "handed it to me as I was dashing back to court one day and, rather stupidly I now realise, I signed it in blank." He said that he did not suspect that she would complete it inaccurately to ensure that she qualified for a work permit. He drew attention to some spelling mistakes typed in the Notice, suggesting that he would not have subscribed to them. It is difficult to reconcile the Chief Justice's statement that he signed one form in blank with his acceptance under cross-examination that he signed no less than three forms in blank, although he had "no positive recollection" of having done so. In re-examination, however, he said;

"I recall distinctly now Ms Williams accosting me in the kitchen of the house as I was dashing back to court, saying --- and I can't remember that she said the [ETB], but they need me to complete these forms. Do you mind signing them, Mr Schofield? And I did."

- 3.5 With some hesitation we are inclined to accept that the Chief Justice signed the ETB forms in blank, that being consistent with what he told the Governor in a letter dated 16 June 2000 (see paragraph 3.11). However, the Chief Justice accepted in cross-examination that the office of a judge carries with it particular responsibilities in relation to the handling of personal affairs, all the more so in the case of the office of Chief Justice; that public confidence in that office demanded the highest standards of integrity; and that the holder should exercise particular diligence to ensure compliance with the laws which he is administering. He further accepted that his failure to obtain a work permit for Ms Williams when first employed in Gibraltar was in breach of Regulation 7 of the Employment Regulations 1994, and constituted a criminal offence by him under Regulation 5, which placed the obligation to obtain such a permit on the employer. Having left it to Ms Williams to complete the ETB forms, she was his agent in doing so, the responsibility for the accurate completion of the forms resting with him. He recognised that the forms were important, with legally significant consequences. He also accepted that it was an offence to make a false statement for the purpose of obtaining a work permit, and that the purpose of the notification of details of the proposed employment was to enable the ETB to consider whether the person in question should be granted a work permit. He could not recall taking any steps to check the position in regard to Ms Williams obtaining a work permit, but accepted that it was inconceivable that, if he had taken such steps, he would not have realised the true legal position. He also agreed that it was a matter of some importance that he complied with the laws of Gibraltar, given that he was coming to a new jurisdiction as Chief Justice. He accepted counsel for the Government's summation of his conduct that it amounted to "a reckless disregard" on his part for legally important forms.
- 3.6 Throughout Ms Williams' employment by the Chief Justice in Gibraltar he did not deduct any PAYE nor pay any social security contributions. He said in his fourth witness statement that she did not qualify to pay income tax because of her low salary and he made no deductions from her salary.

Nor, he said, did she qualify for the purpose of social insurance contributions. In his fifth witness statement he said that she worked less than 15 hours per week and earned £150 per month, adding "This brought her outside the limits for social security and PAYE contributions". He went on to point out that he had never been contacted by any Government department about any liability in respect of Ms Williams.

- 3.7 An examination of the relevant regulations relating to PAYE (viz. the PAYE Regulations 1989) showed, as the Chief Justice accepted in cross-examination, that there was a clear obligation on an employer to deduct PAYE at source, whether or not the employee had obtained a tax code from the Income Tax Office. The Chief Justice further accepted that the obligation did not depend on the number of hours worked by, or the amount of remuneration paid to, the employee. While there were tax thresholds for the employee's personal tax position, they were separate from the employer's obligation to pay PAYE. In regard to social insurance contributions, it was not in dispute that even if an employee worked less than 15 hours a week the employer was obliged to pay social insurance contributions at a reduced rate, these being the employment injuries element of the composite social insurance contribution.
- 3.8 In cross-examination the Chief Justice accepted that he was wrong about social insurance contributions, and that his counsel was in error in asserting that the Chief Justice was correct in making no deductions from the low salary of Ms Williams. He maintained that he believed in 1997 that there was no PAYE liability or liability for social insurance contributions, but when asked for the basis of that belief he was unable to provide an answer. He accepted once more that it was inconceivable that, if he had taken steps to check what was his position as an employer in relation to PAYE and social insurance contributions he would not have realised the true position in relation to those obligations. He confirmed that at no stage before 2000 did he take legal advice about Ms Williams or Ms Hermina Danvers to whom we will now turn.
- 3.9 By a written contract dated 9 October 1997 the Chief Justice employed Ms Danvers as a housekeeper/cook at a monthly salary of £450. On 9

December of that year he obtained a work permit in respect of her. She was so employed by him from December 1997 to April 2000. In April 2000 she lodged a complaint with the Ombudsman and shortly afterwards with the TGWU which took up her case that she was paid less than the Gibraltar minimum wage - an allegation which was never substantiated.

- 3.10 On 4 May the Chief Justice met the Commissioner of Income Tax to discuss his failure to deduct income tax from Ms Danvers' earnings and told him that he had mistakenly assumed that no tax was payable. An article in *Vox* on 26 May 2000, under the heading "M'Lord, you have broken the law", said that it was understood that the Chief Justice was due to pay arrears of income tax and social insurance contributions in respect of Ms Danvers over a period of about three years. On 30 May 2000 the Chief Justice wrote to the Governor, David Durie CMG, apprising him of the situation in relation to Ms Danvers. He said that he had approached both the Income Tax Department and the Department of Social Security, and that he had already paid the amount assessed by the latter. The Governor replied on 1 June 2000, saying that he would be making his own assessment of the matter, and issued a public statement in which he said that it was important that the issue should not be allowed to bring the administration of justice into disrepute. With a view, therefore, to making his own assessment he had asked the Chief Justice to let him have all relevant information relating to the circumstances surrounding the employment of Ms Danvers. Following the meeting on 4 May the Commissioner of Income Tax indicated to the Chief Justice that he was prepared to accept payment of tax in the sum of £2,473.20, the payments to Ms Danvers being treated as grossed up in accordance with the appropriate code. The accountant of the Chief Justice looked at the Commissioner's figures and suggested that only £990.30 was owed. This was paid on 28 June.
- 3.11 The Chief Justice in his oral evidence said that it was inconceivable that he did not mention Ms Williams also to the Commissioner of Income Tax, but we do not accept this as there is no mention of her earnings and tax position in the correspondence with the Commissioner. Further, the

position of Ms Williams became public only following an article in *Vox* of 9 June 2000, under the heading "The Chief Justice must now resign", in which it was suggested that the Governor should also look into the Chief Justice's employment of Ms Williams, asking "Were her contributions for social insurance and income tax paid up?". In a further letter to the Governor on 16 June 2000 the Chief Justice said that he had signed Ms Williams's work permit application form in blank but that she had completed it giving her old terms of service. His recent researches revealed that he might have been liable to pay social insurance contributions in respect of her.

- 3.12 When by late June the Governor had not reached a decision, some dissatisfaction was expressed to him at his delay. On 28 June 2000 five Queen's Counsel sent a memorandum to him in which they said that it was a matter of regret that some four weeks after announcing that he would assess the allegations against the Chief Justice he had not announced whether he had found any grounds for establishing a Tribunal of Inquiry under the Constitution. They said that the delay had resulted in "division between those whose views on the quality and integrity of the administration of justice permit them to regard the allegations as a storm in a teacup and those who cannot so regard them".
- 3.13 On 3 July 2000 the Chief Justice wrote to the Governor, informing him that he had placed the matter of Ms Williams before leading counsel. The letter continued:

"Having perused the legislation with some care counsel has advised that by reason of its terms and the hours she worked she does not come within the legislation. [Ms Williams] did not earn sufficient wages to come within the PAYE legislation."

3.14 The Chief Justice was asked about the contents of that letter in cross-examination. He said that the counsel referred to was either Desmond de Silva QC in London or Guy Stagnetto QC in Gibraltar. The Chief Justice accepted that in 2000 he had not asked his legal advisers to advise about his obligations in 1997; that was because he felt the Williams matter was closed. He further accepted that the picture presented to the Governor

was that there was never any obligation on the Chief Justice to deduct PAYE from Ms Williams' salary because she fell outside the legislation, and that it was impossible to square this with the legislation. Although this Tribunal ruled that the Chief Justice waived privilege in respect of the legal advice he received and although the Chief Justice had said through his solicitors by letter dated 14 July 2008 to the Solicitors to the Tribunal that Mr Stagnetto QC had advised him in relation to Ms Williams, no instructions to Mr Stagnetto nor record of any advice to the Chief Justice were found in Mr Stagnetto's file. Indeed in an e-mail dated 21 July 2008 from Mr Stagnetto to the Chief Justice's solicitors, Mr Stagnetto emphasised the limited nature of his instructions from the Chief Justice. He said that the Chief Justice had informed him that Ms Williams' pay did not reach the threshold to bring her within the PAYE legislation and that he replied to the effect that, if that was so, the Chief Justice had nothing to worry about and should inform the Governor accordingly. The Chief Justice's solicitors said that Mr de Silva had no memory of any advice he might have given to the Chief Justice.

- 3.15 In his fourth witness statement the Chief Justice said, in relation to Ms Williams, that he co-operated completely with the Governor "and gave him a full explanation". This is plainly untrue. In his oral evidence he attempted to maintain that his explanation was full and fair "in relation to the advice I received". However, he went on to recognise, though somewhat reluctantly, that his letter to the Governor of 3 July 2000 was misleading in that it indicated that he had been advised that he had never been under any liability in respect of PAYE, whereas the advice was that he had no outstanding liability as at the date in 2000 when the advice was given.
- 3.16 On 23 August 2000 the Chief Justice, who had been informed by the Governor of what he was proposing to say in respect of Ms Danvers, wrote to the Governor urging him also to make a determination in relation to Ms Williams. By letter dated 31 August 2000 the Governor wrote to the Chief Justice, enclosing a copy of the statement which he proposed to

make the next day but which in the event was not issued for over a month and then in a revised form. The letter included:

"In your letter of 23 August you raised the matter of Ms Williams. Since at an earlier date you chose not to provide me the extra information for which I asked in relation to Ms Williams, I continue to rely on the assurance which you gave me on 12 July in respect of her. I have conducted no separate investigation in her regard."

The allusion to extra information appears to be a reference to a passage in the letter from the Chief Justice of 16 June 2000 to the Governor, where he told him that he was unsure whether he could give any authority for the release of confidential information by the Income Tax and Social Security Departments.

3.17 It was not until 4 October 2000 that the Governor announced his decision. In it he said that Ms Danvers' employment had been registered and "all outstanding PAYE payments and social security contributions have now been paid". He continued:

"It is regrettable that matters were not regularised at an earlier stage but I have accepted that the Chief Justice did not deliberately seek to avoid his obligations.

The Chief Justice has also assured me, in relation to another former employee of his, Ms Williams, that he has no outstanding liabilities.

I have concluded, in view of the information and assurances which I have received, that it would not be appropriate for me to take any formal action in exercise of my constitutional powers."

3.18 On 9 October 2000 the Government issued a press release in which it stated that since the Governor's statement stated only part of the facts, the Government considered that the public interest required that the facts of the cases of Ms Williams and Ms Danvers, as known to the Government, be placed in the public domain. Those facts included that in the case of Ms Williams the Department of Social Security had no record of social insurance contributions having been paid in respect of her employment and that the Commissioner of Income Tax had no record of tax having

- been paid in respect of her employment and had no record of her employment. In relation to Ms Danvers it said that PAYE payments and social security contributions in respect of her employment during the period December 1997 to 3 April 2000 were made in June 2000.
- 3.19 In his fifth witness statement the Chief Justice said that Ms Williams was outside the limits for social security and PAYE contributions. In crossexamination he said that he realised his mistake only when he heard the Chief Minister give evidence. In the same statement he said that the Government's press release seemed to suggest that contributions were due and unpaid in respect of Ms Williams and described that as a misleading innuendo since no contributions were due and unpaid. However, in cross-examination he accepted that there was no misleading innuendo in this press release, which was entirely accurate. When asked why he had instructed his counsel to run the maids issue as part of the argument that the Government was acting in bad faith against him, he said that it was because he genuinely believed at that stage that there was no liability in respect of social insurance contributions or PAYE. It seems to us that, if he had been taking the maids issue as seriously as he should have done, he would have found out the true position before he gave formal evidence in a witness statement that he had had no obligation to make the contributions, and, even more seriously, an accusation that the Government had been acting in bad faith.
- 3.20 There is one other feature of the maids issue which is of some relevance. The Chairman of the Bar in 2000, Robert Vasquez, gave evidence in his witness statement of speaking on the telephone to Mrs Schofield, with whom he was then friendly, on this topic. On 29 May 2000 Mr Vasquez made a note of a telephone conversation which he had with her. In his oral evidence he corrected and expanded on his witness statement. First, he said that in April 2000 he had what he described as "a very odd conversation" with her when she asked him questions about the payment of domestic staff in Gibraltar. There was, he said, a second conversation late on 29 May 2000 when she telephoned him from London. He made a note of the conversation the next day, which he provided to the Chief

Minister because of what Mrs Schofield said about the mother of the Chief Minister, who was Mr Vasquez's aunt. Mrs Schofield said to Mr Vasquez that she was wanting to hear what people were saying about the press reports on Ms Danvers. She said that the employment of illegal domestics was widespread and that generally people did not pay social security for domestics. In this context she referred to the Chief Minister's mother, but Mr Vasquez told her that that accusation was baseless. She said it was a can of worms but if people wanted it opened, she would open it up and would go not to the local gutter press but to the international press with everything. She twice posed the question:"[W]ould Gibraltar's finance centre withstand such publicity?" She said that she was fighting for her husband and children and that if Gibraltar wanted to remove him it would have to go through the whole procedure all the way to the Privy Council and she would fight all the way.

- 3.21 Mrs Schofield in her second witness statement refers to this note as "not on all fours with the conversation" which she had with Mr Vasquez. However, she said that his statement did correspond with her recollection of the conversation save for two details which she did not recall saying. She also said that she told Mr Vasquez to tell the Chief Minister that if he fought her husband in the gutter she would meet him in the gutter. The Chief Justice in cross-examination accepted that his wife had told him that. We have no hesitation in accepting the note as an accurate record of the conversation between Mrs Schofield and Mr Vasquez.
- 3.22 There is another piece of evidence that accords with Mrs Schofield having said that she was prepared to fight in the gutter for her husband. The Attorney General, Reginald Rhoda QC, said in his witness statement that at the time when the maids issue was in the press and after court one day the Chief Justice in his chambers told him that Mrs Schofield had said to him that the Chief Justice could not be involved in a public controversy but that she could get down into the gutter and fight. The Chief Justice in his fifth witness statement said that the language attributed by the Attorney General to him was "not my language". However, the Attorney

General gave evidence of having a distinct recollection of what the Chief Justice had said. As he explained:

"The importance of it is that it has always stuck in my mind when I saw what was happening thereafter, that what was happening thereafter was putting into practice the very thing that the Chief Justice had said to me, that Anne said she can get down in the gutter and fight."

We have no doubt but that the Attorney General's evidence is to be preferred to that of the Chief Justice on this point.

3.23 The maids issue and the Governor's decision caused considerable controversy in Gibraltar and troubled legal practitioners in particular. The Opening of the Legal Year was boycotted by the majority of barristers and solicitors. The Chairman of the Bar traditionally makes a speech at the ceremony. Mr Vasquez, who had been Chairman of the Bar since 1999, formed the view that in his speech at the ceremony on 6 October 2000 he should refer to the issue relating to Ms Danvers, as the Governor's statement on 4 October 2000 was in his opinion not entirely satisfactory. He drafted what he proposed to say and called a meeting of the Bar Council so that he could consider the views of the members on his draft. He intended to resign as Chairman if there were objections. In his draft he said that the Governor's statement fell short of repairing the prejudice caused and restoring confidence in the administration of justice, and that he did not know if the difficulties could be surmounted. No one at the meeting supported the making of a statement in terms of the draft. However, he was persuaded not to resign but instead to make no speech, simply moving the Opening of the Legal Year. He agreed to that course. After the ceremony the Chief Justice gave a drinks party following which the Bar Council gave a lunch for the judiciary and others. After the lunch as the Chief Justice left, he said to Mr Vasquez: "Robert, don't rape the Constitution". Mr Vasquez said in cross-examination that the Chief Justice was extremely gruff and angry with him. He considered the remark to be a "very, very, very pointed comment to show [the Chief Justice's displeasure". He found it threatening. As was pointed out by the Signatories, a similar remark about the "rape of the Constitution" was

- later made by Mrs Schofield in her e-mails to Mr James Neish and Mr Peter Schirmer (paragraphs 5.18 and 5.32).
- 3.24 The Chief Justice accepted in cross-examination that he had used the phrase "rape the Constitution", and that it was not the most dignified of expressions. In our view it was not consistent with the dignity and status of the office of Chief Justice.
- 3.25 The connection between the maids issue and what happened at the ceremony was readily made by the press, as can be seen from an article in the *Gibraltar Chronicle* of 7 October 2000, headed "Scofield maids controversy: Bar Council drops out of legal year speech". *Vox* on 13 October 2000 carried an article headed "Chief Justice must now go: either he resigns or he must be asked to leave".
- 3.26 In his witness statement the Attorney General described the maids issue as causing "reputational damage" to the office of Chief Justice. It was pointed out to us that in a letter to Mr George Kegoro of the International Commission of Jurists (ICJ) Kenya dated 25 April 2007 (paragraph 5.35) Mrs Schofield said that the maids issue had been used to call for the Chief Justice's resignation. In his oral evidence he said that he would not go so far as to say that the maids issue was an attempt to discredit him, although he acknowledged fully that he had made grave errors in relation to it. Nevertheless he did not consider in 2007 dissociating himself from any suggestion that the Government had used the maids issue as part of an attempt to drive him from office "because once I embark upon that, I then have to embark upon qualifications, explanations, which may lead us into further problems".
- 3.27 We agree with the Attorney General that the Chief Justice's conduct in relation to the maids issue caused damage to his office. On his own admission he committed a series of offences relating to PAYE, social insurance contributions and the work permit for Ms Williams, exhibiting a reckless disregard for the completion of important forms. When he employed Ms Danvers he again committed offences relating to social insurance contributions and PAYE, only belatedly making payment of

what was due. He had no defensible basis for his ignorance of the law. He never instructed his lawyers in 2000 or at all to ascertain the true position in 1996 – 8. We regard his communications to the Governor as less than full and frank. They were positively misleading in respect of the legal advice he purported to receive and pass on. We cannot see how any lawyer, let alone leading counsel, could possibly have advised that Ms Williams did not come within the legislation if that lawyer had given careful perusal to the legislation. From Mr Stagnetto's description of his instructions, it is apparent that the blame rests with the Chief Justice in giving incomplete and incorrect instructions. The fact that he sought to treat this matter as evidence of the Government's bad faith when he had failed to investigate the true position in law speaks volumes about the lack of judgment of the Chief Justice. His explanation for failing to dissociate himself from remarks of his wife that the Government's approach to the maids issue had been part of a campaign to get rid of him from office does not impress us.

Instructions to Registrar of Supreme Court in respect of expenditure in May 2000

3.28 The costs of publicly funded entertainment are met from a central vote under the control of the Chief Secretary. In April 1998 the Chief Secretary issued a general circular to the effect that authority to incur expenditure on official entertainment had to be sought in advance. On 20 April 1998 the Chief Justice replied to the circular through the Registrar, asking for confirmation that he could continue to entertain in his official residence, "obtaining consent to do so in advance of each occasion". On 24 April 1998 the Chief Secretary gave that confirmation. The Chief Justice through his wife arranged a cocktail party at his residence on 9 May 2000 and instructed the caterers and the suppliers of drinks, Anglo Hispano, but, the Government said, did not seek prior approval for the commitment to expenditure. On the Chief Justice's instructions the Registrar applied for the consent of the Chief Secretary, but this was refused the day before the party. The Government said that the Chief Justice then instructed the Registrar to meet the costs from the Supreme Court funds for "general

and office expenses". The Registrar on 15 June 2000 wrote to the Accountant General saying that the Chief Justice considered that the cost should come from the general expenses vote in view of the fact that permission for the party had already been refused by the Chief Secretary. On 24 October 2000 she wrote to the Chief Secretary setting out what had occurred and to put on record that she had no input into the arrangements.

- 3.29 The Chief Justice in his first witness statement said that he had no memory of the incident. In his second he said, having seen the documents, that it was his practice to introduce any new Governor to the members of the judiciary and Bar. When funding was refused under the entertainment budget, he asked the Registrar whether it could be paid for from the general office budget. The Registrar, he pointed out, acted as the accounting officer for the Supreme Court and had the power to disallow claims and a duty to tell the Chief Justice if he acted outside the financial regulations. To his recollection, the Chief Justice said, he hosted the party at his own expense. In his fourth witness statement he said that, having seen a letter from Sir David Durie to the Solicitors to the Tribunal dated 3 June 2008, he was reminded that the Governor sent him a cheque towards the costs of the event.
- 3.30 Ms Dawson accepted in cross-examination that she was in charge of the budget for the judiciary and was answerable for expenditure on entertainment.
- 3.31 We conclude that this issue is of only limited significance to the general issues falling for our consideration.

The allegations of interception of the Chief Justice's telephone communications in 1999 and 2000

3.32 Between October 1999 and November 2000 the Chief Justice raised concerns with the police about the security of the telephones at his residence (as well as about suspected interference with a letter to him and

about the presence on one occasion of four people caught on his own CCTV loitering outside his house). On 30 May 2000 the press reported that he and his wife had asked the police to investigate suspicions of telephone tapping at their home. The newspaper *Panorama* reported under a banner headline:

"Police asked to investigate phone tapping claim at home of Chief Justice 'Campaign to hound my husband out of office' – Mrs Schofield."

Mrs Schofield had given an interview in which she said: "We made our initial report last October and nothing was done". The article commented that that was at the time of the crisis that developed in the wake of the stand taken by the Chief Justice on the independence of the judiciary. The article referred to the maids issue and to Mrs Schofield's comment: "They are trying to discredit my husband. They are trying to hound him out of office".

- 3.33 There were reports also in the UK press. In the *Sunday Express* on 11 June 2000 Mrs Schofield was reported as having accused the Government of tapping her phones, and in the *Sunday Business* of the same date the Chief Justice was said not to have made any public comment about the alleged phone tapping but to have told friends that he believed he was under surveillance because of a clash with the Chief Minister over claims of political interference in the judiciary's independence. That article drew attention to the fear of politicians and lawyers that the issue threatened Gibraltar's reputation as a growing financial centre.
- 3.34 On 8 June 2000 Mr Montado wrote to Mrs Schofield asking whether she had made a statement to the *Sunday Express* accusing the Government of tapping her telephone and, if not, whether she would let him know what steps, if any, she proposed to rectify the matter. Her limited reply on 12 June 2000 merely said that she had not spoken to the *Sunday Express* either directly or indirectly.
- 3.35 On 22 November 2000 the Chief Justice wrote to the Commissioner of Police about his concerns. On 28 November 2000 the Commissioner replied that in respect of the telephones, the police had conducted several

investigations but that no evidence had emerged supporting those complaints and he could see no operational justification for continuing any investigation. The Commissioner said of the allegedly tampered letter that the problem was probably due to inefficient machine sealing by the senders. He said of the persons seen on CCTV that it had not been possible to identify them and that they had not actually ventured upon the Chief Justice's property.

- 3.36 The Chief Justice said in his first witness statement that there was an unexplained problem with his telephone and that he called in a security company who made a report. He further said that he did not suggest any police involvement in phone tapping but that this was a private concern which had no implications for the performance of his duties. In crossexamination he said he had read the *Panorama* Report at the time. He acknowledged that he would have been very concerned if Mrs Schofield had been misquoted in saying that an attempt was being made to discredit him and hound him out of office. He had not considered making a public statement that it was not his view that he was being hounded out of office. He said that he and his wife never regarded the telephone tapping issue to be in any way the responsibility of the Government. As for the remarks attributed to him in the Sunday Business, he said that it was not his view that he was under surveillance because of a clash with the Chief Minister over claims of political interference in the judiciary's independence. He had not told friends that that was his view and he did not believe the Government was involved. He did not think it worthwhile to correct the position as represented by the *Sunday Business* which was not published in Gibraltar. However, he accepted that the international reputation of Gibraltar was of considerable importance to the financial and legal services operating in Gibraltar but said: "once one starts to rectify a report such as that, one gets into all sorts of difficulties".
- 3.37 In her second witness statement Mrs Schofield said that the decision to raise this matter in *Panorama* was hers and that her intention was to raise awareness in a matter of public interest.

3.38 There is no evidence that the Chief Justice's telephone was tapped. It was Mrs Schofield who chose to link the allegation of telephone tapping with the unsubstantiated allegation of attempts being made by the Government to hound the Chief Justice out of office. His unwillingness to correct what had been attributed to him and to dissociate himself from his wife's comments on this issue was not satisfactorily explained by him. He must have been aware of the damage thereby caused to Gibraltar's financial and legal reputation.

The defence advanced on behalf of the Chief Justice on his prosecution in 2001 for an offence under the Motor Vehicles Test Regulations.

3.39 On 28 July 2000 the police reported the Chief Justice for two offences: one was driving a motor vehicle without a MOT test certificate, contrary to the Motor Vehicles Test Regulations; the other was for having no valid road tax. On 17 August 2000 the police wrote to the Chief Justice saying that on that occasion a lenient view had been taken by the Commissioner of Police who had decided to issue him with a caution (the caution for the lack of road tax was later withdrawn). The Chief Justice received the letter of 17 August only on 26 August. On 25 August 2000 the Attorney General wrote to his solicitors, Stagnetto & Co., asking whether he was prepared to accept the caution. On 30 August 2000 they replied that the question of accepting or rejecting the caution did not arise on the Commissioner's letter. On 4 September 2000 the Commissioner wrote to Stagnetto & Co. stating that it is was within the discretion of the Chief Officer of Police to issue a caution if the necessary criteria were met. One of them was that the person to be cautioned accepted his guilt and agreed to be cautioned. The Commissioner said that in respect of the MOT certificate, which was out of date by over 6 months, he was still prepared to continue by way of formal caution if the Chief Justice was prepared to accept his guilt; but if he was not prepared to do so, he was left with no alternative but to progress the matter by other means. On 15 September 2000 Stagnetto & Co. replied, asserting that the Chief Justice was not required to accept his guilt or agree to be cautioned. On 22 September 2000 the Commissioner

issued a summons against the Chief Justice. According to the Attorney General in his witness statement, on the eve of the hearing he sought to avoid a detrimental situation in a small jurisdiction in which he would be prosecuting the Chief Justice. He offered that if the Chief Justice accepted a caution, costs against him would not be sought and he could make a public statement. The offer was not accepted.

- 3.40 Between 26 and 28 July, and on 14 and 15 November, 2001 the case was heard by the Stipendiary Magistrate, Mr Anthony Dudley (as he then was). The Chief Justice was represented by leading and junior counsel. The Attorney General described the hearing as effectively turned into 'a state trial'. At the invitation of Mrs Schofield, various parts of the hearing were observed by representatives of the ICJ and of the International Bar Association (IBA).
- 3.41 An application was made on behalf of the Chief Justice for a stay of the prosecution on the basis that it amounted to an abuse of the process, on the ground that the letter of 17 August 2000 had disposed of the matter finally and unequivocally, and contained no element of conditionality; it therefore amounted to a promise or representation which precluded further action. The Stipendiary Magistrate, who commented in his ruling that the application was "somewhat surprising given the nature of the alleged offence", found that the Chief Justice, albeit "acting unreasonably", was entitled to refuse to indicate his acceptance of the caution and to treat the letter as a promise or representation that the matter had been dealt with and that no proceedings would be issued against him. However, he went on to find that the Chief Justice's legitimate expectation, arising from the letter of 17 August which the Chief Justice received only on 26 August, that no proceedings would be taken against him would have been very short lived because of the letter dated 25 August 2000 from the Attorney General. He therefore held that in the circumstances of the case, and there being no evidence of bad faith on the part of the Crown and in the absence of exceptional circumstances or compelling reasons, the application for a stay should be dismissed. We should add that we are satisfied that, despite an indication to the contrary

- in the Stipendiary Magistrate's ruling, counsel for the Chief Justice did not submit that he should receive special treatment by reason of his office.
- 3.42 In the light of the Report of Mr Justice Nicholson as its observer the ICJ was satisfied that the court procedures conformed to international standards of due process, as it announced in a press release on 10 August 2001. We have not seen his report, but it is referred to in some detail in the Report to the IBA. It is apparent from the latter that he concluded that the trial was conducted "impeccably", that the attitude and approach of the Stipendiary Magistrate was "beyond reproach" and that the trial "was completely and absolutely a fair one...both in relation to the procedure under which the trial was conducted and the manner in which it was conducted". He noted that nothing arising in the defence case at trial raised any issue that the prosecution of the Chief Justice was motivated by or the product of executive harassment of the accused.
- 3.43 The further hearing on 14 and 15 November 2001 was concerned with a submission by counsel for the Chief Justice that the Governor, when bringing into force the MOT Regulations, had acted ultra vires. Following the rejection of that argument by the Stipendiary Magistrate on 15 November 2001 the Chief Justice was found guilty of driving a vehicle without a valid MOT certificate, and fined £50.
- 3.44 The hearing on 14 November 2001 was attended by Lord Hacking as an observer appointed by the Human Rights Institute (HRI) of the IBA. He was instructed to report to the IBA on whether there appeared to be any breaches of the human rights conventions relating to the trial or to the prosecution of the Chief Justice. In paragraph 18 of his report dated 12 February 2002, he referred to a number of allegations made by or on behalf of the Chief Justice relating to his relationship with the Governor, the Government and in particular the Chief Minister and the Attorney General. He also referred to notes taken by Dr Karen Brewer of the Commonwealth Magistrates and Judges' Association (CMJA) of a "briefing meeting" which she held with Mrs Schofield between July 2000 and November 2001. In paragraph 19 he stated that he understood that representations had been made to the HRI to the effect that the

proceedings were politically motivated and improperly brought. In paragraph 20 he referred to certain differences between the Chief Justice and the Chief Minister and asserted:

"In these differences...the Chief Minister called, on at least one occasion, for the Chief Justice's resignation. It also appears that there were calls from the Chief Minister or from other quarters of the Executive Council that the Chief Justice should be subject to an investigation under section 60(4) of the Gibraltar Constitution."

We are unaware of any factual basis for those assertions.

- 3.45 Despite those assertions, Lord Hacking in his report to the IBA concluded that there were no grounds for holding that the prosecution had been improperly brought or tainted by political or other improper considerations. He said that the trial was properly and fairly conducted, and he described the conduct of the Stipendiary Magistrate as "exemplary". He was, however, highly critical of the Chief Justice's conduct, which he said was "deeply puzzling". Among the questions he raised in paragraph 39 were the following:
 - "(ii)...If the Chief Justice is not prepared to abide by [the MOT law] why should any other citizen of Gibraltar do so?
 - (iii) If the Chief Justice agreed that he had an invalid Motor Test Certificate when stopped on 28th July, why did he not accept the caution procedure which was twice offered to him, and let matters rest there?
 - (iv) Why did he instruct his lawyers to present the 'abuse of process' argument (and in doing so take up 2 ½ days of the Court's time) when all the time he was admitting that he did not have a valid MOT Certificate?
 - (v) Why did he instruct his lawyers to present the 'ultra vires' argument which, if correct, would have (or will have) the consequence of obtaining an acquittal for him while leaving every other citizen of Gibraltar, also wrongly convicted under offences created in the Motor Vehicle Test Regulations, without redress?"
- 3.46 Lord Hacking said that his fundamental concern, in considering the Chief Justice's conduct, related to his position as Chief Justice. He referred in

paragraph 40 to the 'Delhi Approved Standards' of the IBA on "Minimum Standards of Judicial Independence" and to the responsibility placed on every judge "always to behave in such a manner as to preserve the dignity of his office". He said (in paragraph 41):

"When a Chief Justice admits that he was not in possession, as in this case, of a valid MOT Certificate, then he should surely look to the dignity of his office before presenting the 'abuse of process' argument".

Lord Hacking (in paragraph 42) said that another disturbing aspect of the Chief Justice's decision to contest the charge, and to have his defence conducted as it was conducted, was the burden which the case had placed on the judicial system over which he presided and on the Stipendiary Magistrate and he referred to the damage inflicted on Gibraltar's judicial system.

- 3.47 Lord Hacking in paragraphs 44 and 45 considered the conduct of the Crown. He thought that it would have been a dereliction of duty for the Crown, upon being unable to reach agreement with the Chief Justice over the caution, not to have proceeded forward by way of prosecution. He concluded in paragraph 47 that there were no grounds for holding that the prosecution was improperly brought or tainted by political or other improper considerations.
- 3.48 In his first witness statement the Chief Justice denied that he refused to accept a caution, denied that any defence argument was in any way improper, and said that his view as a lawyer was that for him to have been cautioned and then asked to admit guilt or face a prosecution was improper and potentially an abuse of process. He also said that Lord Hacking did not comply with the independent trial observers' guidelines as he was not present for the whole of the hearing and he had conversations with the prosecution but not with the defence.
- 3.49 The Chief Justice in his fifth witness statement did not dispute that his motor vehicle licence and MOT certificate were both out of date, but relied on the fact that a period of grace for obtaining vehicle licences had been announced and was set to end on 31 July 2000. A note made by Mr

Mendez, dated 16 August 2000, and addressed to the Chief Justice states that in May 2000 he had made enquiries on behalf of the Chief Justice with the police as to whether the Chief Justice, who had a MOT appointment for his own car booked for late August 2000, could continue to use the vehicle while not displaying the new disc. He was informed that the MOT appointment was sufficient to enable the Chief Justice to continue to use the car, and he so informed the Chief Justice. A further note dated 21 August 2000 and made by Ms Dawson stated that the Chief Justice in late July 2000 had requested her to double check the rules relating to the MOT/licensing of the Chief Justice's car and that she had passed on to him the advice which she had obtained, which was that after 31 July 2000 the fact that a MOT appointment existed did not allow the owner of a vehicle to drive that vehicle unlicensed.

- 3.50 The Chief Justice stated that he considered, and still considers, that the decision to prosecute him was grossly unfair, a view which he repeated more than once in his oral evidence, and an abuse. Given his firm view on the unfairness of the prosecution, he took counsel's opinion and instructed him to defend the issue on the discrete point of abuse of process. He further said that when junior counsel advised him that the ultra vires argument was strong, he decided to instruct him to run that point also. He said in his third witness statement that his lawyers wanted him to widen the issues by bringing in Mr Mendez as a witness but he instructed them that he was not prepared to put him in a position of having to testify. In his fifth witness statement he said that the police had denied the conversation with Mr Mendez, and he did not want Mr Mendez's credibility to be put into question before the Magistrate any more than he wanted the Magistrate to have an embarrassing decision to make in regard to that credibility.
- 3.51 In cross-examination the Chief Justice accepted that he knew before the trial commenced that the ICJ and the IBA were to send observers to the trial at the request of his wife, but he said that he did not discuss with her their briefing or know its content. He knew that his wife thought the prosecution was politically motivated. He thought the prosecution

- grossly unfair. When asked by the Tribunal to explain how what Mr Mendez and Ms Dawson had ascertained was of assistance to us, given that the point was not taken by way of defence at the trial, the Chief Justice said that he wanted to explain to us that he could have dealt with these factual matters but chose not to embarrass Mr Mendez, Ms Dawson and the Stipendiary Magistrate.
- 3.52 We are satisfied that the Chief Justice did refuse to accept the offer of a caution from the Attorney General and that he did so because he was content that the trial should proceed on his chosen grounds of defence in the presence of international observers recruited by his wife with the aid of unsubstantiated allegations against the Government. If he had thought it inappropriate that the observers should attend the trial, with the implication that the prosecution might be improper or the trial processes inadequate, he could have told them in advance that there was no need to come. We do not accept that he did not know from his wife the content of the observers' briefing, given his strong feeling of gross unfairness and his knowledge of her views. We share Lord Hacking's puzzlement as to how the Chief Justice thought his behaviour to be consonant with the proper conduct of a Chief Justice in a small jurisdiction and with the dignity of his office. His conduct of his defence betrays a remarkable lack of judgment and sense of proportion and a disregard for the damage done to the administration of justice in Gibraltar. We do not find of any assistance the fact that arguably the Chief Justice might have run a further defence based on Mr Mendez's and Ms Dawson's evidence when he chose not to do so.
- 3.53 Before leaving this issue we should deal with two other matters. The Chief Justice appealed against the Stipendiary Magistrate's decision and in response the Crown cross-appealed against his decision on costs. In early January 2002 the Attorney General received a telephone call from Mr Picardo, junior counsel for the Chief Justice. It is not in dispute that they discussed an extension of time for filing comments on the case stated, and a basis on which both appeals might be withdrawn. However, Mr Picardo thereafter wrote in a letter to the Attorney General dated 9 January 2002

that the latter had asked him to communicate to the Chief Justice that he should not fear that a failure to pursue his appeal might have repercussions on the renewal of his contract, and that in any event, if the Chief Justice in the course of discussions with the Governor were to raise the possibility of withdrawing his appeal, other job opportunities from other jurisdictions in the Commonwealth might be put to him as an alternative to or after the renewal date. The Attorney General replied on 10 January 2002 that he had made clear his personal opinion that a withdrawal of the appeal by the Chief Justice was not likely to have any repercussions one way or the other on the question of the contract. He also said that, in answer to Mr Picardo's asking whether there might be other job opportunities elsewhere in the Commonwealth, he made clear that that was a matter which the Chief Justice would have to raise directly with the Governor. Mr Picardo responded by letter dated 23 January, saying that he did not say that the Chief Justice was concerned that the withdrawal of his appeal might negatively influence the question of renewal of his contract, and that his recollection was that it was the Attorney General who raised the possibility of job opportunities elsewhere in the Commonwealth.

- 3.54 To this should be added the account given by the Governor in a letter dated 9 April 2002 to the Chief Justice of the Attorney General's recollection (which he confirmed in his written statement and orally). This was that he was adamant that Mr Picardo had stated that the Chief Justice was concerned that the withdrawal of his appeal might negatively affect the question of his reappointment. Mr Picardo had then suggested that the Chief Justice be reappointed for 3 years and, after a respectable interval, should be found a position elsewhere in the Commonwealth. It was at that stage, the Attorney General recollected, that Mr Picardo asked for the conversation to be off the record as he was not instructed to discuss those matters.
- 3.55 We should also refer to the terms of a letter to the Chief Justice dated 4 September 2002 from Mr (now Sir) Desmond de Silva QC, one of the counsel whom the Chief Justice had consulted in connection with the

maids issue. The letter sets out Mr de Silva's recollection of a conversation between him and the Attorney General in about August 2000 when he was on holiday in Sotogrande where the Attorney General had a house. The conversation had turned to the Chief Justice. The Attorney General told Mr de Silva that, in addition to the matter of Ms Danvers, the Chief Justice had been stopped by the police for driving without road tax and a MOT certificate. The Attorney General had said that if the matter were to be uncontested and disposed of with the minimum of fuss, then no doubt the Chief Justice would receive every assistance and help if he were to be seeking a judicial position in England or elsewhere.

- 3.56 In his witness statement the Attorney General recollected his having had a conversation with Mr de Silva in which the Chief Justice's position was discussed in general terms. Mr de Silva had suggested that it would be a good idea if the Chief Justice could be found something elsewhere. He had responded that it would be a good idea. He was not offering the Chief Justice help in finding a position in England or elsewhere, which he would be in no position to do. In his oral evidence he disagreed with the suggestion in Mr de Silva's letter that he had suggested a quid pro quo, namely that if the Chief Justice did not contest the case against him, he would be helped. He said that that certainly did not occur. He maintained that it was his recollection that it was Mr de Silva who raised the matter of something else being found for the Chief Justice.
- 3.57 In regard to the difference of recollection between Sir Desmond de Silva and the Attorney General we accept the evidence of the latter who was an impressive witness. Sir Desmond de Silva did not give evidence. It is improbable that the Attorney General would indicate the possibility of job opportunities elsewhere in the Commonwealth when that was not a matter for him or within his power, but was one for the Governor and the FCO. The same applies to the difference of recollection between the Attorney General and Mr Picardo. We accept the evidence of the Attorney General that it was Mr Picardo who raised the possibility of job opportunities for the Chief Justice elsewhere in the Commonwealth.

CHAPTER 4

2002-2006

This chapter deals with:

- the comments made by the Chief Justice in open court in February 2002 following and relating to his appointment as Chief Justice for one year (paragraphs 4.1 4.26)
- the conduct of the Chief Justice in respect of certain proceedings in November 2004 (paragraphs 4.27 4.31)
- the conduct of the Chief Justice on the departure of the former Governor Sir Francis Richards in July 2006 (paragraphs 4.32 4.37)
- the involvement of the Chief Justice in debate over the 2006 Constitution (paragraphs 4.38 4.62).

The comments made by the Chief Justice in open court in February 2002 following and relating to his appointment as Chief Justice for one year

Prior events

White, informed the Chief Justice that his appointment had been approved, and that he was offered the appointment on contract terms for an initial period of 3 years with the possibility of renewal or extension, depending on the circumstances prevailing at the time. At the Chief Justice's request the suggested contract terms were amended to include reference to section 60 of the 1969 Constitution then in force, which gave him security of tenure until the age of 67. On 8 March 1996 the Chief Justice signed a contract which incorporated the amendment which he had requested. By warrant of appointment dated 8 February 1996 the Governor appointed the Chief Justice for a term expiring on 7 February 1999. In 1999 the Chief Justice refused to sign a new 3-year contract

because he wished to make the point that his tenure of office was not subject to a 3-year contract. On 8 February 1999 the Governor issued a further 3-year warrant.

4.2 In his first witness statement the Chief Justice said that in the months leading up to the expiry of the initial 3 year period he was "called to two or so meetings" by the then Governor, Sir Richard Luce, at which the Governor suggested that he accept a 6 months' extension of contract. He further said that after he had handed him the new warrant for 3 years the Governor told him that his suggestion that he accept a limited contract came as a result of representations by the Chief Minister that his contract be not renewed. On this point the Chief Justice referred to his letter to the Governor, Mr (now Sir) David Durie dated 8 October 2001 in which the Chief Justice stated:

"In the period leading up to the expiry of the last warrant of appointment, I recall that the then Governor, Sir Richard Luce, called me to a series of meetings suggesting on a number of occasions that I consider and reconsider my position only to have to admit at the end of the day that the Chief Minister had made representations that I should vacate Office".

- 4.3 The Chief Justice in his examination in chief maintained that in that period Sir Richard Luce had said to him that perhaps he might take a six months' extension, and had explained that he had received representations from the Chief Minister that the Chief Justice's contract be not renewed. In cross-examination he agreed that he viewed this as quite a grave matter at the time, but did not put it on record at the time by writing to Sir Richard Luce or to the Government. His explanation was that he did not think anything could be gained by doing so. The only record of the exchange with Sir Richard Luce came almost three years later in the letter which he wrote on 8 October 2001. In cross-examination he accepted that the Governor had spoken in terms of the contract not being renewed rather than vacating office.
- 4.4 Lord Luce, as Sir Richard Luce became, stated in his witness statement that as Governor he had held discussions with the Chief Justice in strict

- confidence and trust. He was disturbed that the Chief Justice had decided to betray that trust some 8 years later and, without notice to him, had sought to report on his recollection of what he had said to him. Whilst unable to recall the exact verbatim details of their discussion, Lord Luce said that the Chief Minister never sought to undermine the Governor's sole responsibility to decide on the Chief Justice's contract.
- 4.5 The Chief Minister in his witness statement said that he had never asked Lord Luce, or any Governor, not to renew the Chief Justice's contract, or to issue one of a reduced or particular length, still less 6 months. In his oral evidence, he confirmed that he had never made any observation or submission that the Chief Justice be removed. In the statement which the Chief Secretary, Mr Richard Garcia, presented on behalf of the Government he referred to the statement of the Chief Justice in his proceedings for judicial review, which contained a similar account of his conversation with Sir Richard Luce. The Chief Secretary said that Lord Luce had confirmed to him that he had never said any such thing to the Chief Justice, that he had never offered the Chief Justice a six months warrant of appointment, and that the Chief Minister had never said anything to him seeking the non-renewal of the Chief Justice's contract or his removal from office and that Lord Luce had never said any such thing to the Chief Justice. In his oral evidence the Chief Secretary said that he had verified this over the telephone with Lord Luce, who had told him categorically that on no occasion had the Chief Minister made any such representation to him.
- 4.6 We have not had the benefit of Lord Luce giving evidence before us. Some of what he said betrays confusion between a new warrant, which is the Governor's concern, and a new contract, which is the Government's concern. However, the only record supporting what the Chief Justice said he was told by the Governor was his letter of 8 October 2001 written to a subsequent Governor and then in terms which he accepted were somewhat different from what he now says he was told by the Governor. Further, we find it surprising that, in regard to a matter which he described as "very serious" and "of grave concern", he did not

immediately put on record in 1999 that he considered the Chief Minister's involvement to be grossly inappropriate. We prefer the evidence of Lord Luce, the Chief Secretary (on what he was told by Lord Luce), and the Chief Minister, to that of the Chief Justice on this point.

February 2002

The dates in this section of the Report relate to that month

4.7 On 8 February the then Governor presented to the Chief Justice a warrant of appointment for a further period of one year. The Chief Justice on 11 February wrote to him in a tone notable for its hostility and threat, saying:

"For you to purport to reengage me for one year sends out a signal to the public which is calculated to undermine me in Office and which is calculated to undermine public confidence in the judiciary. Security of tenure is one of the cornerstones of judicial independence and should you persist in your intention to purport to limit my tenure of Office I shall be obliged to correct public perception and to reassure the public that the judiciary is inviolate and has the protection of the Constitution".

On 13 February the Governor replied that the long-standing practice of issuing time limited warrants was not intended to, nor could it, affect the provisions of the Constitution.

4.8 In the meantime on the morning of 12 February there was a hearing in chambers before the Chief Justice in the course of a criminal trial, *R v Clinton*. Three counsel were present: Mr Charles Pitto (now the Stipendiary Magistrate) for the prosecution, Mr David Hughes for two of the defendants and Ms Anne Balestrino for another two. There is no minute book for a hearing in chambers. However, according to his supplementary witness statement, Mr Mendez the Deputy Registrar, took some manuscript notes of the hearing which were then type-written (the manuscript being later destroyed). The typed notes record the following remarks:

"Chief Justice: Last Thursday, during this trial, Governor informed me that my appointment as Chief Justice had come to an end, which of course I do not accept. On Friday His Excellency

purported to give me a further appointment for one year. Constitution leaves me in office until 67 years of age. The effect of the Governor's purported action would leave me without independence following loss of security of tenure of office. My office would become dependent on the executive. I have told the Governor that he may not offend the Constitution or violate my independence. If the Governor has not withdrawn his purported action by close of business today, I shall want the Attorney General to be here, to address me on the validity of the Governor's purported action. I may feel that I must abandon this case and indeed suspend all sittings of the Supreme Court.

Hughes:

You are right. An opposite view to yours would not even be arguable.

. . .

This is a grave matter. We may have to run arguments on this. You are correct in your views.

Chief Justice: Hughes, how did you know so much about this? Is it out?

Hughes: It was discussed at the Bar Council. Maybe you should ask the Chairman of the Bar to be present tomorrow as amicus curiae.

Chief Justice: If the executive interferes, I must have this out. An important organisation in London is willing to take the matter up, but I want to deal with it myself. This is a crisis not of my making. It is interference with administration of justice. I would abandon this case to avoid an appeal."

The Chief Justice is then recorded as saying that the Attorney General had 24 hours' notice that the Chief Justice wanted him there the following morning, and as repeating the warning that the Governor had until close of business that day to withdraw his purported action. Mr Pitto requested a copy of the note for the Attorney General, which was faxed to him on the same day.

- 4.9 On 13 February there was a hearing in court in the absence of the jury. The same counsel were present as on the previous day. The Attorney General did not attend. A transcript was made of the hearing. The Chief Justice made a statement about his security of tenure and the "purported warrant of re-engagement for one year". He referred to the decision of the High Court of Justiciary in *Starrs v Ruxton*, to his own view that he had the protection of the Constitution, and to the assurance from the Governor that the practice of issuing three year warrants was not intended to and could not affect the provisions of the Constitution, but said that he had a duty to bring the matter to counsels' attention in case it affected the propriety of the proceedings before him. Mr Hughes (with the agreement of Ms Balestrino) agreed with the view of the Chief Justice and said that the defence did not wish to challenge the proceedings. He said that the Chief Justice was independent and impartial, but that if he were not, not only would the current trial have to cease but all sittings of that Court would have to cease. Mr Pitto for the Attorney General said that the Crown did not dispute the Chief Justice's right to sit. The trial then proceeded to its conclusion.
- 4.10 The issuing of a one year warrant was reported in the press. In an edition confusingly dated "Wednesday 12th February 2002" (the 12th February was a Tuesday) the *Gibraltar Chronicle*, under the heading "Crisis looms as Chief Justice and Convent (the Governor's official residence) clash", referred to the Governor's decision to limit the extension of the Chief Justice's employment to one year and said that a major row was brewing, involving a crisis meeting at the Convent, and that the Chief Justice had apparently signalled that he was challenging the Governor's decision. The *Gibraltar Chronicle* continued:

"The row is set to shake the establishment and may even suspend the Supreme Court's functions. It is understood that the judge has indicated privately that he believes the decision could undermine the independence of his standing in the Court and threaten his Constitutional position".

While it may be that the edition containing the article was published on 13 February, the newspaper was not purporting to report what had happened in the proceedings in chambers on 12 February. That prescient article, as Counsel to the Tribunal aptly described it, was plainly the product of information from a source close to the Chief Justice.

4.11 The New People on 15 February carried an article bearing the heading "Chief Justice in constitutional row with UK" and the subheading "British Government accused of trying to remove Chief Justice". It quoted from a letter sent by Mrs Schofield to the then Foreign Secretary, the Rt Hon Jack Straw MP, in which she claimed that there had been harassment by those hoping that it would lead to the Chief Justice departing of his own accord. She also said in that letter:

"Your instructions violate my children's rights. My children's rights to family life, home are protected by law and it is my intention as a mother to ensure that no further damage is caused by Her Majesty's Government or by the Governor of Gibraltar to the children".

The Chief Justice said in his first witness statement that he had no hand in the raising by his wife of the issue with the Foreign Secretary and the press.

The New People also reported that it understood that Mrs Schofield was considering judicial review to restrain the British Government from interfering with her children's rights and that she might also seek damages against the British Government for pain and suffering caused to the children and herself by unconstitutional threats from British Government officials in Gibraltar.

4.12 The first matter which we have to consider is the authenticity and the accuracy of Mr Mendez's note of the hearing in chambers, which were challenged by the Chief Justice and Mr Hughes.

- 4.13 The Chief Justice in his fourth witness statement said that it could only relate to a discussion he had with counsel in chambers, not on 12 February, but on 11 February. He said that the tone of the note was wrong. It was "unthinkable" for him to threaten to suspend all sittings of the Supreme Court though he accepted that he might well have raised the idea that his own sittings would be suspended until the issue was resolved. He said that he would not have mentioned to counsel that he had been in touch with an important organisation in London, but he recollected that he had contacted the ICJ. The tone of the final paragraph with its warning to the Governor was, he said, "bewildering".
- 4.14 In his first witness statement Mr Hughes said that he was "reasonably confident" that the discussion was not as set out in the transcript. He was confident that the Chief Justice did not suggest that there were organisations in London who were willing to look into the matter and that the Chief Justice did not make any threat, such as a threat to suspend sittings of the Supreme Court. He would have remembered any such statements. He did not understand how the transcript could have been produced and that a clerk did not take a note. He pointed out that the note showed it was faxed on the Registrar's machine and on 12 February, shown as the date of the hearing in open court. He was "sure" that that did not happen until 13 February. In cross-examination he said he recalled counsel in the case going into the Chief Justice's chambers on two separate occasions. The first he thought was not a long meeting on Friday 8 February, when the Chief Justice brought to everyone's attention the fact that he had been given a one year warrant and said that he might need to hear counsel on it. He had a distinct recollection of a weekend intervening between the two occasions. He recalled a discussion of that matter at the Bar Council, which he said must have occurred on that day or the following Monday 11 February. In his oral evidence about the hearing in chambers on 12 February, he was less categorical than in his witness statement. He could not say that he was 100% certain that the Chief Justice had said that he might have to suspend the sittings of the Supreme Court, but he could say that there was no threat because it was not that type of discussion. He did not remember the Chief Justice getting

angry. He accepted that it was possible that the Chief Justice could have referred to an important organisation in London in a way which caused Mr Mendez to remember it. He did not persist in his points on the chronology.

4.15 Mr Mendez in his supplementary statement stood by the accuracy of his note. He said that clerks did take notes, and that judges never saw the countless minutes and transcripts that they produced. He had his manuscript notes typed because he could see that the whole incident had far-reaching consequences and that he foresaw he would be asked for that transcript before long, as in fact happened. The Registrar's fax machine was used all the time, but she was not present at the hearing. The trial resumed on 12 February and again the next day. In cross-examination he adhered to his evidence about the accuracy of his note.

It was normal to give a record to a party if that was requested. He accepted that he had not checked with others that his note was accurate and that he probably should have told the Chief Justice that he had given a copy to the Attorney General. When it was put to him that the Chief Justice did not say, or would not have said, that he might feel that he must abandon the case and suspend all sittings of the Supreme Court, Mr Mendez said:

"Well I'm very satisfied with the accuracy of this note and I repeat those are things that would never occur to me, and I certainly would not invent them and put them in just like that."

When it was similarly put to him that the Chief Justice had no recollection of making any statement about an important organisation in London and did not think it was ever said, he responded that important organisations in London would not have occurred to him. He added that on that morning the Chief Justice was "a very, very angry Chief Justice".

4.16 Mr Pitto said that the first occasion on which he heard that there was any issue relating to the Chief Justice's warrant occurred was when he was approached in court by Mr Hughes and asked if he knew that the warrant was about to expire, to which he replied that he did not. Mr Hughes then

tried to persuade Mr Pitto, as the advocate appearing for the Crown, to raise the point in open court. Mr Pitto replied to the effect that if the Chief Justice sat in court, then he was entitled to sit in court. After that conversation counsel were summoned to the Chief Justice's chambers. He said that his recollection of what transpired there was corroborated by the note of the hearing, and there was nothing in it which struck him as inaccurate or misleading. He clearly recalled that he left the hearing with a sense that there was a likelihood or real danger that the Chief Justice would not sit beyond 14 February. To the best of his recollection a suspension of all sittings was mentioned. He distinctly recalled the Chief Justice asking Mr Hughes how he knew so much about this matter and Mr Hughes replying that it had been discussed at the Bar Council. In crossexamination he said that he did have a clear recollection of what was said in chambers on 12 February, in that he recalled the threat of a suspension, be it of all hearings or at least hearings before of the Chief Justice, because he recalled communicating that to the Attorney General immediately afterwards. He also recalled the Chief Justice asking Mr Hughes about his knowledge of the warrant, and Mr Hughes's response. He had no recollection of reference being made by the Chief Justice to a London organisation.

4.17 Bearing in mind the Report in the *Gibraltar Chronicle* and having had the opportunity to observe the witnesses and their demeanour, and having regard to the fact that Mr Mendez, an experienced note-taker, is the only witness to have made a contemporaneous note and had it typed very shortly afterwards, we would accept that what he recorded was actually said, unless we found Mr Hughes's attack on Mr Mendez's credibility to be persuasive. As to that attack, we reject every single argumentative point taken by Mr Hughes as being without foundation. Mr Mendez seemed to us to be patently honest, frankly accepting errors which he had made in his evidence but maintaining his account when he was sure that he was correct. It follows that we also reject the Chief Justice's account of what was said. We find that he was indeed very, very angry that morning and did say what he is recorded as saying, including the warning to the Governor.

4.18 The next matter which we consider relates to the circumstances in which the discussion in chambers took place. Mr Pitto was intrigued and suspicious at the insistence with which Mr Hughes brought up the matter of the warrant, and at the underlying assumption that there was a potential problem with the warrant. He said:

"Why should it be assumed that there was a problem with it? Why should it be worth raising?"

He said that, in the light of the exchanges in court, what happened in chambers made him uncomfortable and suspicious. The whole situation was unique in his career.

- 4.19 In this context the Attorney General's oral evidence is also of relevance. He said that he had unusually not acceded to the Chief Justice's request to be present on the morning of 13 February because, after discussing it with Mr Pitto, they both felt "that there was a little bit of a set-up in this between Mr Hughes and the Chief Justice". He was of the view that he was being called in front of the Court to embarrass the Governor in some way. It may be noted that in his witness statement Mr Mendez described Mr Hughes as "in and out of the Chief Justice's chambers for private conversations constantly", Mr Mendez and the staff joking "Mr Hughes is upstairs again".
- 4.20 The Chief Justice in his oral evidence flatly denied that there was concerted action by him and Mr Hughes. He maintained, with the support of Mr Hughes, that it was entirely appropriate for him to raise an issue regarding the regularity of proceedings before him. Mr Hughes denied visiting the Chief Justice to speak to him privately, save on one occasion relating to his professional future in joining chambers in England. He speculated that the reason why Mr Mendez had given evidence about him visiting the Chief Justice in chambers was that he (Mr Mendez) worked for the Government. Mr Hughes said it would be a very brave civil servant who was not inclined to be helpful to the Government and was not afraid of the consequences of telling the truth. However, Mr Hughes's evidence that he knew about the warrant because it was

- discussed at the Bar Council is not borne out by the Bar Council minutes. These show that the matter was not discussed by the Bar Council until an emergency meeting at 4.30 pm on 12 February, that is to say after the hearing that morning in chambers. We think it regrettable that he should have chosen by way of speculation to attack the veracity of Mr Mendez, a senior civil servant.
- 4.21 It is plain that the Chief Justice was deeply angered by the Governor giving him a warrant for only one year. We heard no evidence as to why the Governor chose to do that rather than give the Chief Justice the usual 3-year extension. It may be that the Governor was dissatisfied with the Chief Justice's conduct and wished to mark his disapproval in some way. But, as the Chief Justice rightly believed, his security of tenure until the age of 67 was given by the Constitution and indeed had been incorporated into his contract with the Government. The Governor told him that the practice of issuing time-limited warrants could not affect the Constitution. In *R v Clinton* no counsel for any of the parties suggested that the Chief Justice's tenure was affected, and it was clear that the decision in the *Starrs* case was distinguishable. The Chief Justice said that he had raised the issue in open court to prevent an appeal on the point, but the procedure he chose was not apt to resolve the suggested problem. Even if defence counsel conceded the point before the Chief Justice, different counsel could take the point on appeal, and defendants in other cases could take the same point and have it resolved authoritatively.
- 4.22 Mr Hughes produced a second witness statement after seeing Mr Pitto's witness statement and reading the transcript of Mr Pitto's oral evidence and the transcript of Mr White's closing submissions. In it he said that Mr Pitto's evidence that on 12 February he had asked Mr Pitto to raise with the court the question of the warrant did not accord with his own recollection. However, he said that it was difficult to be absolutely certain about the sequence of events nearly 6 ½ years later, but he did not accept that there would be anything suspicious about it as he thought it more appropriate for the prosecution to raise the point. We have no doubt that

- the evidence of Mr Pitto, who had good reason to recall Mr Hughes's surprising request, is to be preferred to that of Mr Hughes.
- 4.23 Mr Hughes referred in his second witness statement to his earlier evidence that the matter of the warrant was raised by the Chief Justice twice in chambers. He adhered to that evidence and said that he did not know of the issue of the warrant before the Chief Justice asked counsel into chambers for the first time. On his evidence that would have been on 8 February. Mr Pitto's evidence was clear on this point and not subject to cross-examination by counsel for the Chief Justice. Mr Pitto first heard of the issue about the warrant when told by Mr Hughes on 12 February. The Chief Justice in cross-examination wondered whether Mr Hughes was right to say that the matter of the warrant was raised twice; but he was not sure. The only way Mr Hughes's distinct recollection of the issue being raised briefly by the Chief Justice in chambers on 8 February can be reconciled with Mr Pitto's evidence is to accept that the Chief Justice did indeed mention the issue to Mr Hughes on 8 February but also to accept that Mr Pitto was not then present, and we so find.
- 4.24 Mr Hughes in his second witness statement accepted that his recollection that before 12 February there was a Bar Council meeting at which the issue of the warrant was discussed might be incorrect. He said that his recollection of events was different from that of Mr Pitto, and that he could not possibly have said that he had learnt of the issue at a Bar Council meeting. However, Mr Mendez's minute on this point is clear and Mr Pitto's recollection accords with that minute. We accept Mr Mendez's and Mr Pitto's evidence.
- 4.25 Mr Hughes in his second witness statement said that it was simply untrue that he was involved in any kind of set up involving the warrant. We are unable to accept that evidence. We find that the Chief Justice did mention the issue of the warrant to Mr Hughes, but not to Mr Pitto, in the Chief Justice's chambers on 8 February when he indicated that he might need to hear from counsel on the issue. Mr Hughes was known to the Chief Justice as counsel who might take constitutional points. The artificial-sounding question and answer recorded by Mr Mendez as to Mr Hughes's

knowledge together with Mr Hughes's incorrect reference to a Bar Council discussion point to the fact that there was such a set up. Mr Pitto's and the Attorney General's suspicions were fully justified. The hearing before the Chief Justice in chambers was contrived to enable him to give vent to his outrage at the Governor's action.

4.26 There is one other feature of this matter to which we would draw attention. The angry tone of the Chief Justice's letter of 11 February to the Governor and the leak of his thinking to the *Gibraltar Chronicle* as appearing from its article dated 12 February is matched by the stridently indignant tone of Mrs Schofield's letter to the Foreign Secretary and the article in *The New People* of 15 February. The perception thereby given publicly is of concerted action by the Chief Justice and Mrs Schofield in response to the Governor's action.

The conduct of the Chief Justice in respect of certain proceedings in November 2004

4.27 On 18 November 2004, an application in the civil case of *Rodret AB v* Osloford came on for hearing before the Chief Justice. A Spanish property development company (the buyer) had negotiated with Rodret for the purchase of the shares in three Gibraltar companies (including Osloford) and the assignment of loans by Rodret to the companies. Prior to completion the buyer had claimed that the loans were fictitious, and applied successfully to a Spanish court in Malaga for an order requiring Rodret to transfer the shares for 11.5 million Euros and a declaration that the loans were fictitious. Rodret appealed in the Spanish court. In order to protect its position it also issued proceedings against the Gibraltar companies and obtained a consent judgment against each of them for the full value of the loan to it. The consent judgments were then registered in Spain as Spanish judgments. The buyer applied in the Gibraltar Supreme Court to have the consent judgments set aside. It joined Rodret as the sole respondent to the application.

- 4.28 The application had been set down for a 2-day hearing to be held on Thursday 18 and Friday 19 November, as the parties were notified on 8 November by the court. It is not in dispute that the hearing was to be in Court 2 (called "the Library") and that the Library was used by the Magistrates Court on Fridays.
- 4.29 At the outset of the hearing the Chief Justice took a number of points. The first was that the Gibraltar companies were not, but should be, represented. Although none of the parties had taken the view that this was necessary, Mr J E Triay QC, for Rodret, undertook to obtain instructions from the Gibraltar companies "so as not to make an issue of non representation" (according to notes made by a clerk in the court minute book). The second point recorded in the clerk's notes was that the court was available only for 18 and the afternoon of 19 November; and that counsel would "have to explain everything" as the Chief Justice had not had time to read all the papers. In an e-mail to his clients the next day Mr Javier Triay (who had appeared with Mr J E Triay for Rodret) stated that the Chief Justice told counsel that he expected them to take him slowly through the relevant facts and history in detail. Contrary to the notes made by the clerk, Mr Javier Triay reported to his clients in the email that the parties were given only the option of having the case heard on the morning of 18 February and then having the case part heard until December. The third point was whether the proceedings in Gibraltar should be stayed until the appeals in Spain had been heard. It was clear that neither side wanted a stay. Mr J E Triay made it clear that he did not want a split trial. The parties agreed to an adjournment. According to his witness statement Mr J E Triay formed the impression that the Chief Justice was looking for any opportunity to postpone the hearing of the case, first by reference to the parties' representation, then the courtroom availability, then not having read the papers and lastly a stay on the ground of the appeals in Spain.
- 4.30 The following week Mr Javier Triay was informed that the Chief Justice had been seen leaving Gibraltar on a flight to the United Kingdom before the weekend.

4.31 Despite suggestions that the Chief Justice deliberately misled the parties as to the ability of the court to deal with their case we think it clear that the Chief Justice did not mislead or deceive anyone in the course of the hearing on 18 November. What he said about the time available that day and the next day appears to have been open to misinterpretation in view of the differing recollections, but we are inclined to accept the evidence of the clerk's minute that the Chief Justice said that the available time was that day and the following afternoon. However, we are surprised that the Chief Justice did not offer to make up the time which would be lost through the unavailability of the Library on the morning of 19 November by sitting longer hours. Because of the series of unimpressive points taken by the Chief Justice he appears to have given to Mr JE Triay and Mr Javier Triay the impression he was determined that the case should be adjourned, but we do not find that this was to enable him to be on a plane out of Gibraltar before the weekend. The evidence on that point is too inconclusive.

The conduct of the Chief Justice on the departure of the former Governor Sir Francis Richards in July 2006

- 4.32 On 17 July 2006 Sir Francis Richards, the outgoing Governor, left Gibraltar by sea. There was a formal departure ceremony at the naval dockyard. The Deputy Governor, Mr Philip Barton, said in his witness statement that Sir Francis had given clear instructions as to how he wished his departure ceremony to be organised. In accordance with those instructions Mr Barton arranged for there to be two groups present to say goodbye to Sir Francis and Lady Richards, one of local dignitaries and officials, the Chief Justice being placed at the head of that group, and another group placed closer to the gangplank, consisting of Mr Barton and his wife and the Chief Minister. Sir Francis wanted the final person to say farewell to him to be a representative of the people of Gibraltar.
- 4.33 Traditionally Governors are sworn in by the Chief Justice. Mr Barton asked the Chief Justice to swear him in as Acting Governor and this had

been arranged for a time shortly after the Governor's departure, that is to say at 2 pm on the same day at the Convent. Mr Barton said that under sections 20 and 22 of the 1969 Constitution when the office of Governor was vacant an Acting Governor needed to be appointed and sworn in formally. He was concerned that until he was sworn in, there would be no one in Gibraltar in a position to exercise the executive powers of the Governor, for example in the event of an emergency.

- 4.34 The Chief Justice was unhappy with the way the departure ceremony had been organised and in particular with the Chief Minister being placed, as the Chief Justice thought, ahead of him in breach of protocol. Mr Mendez described him as returning "fuming". On the Chief Justice's instructions Ms Annabelle Desoiza, the Deputy Registrar, telephoned Mr Barton to cancel the swearing in arrangements. At Mr Barton's request the swearing in was arranged for the next day but only on conditions stipulated by the Chief Justice, viz. that Mr Barton put in writing why he had made the departure ceremony arrangements in the way he did and that Mr Barton should come to the Chief Justice's chambers rather than the Chief Justice attending on Mr Barton. Mr Barton did what the Chief Justice requested. When Mr Barton saw the Chief Justice the next day, he was criticised for what he had written about the arrangements. The Chief Justice said that it was not for Her Majesty's representatives in Gibraltar to change the order of precedence laid down by the Queen.
- 4.35 The Chief Justice in his third witness statement said that it had never been the practice that on the departure of a Governor the Deputy Governor was sworn in. He said that he required Mr Barton to come to his chambers to demonstrate his dissatisfaction with the way Mr Barton had organised the departure of the Governor.
- 4.36 Mr Mendez said in his witness statement that it was not unusual for the Deputy Governor to come to the Chief Justice's chambers to be appointed as Acting Governor for short absences on the part of the Governor but that for periods of interregnum it was customary for the Chief Justice to go to the Convent and administer the oath there.

4.37 In our view every incident in this episode reflects badly on the Chief Justice. His petulant behaviour, at a time when there was no Governor, towards the senior representative of the Queen in Gibraltar, the Acting Governor, was disgraceful. He was wrong to suggest that it had never been the practice to swear in the Acting Governor. His refusal to adhere to the arrangements which he had made with Mr Barton to swear him in as Acting Governor took no account of the fact that thereby Gibraltar was left for the time being without anyone able to exercise emergency powers. His acknowledgment that by imposing conditions on Mr Barton for him to be sworn in he was demonstrating his dissatisfaction with the arrangements for the departure of the previous Governor shows that his actions were governed by pique in a manner wholly inconsistent with the dignity of his office. His dissatisfaction stemmed from the fact that Mr Barton had complied with the wishes of the outgoing Governor as to what should happen on his departure. Given Mr Barton's explanation as to the wishes of the outgoing Governor and the positions by the gangplank occupied by those present, the notion that this was in breach of the Queen's wishes on protocol is fanciful, and we cannot help noting that it was because the Chief Minister was placed in a position which the Chief Justice thought gave the Chief Minister precedence that caused the Chief Justice to be so angry.

The involvement of the Chief Justice in debate over the 2006 Constitution

The dates in this section of the Report refer to 2006, unless otherwise indicated.

4.38 In 1999 a Select Committee had been established by the Gibraltar House of Assembly to consider the reform of the Constitution. An open invitation had been issued to all members of the public or interested parties to submit their views. Following the adoption of the Report of the Select Committee on 27 February 2002 the draft text of a new Constitution had become the subject of negotiation between a UK delegation and a Gibraltar cross party delegation. The judiciary had not responded to the invitation to submit views. The Chief Justice said in evidence that he

preferred "to put his representations through the Governor". However, the period for receipt of representations had ended in February 2001, and the first indication of his communication with the Governor on this subject had been on 23 March 2005 when he submitted to him a paper with recommendations for the amendment of the Constitution. He had sent an advance copy to the Chief Minister on 7 March of that year, pointing out that "the recommendations are made after extensive consultation with my colleagues and are those of the whole judiciary". We accept the evidence of Mr Charles Pitto, the Stipendiary Magistrate, that he did not have the opportunity to consider them in detail before they were submitted. The paper had recommended a Judicial Service (or Appointments) Commission (JSC), with a membership of six, including the Chief Secretary, the Attorney General and the Chairman of the Bar Council. In late 2005 the Chief Justice sought a meeting with the legal adviser to the FCO and the legal adviser to the UK delegation. He was offered instead a meeting with another FCO official. He declined the offer on the ground that it would be inappropriate for him to meet an official who was not a legal officer but a "political" member of the UK delegation. In due course on 9 January 2006, after consulting the members of the Gibraltar delegation about a number of issues, the Chief Minister wrote to the Chief Justice enclosing a detailed matrix which showed the position of the Government in regard to the recommendations which had been made in the paper. His letter ended with the words: "I will, of course, let you know as soon as it becomes apparent, what is the likely outcome of the negotiations with the UK as they affect issues affecting the judiciary". This led to the Chief Justice making further submissions on 21 February with a modified proposal for membership of the JSC, and a recommendation that it should be provided that the Chief Justice was the Head of the judiciary. In the event the Chief Minister did not let the Chief Justice know the likely outcome of the negotiations. We will refer at paragraph 4.51 to his explanation for this.

4.39 After further negotiations, on 17 March the UK and Gibraltar delegations reached an agreement on the text of a new Constitution. On 20 March the Chief Justice telephoned the Deputy Governor, Mr Barton, and asked him

about the status of the text. He was informed that, having been agreed between the delegations, the text had gone to the Foreign Secretary, after which it would be made public. It appears that the Chief Justice did not press for a copy, and made no subsequent request for one. On 27 March the agreement on the terms of the final draft text was formally and publicly announced. A Government press release on 5 July announced that, following the confirmation by the UK Government that Gibraltar's Constitutional Referendum would constitute an exercise of the right to self-determination, the two governments had published the text of the proposed new Constitution. It was available on the Government website, and hard copies could be obtained from 6 Convent Place. This information was also published in the *Gibraltar Chronicle* on 6 July, in an article which referred to the fact that the Gibraltar Parliament had united behind a 'yes' vote, and that it was anticipated that the proposed referendum would take place in September.

4.40 Despite this the Chief Justice did not obtain a copy the text until about 4 August. In his letter to the Chief Minister dated 11 August, to which we refer in the next paragraph, he said that he had "managed, late last week, to download the Draft from the Government website." He claimed in evidence that this was due to a persistent difficulty in obtaining access to it on the website. We find this hard to believe. We also find it hard to understand why he had taken no steps to obtain a hard copy in the interim. When the Chief Justice read the text he found that it contained a number of provisions which he had not seen before. In regard to the JSC, four of its seven members would be appointed by the Government and by the Governor; and the President of the Court of Appeal, and not the Chief Justice, would be its Chairman. It should be noted that a similar, but not identical, proposal about the membership of the JSC had been drawn to the attention of the Chief Justice in January 2006, on which he had made representations in February. Article 57 (2) would enable the Governor, with the advice of the JSC, to make and confirm various judicial appointments, and exercise disciplinary control over subordinate officeholders. Article 57 (3) made provision for the Governor, with the prior approval of the Secretary of State, to disregard the advice of the JSC

- where this was prejudicial to Her Majesty's service. Article 64 (7) provided that, notwithstanding sub-article (1), which provided security of tenure to the age of 67, a Chief Justice or a Puisne Judge could be appointed for a limited term, on expiry of which the office would become vacant. The Chief Justice almost immediately got in touch with the CMJA, which submitted its observations on the text to the FCO on 8 August.
- 4.41On 7 August the Chief Justice wrote to the Chief Minister stating he had "just had a sight" of the draft on the website. He said that he and other members of the judiciary remained concerned at the draft, particularly sections 57 and 64(7), which were considered to "undermine the independence of the judiciary rather than strengthen it". He sent a copy of his letter to the Acting Governor and the members of the Select Committee. He followed this with a further letter to the Chief Minister dated 11 August, which enclosed "the submissions of the Chief Justice, the Puisne Judge, the Stipendiary Magistrate and the Registrar on the draft Constitution". The submissions expressed particular concern about the overriding control conferred by Article 57(3) on the Governor acting on the advice of the Secretary of State. The independence of the judiciary flowed from the appointments process. It was fundamental that such process was independent and not in the ultimate control of the executive. Giving a power of veto "contravenes universally accepted principles on the separation of powers". The majority of the members of the JSC would be appointed by the executive. The Chief Justice, as head of the judiciary, should be its Chairman. The draft did not guarantee judicial independence. It failed to demarcate the separation of powers between the executive and judiciary. It failed to deal with the question of who was head of the judiciary, or to define the office and responsibilities of the Chief Justice. The submissions concluded with the statement that the judiciary would consider all possible steps it might take to prevent these errors of principle from being promulgated, "including, reluctantly but if necessary, a petition to Her Majesty through Her Privy Council". This was evidently a reference to special reference under section 4 of the Judicial Committee Act 1833.

- 4.42 A meeting had taken place on 11 August between the Chief Justice, Mr Pitto and the Registrar, Ms Annabelle Desoiza, at which the Chief Justice had produced a draft which set out a number of concerns in regard to the text and concluded with a reference to a possible petition to the Privy Council. The document purported to represent the views of the whole of the local judiciary, even though the Puisne Judge, Mr Justice Dudley, had not seen it. Moreover, Mr Pitto had expressed particular concern, both at the meeting and in a subsequent telephone call, at what he considered to be the inappropriateness of including a reference to the Privy Council. The Chief Justice had responded by indicating that he considered that the Government would do nothing without this "bite" or "threat". He submitted the representations in the name of the judiciary without including any express indication of Mr Pitto's reservations. There is no indication that he had received any legal advice in support of a possible petition to the Privy Council.
- 4.43 On 15 August the *Gibraltar Chronicle* carried an article entitled "Chief Justice recruits Commonwealth support in bid to provoke review of draft Constitution". On the same day *Panorama* carried an article headed "Chief Justice in campaign against new Constitution". The *Gibraltar Chronicle* of 17 August carried an article headed "Opposition urges Constitution be held back pending judiciary clarification".
- 4.44 On 25 August 2006 Hassans wrote on behalf of the Chief Justice to the Acting Governor, Mr Barton, and the Chief Minister. They expressed "the deep concerns of the judiciary about some aspects of the proposed new Constitution", stating: "As the current proposals now stand, it is the view of the Chief Justice and his colleagues that some of its provisions would adversely affect judicial independence and the rule of law in Gibraltar, and public confidence in the due administration of justice by our courts". They also claimed that it was unacceptable that the governments had published a draft Constitution without proper consultation with the Chief Justice and the judiciary. On behalf of the Chief Justice they asked the governments "to confirm (i) a willingness to urgently engage in such a process of meaningful and transparent consultation; and (ii) that no steps

shall be taken to expedite the referendum while the consultation is pending". If the two governments were not willing to undertake a proper process of consultation, they said it would be necessary to have recourse to the Judicial Committee of the Privy Council. The most convenient course would be for the two governments to request the Privy Council to refer the Chief Justice's concerns to the Judicial Committee under section 4 of the Judicial Committee Act 1833. The Chief Justice was advised by Lord Lester QC. The *Gibraltar Chronicle* of 30 August carried an article entitled "Lawyers appointed to resolve judiciary doubts". Hassans wrote a further letter for the Chief Justice to the Acting Governor and the Chief Minister on 5 September, in which they stated:

"Unless by noon on Friday 8th September the Gibraltar and United Kingdom Governments confirm that they agree to jointly request Her Majesty to refer the Chief Justice's concerns (in terms to be agreed) to the Judicial Committee for their consideration under section 4 of the Judicial Committee Act 1833, the Chief Justice will feel compelled to Petition the Privy Council directly and unilaterally, although of course, both Governments will be served with a copy of his Petition."

In his evidence the Chief Justice accepted that he was not aware of any precedent anywhere in the Commonwealth for a Chief Justice petitioning the Privy Council on a constitutional issue in this manner.

4.45 In letters to the Chief Minister and the Governor dated 30 August the Bar Council stated in regard to the composition of the JSC:

"The Council felt that the Chairman or other representative of the Bar Council and a member of the public should be appointed to the [JSC]. The Council did not find it objectionable that the President of the Court of Appeal should chair the [JSC]. The Council felt that the Chief Justice should also be a member of that Commission notwithstanding the views of the judiciary that he should not unless he is also the Chairman. The Council agrees with the judiciary that the period of appointment of the [JSC] members should be specified as should other relevant provisions necessary for good governance, certainty and transparency"

The Bar Council objected to Article 57(3) on the ground that it was capable of neutralising the obligation of the Governor under Article 57(2) to act in accordance with the advice of the JSC. On 6 September the Chief Minister and the Acting Governor jointly replied to this letter. They pointed out, inter alia, that Article 57(3) had been included at the behest of the UK side, and that it formed an important element of the UK position in the negotiating process, as part of the UK's agreement to the establishment of the new JSC, which would significantly reduce the Governor's powers. Nevertheless the UK Government "would only envisage the power being used in extremely rare and exceptional circumstances", of which examples were given.

4.46 On 7 September the Chief Secretary, Mr Montado, wrote directly to the Chief Justice in response to Hassans' letter of 25 August. He pointed out that a new Constitution was in the first instance a matter for the UK legislature, while it would be for the judiciary to adjudicate on any challenge to it. He stated that the Government shared his views about the importance of "judicial independence, in all its aspects and dimensions". He repeated the substance of the joint letter to the Bar Council, including the limited circumstances in which Article 57(3) might be used. He rejected the complaints of inadequate consultation, explained how certain concerns would be met and stated that a reference to the Privy Council was neither necessary nor appropriate. In a letter of the same date to the Chief Justice the Assistant Deputy Governor, on behalf of the UK Government, rejected the Chief Justice's concerns and the suggestion of a reference to the Privy Council. The letter concluded by saying that, notwithstanding that the Chief Justice had, regrettably, turned down previous opportunities to talk to them, UK officials would be willing to see him in London to explain in more detail the UK Government's position on the judicial aspects of the draft Constitution. The Chief Justice did not take up this offer and attempt to achieve his aims by private negotiation rather than, as will appear later in this chapter, by public confrontation.

4.47 At the Opening of the Legal Year on 6 October most of the address of the Chief Justice was devoted to the draft Constitution. He stated that he was taking the opportunity to address the whole Bar on the position of the locally-based judiciary. He had taken the unusual step of distributing the judiciary's submissions on the final draft. Having referred to an address to the CMJA by the Lord Chief Justice of England and Wales on the consultation process leading up to constitutional reform in the UK, he said:

"How then in Gibraltar do we find ourselves in a position where the judiciary is making submissions on a Draft Constitution which has been agreed between the Governments of Gibraltar and the United Kingdom? The fact is that the judiciary was not consulted at all on the provisions which give it the most cause for concern ie Articles 57 (2)(c), 57(3) and 64 (7)."

He said that the judiciary had made its initial submissions in March 2005. He had attempted without success to meet the legal adviser to the FCO who had made no attempt to contact him. On 9 January 2006 the Chief Minister had written to him providing a note of various provisions affecting the judiciary, but had not sent the text of those provisions. He quoted the closing words of that letter, pointing out that, despite the Chief Minister's assurance, he had not heard from him about the outcome of negotiations with the UK Government.

4.48 After the announcement of agreement in March 2006, the Chief Justice said, he had immediately contacted the Deputy Governor for a copy of the final draft, but was told he would be able to see it when it was made public. He had not received a copy of the draft from either government. He had not seen the draft until it was accessed by him on 7 August. He added:

"It would appear that it was forgotten or ignored that as the Third Branch of Government the judiciary should have been fully consulted and engaged during the process which led to the Draft".

On 11 August he had made submissions to both governments, which were the submissions of himself, as Chief Justice, the Puisne Judge, the Stipendiary Magistrate and the Registrar. They had sought consultation with the governments, and had advised that the judiciary "was considering all possible steps to prevent what it regarded as errors of principle being perpetrated". There had been no response from the Chief Minister about a meeting. He referred to the letters from Hassans whom he had briefed as his solicitors. The fact that there had been no response or acknowledgement by 5 September was "disconcerting". He continued:

"What happened next was positively unnerving. The Governments of Gibraltar and the UK, in a letter under both Coats of Arms, replied to the Bar Council of Gibraltar in a letter dated 6th September and signed jointly by the Acting Governor and the Chief Minister. In that letter the Governments dealt with the points raised by the Bar Council in its letter of 30th August item by item and stating that they would be content for the Bar Council's letter and the response to be made public.

On the 8th September, I received a response to the judiciary's submissions albeit that the letters were dated the 7th. This was at a time that the Acting Governor knew I was on my way to Canada. The Government of Gibraltar's letter was signed by the Chief Secretary. It dealt with the judiciary's submissions item by item but said that there was "no need or justification for any further process of consultation with you". The reply from the office of Governor is signed by the Assistant Deputy Governor. It deals at length with the process of alleged consultation and then asserts that the UK government rejects the judiciary's allegations that the new Constitution would adversely affect judicial independence. The letter does not answer, item by item, the judiciary's submissions although it does refer to the assurances given to the Bar Council. The letter also says that "UK officials would be willing to see you in London to explain in more detail the UK government's position on the judicial aspects of the draft Constitution" but does not suggest that this would be part of a process of consultation.

What I find unnerving about this turn of events is that the Acting Governor and Chief Minister would prefer to make their positions known through the medium of the Bar Council and then respond to the judiciary through their officials. Does this bode well for the judiciary in the consultation process promised on the Judicial Services Ordinance? It

must be borne in mind that it is the judiciary's independence which is under threat, not that of the Bar Council. And what of the letter to the Bar Council under signature of the Acting Governor and the Chief Minister? One cause for concern is that the Draft Constitution provides for a JSC comprised of three members of the judiciary and two appointments of the Governor acting in his discretion and two appointments of the Governor acting in accordance with the advice of the Chief Minister. This, says the judiciary, makes for a JSC with four out of seven appointments by the executive. Not so, says the executive: it makes for a JSC with three Judicial members and two from the UK Government and two from the Gibraltar Government. But in their letter to the Bar Council of 6th September the Governments of the UK and Gibraltar have demonstrated that on judicial matters they are prepared to act jointly. What is, then, to prevent them from joining forces when it comes to appointments to the JSC to ensure that the executive gets its way in JSC matters?"

4.49 The Chief Justice also said in regard to the JSC:

"I am also concerned that the Chief Justice is a member, but not its Chairman. This means that the Head of the judiciary will be shut inside the JSC, bound by its confidentiality, and, despite his wider constitutional responsibilities, unable to take forward any concerns over the workings of the JSC that he may have".

4.50 The Chief Justice went on to comment that the effect of Article 57(2)(c)would be to put junior members of the judiciary under the control of the executive. He knew, he said, that the Registrar was particularly concerned about her vulnerability, given that her duties straddled the administrative, quasi-judicial and judicial. Section 57(3) put the ultimate control over appointments to the judiciary in the hands of the executive. He was not sanguine about the governments' assurances that the power of veto would only be used in exceptional circumstances and was in any case subject to judicial review. He then said:

"The issues involved are fundamental and will affect future generations of Gibraltarians. It is for the judiciary to protect its independence so that it may in turn protect the rule of law. As Head of the judiciary I have a

duty to ensure that the Constitution together with the assurances do indeed provide the necessary safeguards. I am therefore in the process of putting together a team of constitutional experts who will give an independent opinion on whether the Constitution does provide for an independent judiciary given the recent communication from the UK and Gibraltar Governments. I shall make that opinion available to both Governments, to all members of the Gibraltar Bar, and will also make it public. If there are still issues of concern I have advised the UK Government that I shall take up their offer of a meeting".

- 4.51 In oral evidence the Chief Minister explained why, despite the terms of his letter dated 9 January, he did not inform the Chief Justice about the likely outcome of the negotiations between the Gibraltar and United Kingdom delegation as they affected the judiciary. He said he considered that he was simply to respond in relation to matters set out in the matrix, that he had always seen section 57(3) as giving only a power of veto, and that it was one of the matters which arose in a very short timescale at the end stage of the process.
- 4.52 There are a number of respects in which we find the conduct of the Chief Justice in making this speech wholly inappropriate.
- 4.53 First, although he purported to speak for the locally-based judiciary, it is clear that he had not consulted them about his address, provided them with advance copies or obtained their consent to the public distribution of the submissions of 11 August. He did not consult them before engaging Hassans or instructing them to write the letters of 25 August and 5 September. He was acting on his own initiative.
- 4.54 Secondly, it is clear to us that the Chief Justice was proceeding on the basis of an exaggerated conception of the status of the judiciary in regard to the negotiations. We have already noted that, when referring to the fact that he did not see the draft until August, he said: "It would appear that it was forgotten or ignored that as the Third Branch of Government the judiciary should have been fully consulted and engaged during the process which led to the Draft". It is evident that he considered that the judiciary had a

- right to be consulted at every stage, as surely as if they were a party to the negotiations.
- 4.55 Thirdly, while there may have been some justification for the Chief Justice thinking that he should have heard earlier about the provisions which he found objectionable, his address was deliberately worded in such a way as to give rise to the implication that he was accusing both governments of having acted in bad faith, and we so find. His account of the consultation process was deliberately slanted and misleading. His remarks implied that information about the inclusion of these provisions in the draft Constitution had been deliberately withheld from him. It is plain that at the time his remarks were so understood. On 7 October 2006 the Gibraltar Chronicle carried an article headed "Chief Justice accuses Gib and UK of ignoring judiciary's role in constitutional talks". In evidence the Chief Justice said that his remarks had been misunderstood. He claimed that he could not have put out any kind of corrective statement. He said: "The difficulty is that I am not responsible for how the press interprets my remarks and if I then go into further explanations, then they may be open to further interpretations". We find that explanation unconvincing and disingenuous, and reject it. We are also satisfied that the Chief Justice deliberately implied that the governments might act together in bad faith "to ensure that the executive gets its way in JSC matters". He also implied that the letters dated 7 September had been deliberately sent to him at a time when the Acting Governor knew he was on his way to Canada, and hence that there had been bad faith in this respect also. The Chief Justice gave evidence that he had merely felt that the timing of the letters was "unfortunate", adding "I was certainly hampered in providing a response". Mr Philip Barton, who was Acting Governor at the time, said in evidence that any such implication of bad faith was untrue. The timing of the response to Hassans' letter of 25 August was, he said, determined by the time which it took to prepare it and by the deadline of 8 September in Hassans' letter of 5 September. Mr Barton was a careful and impressive witness, and we accept his evidence.

- 4.56 Fourthly, in his address he made no effort to present a balanced view of the text of the proposed Constitution, which (as explained in chapter 1) represented a considerable improvement for Gibraltar in comparison with the previous one, or of the process of negotiation in the course of which concessions might have to be made on either side in order to reach agreement. Furthermore, in his evidence the Chief Justice accepted that perhaps it would have been fairer if he had reflected the fact that the proposed Constitution had been produced following cross-party negotiation and was now supported by all political parties, and that it should not have been described as simply "agreed between the governments of Gibraltar and the United Kingdom".
- 4.57 Fifthly, his address contained a number of other significant omissions. He did not refer to the fact that, at least as regards the question of the composition of the JSC, he had been consulted. He did not take account of the reservations which Mr Pitto and Mr Justice Dudley had about the judiciary's submissions, saying that this was because if he had "given every reservation, every point in this address it would have been never ending".
- 4.58 Sixthly, and perhaps most importantly, the address was unmistakeably polemical in tone. He said that he wanted the Bar and the public to know why he way taking action at that relatively late stage. The action he took was the opening of a public campaign, to influence public opinion and put pressure on the Governments. In evidence he said: "I felt that if I called upon senior advisers to give a public opinion, it may change the minds of those responsible...the drafters of the Constitution". This formed a continuation of a process which had begun with Hassans' letters, the enlisting of Lord Lester as a legal adviser and the threat of a petition to the Privy Council. According to the evidence of the Chief Justice, Lord Lester took "a great hand in drafting the petition" under section 4 of the Judicial Committee Act 1833. However, according to his fifth witness statement, Lord Lester changed his mind and did not consider such a course should be pursued. Thereafter the Chief Justice sought the advice of Sir Sydney

- Kentridge QC, Keith Starmer QC and Richard Tur, which was in turn the subject of publicity.
- 4.59 In evidence the Chief Justice accepted that he had been keen that this advice should be published before the referendum. This was reflected in the press. On 27 October the Gibraltar Chronicle included an article under the title "Chief Justice expects experts' Report ahead of referendum". This was three days after the Government announced that the referendum was to take place on 30 November. On 16 November the Chief Justice published the opinion which he had obtained from the three legal experts. They considered that the JSC appointments were weighted too heavily in favour of the executive, ie the Governor and the Government. They questioned why the JSC should be chaired by a "visitor", and expressed reservations about Article 57(3) where judicial review would be of little practical effect. On 16 November both governments issued press releases roundly rejecting this opinion. The Gibraltar Chronicle of 17 November carried a statement by the Chief Justice in which he announced the promised release of the opinion. Members of the public who wished to obtain copies could contact the Registrar. In the same newspaper was an article entitled: "Experts express 'serious reservations' with judicial aspects of draft Constitution".
- 4.60 On 20 November the Chief Justice and Sir Sydney Kentridge attended a meeting at the FCO. As might have been expected, there was no question of the proposed constitution being withdrawn or amended. The Chief Justice had known, since the time when they were approved by the Foreign Secretary, that the provisions in the draft Constitution were, in the words of his first witness statement, "set in stone". Two days later the *Gibraltar Chronicle* reported: "F & CO remains firm on constitution's text". However, arising from the discussion it was agreed that an addition would be made to the explanatory note, which does not form part of the Constitution, stating that the executive powers of the JSC were "subject only to an exceptional power of veto by the Governor". The Chief Justice regarded these words, in comparison with the written assurance which had been given to the Bar Council, as of far greater weight. However, his

- reliance on the explanatory note was misplaced as it added little, if anything, to that assurance.
- 4.61 On the following day the *Gibraltar Chronicle* carried an article "Yes or No constitution debate warms. Opposition backs judiciary over concerns". On 24 November *Vox* featured "Clash over constitution could be taken to Privy Council". In the referendum on 30 November the new Constitution was approved.
- 4.62 In our view the Chief Justice behaved improperly by undertaking a public campaign in regard to part of the subject of the referendum. The fact that his concerns related to the judiciary or were shared by legal experts is nothing to the point. What is objectionable was the manner in which he sought to advance his views. The little that he did achieve was the result of his engaging in private discussion on 20 November. His involvement of solicitors and other legal advisers and the threat of legal proceedings in support of his campaign were in any event unnecessary and grossly disproportionate.
- 4.63 It is also right to point out the consequences of that campaign, which were all too plain in the evidence. The publicity which the Chief Justice sought gave the press 'a field day'. The Opposition seized on his remarks. Consensus changed to dissent. As the Chief Minister commented: "It gave the Opposition a hook on which to play politics with the Government on the referendum, when the new Constitution was a huge improvement". The Attorney General said: "The way the referendum fell out was that it almost became a vote of confidence in the Government. [The Chief Justice] was perceived as being very much part of the no campaign". Mr James Neish QC, who described how in August or September he had sought unsuccessfully to discourage the Chief Justice from, as he put it, "another ill-judged, non productive, damaging public confrontation", said: "We are a small jurisdiction which has been under close political scrutiny from a hostile neighbour. We rely on the reputation and integrity of our judicial system to be able to prosper as a financial centre. We need the users of Gibraltar to have confidence in the judiciary, and to suggest that there was some fundamental flaw in the constitution that was being

proposed was in my view a totally disproportionate and damaging line to take". Lastly, we have no doubt that the Chief Justice could never have sat in a case in which the disputed provisions required to be interpreted or applied.

CHAPTER 5

FEBRUARY -MAY 2007

This chapter is concerned with events relating to, and arising from, the following:

- the Judicial Service Bill (paragraphs 5.1 5.35)
- Mrs Schofield's libel action (paragraphs 5.36 5.42)
- the recusal application in April (paragraphs 5.43 5.93)
- the recusal application in May (paragraphs 5.94 5.102).

The dates in this chapter of the Report refer to 2007, except where otherwise indicated.

The Judicial Service Bill

The Chief Justice

- 5.1 Both the Chief Justice and his wife were active in commenting on the proposal for a Judicial Service Act. Since we need to examine what happened in some detail, it is convenient to deal first with the direct involvement of the Chief Justice, and thereafter with that of his wife.
- 5.2 On 20 February the Chief Minister wrote to the Chief Justice enclosing an advance copy of the draft Bill and of the consultation paper relating to it. The letter and its enclosures were delivered to the Chief Justice by hand that day. In his letter he said that he was sending the draft in advance as he understood that the Chief Justice would be travelling that week to Argentina. He explained that there would be a six week consultation period after which any amendments arising from the consultation process would be introduced during the Committee stage. At the end of the letter he wrote:

"Finally, since no other consultee will yet have seen the draft Bill, I must ask that you treat it confidentially and "for your eyes only" until the consultation paper issues next week."

- 5.3 Section 5 of the draft Bill made provision empowering the President of the Court of Appeal and the Chief Justice to make written representations before Parliament on matters that appeared to them, or either of them, to be of importance relating to the judiciary, or otherwise to the administration of justice in Gibraltar. According to section 6 there would be a new office of President of the Courts of Gibraltar, which would be held by the President of the Court of Appeal. Under sub-section (2) he would have "overall responsibility -
 - (a) for representing the views of the judiciary of Gibraltar to Parliament, to the Minister and to the government generally;
 - (b) for the maintenance of appropriate arrangements for the welfare, training and guidance of the judiciary of Gibraltar within the resources made available by the Government;
 - (c) for the maintenance of appropriate arrangements for the allocation of work within courts."

In regard to sections 5 and 6 the consultation paper stated:

"The Government believes that the judiciary is a primary player in the protection of its independence and should therefore have an appropriate and dignified method of making its views known to the Parliament, and through the Parliament to the public at large. Accordingly, section 5 provides for a mechanism whereby the President of the Court of Appeal and the Chief Justice can have their written representations laid before Parliament.

Although the Court of Appeal is an itinerant court, it is nevertheless a Gibraltar Court and its President is therefore the most senior member of Gibraltar's own judiciary. The Government no longer considers it appropriate, in the context of the new Constitution, for the Court of Appeal to "look like" an external court (even though it has never actually

- been that). Accordingly, section 6 establishes the President of the Court of Appeal as the President of the Courts of Gibraltar."
- 5.4 In evidence the Chief Justice said that he was stunned when he received the draft Bill. His initial reaction was that section 6 was not in the interests of justice. He added: "And yes, it was the Chief Minister's way of ridding him of having to deal with me as head of the judiciary ... Clearly from this I would be put in a position where I had to accept that I had in effect been demoted from head of the judiciary or resign". He immediately sought a meeting with the Governor to discuss the matter, but the latter was not aware of the draft Bill or the consultation paper as these had not yet been issued for consultation.
- 5.5 The Chief Justice wrote to the Chief Minister in a letter hand delivered the following day, 21 February, expressing concern about, as he put it, the functions of the Chief Justice being put in the hands of a visiting President of the Court of Appeal and despite the Chief Minister's request for confidentiality, stating that he felt that he must discuss the draft with members of the Court of Appeal while they were in Gibraltar during that week. In the light of his expressed intention to discuss the draft Bill with the Court of Appeal, the Government decided to bring forward the issue of the consultation paper by the Chief Minister, in his capacity as Minister for Justice, to that same day. On 21 or 22 February the Chief Justice consulted Mr James Levy, a senior partner of Hassans concerning the constitutionality of the proposed Act. According to his evidence Mr Levy advised him to write to the Chief Minister laying down a marker that he had his concerns about "the constitutional and contractual implications". This led to the Chief Justice writing again to the Chief Minister on 22 February, stating that he had had very little time to review the draft before he went on holiday. Accordingly he reserved any detailed comment to his return. The letter continued: "However, my immediate reaction is that I have reservations about the constitutionality of clause 6. Furthermore, the provision may well have contractual implications for me".
- 5.6 On 23 February, without further reference to the Government, the Chief Justice instructed the Registrar to circulate copies of the consultation

paper and the draft Bill to all members of the Bar (despite the fact that the consultation paper provided for circulation only to the Bar Council, as representing the Bar). In her covering e-mail the Registrar stated that the Chief Justice felt that the proposed Act might have wider constitutional implications, and, before he responded, would welcome the views of individual members of the Bar, which should be submitted no later than 20 March. In his evidence the Chief Justice stated that he had understood that in regard to the 2006 Constitution the Bar Council had not gone out to wider consultation with its members. He wanted to have the input of the whole of the Bar, and not every one was a member of the Bar Council (in due course he received only two papers from members of the Bar). He had not asked the Chief Minister to extend the list of consultees. He explained that his understanding of the last paragraph of the Chief Minister's letter of 20 February to which we refer in paragraph 5.2 was that, once all the consultees had received the draft Bill and the consultation paper, those documents were no longer confidential. He accepted, in the light of instructions given to his solicitors in connection with obtaining legal advice (see paragraph 5.10), that he must have been aware that the Opposition members of the Parliament would not have seen the draft Bill until 29 March. "It may be", he said, "something that I missed". He understood that once all the consultees had received their copies, the documents were no longer confidential, adding: "But clearly I may not have given that aspect of it the required thought". He denied that he had given any thought to conflicts of interest within the Bar Council, though he admitted that it was his personal view that certain of its members were partisan. He referred in particular to Mr James Neish QC, the Chairman, as being "a good friend of and ally of the Chief Minister". He accepted that he had discussed that topic with his wife over a period of time.

5.7 We find the conduct of the Chief Justice in bypassing the Bar Council by circulating copies of the draft Bill and the consultation paper to the whole of the Gibraltar Bar – and in doing so without further reference to the Government – to be not merely high-handed but deliberately provocative. We find that he was at the time well aware of the nature of the

consultation process, and in particular of the fact that the consultation process provided that opposition members of Gibraltar Parliament would first see the draft Bill on 29 March, when it was due to be debated in Parliament, but that he had felt no compunction in ignoring it.

5.8 On 24 February the Chief Justice left Gibraltar with his daughter Amanda for a three-week holiday in Argentina. On 28 February the Chairman of the Bar Council wrote to the Registrar asking why the Chief Justice had sought to "bypass the Bar Council". The Registrar replied on the same day stating that she did not know the reasons for the Chief Justice's instructions. On 5 March Sir Paul Kennedy wrote on behalf of the members of the Court of Appeal to Mr John Reyes, the Chief Minister's Legal Secretary, saying that the members of the Court of Appeal saw no reason why a senior judge such as the President of the Court of Appeal should not be given the office of President of the Courts of Gibraltar, but suggesting that, as the President of the Court of Appeal was not permanently resident in Gibraltar, section 6 should be redrafted to make it clear that the direct (as opposed to the overall) responsibility for the day to day discharge of the duties set out in section 6 (2)(b) and (c) lay with the Chief Justice. Sir Paul Kennedy concluded his letter by saying:

> "As the Chief Justice is in South America and as the consultation period is short we have not been able to consult with him in relation to the content of this letter but I am sending a copy of it to await his return."

5.9 Following his return from Argentina on 15 March the Chief Justice wrote to the Chief Minister on 26 March expressing his concern about the constitutionality of section 6 and section 38(3), the latter dealing with the exercise of disciplinary power over a senior office holder. He stated that he had commissioned an opinion from leading counsel in London. In regard to the practical implications of section 6, he questioned whether it could possibly be in the interests of the administration of justice for overall responsibility for the judiciary to be transferred from a full-time, resident Chief Justice with full security of tenure to a part-time, non-resident President of the Court of Appeal on a relatively short-term contract.

- 5.10 By e-mail from his English solicitors dated 26 March the Chief Justice sought the advice of Rabinder Singh QC and Alex Bailin on whether the proposed Act would be ultra vires, whether action could be brought in the United Kingdom (as he preferred) and whether he had a claim for constructive dismissal, on the basis that, as stated in the Instructions, "Since 1998, there have been major disagreements between Derek and the Chief Minister who has been trying to get rid of him, so far without success". On 28 March they provided a written opinion on the draft Bill. They advised that section 6 would be unconstitutional since "it undermines the core constitutional principle of the independence of the judiciary, which itself forms part of the rule of law". An amendment to provide that for the Supreme Court and lower courts the Chief Justice should have direct responsibility for the matters specified in subsection (2)(b) and (c) would in their opinion create "an artificial and unworkable division of functions". They considered that the independence of the judiciary, enshrined in the Constitution, required that the President of the Courts must have a degree of security of tenure which the Court of Appeal judges did not have, and the Chief Justice did.
- 5.11 On 2 April Sir Murray Stuart-Smith wrote to Mr Reyes referring to a letter from Mr Justice Dudley, in which he had expressed reservations about section 6 of the draft Bill, suggesting that it might be undesirable to place the titular headship of the Gibraltar Courts on an office to which a Gibraltarian could not aspire, and expressing the view that it was in any event impracticable for a non-resident judge effectively to discharge the functions specified in section 6(2). Sir Murray Stuart-Smith said in his letter that he had discussed the matter with the other members of the Court of Appeal, and that they saw force in the view expressed by Mr Justice Dudley.
- 5.12 On 3 May the Government issued a press release announcing the publication of the Bill. It explained that following a process of consultation the draft had been modified to reflect some of the views which had been expressed by consultees. The modifications included one to section 6 by which, subject to the President's overall responsibility, the Chief Justice

- would have direct, day to day responsibility for the matters mentioned in paragraphs (b) and (c) of subsection (3), previously subsection (2), for the Supreme Court and all lower courts.
- 5.13 The Judicial Service Bill was debated on 15 June, and was passed with commencement on 5 July. As in the original draft, the President of Court of Appeal remained President of the Gibraltar judiciary. Subsection (3) of section 6, which was in the same terms as sub-clause (2) of the Bill, was followed by subsection (4) which was in the same terms as the modification to which we referred above. The Act received the assent of the Governor on 27 June.

Mrs Schofield

5.14 At around lunchtime on 20 February the Chief Justice discussed the draft Bill and the consultation paper with Mrs Schofield. In answer to his counsel he said: "I simply disclosed section 6 to her because I felt it may affect our stay in Gibraltar, and I felt it right that I should share that with my wife". In cross-examination he said it was his clear recollection that he did not disclose to her the remainder of the Bill or the consultation paper. When he was asked if he knew how she came to be in possession of a copy of the draft Bill on 22 February when she consulted Mr Charles Gomez (see below), he said he did not "provide" her with a copy of the Bill, meaning that he did not say "here is the Bill", though the consultation paper was "a very hot topic of conversation" at their home. The draft Bill was left in the study at his home, he thought. In his third witness statement the Chief Justice said:

"I know that my wife was deeply upset about section 6 of the draft and the way I had been treated after eleven years of service to Gibraltar. She expressed her dismay that I had not been forewarned of the intention to remove me as Head of the judiciary so as to enable me to decide whether to make a dignified exit or accept the changes. She was also concerned about the implications on my family life ...".

In cross-examination the Chief Justice said that Mrs Schofield was wounded in two areas: by section 6 and by the summary termination the same day of her consultancy with Hassans. He described this as a very traumatic time for her particularly. On 22 February she instructed Mr Gomez to investigate the possibility of issuing proceedings challenging the proposed Act. Mr Gomez in his second witness statement said that she brought a copy of the Bill.

5.15 On 23 February *Vox* published an article referring to section 6 under the heading "Now Caruana Attempts to Twist the Law – Plans for New Judicial Service Act Erodes Role of Chief Justice". It referred to a "leaked draft", which the Chief Justice had received, and to the Chief Justice immediately seeking a meeting with the Governor. An unnamed source was quoted as saying:

"These measures clearly usurp many of the present duties and responsibilities of the Chief Justice and would demote him to a lesser role – particularly as the President's is a part-time position held by a retired English judge ... often one who has been put out to pasture."

The article went on to say:

"Local lawyer and politician Charles Gomez yesterday confirmed that he was looking to initiate proceedings to declare the act [sic] invalid after being consulted by a "concerned citizen" who had seen a copy of the draft legislation".

5.16 The Chief Justice, when asked in examination-in-chief whether he had suggested or encouraged Mrs Schofield to "go public" in relation to section 6, said: "No. It came as a surprise". In cross-examination he said he became aware that she had been to see Mr Gomez: it may well have been before he left for Argentina. He could not recall whether he was aware of the article before he left. He did not recall discussing it with her at that time. At most at that stage he would have "skim-read" it. He agreed that even a skim read would have made it clear that the article contained some very serious allegations. He was sure that he did not discuss with his wife whether she had provided *Vox* with a copy of the draft Bill. Recently he asked her whether she was the source of the quotation. He said he thought that her answer was "Absolutely not".

When it was put to him that it might be thought somewhat implausible that he had not discussed any of these matters with her before he left for Argentina, he replied:

"Well, you may suggest that, but I have explained that I have a very ambiguous attitude towards the weekly newspapers, and I may or may not have even seen it."

- 5.17 In all the circumstances, and given the reference in the article to the Chief Justice having "received the leaked copy of the draft on Monday or Tuesday of this week", we have no doubt, and we find, that Mrs Schofield and Mr Gomez with her consent provided *Vox* with the copy of the draft Bill. She was the "concerned citizen" in the article. Further, we do not accept the Chief Justice's evidence that he only discussed section 6 of the draft Bill with Mrs Schofield; that he did not know how she came by a copy of the draft Bill; that it came as a surprise to him that she "went public" in the *Vox* article; and that he may not even have read the article, and certainly did not discuss it with her, before he left for Argentina. We find that the Chief Justice showed Mrs Schofield the draft Bill and the consultation paper as soon as he received them (i.e. on 20 February); that his discussion with her about them was not limited to section 6 (although no doubt section 6 was the focus of their attention); that he left the draft Bill in his house where it was available to her: and that it was no surprise to him that she "went public" in the way she did. We further find that it was at the material time the expectation of the Chief Justice that Mrs Schofield would "go public" on the matter of the draft Bill; and that he was, at the very least, content that she should do so and supportive of her in that respect. Nor do we accept his evidence that he did not discuss the *Vox* article with Mrs Schofield before he left for Argentina. The overwhelming likelihood is that they discussed it together before he left for Argentina, and we have no hesitation in finding that they did so. We regret to say that we found the Chief Justice's evidence to us on each of these matters to be deliberately evasive.
- 5.18 On 25 February, that is to say, the day after the Chief Justice left for Argentina, Mrs Schofield sent an e-mail to Mr Neish about the

consultation paper and a forthcoming meeting of the Bar Council which was to discuss the draft Bill. She referred to the fact that one of the functions of the Bar Council was to consider all recommendations and other matters referred to it by any authority, lawyer or member of the public. "As a member of the public and a person who may be affected by any constitutional and contractual implications of the draft Bill" she recommended that, before discussing the consultation paper and draft, the Bar Council considered and made recommendations on a number of matters. These included a recommendation that every member of the Bar Council "discloses personal, political and business relationship with the Government of Gibraltar, Minister for Justice or any other member of the Gibraltar Government", and that "members disclose whether they had discussed the proposed Bill prior to its being referred to the Bar Council for consultation and if so, when". She went on to request the Bar Council to consider (among other things) "whether the Gibraltar Parliament had power under local legislation in the current constitutional arrangement to restructure the court structures set up under the Constitution", and "whether the role of the Chief Justice can be re-defined by local legislation". The e-mail concluded as follows:

"I consider the matters raised as of utmost importance. I am seeking a legal opinion as to whether the draft violates the Constitution and by targeting retired judges of the court of appeal is discriminatory and unconstitutional. Whether it is intended to cleanse the office of head of the judiciary at the moment to ensure the holder or the current and future holders of the office of head of the judiciary whose experience may be from outside the UK are excluded and only retired English judges can hold the office.

Looking forward to hearing from you. Please be advised that unless I hear from you by the end of the day tomorrow I shall have no alternative but to release this request to the public through the press to raise awareness on the need for informed debate by the Bar Council, the Public and Parliament on the proposed Bill. Section [6] to my mind is an "attempted rape" of the Gibraltar Constitution and the Chief Justice's office and contract. In my view it is intended to force a resignation of the

CJ unless he accepts a demotion or to force him to sue in which case we shall hear calls for him to resign."

- 5.19 On 27th February Mr Neish replied on behalf the Bar Council stating that it was well aware of its functions and responsibilities and would form its conclusions independently and express such views to the Minister for Justice as it considered to be in the best interests of the Bar.
- 5.20 On 2 March *Vox* carried an article about the Chief Justice's decision to circulate all members of the Bar with copies of the draft Bill, under the headline "Chief Justice loses control of courts - Schofield bypasses Bar Council in legal clash". It stated that the Chief Justice doubted the impartiality of the Bar Council, believing that several of its members were partisan, and supported the Chief Minister. In a parallel move his wife had written to the Bar Council asking its members to declare any interest they might have which could affect their attitude to the Chief Minister's moves against the independence of the judiciary. The article claimed that Mrs Schofield clearly believed that if any of the Bar Council had such links they should effectively recuse themselves from any deliberations on the proposed Act and its implications. We accept Mrs Schofield's evidence in her first witness statement that she was in Kenya when this article was written, and that she had no hand in it. On 8 March she sent a further email to Mr Neish in which she asked for confirmation that the Bar Council acknowledged her right to ask it to consider various facts in accordance with its published functions.
- 5.21 We accept the evidence of the Chief Justice (which was corroborated by his daughter Amanda) that he was not made aware of the correspondence which had taken place between Mrs Schofield and the Bar Council until after his return, and that there was no discussion between himself and Mrs Schofield about such correspondence while he was in Argentina. When he was asked by counsel to the Tribunal how he came to learn of its existence, he said:

"She [Mrs Schofield] told me that she had sent e-mails, and I said: "Oh dear, what have you done now? I don't want to know."

He accepted she probably had told him the gist of the e-mails which she had sent. He said that she had pointed the Bar Council to matters of principle. He did not recall her telling him that she had suggested to the Bar Council that the draft Bill was an attempted rape of the Constitution, or that she had raised a number of questions about the ages of the current members of the Court of Appeal. Asked whether she had told him that she had sought a response in 24 hours, failing which she would go public, he replied:

"No, I don't recall that ... I really took very little interest in it. She was well-known to the Bar Council. The Bar Council was well known to her. I anticipated that the e-mails would be forthright, but I would expect the Bar Council to understand, given the circumstances..."

In response to further questions he said:

"I did not pay a great deal of attention to this. I felt that that issue would blow over ... that it would all be ironed out ... I did not anticipate it could possibly escalate".

When he was asked if he was even interested in looking at these e-mails he responded:

"No. At that stage I felt that was an issue between my wife and the Bar Council".

Asked whether he was not interested at all in seeing precisely how the Bar Council responded to those points of principle that she had raised, he answered:

"I would have been interested to see the Bar Council's response to the consultation paper, but not to my wife ... I think she told me, and I cannot recollect exactly, I have no doubt she told me that she had received a bland reply from them".

This conversation was, he thought, in the late afternoon of 15 March.

5.22 We regard the above passages of evidence by the Chief Justice as wholly unconvincing, and we do not accept them. In particular, we consider the Chief Justice's assertions that he "really took very little interest" in the

correspondence which had taken place between Mrs Schofield and the Bar Council, and that he "was not interested" in seeing how the Bar Council had responded to the points of principle which she had raised, to be frankly incredible. We find that the Chief Justice was, as indeed one would have expected of him in the circumstances, concerned to acquaint himself fully, and that he did acquaint himself fully, with the terms of the correspondence which had taken place. We also reject his evidence that he felt that the issue would "blow over". The overwhelming probability is, and we find, that he was fully aware that his wife intended to pursue the matter as vigorously as she could, as indeed she did.

5.23 Later on 15 March (at 7.58 pm to be precise) Mrs Schofield sent a further email to Mr Neish saying that she had received information that he or members of his firm had been involved in drafting the Bill and asking whether this information was correct and, if so, whether it was disclosed at the meeting of the Bar Council when the draft Bill was considered. She continued:

"If not, why not? Do you agree that if the information is correct, you or numbers of the firm in the council had a conflict of interest?

May I assert once again I have a right to make the request in this e-mail and any assertion by any member of the council to the contrary amounts to discrimination against me."

5.24 Mr Neish responded to Mrs Schofield's e-mail of 15 March by letter dated 3 April. In the light of what followed, the text is quoted in full, as follows:

"I refer to your e-mail of 15th March 2007, which was considered by the Bar Council last Thursday.

We regard your conduct in this matter as extremely serious. Not only do you try to interfere in the affairs of the Bar in a way which you have no right to do but you set about your task in a hugely confrontational and offensive way with no compunction whatever about raising questions about the integrity of members of the Council, and mine in particular. Your hostility is manifest.

This is unprecedented conduct in Gibraltar by the wife of a Chief Justice. Notwithstanding the ludicrous fiction under which you have sought to interfere, i.e. "as a member of the public", the inescapable fact is that you are the wife of the Chief Justice acting in your common interests. Your actions cannot be dissociated from the Chief Justice and, I regret to say, impact upon his position. This is an issue which will be addressed separately.

You state in your e-mail of 15th March 2007 that "you have received information...". Is this what we have come to – the wife of the Chief Justice being accessible to "informants" and relying on their "information"? Please note that your informant is either incompetent or malicious or both. Neither I nor any member of my firm have been consulted about the provisions of the Judicial Service Bill nor have any of us been involved in any way in drafting it. The first time I or any member of my firm saw the Bill was when it was sent to me as part of the Consultation Paper.

We regret that you should have created a situation which compels me to write to you in these terms. However, we cannot stand by and allow the Bar to be subjected to conduct of this kind which can only be intended to inhibit the Bar's freedom of expression and undermine its independence."

5.25 On 3 April Mr Neish wrote to the Attorney General:

"The Bar Council has asked me to write to you in your capacity as leader of the Bar in Gibraltar and to forward copies of e-mails exchanged between Mrs Schofield and the Bar Council between 25th February 2007 and 3rd April 2007, and of an article in the Vox issue of 2nd March 2007, for such action as you may consider appropriate.

As the Chief Justice's wife whatever Mrs Schofield has written or done is liable to be construed as having the express or implied approval or knowledge of the Chief Justice or as expressing their common views. The Chief Justice has not distanced himself from Mrs Schofield's e-mails or their contents.

In the circumstances the situation is unprecedented and needs to be addressed. Not only has there been an attempt to interfere with the workings of the Bar but allegations/innuendos have been made against

members of the Bar Council questioning their integrity. The implications for the future administration of justice are obvious and do not need to be spelt out."

5.26 On the same date Mr Neish wrote to the Chief Justice as follows:

"I am writing to inform you that the Bar Council has taken a very serious view of the e-mails addressed to it by Mrs Schofield. It has accordingly decided to send copies of the e-mails exchanged, together with the issue of Vox dated 2nd March 2007, to the Attorney General, as the leader of the Bar in Gibraltar, for the matter to be pursued as appropriate.

Mrs Schofield is not an ordinary member of the public – she is the Chief Justice's wife. Her e-mails were therefore liable to be construed – as in fact they have been – as enjoying your express or implied approval or as reflecting your and Mrs Schofield's common views. We note that you have not distanced yourself from those e-mails."

5.27 The evidence of the Chief Justice was that it was only when he received Mr Neish's letter that he gave any detailed consideration to the text of the e-mails of 26 February, 6 March and 15 March. As he put it:

"When I saw that Mr Neish was taking this further, I felt then that I ought to have a look at the e-mails ... I think I had to ask my wife for copies of them. I must say, even then, I skim-read them ... I thought I had better have a look."

We reject that evidence. We have no doubt, and we find, that by the time the Chief Justice saw Mr Neish's letter he was already well aware of the terms of the e-mail correspondence.

5.28 In answer to further cross-examination, the Chief Justice did not accept that the final paragraph of Mr Neish's letter contained an implicit invitation to him to distance himself from the statements in Mrs Schofeild's e-mails. He understood Mr Neish to be "making a statement ... He was just making a note that I hadn't". Asked whether he had considered whether he should distance himself from the e-mails, he said:

"I have made it clear throughout, I think, that it is well known to the Bar and it is well known to Gibraltar that my wife is an independent-minded individual. I did not consider in the context of the Judicial Services Act that public opinion would consider that her views were necessarily mine. And I do not consider it appropriate either to associate or disassociate myself."

He said that he did not even consider putting out a statement to the effect that her views should not be taken as his. Later in his cross-examination, the Chief Justice confirmed that he had by then formed the view that the Chief Minister was trying to rid the jurisdiction of him as Chief Justice, and that Mr Neish was partisan. However, he had not considered whether Mr Neish might have a conflict of interest. He did not think that Mr Neish's partisanship would necessarily go so far as to support the Chief Minister. He went on to say that the matter was being taken up by the Bar Council in "inappropriately serious form". He said that he "felt that the matter could have been laid to rest very easily within the Bar Council ... by simply writing a bland response" to his wife. The Chief Justice was then asked again whether it had occurred to him that he should distance himself from Mrs Schofield's e-mails, to which he replied: "As I have said, ... at that stage I would not give conscious thought to issuing a statement". It was not until he was asked to recuse himself (see paragraph 5.43 et seq) that he made a conscious decision not to respond or react.

5.29 It is an understatement to say that we find that the Chief Justice's evidence in relation to Mr Neish's letter, and his reaction to it, to be less than satisfactory. It must have been clear to any reader of the final paragraph of the letter, and most especially to the Chief Justice of all people, that the Bar Council considered that he ought to have distanced himself from Mrs Schofield's e-mails, and that by not doing so he had provided tacit support for the statements contained in them; and that in that paragraph the Bar Council was inviting him to remedy the situation by making a statement distancing himself from those statements. To say of that paragraph that Mr Neish was merely making a statement – that he was merely noting – that the Chief Justice had not distanced himself from Mrs Schofield's e-mails is wilfully to close one's eyes to its clear import. We regard it as highly regrettable, to put it no higher, that the Chief Justice

should have chosen to do so in his evidence before us. This is the more so given that in answer to a question from the Tribunal (Sir Jonathan Parker) the Chief Justice had no hesitation in accepting that for him, in his capacity as Chief Justice, to have sent e-mails to the Bar Council in the terms of Mrs Schofield's e-mails would have been neither proper nor appropriate. In any event, we reject the Chief Justice's evidence in this respect as palpably untrue. His unquestioned professional ability is ample testament to his intelligence. We have no doubt that, as any intelligent person must have done, he fully appreciated the implications of the final paragraph of Mr Neish's letter.

- 5.30 We also find it extraordinary that, according to his evidence, on receipt of Mr Neish's letter he did not even consider making a statement of some kind to the effect that Mrs Schofield's views should not necessarily be taken as representing his own. If that were true, it would, to put it no higher, reveal a complete inability on his part to appreciate that, for better or worse, others might perceive his silence on this issue as identifying him – and hence his office as Chief Justice – with statements which, as he readily accepted, would have been improper and inappropriate had he made them himself. That said, however, we find his evidence in this respect to be untrue. We are satisfied, and we find, that at the material time the Chief Justice well knew that, in the perception of others, his silence would almost certainly be interpreted as support for Mrs Schofield's views – as indeed the Bar Council had clearly implied in the final paragraph of Mr Neish's letter – and that he was more than content that should be so.
- 5.31 In the afternoon of 3 April 2007 Mrs Schofield e-mailed Mr Neish asking for copies of his letter to the Attorney General and of the *Vox* article referred to in that letter. Later that day she e-mailed him again saying she had now seen a copy of the *Vox* article, asking him how much of it was attributed to her and suggesting that members of the Bar Council were responsible for disclosing the contents of her e-mails to *Vox*. She continued:

"The Chief Justice has nothing to do with the exercise of my rights. The attitude of the Bar Council that a wife reflects the views of her husband is very dated and laughable. It is so caveman and dinosaur-like that I would like to meet you all in a court of law on this. Perhaps most of you are unused to a wife who asserts her rights.

I wait to see what the AG can do about the exercise of my rights. I can assure you, none of this intimidates me. I am not the only one who shares the view that the Chief Minister wants to get rid of the CJ. That is where I come in, whether the Bar Council likes it or not.

I shall continue to express my views on matters of public interest."

5.32 On 4 April Mrs Schofield sent an e-mail to Mr Peter Schirmer of *Vox* newspaper, in which she referred to his having confirmed that they had not discussed her e-mail to the Bar Council, and that he had done his own research with members of the Bar. She went on to say that the stand taken by the Bar Council had been intended to intimidate her. It was unbelievable that they asserted that the views of a wife must be those of her husband. She was now releasing her e-mail as members of the Bar had been discussing it. She complained that her use of fundamental rights was being used against her husband. She continued:

"I want to assert that I have rights like every citizen in Gibraltar unless we are to accept we are living in a similar situation as Mugabe's Zimbabwe. I have also received a letter from the Bar Council stating that I am no ordinary member of the public. I say, I am under the law.

My e-mail raised serious matters of public interest and I consider the action by the Bar Council an attempt to harass me into silence. In a matter where I assert that the Chief Minister now clothed as Minister for Justice has been trying to get rid of the Chief Justice, the use of consultation and legislation to legitimise this long held ambition is an attempted rape of the Constitution and the Chief Justice. It is the most blatant example of interference with the judiciary and I as a human rights lawyer will not sit back as some fall into the Chief Minister's clever attempt. I also regard this attempt as affecting my rights and nobody other than those who side with the attempt can criticise me for speaking out. The issue of conflict of interest is relevant to this matter and

members of the council should not be afraid to declare such interests. I regard the exchange of communication taking place a matter of public interest. I would ask that you publish at the earliest possible opportunity this letter and clarification as to how you carried out your research.

I trust I am in a free country and I have freedom of expression. I would also ask that you carry out an opinion poll to establish how many as members of the Bar and the public regard the Gibraltar Bar as independent.

Please feel free to publish this e-mail together with the e-mail to the Bar Council."

In due course there was an article in *Vox* on 13 April which was headed: "Intimidation: Chief Justice's wife accuses Bar Council. Are we living in Mugabe's Zimbabwe?" It featured, with her permission, the text of the e-mail which Mrs Schofield had sent to Mr Schirmer on 4 April.

5.33 On 5 April Mrs Schofield sent a further e-mail to Mr Neish. After acknowledging his letter of 3 April, she went on:

"That a request for you to confirm information which I had received should call for this response from a Bar Council clearly shows I have hit a sacred cow. The Issue of conflict of interest in a matter which involves the Chief Minister and the Chief Justice and which as much we all pretend is about consultation would be obvious to a nursery school child is about constitutionality and independence of the judiciary will not be swept away by accusations of my interference. This is simply spin. I am unmoved by spin. It is a game for politicians, not lawyers.

The conduct of the Bar Council raises issues of conduct unbecoming of members of the Bar. The refusal for you as a close personal friend of the Chief Minister and members of your firm who do a lot of work for the Government to recuse yourself from voting and making decisions on this issue is conduct which brings the Bar into disrepute, undermines independence of the Bar and public confidence in the Bar.

I note that there is no Code of Conduct for the Bar in Gibraltar and I am seeking an Opinion as to whether the UK Code applies. If it does, please

take note that I shall be filing a complaint against the members of the Bar Council who approved the draft.

Your letter also raises issues as to how many members of the Bar Council approved this letter and the letter to the Chief Justice. Your letter gives the impression that all members supported the decisions to write. Can you confirm how many members voted for this action. You will agree with me that your letter would be misleading if only a few members agreed with the decision. Can you provide details of how many members of the Bar Council attended the meeting on Thursday? If most of the members did not approve, do you agree that your letter is misleading?

Please note that I shall not hesitate to seek a declaration that the assertions by the Bar Council on my right as a wife violate the European Convention and the Gibraltar Constitution. This would definitely answer the question as to what Gibraltar has come to. In the meantime, I can tell you what Gibraltar has come to. No Chief Justice in Gibraltar has been put through what the Chief Justice has been put through since the current Chief Minister came to power. Nor has the judiciary been under attack in the way that has taken place in the last 10 years. S6 of the draft bill is a direct attack on independence of the judiciary. No matter how much we spin or try to deflect the real issue behind this section.

I would welcome an opportunity in a court of law to lift the veil on the true intention of the section.

By the way I am a known campaigner for the independence of the judiciary in many parts of the Commonwealth and so far as I am concerned there are no sacred countries. You can rest assured if my husband was undermining independence of the judiciary, I would not hesitate to speak out.

I shall be placing your letter before my lawyers on Tuesday. I shall also be instructing my lawyers to ask the current Attorney General to recuse himself from dealing with any issues relating to me and my husband.

Please note that I am ready and willing to protect my rights and stop abuse of office as far as this matter is concerned."

- 5.34 The Chief Justice said in evidence that he was not aware of this e-mail at the time when it was sent, and claimed that he was first aware of it when the proceedings before this Tribunal started. In view of what he already knew about his wife's communications with the Bar Council, let alone the fact that she was purporting in the e-mail to act in his interest, and the fact that they used the same e-mail address, we reject this evidence. It is to be noted that she stated that she intended to invite the Attorney General to recuse himself from dealing with any issues relating to her or to the Chief Justice. In evidence the Chief Justice said that she had no authority to do such a thing, and that he had not discussed that possibility with her. Despite circumstantial evidence to the contrary, we accept the latter evidence.
- 5.35 We should add that on 25 April Mrs Schofield sent a fax to the ICJ Kenya, copied to the UK Foreign Secretary, the Governor, the Chief Minister and the Leader of the Opposition in Gibraltar. It set out four reasons why she said that the Chief Minister had an ambition to get rid of the Chief Justice, as follows:
 - "(i) In 1998 the Chief Justice at the Opening of the Legal Year announced that there had been attempts to interfere with the judiciary. In 1999 my husband told me that he had been informed by the then Governor Sir Richard Luce (now Lord Luce) that the Chief Minister (now Minister of Justice) had made representations for the Chief Justice's contract not to be renewed. It was suggested that he should renew for six months. My husband and I discussed the matter and he assured me that he had security of tenure until the age of 67. We have planned our lives on that basis.
 - (ii) You may recall that there was an issue relating to our maid. This was immediately followed by an incident involving the Chief Justice's private vehicle. The police intended to caution the CJ on driving without a licence and not having an MOT. The licence matter was dropped as that was a period of grace. The CJ was prosecuted over the MOT. He pleaded guilty. In England and Wales at that time a judge was not required to report an MOT offence. These incidents were used to call for the CJ's resignation. He is still in post.

- (iii) In July 2000 as the MOT matter was going on the Attorney General went to see English counsel who was advising my husband: and in his words the AG said "were this matter to be not contested and to be dealt with the minimum of fuss then no doubt the Chief Justice would receive all assistance and help were he should be seeking a judicial post in England and elsewhere"
- (iv) During the hearings of the MOT case the Lord Chief Justice of England and Wales called my husband and told him that he had met the AG in a conference and that he had mentioned the case. He went further to say to my husband that if he did not contest the case he would be considered for a judicial post in England."

She referred to section 6 and said that her husband either "accepted this set up or resigned" (see also paragraph 3.26). This letter was featured in an article in the *Gibraltar Chronicle* on the following day. An article in *Vox* of 27 April reported a phone call from Mr Kegoro of ICJ Kenya with his response. On 2 May he faxed and wrote to the Governor and alleged an infringement of the independence of the judiciary and a breach of the Constitution.

5.36 On 4 May the Chief Minister gave a media interview about the consultation process relating to the Bill, in the course of which he commented:

"I cannot say that the Chief Justice has limited the expression of his views to the consultation process and in the spirit of the consultation process. You know, otherwise we would not have had the unseemly rows that there have been in the Press, for what was supposed to be a consultation process at a time when all the parties consulted had the opportunity to express their views to the Government. Instead, things have found their way into the public domain through the most inappropriate of Press comments and that is deeply to be regretted."

The Chief Justice wrote a letter to the Chief Minister on 21 May, in which he challenged a number of statements which the Chief Minister had made in the interview. In regard to the passage which we quoted he said that these comments had given a wholly wrong impression. He emphasised

that he had mentioned contractual matters in his letter of 22 February in order to lay down a marker at the earliest possible moment, and that marker was still in place. He continued:

"Perhaps you would be kind enough to clarify why you say I did not respect the spirit of the consultation process. It cannot be that it is simply because I disagreed with clause 6 because all other consultees' submissions which I have seen disagree, in some degree, with the original draft. Furthermore, in this passage you seem to connect what you describe as "unseemly rows" in the press with my submissions. As yet I have made no public comment on the Bill."

In his oral evidence the Chief Minister defended the comments which he had made. They were much more gentle than the Chief Justice's conduct had warranted. He could have said that the Chief Justice had not respected the confidential nature of the consultation, and had passed information to his wife which only he could have known. He was not willing to pretend in public that he thought that the Chief Justice had behaved properly. The consultation paper had resulted in allegations in the press attributed to his wife that the Government were hounding him out of office. He must have known that the allegations being put into the press were sourced to him. He had undermined the consultation process. We regard the Chief Justice's statement in his letter to the Chief Minister that he had as yet made no public comment on the Judicial Service Bill as highly disingenuous. We are in no doubt that the "unseemly rows" which had occurred in the Press formed part of his intention in opposing the Bill.

Mrs Schofield's libel action

5.37 On 10 April Charles Gomez & Co wrote a letter before action on behalf of Mrs Schofield to the Bar Council, saying that the allegation of improper motivation on her part contained in Mr Neish's letter dated 3 April amounted to a serious defamation which was false and unjustified and which had caused Mrs Schofield considerable distress and

- embarrassment. The letter went on to ask for a list of all those to whom the allegations had been repeated, and to require an immediate undertaking on behalf of all Council members not to repeat them, failing which they were instructed to apply for an injunction without further notice.
- 5.38 The Bar Council met at 5.30 pm on 12 April and decided unanimously that no undertaking should be given, that a response should be sent to Charles Gomez & Co, and that if Mrs Schofield were to issue the threatened proceedings Mr Neish should ask the Chief Justice to recuse himself from hearing any matters involving members of the Bar Council or their firms, namely Triay Stagnetto & Neish, Triay & Triay, Hassans and Attias & Levy.
- 5.39 On the same date and following the meeting Mr Neish, on behalf of the Bar Council, responded by letter to Charles Gomez & Co denying that Mrs Schofield had any cause of action; maintaining that in any event the Bar Council had a duty to inform members of the Bar at the forthcoming AGM of the Gibraltar Bar Association (which was due to take place on 17 April) about what had transpired and to make available copies of all correspondence exchanged; and declining to give any undertaking. The letter concluded by stating that the Bar Council would resist strongly any proceedings which Mrs Schofield might institute.
- 5.40 On 13 April Mrs Schofield filed an action claiming damages and injunctive relief against Mr Neish as sole defendant, based on an alleged libel in his letter to her dated 3 April (2007 No 85). In the Particulars of Claim he was described as the Chairman of the Bar Council. In those particulars she alleged that the words "the ludicrous fiction" in the letter meant, and were understood to mean, that her communications with the Bar Council were "a dishonest and cynical manoeuvre and that she was motivated exclusively by a desire to protect her personal interests by inhibiting the Bar's freedom of expression and undermining its independence". She also stated in those Particulars:

"The Defendant circulated his letter not only to those Bar Council members who attended the meeting at which a response to the Claimant was agreed but also to those members who were not present at that meeting and had no part to play in the drafting or approval of that letter."

She alleged that there was a threat of repetition of the libel at the meeting of the Bar Council on 17 April, and sought an injunction to restrain repetition. According to Mr Gomez, Mr Neish was selected as the defendant as he was the only person who was known to have been involved in the drafting of the letter complained of.

- 5.41 Mrs Schofield's action against Mr Neish was reported in the *Gibraltar Chronicle* of 16 April under the headline: "Row turns sour as Chief Justice's wife serves writ on Bar Council Chairman". On 17 July she applied for various declarations, to which we refer later at paragraphs 6.17 6.18. The action has not yet been determined. We say nothing as to its merits.
- 5.42 The evidence of the Chief Justice was that he knew nothing of the libel proceedings until after they had been issued and that he could not recall any prior discussion with his wife of her intention to issue them. However, he agreed that it was possible that she told him that she had sent a letter before action although he could not specifically recall her having done so. When he was asked whether he should have dissociated himself from the libel action, he replied that this would not have been appropriate. It would not have been proper for him as Chief Justice to interfere with his wife's legal rights. It would have sent out a message of disapproval. As for his speaking privately to Mr Neish, the latter would know, he said, that he had nothing to do with the action. Mr Neish knew her as an independent minded person. We consider the Chief Justice's evidence to be highly implausible. We find that he was aware of Mrs Schofield's intention to issue libel proceedings, although we are willing to accept that he was not aware of the detailed nature of those proceedings until after they had been issued. His explanation for not having dissociated himself from her action was evasive.

The recusal application in April

- 5.43 It is now necessary for us to retrace our steps to the stage at which the Bar Council decided to seek the recusal of the Chief Justice (see paragraph 5.38 above)
- 5.44 In discussions between members over the weekend of 14/15 April the Bar Council confirmed that, notwithstanding that the libel proceedings had been issued against Mr Neish alone, its decision that Mr Neish should ask the Chief Justice to recuse himself from hearing any matters involving members of the Bar Council or their firms should stand. In his witness statement Mr Stephen Catania, a partner in Attias and Levy, described the rationale for this decision. He pointed out that by 12 April Mrs Schofield had threatened to issue proceedings against all members of the Bar Council. He continued:

"Her letter before action was addressed to all but then James Neish was singled out and proceedings were issued against him alone. We had agreed that we would take a joint stance, and my perception and I believe that of the others was that we saw legal proceedings issued by her as having been commenced with at least the approval of the Chief Justice. In the light of this, the members of the Bar Council agreed that we could not go on to appear in court before him and that we would ask for his recusal at the first opportunity."

The events of the morning of 16 April 2007

5.45 A case management conference in *Sonia Bossino v Attorney General* was listed to be held before the Chief Justice at 9.30 am on 16 April. This was the first case following the Bar Council resolution in which one of the signatory firms was involved. In that action Hassans acted for the claimant. Immediately before the hearing Mr Neish handed to the Chief Justice a letter, dated that day, which informed him of the decision of the Bar Council. According to the transcript, Mr Neish appeared on behalf of the Bar Council and, along with Ms G Guzman of Hassans, for the claimant (Mr Neish accepted in evidence that he had not been instructed

to appear for the claimant, but said that Ms Guzman had wanted someone from the Bar Council to make the application). He informed the Chief Justice that the members of the Bar Council would be asking him to recuse himself from hearing any matters involving members of the Bar Council or their respective firms. The application was based, he said, on the exchange of e-mails between Mrs Schofield and the Bar Council. The Chief Justice inquired how the e-mails affected him. Mr Neish responded that the e-mails did affect him, and that he was not prepared to argue the matter fully "because of a turn of events where there is obviously a clear breakdown in relations" between the Chief Justice and Mrs Schofield on the one hand and the Bar Council on the other (in evidence, Mr Neish explained that this referred to the series of e-mails from Mrs Schofield culminating in the issue of the libel proceedings).

5.46 After some discussion, the Chief Justice commented that he had only looked at the e-mails in question for the first time some 7 to 10 days previously, and that he had not seen them before, nor had he taken any interest in them. He pointed out that he had only recently returned from a holiday in Argentina with his daughter. The following exchanges then took place between the Chief Justice and Mr Neish:

"Chief Justice: [W]ould you in fairness read out the basis upon which Mrs Schofield wrote to you.

Mr Neish: Mrs Schofield says: "As a member of the public and a

person who may be affected by any constitutional and contractual – and contractual – implications of the Bill".

Chief Justice: Yes.

Mr Neish: Contractual, affecting yourself, My Lord.

Chief Justice: Yes, affecting herself.

Mr Neish: No, affecting you, My Lord. Contractually, your

contractual position as Chif Justice. So she is not speaking

as far as the constitutional aspect is concerned, she is speaking about your Lordship's own position. Members of the public, My Lord, were not consulted by the Chief Minister in his consultation paper. The consultees were the Bar Council ...

Chief Justice: Yes.

Mr Neish: ... your Lordship, the Puisne Judge and the Stipendiary

Magistrate. So in those circumstances, My Lord, members of the public do not know what is going on unless they are told by one of the consultees. And I understand that Your Lordship had prior access to the consultation paper before

Your Lordship left for Argentina.

Chief Justice: In fact I did not ... Oh yes I did, as did all members of the

Bar, because I circulated all members of the Bar.

Mr Neish: Yes, this is the ...

Chief Justice: Let us not go into that matter at this stage because what

you are doing, with greatest respect, is winding the situation up. What I am suggesting is that we need proper argument on this matter and I do not at this stage, on the basis of what you have said, intend to recuse myself. If members of particular firms wish to pass their briefs over to someone else in, say, undefended divorce matters, or if they wish to apply in individual cases for adjournments, that is a matter for them. But at this moment in time, if you are not prepared to argue the matter, let us then adjourn it to a day when you will come

fully armed and fully prepared."

5.47 The Chief Justice ultimately agreed to adjourn the recusal application to 2.30 pm on the following day, 17 April (see paragraph 5.81).

5.48 In his first witness statement Mr Gomez described the events which took place on the morning of 16 April 2007 as "... a means of causing the Chief Justice as much damage as possible for the purpose of forcing Mrs Schofield to withdraw her libel action". In evidence he said that the fact that Ms Bossino, as he understood, had never been consulted about the recusal application aggravated his concerns that its purpose was to bring pressure on Mrs Schofield to drop her libel action.

5.49 In evidence Mr Neish said that the recusal application was necessary because Mrs Schofield's e-mails had expressed views that were totally detrimental to members of the Bar Council, accusing them of being incapable of reaching honest and objective decisions, and they were expressions which, in their view, impacted on her husband's ability to sit in cases in which they were involved. It was "absurd" to suggest that the purpose of the resolution to make the application was to stifle Mrs Schofield's libel action and to cause damage to the Chief Justice himself. In cross-examination it was put to him that, had Mrs Schofield not commenced proceedings, the Bar Council would not have decided to make the application. He replied:

"It wasn't that we were asking the Chief Justice to recuse himself in retaliation for his wife issuing libel proceedings against the Bar. That is not the way we focused it. We focused it on the basis that here is the wife of the Chief Justice suing publicly, or threatening to sue in a public way, the Bar as a whole. We can't possibly appear before the Chief Justice now. That was our reason. Whether it was mistaken or not, that was our reasoning. What I must emphasise is that contrary to this proposition that is being bandied about that we asked the Chief Justice to recuse himself in retaliation for his wife's issuing of proceedings is simply not the case. ... It was the final straw in a sequence of actions."

5.50 We accept that evidence and find that, as Mr Neish put it under cross-examination, so far as the Bar Council was concerned the libel action was not the "causa belli"; it was "the last straw ... the last drop for the cup to overflow". Whether it was procedurally correct or not (and that is not a matter which need concern us), the recusal application was, we find, born out of a genuine fear on the part of the members of the Bar Council that in the circumstances, culminating as they did with the commencement of Mrs Schofield's libel action, they and their clients would not in the future receive a fair hearing from the Chief Justice.

The events of the afternoon of 16 April

5.51 Following the above hearing Mrs Schofield telephoned Mr Gomez to inform him of the proposed recusal application. Mr Gomez had not been

- present in court when the matter of recusal was raised, and was previously unaware that an application for this purpose was to be made. She instructed him to appear on her behalf at the adjourned hearing of the application. She also informed him that the editor of the *Gibraltar Chronicle* had been present in court that morning.
- 5.52 Some time during the morning or early afternoon the Chief Justice was contacted by two partners of Hassans, who invited him and Mrs Schofield to meet them. The Chief Justice passed on this request to Mrs Schofield, and they agreed that it would not be proper for the Chief Justice to attend such a meeting as Hassans were involved in the recusal application. Mrs Schofield met them later that afternoon. In her first witness statement she stated that they informed her that at the forthcoming AGM of the Gibraltar Bar, which was due to be held the following day, 17 April, a vote of no confidence in the Chief Justice was to be proposed. Mrs Schofield went on to say that she took the view that the purpose of the vote of no confidence was to put pressure on her because she had commenced the libel proceedings. Immediately following that meeting Mrs Schofield went to see Mr Gomez and instructed him to contact the Supreme Court Registry to seek an appointment before the Chief Justice "to inform him of the information".
- When Mrs Schofield arrived in Mr Gomez's office following her meeting with the partners of Hassans, she was, according to Mr Gomez, in an agitated, anxious state. Having been informed by her of the proposed motion of no confidence in the Chief Justice he concluded, to quote from his first witness statement, that Mr Neish and the other members of the Bar Council "were embarked on a manoeuvre to permanently damage the Chief Justice's standing as a reprisal for the defamation proceedings". Based on that perception of the purpose of the recusal application, he advised her that in his view the application was a serious and unprecedented abuse of the court's process. He also took the view that the motion of no confidence was likely to be passed. In his first witness statement, he said:

"Mrs Schofield instructed me to bring this situation to the attention of the Chief Justice for his information and guidance generally and thereafter to apply to Mr Justice Dudley for an injunction to restrain the tabling of the no confidence motion."

- 5.54 In accordance with Mrs Schofield's instructions, Mr Gomez telephoned Mr Mendez, the Deputy Registrar. According to a memorandum of Mr Mendez, which we accept as accurate, Mr Gomez telephoned him shortly after 4 pm and told him that he was with Mrs Schofield, and that they wished to appear before the Chief Justice urgently, without papers. Mr Mendez informed Mr Gomez that the Chief Justice had left his chambers for the day, but that Mr Justice Dudley was available. Mr Gomez responded that the matter concerned "Mrs Schofield's case" and that they did not wish to appear before Mr Justice Dudley as Mrs Schofield did not wish to cause him any difficulty or embarrassment (a reference to the fact that it was anticipated that Mr Justice Dudley would in due course be hearing Mrs Schofield's libel case). Mr Gomez urged Mr Mendez to telephone the Chief Justice at home "as is done in the case of urgent injunctions". Mr Mendez tried to contact the Chief Justice at home, but without success. When Mr Gomez was informed of this, he suggested that he and Mrs Schofield should wait, "in case the Chief Justice returns to chambers".
- 5.55 Not long after this the Chief Justice arrived in chambers, evidently as a result of a telephone call from his wife who informed him that he might be needed in chambers later that afternoon. It may be noted that the Chief Justice made no mention of this call in any of his witness statements. In his first witness statement he said:

"I was told in the afternoon that a very senior member of the Bar, Charles Gomez, wanted to see me urgently on a without papers matter and I think (although I am not sure) that I was told my wife was with him."

5.56 This is a misleading statement in so far as it suggests (a) that he was first told of the possibility that he might be needed in chambers by a third party (e.g. Mr Mendez), and (b) that he was unsure whether Mrs Schofield was involved.

When counsel to the Tribunal put to him that, by failing to mention his telephone conversation with Mrs Schofield, he had thereby failed to present a full picture of what occurred, he responded that he had forgotten about that conversation and had been reminded of it only recently. However, we cannot avoid the conclusion that the absence of any reference to this conversation was due, not to any lack of recollection on his part, but rather to his sensitivity to the criticism that his conduct on that afternoon was inappropriate and improper.

- 5.57 The Chief Justice gave a variety of accounts as to his understanding as to why he was required in chambers. In his first witness statement he said he thought that Mr Gomez wanting to see him "would be to do with the recusal application as I knew that I could not hear any application in the libel proceedings". In his third witness statement he said that he had "no idea" to what it related. In cross-examination he insisted that he did not ask Mrs Schofield for any details as to why he might be needed in chambers, but said that he assumed that it was something to do with her meeting earlier that day with the partners of Hassans.
- 5.58 In examination-in-chief, Mr Gomez described the purpose of his and Mrs Schofield's attendance on the Chief Justice as follows:

"The principal point, obviously, was to inform the Chief Justice of what was happening. We perceived that if notice was given of the proposed vote of no confidence, and that if it was made public, that would have a severe impact on the Chief Justice and, by extension, to our case, to the libel case. You can say – in fact I think it has been confirmed – that the entire matter was linked. In other words, the correspondence between Mrs Schofield and Mr Neish, the allegedly libellous statements made by Mr Neish, the issue of proceedings, and immediately thereafter the recusal application, and finally as part of ...what seemed to us to be a campaign linked to the libel action, a vote of no confidence at the Bar Council.

The first and dominant intention was to bring this to the Chief Justice's attention, to see whether he might have to take advice or steps on his own account. What I had at the back of my mind was that perhaps he would have referred the matter to the Attorney General, seeing as I considered

the matter to be an abuse of process, and a way of circumventing the judicial process, by the defendant himself. So that was the main concern. In other words, that the Chief Justice should be aware of what was happening.

Secondary to that was to obtain guidance generally as to how he considered we might best proceed in order to protect the integrity of the judicial system basically in so far as it affected the libel case."

Asked how he considered it appropriate to appear before the Chief Justice on behalf of his wife, Mr Gomez replied that he regarded it as "essential" that he should do so. Asked why, if it was at the back of his mind that the Chief Justice might refer the matter to the Attorney General (a possibility which he had not mentioned in his witness statements), he had not considered contacting the Attorney General himself, Mr Gomez replied that he had considered at the time that the appropriate course would be for the Chief Justice to do so.

- 5.59 It is at least apparent from the above evidence that Mr Gomez's purpose in seeking an appointment before the Chief Justice was to protect Mrs Schofield's position as claimant in her libel action, and to obtain the Chief Justice's guidance as to how that might best be achieved. Indeed, this is confirmed by the Chief Justice's own evidence, in his third witness statement, that Mr Gomez "wished to seek guidance as to how to protect [Mrs Schofield's] interests in relation to the libel action". We are satisfied that the urgency with which Mr Gomez and Mrs Schofield sought an appointment before the Chief Justice was dictated entirely by their understanding that at the AGM of the Bar which was due to take place the following day a vote of no confidence in the Chief Justice was to be proposed; and that the proposed vote of no confidence was the focus of the guidance which Mr Gomez was seeking from the Chief Justice.
- 5.60 How did the Chief Justice perceive the proposed attendance on him of Mr Gomez and Mrs Schofield? Asked by his counsel why he decided to receive them he replied:

"Because I knew that my wife had had this meeting with the partners [in Hassans]. I also knew that neither Mr Gomez nor my wife would seek that I should do anything improper, and I felt if I had to hear what they had to say in an informal context, it may be that I would have to convene the court later that day. So I felt it sensible to hear them."

In cross-examination he expanded somewhat on that evidence, saying:

"When I learned that there was an application by Mr Gomez which involved my wife, I had to consider: was it appropriate that I should see them in chambers or should I say no. Of course, it went through my mind, well, if my wife comes home and over a cup of tea relates events which require me to convene the court, then we have to go back after court hours to do it. So...,knowing full well that my wife and Mr Gomez would not expect me to do anything improper, and of course knowing full well that I would not myself do anything improper, I decided I would hear them in chambers."

Later in his cross-examination, the Chief Justice confirmed that it was "within [his] contemplation that there was a possibility of some sort of application". He continued: "I really did not know what it was all about, and I had them in and I sat there to hear them". Asked why he had not simply refused to hear them, the Chief Justice replied:

"It was quite clear to me that I could make no order. It was also quite clear to me that once he mentioned the recusal proceedings – that I was the person seised of the recusal proceedings – I wanted to know what information he was going to impart to me...I had been faced with something which not many judges have ever faced, and that is the Chairman of the Bar coming in and telling you that a number of members of the Bar are seeking my recusal because the Chairman of the Bar had been served with libel proceedings. If there was something that I ought to know, then I felt I ought to hear it."

5.61 Later during the Chief Justice's cross-examination the Tribunal (Sir Jonathan Parker) returned to this topic, and the following exchanges took place:

- "Q. I just wanted to ask you about your state of mind before the attendance of Mr Gomez and his wife before you in chambers. I understand your evidence to be that this was to be in some way a formal application, or formal appointment of some sort?
- A. Yes, there was a possibility that it was. Certainly something was going to be formally communicated to me.
- Q. Well, I think you said in answer to Mr Fitzgerald that you preferred it in effect to be on a formal basis because you felt that if you had to hear what she had to say in an informal context it may be that you would have to convene the court later that day.
- A. Yes. Clearly if there was something that had to be formally imparted to me which my wife and Mr Gomez considered ought to be done in a formal context, I ought to hear them in a formal context...rather than hear it at home and perhaps come to a decision that something ought to be done.
- Q. So you were receiving them into your chambers in your official capacity as a judge?
- A. Yes.
- Q. And I also understand from your evidence although correct me if I have it wrong that you were not at all clear exactly what application, if any, was to be made, or what would transpire when they arrived.
- A. Absolutely.
- Q. And you were also aware, as you said, of the judicial guidelines, and you had personal experience of this in Kenya and so forth.
- A. Yes.
- Q. My question is this: in those circumstances did it occur to you that it might be, to put it no higher, unwise for you to entertain any sort of application, address, whatever it may be, from Mr Gomez

- who was counsel for your wife, who was also present in your chambers at the time? Did that occur to you?
- A. I knew that my wife knew the constraints upon her appearing before me. So I felt that she must have considered those constraints before seeking to come before me.
- Q. Well, can you answer my question, please? Did it occur to you that it might be highly unwise for a judge to entertain this sort of proceeding?
- A. I considered that there were dangers, and that is why I asked the Registrar to come in.
- Q. But you thought it appropriate nevertheless to proceed with the hearing, if I can call it that?
- A. Yes, I did."
- 5.62 We are at a complete loss to understand how any judge, let alone a judge of the Chief Justice's intelligence and experience, could conceivably have thought it either appropriate or proper to entertain any kind of formal attendance on him by his spouse, who was currently a litigant in his own courts, and/or by her lawyer. Plainly it was both inappropriate and improper for him to have done so, as any judge with a degree of sensitivity to the requirements of his office must have appreciated. No judge should ever allow himself to be placed in the position in which the Chief Justice found himself on this occasion. Furthermore, we find it profoundly disturbing that, when giving evidence before this Tribunal, the Chief Justice still appeared unable to comprehend this. Nor does the fact that he was not clear as to the precise purpose of Mr Gomez's and Mrs Schofield's attendance on him make any difference: if anything, it makes matters worse.
- 5.63 There is a difference of recollection between the witnesses as to the length of the hearing, ranging from 5 minutes to about half an hour. We find, on the basis of Mr Mendez's note, that it lasted about 20-25 minutes. At the outset, the Chief Justice instructed Ms Desoiza, whom he had instructed to attend, not to take notes on the face of it, a most surprising instruction.

She was unsure why he gave that instruction. He gave her no explanation. She confirmed that if she was going to do the clerking for a hearing, she would normally take notes. She told us that, apart from this occasion, she could not recall any other occasion when she had been instructed by the Chief Justice not to take notes, or had been asked to attend a hearing in his chambers. In his first witness statement, the Chief Justice said that as his wife was present in his chambers he specifically asked the Registrar to attend "to ensure that nothing improper took place". However, if that was indeed his concern, one would have expected it to have led him to insist not merely that the Registrar attend but that she took a note of everything that was said. Furthermore, his statement suggests that had his wife not been present, he would not have thought it necessary to ask the Registrar to attend: a suggestion which we find extremely surprising. In cross-examination he said that his intention was that no note should be taken until it became clear precisely why Mr Gomez and Mrs Schofield had attended on him. He referred to the fact that no file had been provided and that no application notice had been lodged. When it was suggested to him that his very uncertainty as to what was about to transpire should have led him to require a note, the Chief Justice replied:

"In hindsight I wish a note had been taken. I did not anticipate that this would blow up as it did. Of course in hindsight I wish we had made a note."

In instructing the Registrar not to take a note of what was said the Chief Justice was, in our view, simply compounding his initial serious misjudgment in allowing Mr Gomez and Mrs Schofield to attend on him formally in his chambers. If it were true that the Chief Justice did not expect the matter to "blow up", then he would, once again, stand convicted out of his own mouth of a complete insensitivity to the requirements of his office. However, we are satisfied that that evidence is not true. The truth, as we find, is that at the time he gave no consideration to the possible consequences of his allowing Mr Gomez and Mrs Schofield to attend on him in his chambers; rather, his focus throughout was on the need (as he saw it) for Mrs Schofield's position as claimant in her libel action to be protected.

5.65 On 2 May the Chief Justice instructed Ms Desoiza to prepare a note of what had happened in chambers. She prepared a note that day from memory and with the aid of a post-it on which she had jotted 3 or 4 words shortly after the hearing. According to the note, which bore the heading of 'Schofield v Neish':

"Mr Gomez addresses the Chief Justice. States that he is here on an application in this case and also as an interested party in the case of *Bossino*. Informs the Chief Justice of events leading up to the information being acquired that vote of no confidence in the Chief Justice is to be proposed at Bar Council meeting to be held tomorrow and explains the impact this is to have on his client's case. Discusses the possibility of an injunction.

Chief Justice asks on what basis is the injunction sought and says that in any event such a matter should be referred to his brother Judge.

Mr Gomez explains the basis on which an injunction could be sought.

Chief Justice – discards the basis as need to write to Bar Council beforehand asking them to refrain from holding a vote of no confidence.

Application concluded - no order made."

5.66 In an attendance note made subsequently but undated, Mr Gomez recorded:

"Attended before the CJ with Mrs Schofield.

Explained that Mrs Schofield was concerned that JN [James Neish] and others may be about to air the issues the subject of the defamation claim before the Bar Association, and that this could prejudice her case.

Mrs S had been called to a meeting by Fabian Picardo and Lewis Baglietto. They had told her that they had nothing to do with the applications to have the CJ recuse himself from cases and that they were worried that there were moves now to ask the Bar Association for a vote of no confidence in CJ arising from the issue of the claim and the underlying issues. They also told her that they had been told that the Governor was going to London for advice on the issue.

I said that since most members of the Association were salaried employees of the big firms it was likely that such a vote would be carried without much difficulty or debate and this would imply to the public, (including potential jurors) that...in bringing the proceedings Mrs S had done something wrong.

I said that we were concerned to bring this to the CJ's attention since it affected him in his office and arose from the litigation but that any applications in the case which we might have to make would be to Mr Justice Dudley.

CJ said that he did not think that AJ [Mr Justice Dudley] should deal with the defamation case either and there was really nothing that he could do or add. He said that he would not make any comment."

5.67 In his Preliminary Response, the Chief Justice gave this account of what transpired:

"On the afternoon of 16th April he [i.e. the Chief Justice] was asked to hear a matter without papers and in private. He had no prior notice of the case or its title...It is not correct that this was an application in the libel proceedings brought by his wife, although it may be that her solicitor [i.e. Mr Gomez] was unclear; rather, it was a request for procedural guidance by his wife's solicitor in relation to the recusal application made that morning. Mrs Schofield was in possession of information that the Bar Council [sic] was to meet the next day to vote on a motion of no confidence in the Chief *Justice.* That motion, if passed, would have prejudged the issues to be determined in the recusal application and would therefore be a contempt of court. Mr Gomez on behalf of Mrs Schofield sought to make an application to the Puisne Judge for an injunction in the recusal proceedings, but since the recusal proceedings were reserved to the Chief Justice, Mr Gomez was not able to go before the Puisne Judge without an order from the Chief Justice. He therefore sought the Chief Justice's guidance as to the proper way to apply for injunctive relief... The Chief Justice viewed any application in the matter as premature and declined to take it further or to put the matter before *the Puisne Judge.*" (Emphasis supplied)

5.68 Mr Gomez stated in evidence that it was never his intention, nor that of Mrs Schofield, that the Chief Justice should make any order at this

hearing, but it is difficult to reconcile that statement with the Chief Justice's understanding (as set out in his Preliminary Response as quoted above) that Mr Gomez was seeking an order of some kind. When that understanding was put to Mr Gomez, he maintained that, to the extent that this passage was attempting to summarise his approach at the time, it was erroneous.

5.69 In paragraph 34 of his third witness statement the Chief Justice summarised what transpired in his chambers on this occasion as follows:

"...the "application" was in the context of matters as they progressed in the recusal proceedings to which she was not a party. As the applicants in the recusal application relied upon the fact that my wife had issued libel proceedings and on the subject matter of that litigation...he was concerned that a contempt might be committed in the recusal application and wanted to know the correct procedure for making an application in the recusal proceedings which were reserved to me. So the subject matter of the application concerned the libel action, but the proceedings were the recusal proceedings. I simply told him that any application was premature. I said that I could not deal with any application in the libel proceedings. I was not asked to make any order and I made none."

5.70 In cross-examination, the Chief Justice gave the following description of what occurred:

"They imparted to me information which I think they could have imparted in private. They chose to do it in the formal context of my chambers. But they could have done that privately. They thought clearly that this was more appropriate – for them to do it in a more formal context...I knew I could make no order, and it transpired that the only application made to me was that the matter should be referred to my brother judge. [Such an application] was mentioned. In fact what happened during the course of the proceedings [was this.] I said: Well what do you want me to do about all of this? I recall distinctly my wife leaning over to Mr Gomez and saying – and I heard her – if anything is done, it has got to be referred to the other judge. That was what Mr Gomez said to me...Mr Gomez did mention that there was the possibility of an injunction to prevent the Bar Council discussing this matter. I did

not [ask him to explain the basis for the injunction] because I would not issue any injunction. I did say to him: Well, have you written to the Bar Council? And he said: I haven't. And I said: Well, all this is premature. ...I did not enter into the merits. I simply said: Have you written the letter? I just asked him the question. I was not going to involve myself in any application for an injunction."

- 5.71 Regrettably, however, the Chief Justice undoubtedly did involve himself in a possible application for an injunction in so far as he engaged with Mr Gomez and addressed Mr Gomez's concerns by inquiring whether he had written to the Bar Council (plainly he had in mind that before applying for an injunction Mr Gomez should first ask the Bar Council to give an undertaking) and by expressing the view that, absent such a letter, any application for an injunction would be premature. There is no evidence that it occurred to the Chief Justice at any stage, as it should have done, (a) to inquire whether advance notice of this hearing before him had been given to the Bar Council, or to Mr Neish, or (b) to consider whether such notice should have been given.
- 5.72 As indicated in Mr Mendez's memorandum to Ms Desoiza, no application notice had at that stage been filed; nor were there any papers. In the circumstances, it is not surprising that Ms Desoiza found the situation extremely confusing. She was not clear what application was being made. She remembers that there was some discussion about the Bar Council. She also recalls that at one point Mrs Schofield's libel action was mentioned, and that the Chief Justice said that he could not hear anything to do with that. In the evidence before us the accuracy of Ms Desoiza's note was not substantially challenged, save that the evidence of the Chief Justice, and of Mr Gomez, was that the Chief Justice had said that he did not think that it would be appropriate for the injunction to be heard by Mr Justice Dudley. In his witness statement Mr Gomez said that, in view of the Chief Justice's comment, Mrs Schofield decided not to apply for an injunction.
- 5.73 Plainly there was a high degree of confusion not only on Ms Desoiza's part but, more significantly, also on the part of both Mr Gomez and the

Chief Justice as to what application (if any) Mr Gomez was seeking to make, and in what proceedings. However, the following facts at least are, as we find, clearly established:

- Mr Gomez appeared before the Chief Justice as the lawyer acting for Mrs Schofield in her libel action;
- Mrs Schofield was present throughout;
- the Chief Justice was there in his official capacity as a judge, and in a formal context (equivalent to a "without papers" application);
- from the outset, the Chief Justice was conscious that Mr Gomez might ask him to make an order of some kind – hence the formal nature of their attendance on him;
- Mr Gomez explained to the Chief Justice that he was concerned to protect his client's position as claimant in the libel action in the face of the perceived threat of a vote of no confidence being passed at the forthcoming AGM of the Bar;
- Mr Gomez had in mind throughout, as a possible means of meeting that concern, that Mrs Schofield should seek an injunction against the Bar Council restraining it from proposing a vote of no confidence, and he sought the Chief Justice's guidance in that respect;
- the Chief Justice engaged with Mr Gomez (and hence with his client Mrs Schofield) at least to the extent that he expressed the view that an application for such an injunction would be premature since the Bar Council had not been invited to undertake not to propose such a vote: and he thereby gave the guidance which Mr Gomez sought.
- 5.74 In the light of the above facts, we regard the Chief Justice's conduct on the afternoon of 16 April as demonstrating a reckless disregard for the requirements and reputation of the office of Chief Justice. In the first place, he should never have allowed Mr Gomez and Mrs Schofield to attend on him formally; the more so since (as we find) he well knew that their request to attend on him was directly connected with Mrs Schofield's

position as claimant in her libel action. In the second place, having mistakenly allowed them to attend on him, he should have taken the first opportunity to bring the proceedings to an end. Instead, he went so far as to offer guidance to Mrs Schofield (via her lawyer Mr Gomez) in relation to the possibility of an application by Mrs Schofield for an injunction against the Bar Council in order to protect her position as claimant in her libel action against its Chairman. All in all, the Chief Justice's conduct on the afternoon of 16 April 2007 constitutes, in our view, judicial misconduct of the most serious kind.

- 5.75 The following morning Mr Mendez faxed Mr Gomez requesting an application notice and fee for the "without papers" application. According to Mr Mendez, the Chief Justice had remarked that he saw no need for an application notice, since no formal application had been made, but he was informed this was in accordance with normal procedures, to which he did not demur.
- 5.76 Later that day Gomez & Co sent Mr Mendez a cheque for the court fee, which Mr Mendez returned as it was without any papers. After further requests on 19 April Gomez & Co lodged an application notice relating to the hearing on the afternoon of 16 April, signed by Mr Gomez, and headed in the matter of Mrs Schofield's libel action. It gave notice of intention to:
 - "... inform the court that arising from the issue of the proceedings herein the Defendant and others propose a vote of no confidence against the Honourable Chief Justice at the forthcoming Annual General Meeting of the Gibraltar Bar Association."
- 5.77 On 25 April Triay & Triay wrote to Mr Gomez informing him that they had been instructed to act for Mr Neish in Mrs Schofield's libel action, and saying that they had advised Mr Neish that the action had no prospect of success. In the final paragraph of their letter, they said this:

"We note that an ex parte application was made to the Chief Justice on April 16th, although from the application notice in our hands it would appear that no relief was sought, which seems an extraordinary state of

affairs. Could you please explain what the precise purpose of the application was, and why this was conducted in the absence of the Defendant? Please also provide us, by return, with a typescript of counsel's note of that hearing which we shall await as a matter of urgency."

- 5.78 In response, Mr Gomez sent Triay & Triay a copy of the attendance note which we set out in paragraph 5.66. Thereafter on 2 May the Chief Justice requested Ms Desoiza to make a note of the hearing on the afternoon of 16 April (see paragraph 5.65). In cross-examination he said that he made this request because he was aware that Triay & Triay were making inquiries, and he considered that it was appropriate that Ms Desoiza should record her recollection of what had occurred on that occasion, "lest there be any mischief about it". He went on to say that he feared that "something more serious could be made of it", referring to the fact that his wife's counsel had appeared before him ex parte. He maintained that he had had nothing to do with the drafting of Ms Desoiza'a note, and said that he had not asked to see it once it had been prepared. Ms Desoiza took the heading of her note from the application notice. When the Chief Justice saw that the note was so headed he contacted her again, sounding (according to her) "horrified", and told her that he had thought that the hearing was in the matter of *Bossino v. Attorney General*.
- 5.79 On 3 May Triay & Triay, acting for Mr Neish in Mrs Schofield's libel action, wrote to Mr Mendez asking for a photocopy of the judge's note of the ex parte application which they understood had been made on the afternoon of 16 April. On 4 May Mr Mendez replied saying: "The Chief Justice states that no notes were taken. He further states that the matter which was brought before him was not *Schofield v. Neish* but the *Bossino* recusal application". This was in accordance with a manuscript note in the hand of the Chief Justice. He said in evidence that he understood that the request related to notes taken by him. We have no hesitation in rejecting that evidence as no more than an afterthought by the Chief Justice, in an attempt to explain what was as he no doubt fully appreciated by the time he gave evidence before us a highly misleading statement. With some hesitation we are prepared to accept that in stating

that no notes were taken the Chief Justice was not seeking deliberately to conceal what had transpired in his chambers on the afternoon of 16 April, but the very fact that he was content to make that statement betrays, to put it no higher, a highly cavalier attitude to what had occurred on that occasion. On 9 May Triay & Triay wrote again to Mr Mendez, expressing surprise that the Chief Justice had taken no notes of the hearing on 16 April and asking for a copy of the note taken by the court clerk in attendance at the hearing. A copy of Ms Desoiza's note was then supplied to them.

5.80 On 17 May Mr Gomez wrote to Mr Mendez saying that, as it had never been intended that the Chief Justice should be involved in the libel action, the filing of the application notice was an error and potentially misleading. He said that it should be amended to bear the title of the *Bossino* case. On 22 May Ms Desoiza responded to Mr Gomez's letter dated saying that she had no power to do what he asked, and suggesting that he might wish to apply to a judge for that purpose. It appears that no such application was made.

The adjourned hearing on 17 April

5.81 On the morning of 17 April Mr Gomez e-mailed Mr Neish concerning the proposed motion of no confidence in the Chief Justice (described by him as "a one-sided affair") in which members of the Bar Association would be expected to pass judgment on matters which were sub judice (a reference to Mrs Schofield's libel action). He expressed the hope that they would have the intelligence not to vote in favour of the motion. He continued:

"I have made my misgivings known to the Chief Justice since he is dealing with your application that he recuse himself from the unrelated case of *Bossino v. A-G* and your application in this case is [premised] on the claim which Mrs Schofield has issued against you. I have asked permission to address the Chief Justice at Friday's adjourned hearing of the *Bossino* case since his decision there could affect my client's rights.

Meanwhile, I repeat my request for your undertaking that you will not allow any discussion at today's Association meeting on our case. I believe that such discussion could not only aggravate the defamations complained of but also constitute a breach of the sub judice rules and might lead to contempt of court."

- 5.82 In the event, no motion of no confidence in the Chief Justice was tabled at the AGM of the Bar held that day. Instead, a vote of confidence in the Bar Council was passed unanimously.
- 5.83 At the outset of the adjourned hearing of the *Bossino* case, Mr Neish informed the Chief Justice that the four firms intended to make a general application for recusal. The Chief Justice observed that Mr Gomez was also in court, having asked to be present for "certain limited reasons". At that point, Mr Gomez addressed the court, as follows:

"My concern arises from a claim which my client, Mrs Schofield, has issued against my learned friend...And the concern is as to how the issues in that case are apparently already being canvassed in Mrs Bossino's case...And clearly in my view that is an abuse of the process of the Court and I mean to stop that as soon as possible."

- 5.84 The Chief Justice asked Mr Gomez for the reference to Mrs Schofield's case. Mr Gomez gave the reference, explaining (quite unnecessarily, since the Chief Justice was well aware of it) that the case was "a defamation case".
- 5.85 The Chief Justice went on to say that, having considered the matter overnight, he had come to the conclusion that the matter required the assistance of an amicus curiae, and that he proposed to instruct London counsel to act in that capacity. He accordingly adjourned the recusal application until 24 April.

The adjourned hearing of the recusal application on 24 April

5.86 On 18 April the Chief Justice attended a conference with Mr James
Dingemans QC in London concerning his possible appointment as amicus
curiae in the recusal application. Mr Dingemans recalls that there was
some discussion about the duties of an amicus curiae, and how a judge

facing a recusal application could adduce evidence relevant to the application. This step was taken by the Chief Justice entirely on his own initiative, without, at that stage, any involvement by or on behalf of the Attorney General. In cross-examination he gave this explanation:

"To my recollection, I had no experience of appointing an *amicus* before. So I did not think it improper for me to identify someone myself when I saw Mr Dingemans. I had a meeting to explain the situation to him. He said: Well, I've got to be instructed through – he may have referred to the Attorney General. It may have been through the Registrar. But he told me he had to be appointed in that way."

- 5.87 At the adjourned hearing of the recusal application on 24 April Mr Neish appeared for the Bar Council, Mr D Dumas QC for the claimant, Sonia Bossino, Mr Yeats for the Attorney General, along with representatives of the four firms involved in the recusal application, including Mr Stephen Catania, a member of the Bar Council and a partner in the firm of Attias & Levy.
- Also present was Mr Gomez who presented a written skeleton argument in support of Mrs Schofield being given permission to be represented at the hearing, on the basis that the arguments of the applicant firms "arise from the very issues which are the subject of her claim against Mr Neish". The skeleton argument maintained that "it would be unfair for issues which are the subject matter of the defamation case to be argued by or on behalf of the Defendant in that case without Mrs Schofield being given the opportunity to contest those arguments, to the extent that they affect her personally in general and more particularly as claimant in that case". It was also maintained that the court would be assisted in its determination of the request for recusal if it heard Mrs Schofield's version of events, in connection with which she wished to clarify certain matters, which were specified.
- 5.89 At the outset of the adjourned hearing, Mr Neish informed the Chief Justice that "following developments in the course of the last week" (referring to the presentation of the Signatories' memorandum) it was no longer intended to proceed with a general application for recusal. He

said that in the circumstances he was content to leave it to the Chief Justice to consider whether it was appropriate for him to recuse himself from hearing any matter in which the Signatories' firms were involved. Mr Dumas was granted a short adjournment to take instructions as to whether to apply on his client's behalf for the Chief Justice to recuse himself. Meanwhile the Chief Justice granted the application of Mr Gomez to remain in court to protect the interests of Mrs Schofield while Mr Dumas made his submissions. The transcript records these exchanges:

"Mr Gomez: My Lord, may I ask to remain in this application, because

as you know I represent Mrs Schofield, who is the claimant in the case against Mr Neish, and I'm very concerned that all this flurry of applications is actually meant to put pressure on Mrs Schofield in those proceedings, and it is for that reason that, if you don't mind, My Lord, I'd like to

stay.

Judge: Oh, absolutely. In fact it is open Court and if Mr Dumas

does slip and say something which you wish to take exception to then I'm sure Mr Dumas would have no objection to at least you making representations. But I don't think we are going down that route, are we?

Mr Dumas: Well, My Lord, my friend has made a scandalous

allegation that somehow we're seeking to put pressure on

his client.

Mr Gomez: We see it another way, My Lord".

5.90 In his third witness statement the Chief Justice observed in regard to this hearing:

"At that stage no objection was made about Mr Gomez appearing. I did not address nor was I asked to address the implications of Mr Gomez' appearance. I was facing a situation which, as a judge of many years' experience, I have not faced before and which I venture to suggest not many judges have faced. Quite frankly my focus was not upon Mr Gomez or his client; it was to preserve sanity in the difficult and unfolding situation".

5.91 In his fourth witness statement the Chief Justice observed in regard to this hearing:

"Mr Gomez was present but he did not make any formal application to be joined to proceedings and I did not rule on any such application. He was not therefore an interested party and was merely given permission to stay at the hearing which was in open court. He did not make any applications on behalf of his client".

This may be contrasted with his acknowledgement in his first witness statement that his wife was seeking the court's protection (see paragraph 5.92 below). In his evidence the Chief Justice accepted that he was not "particularly comfortable" at Mr Gomez being present on behalf of Mrs Schofield.

- 5.92 When the hearing resumed, Mr Dumas informed the Chief Justice that he had not been able to get instructions to make a recusal application. He had had instructions to hold the claimant's case in abeyance pending Mr Neish's general recusal application. The Chief Justice responded by informing Mr Dumas that, on the basis that the general application would be made, he had identified an amicus curiae, namely Mr Dingemans QC, who was set to come to Gibraltar for the hearing of the application. The Chief Justice went on to say that he could understand Mr Neish personally asking him to recuse himself in matters affecting him and his client. He would give instructions to the Registry accordingly. In the absence of a recusal application by Mr Dumas, he would release Mr Dingemans, and there was no need for Mr Gomez to come in as an interested party.
- 5.92 In his first witness statement the Chief Justice observed in regard to this hearing:

"If my wife's actions are raised in proceedings before me, she is entitled to the same protection as any other litigant...No objection was taken by any party to my wife's counsel's intervention".

- This statement seems to us to betray a complete lack of understanding on the part of the Chief Justice of the perception which would inevitably be created if he were to be the provider of such protection.
- 5.93 In treating Mrs Schofield as an interested party in a prospective application that he recuse himself based upon the correspondence which had passed between her and the Bar Council and which had culminated in the issue of her libel action, the Chief Justice was, to put it no higher, giving a degree of judicial credence to Mr Gomez's submission that the purpose of the recusal application was to stifle the libel action (see in particular the Chief Justice's answer "Oh absolutely" in the exchanges set out in paragraph 5.89 above). This he should not have done. Rather, he should have recognised that the continuing involvement of Mrs Schofield in the recusal application was yet another reason for his recusal, and his failure to recuse himself can only have inflicted further damage on his office as Chief Justice. Fortunately, however, since in the event the recusal application was not pursued no further harm was done in that respect.

The recusal application in May

5.94 On 22 May the Chief Justice heard an application for the claimant company in *Gibraltar Commercial Property Co v. Attorney General (suing on behalf of Selina Ltd)* that he should recuse himself from hearing the trial in that case. Mr Stephen Catania acted for the claimant company, which was wholly owned by the Government. Mr Dumas acted for Selina Ltd. On the afternoon of the previous day Mr Catania had telephoned the Supreme Court to ask if counsel could see the Chief Justice in chambers to explain, as a matter of courtesy, that their respective clients had instructed them to make an application that he recuse himself from hearing the trial of the action. This request was referred to the Chief Justice, who wished to know the reason for it. In response Mr Catania supplied skeleton submissions.

- 5.95 In these submissions Mr Catania maintained that the surrounding circumstances were such as to give rise to an objective appearance of bias on the part of the Chief Justice against his clients. The circumstances to which he referred were the various public statements made by Mrs Schofield and the public comment to which they had given rise, from which the Chief Justice had chosen not to dissociate himself. Mr Catania referred in particular to the articles in *Vox* on 2 March (see paragraph 5.20) and 13 April (see paragraph 5.40), and to the article in the Gibraltar Chronicle on 26 April (see paragraph 5.42). Applying the test of bias formulated by Lord Bingham of Cornhill in Magill v. Porter [2001] UKHL 67 at 103, Mr Catania submitted that in the circumstances a fair-minded and informed observer would conclude that there was animosity on the part of the Chief Justice towards the Government, which the Chief Justice perceived as attempting to deprive him of his livelihood, and hence a real possibility of bias on the part of the Chief Justice against the Government and companies associated with it: i.e. against his clients.
- 5.96 At the hearing Mr Catania, on behalf of the claimant company and the Government, informed the Chief Justice that his clients had instructed him to request the Chief Justice to recuse himself from hearing the case, on the basis of "certain public statements by your wife which Your Lordship had not dissociated yourself from" and which "give the appearance of bias". He pointed out that, in contrast to the previous recusal application, this was an application by a client. Mr Dumas adopted a neutral stance.

5.97 The Chief Justice observed:

"If you followed the previous recusal applications, you will have known that I was seeking the assistance of an *amicus*. That situation cannot have changed on this application...And secondly you will, if you were following the previous aborted recusal proceedings, have realised there was an interested party...That means, does it not, that I will have to put this matter out for the *amicus* to be briefed again, and I think it would be proper, would it not, for you to consider notifying the interested party. Not only that, you had not put in an application notice until very late yesterday, and that was because I sought one. I have not seen attached to that application notice any affidavit or statement from your client".

In these circumstances the Chief Justice decided that the application should be adjourned, and that in the meantime he should not proceed with the trial. Before doing so, he insisted, contrary to the submission of Mr Catania, that there was an interested party, namely his wife. He said:

"I think at least the interested party who came into the last recusal application may have a view on that, and may wish to come in...It may well be that the interested party in the other application has a different view and they have a right to be heard...if they wish to come into the proceedings".

Mr Catania submitted that it would be improper if his wife were to be represented. He referred to the Guide to Judicial Conduct of England and Wales, paragraph 7.2.1 as a further ground for the Chief Justice recusing himself. The Chief Justice observed that the only person who could deal with the recusal application was the judge who was seized of the case.

5.98 We see force in the point made by Mr Catania, in the course of his cross-examination by counsel for the Chief Justice, that there was no need for an *amicus* to assist the court in relation to the *Selina* recusal application. As Mr Catania put it:

"The Chief Justice tries cases which have very grievous consequences for a lot of people and he does that without an *amicus*. He hears the parties and he makes decisions. He is a professional judge of many, many years' experience. I would have addressed him as to the law and he could have made up his own mind: simple as that."

- 5.99 In examination-in-chief, the Chief Justice accepted that the *Selina* recusal application was not his "finest hour in court". Indeed it was not. As he now appreciates, there was no conceivable ground upon which Mrs Schofield could sensibly have been treated as an interested party in the *Selina* recusal application. His failure to appreciate this is yet another example of his inability to comprehend the constraints and responsibilities of his office, in particular where Mrs Schofield is concerned.
- 5.100 Thereafter the Government intimated an intention to make a recusal application in the case of *Baldechino v. Attorney General* on the same

evidence as in the *Selina* case. In connection with these applications the Chief Justice issued a statement of facts. We note that in this statement, unlike his statement in his proceedings for judicial review (see paragraph 6.19 et seq), the only reference he made to any attempt on the part of the Chief Minister to have him removed from office was to his alleged representations in 1998 - 1999.

- 5.101 On 7 September Mr Dingemans provided a skeleton argument as amicus curiae in the recusal applications in the *Selina* and *Baldechino* cases. He stressed that it was for the court to determine the applications in the light of all the relevant evidence. He drew a distinction between an appearance of bias against the Chief Minister and an appearance of bias against the Government, in that whilst a fair-minded and informed observer would be likely to conclude that, in any case in which the Chief Minister was to be a witness there would be an appearance of bias if the Chief Justice were to act as the judge, it did not follow that there would be an appearance of bias on the part of the Chief Justice in cases involving the Government of Gibraltar in which the Chief Minister did not appear as a witness and in which no policy decision of the Chief Minister was in issue. The answer to the recusal applications, he argued, depended on factual material relating to the involvement of the Chief Minister in the two actions.
- 5.102 The substantive hearing of this recusal application was fixed to be heard in September 2007, with Mr Dingemans briefed to appear as amicus. In the event it did not proceed since it was overtaken by the suspension of the Chief Justice.

CHAPTER 6

AUGUST - DECEMBER 2007

This chapter is concerned with:

- the cancellation of the ceremony for the Opening of the Legal Year 2007-2008 (paragraphs 6.1-6.8)
- Mrs Schofield's complaint against Freddie Vasquez QC (paragraphs 6.9 6.14)
- the Chief Justice's application for judicial review (paragraphs 6.15 6. 16)
- comparison with orders sought by Mrs Schofield in her libel action (paragraphs 6.17 –
 6.18)
- the Chief Justice's statement in support of the application (paragraphs 6.19 6.36).

The dates in this chapter of the Report refer to 2007 unless otherwise stated.

The cancellation of the Opening of the legal year 2007-2008

6.1 On 11 July Mr Mendez, then Acting Registrar, circulated the following curt message to all chambers in Gibraltar:

"I am directed to inform you that the Chief Justice will no longer be holding a Ceremonial Opening of the Legal Year, this with immediate effect."

This was immediately reported in the press. In the *Gibraltar Chronicle* there was an article headed "Gloom on legal profession as Schofield calls off legal year ceremony". In his witness statement the Attorney General said that the Chief Justice's decision was deeply resented by the Bar, and that the reason for it never became clear. Mr Catania's evidence was that in taking this decision the Chief Justice was "chastising the Bar for confronting him".

6.2 However, two days later the Government issued a press release which stated that, whilst the ceremony was a matter for the courts and the legal

profession, it was also an established part of Gibraltar's history and heritage. It announced that the ceremony would be conducted by Sir Murray Stuart-Smith, in his capacity as President of the Courts of Gibraltar, and would take place on 24 September. In an article which appeared on the same day in the *Gibraltar Chronicle* the Chief Justice was reported as saying that he had "undertaken to give a full explanation of why he has called off the ceremonial opening of the legal year". The ceremony took place on 24 September, conducted by Sir Murray Stuart-Smith.

In his witness statement Mr Mendez stated that at the time the Chief Justice explained his decision by saying that as he was no longer the head of the judiciary it would not be right for him to preside at the ceremony. He also recalled that the Chief Justice asked him for how many years the ceremony had been held, and that when he replied that it had been held for about 60 years the Chief Justice commented that that was not a very long time. In the circumstances, it is hardly surprising that Mr Mendez understood, as he did, that the Chief Justice was intending to discontinue the ceremony not only for the year 2007 but indefinitely. As Mr Mendez put it in cross-examination:

"I don't think he said "this year". [Had he done so] I would have understood it and I would have written that circular in very different terms. As I say, the fact that he was saying: I am not the head of the judiciary, I cannot do it; it was implicit in that that he would never be able to do it, as long as the Judicial Service Act is in place".

6.4 The Chief Justice explained his decision in his third witness statement as follows:

"The opening of the Legal Year is a Supreme Court function...Under the Constitution the President of the Court is not entitled to sit in the Supreme Court and the [Judicial Service Act] had no effect on this position. We now had a Head of the Judiciary who could not preside over the Opening of the Supreme Court Legal Year. I was observing the situation and was still contemplating whether I should issue judicial review against what I considered to be unconstitutional provisions. I was

concerned about the effect that the Opening of the Legal Year ceremony could have on the jurisdiction in the light of events which had occurred during the course of 2007. I was concerned that the event could be used to undermine the office of Chief Justice and I had in mind the boycott of the ceremony by senior members of the Bar in 2000. There was a danger that addresses would be made which would undermine public confidence in the judiciary."

He went on to say:

"I therefore came to the conclusion that it would be best to cancel the ceremony for that year and instructed Clive Mendez to circulate the members of the Bar accordingly. To my dismay, he circulated as if it was a decision for all time. Once there was public criticism I felt that a correction of the error would only fuel the situation and in any event I did not want to embarrass Mr Mendez. In truth I should have checked the circular before it went out and take responsibility for the mistake."

- 6.5 In his fifth witness statement he accepted that he had not discussed the matter in advance with other members of the judiciary. He accepted that he would be open to criticism for not attempting to contact the President of the Courts. He also accepted that he did not handle the situation in relation to the Opening of the Legal Year as well as he should have done. As he put it in examination-in-chief, this episode was "not [his] finest administrative action". In cross-examination the Chief Justice said that, in the light of the announcement that Sir Murray Stuart-Smith would conduct the ceremony, he felt that any statement from him at that stage would have been "totally inappropriate".
- 6.6 It should be noted that although the Chief Justice referred a number of times in the course of his evidence to his no longer being "Head of the Judiciary", no office bearing that title existed in any formal sense prior to the enactment of the Judicial Service Act, nor was such an office created by that Act.

- 6.7 We accept Mr Mendez's recollection of this episode as accurate. We find that the Chief Justice gave Mr Mendez the clear impression that his intention was that the practice of holding a ceremony to mark the opening of the legal year should be discontinued until such time as the office of President of the Courts of Gibraltar should be held by the Chief Justice; that is to say, for so long as section 6 of the Judicial Service Act remained in force unamended. We do not accept the Chief Justice's evidence that his intention at the time was that the ceremony to mark the Opening of the Legal Year should be suspended for the year 2007 only. Such an intention would have been inconsistent with the rationale which he gave for his decision, namely that by virtue of section 6 of the Act the President of the Court of Appeal, and not the Chief Justice, held the office of President of the Courts of Gibraltar. We find that the Chief Justice's intention was that the ceremony should be discontinued for so long as that situation continued; and that, far from being "dismayed" when it became apparent that that was how his decision was perceived by the Government, the Bar and the public, he was content that that should be so. It is our clear impression that it was for that reason, and not by reason of any embarrassment that might be caused to Mr Mendez or by reason of the fact that the ceremony would be conducted by Sir Murray Stuart-Smith, that the Chief Justice did not issue any corrective statement. As for his expressed concern that the ceremony might be the occasion for addresses "undermining public confidence in the judiciary", his real concern was that he should not afford those whom he perceived as hostile to him an opportunity to criticise his conduct or to challenge his views.
- 6.8 This episode illustrates the extent to which the Chief Justice's motives and actions were governed and directed by his perception that the enactment of section 6 was (in Mrs Schofield's words) an attempt by the Chief Minister to hound him out of office; or, as he put it in his statement in the judicial review proceedings (which he signed about a month later) "to diminish my powers as Chief Justice because I would not leave". His decision to cancel the ceremonial Opening of the Legal Year was yet another deliberately provocative act which only served to exacerbate existing tensions.

Mrs Schofield's complaint against Freddie VasquezQC

- On 21 August Mrs Schofield made written complaints against Freddie Vasquez QC to the Registrar of the Supreme Court. The complaints were: first, that he was in breach of the Code of Conduct for Barristers "by making public submissions through the press on issues which are the subject of proceedings before the court in a matter where he represents Mr Neish", ie her libel action; and, secondly, that he had misled the public and sought to undermine the Chief Justice in the eyes of the public.
- 6.10 This arose out of a letter from Mr Vasquez which was published in the Gibraltar Chronicle of 14 August. The letter responded to a letter from Mr Gomez in the same newspaper, in which he had asserted that the Judicial Service Act, and specifically the designation of the President of the Court of Appeal for Gibraltar as head of the Gibraltar judiciary, amounted to an unconstitutional interference with the independence of the judiciary. Mr Vasquez stated in his letter that by convention, and not by any specific constitutional provision, the Chief Justice had traditionally acted as the head of the judiciary in Gibraltar, in a position comparable to that of the Lord Chief Justice in England. However, while the Lord Chief Justice was the senior and most eminent judge in England and Wales, the Chief Justice in Gibraltar was the equivalent only of a High Court judge in England. He went on to point out that in fact the present Chief Justice also regularly heard cases in England as a Recorder, a post more junior than a High Court judge. Modernising changes, he said, had been made in the new Constitution and by the Judicial Service Act. He drew attention to the fact that the President of the Gibraltar Court of Appeal, like his brother judges, was a recent member of the Court of Appeal in England and as such would always be a far more experienced and eminent judge than any Chief Justice of Gibraltar. He pointed out that Mr Gomez did not explain how the alteration of the hierarchy of Gibraltar's independent judiciary impeded or interfered with the ability of any judge in Gibraltar to judge any case, let alone how the Act could possibly amount to an unconstitutional interference with the independence of the judiciary which he so readily asserted.

- Committee. After protracted delay the complaints were heard on 3 April 2008 in the absence of Mrs Schofield who had complained that the members of the Committee, including the Attorney General and the Speaker of the Parliament, were biased against her. The Committee dismissed both complaints on 15 April. In their written decision they observed that they had some concerns as to whether the complaints had been made in good faith. They found it difficult to see what possible harm the complainant could have felt herself to have suffered as a result of Mr Vasquez's letter. They also were surprised that the complainant should have thought fit to complain that Mr Vasquez had contributed to correspondence which her own lawyer had commenced and continued.
- 6.12 In her second witness statement Mrs Schofield said the decision to make a complaint against Mr Vasquez was hers and was in good faith. She claimed that the Attorney General, who chaired the Committee, had a conflict of interest. He had been "mentioned in the applications before the court".
- 6.13 In his third witness statement the Chief Justice accepted that he might have seen the complaints during the drafting process. Mr Vasquez went further and maintained that the Chief Justice had tacitly endorsed the complaint in regard to his office in that he had said it was well made.
- 6.14 We note that in his second witness statement the Chief Justice stated that he took independent exception to Mr Vasquez's letter as it publicly sought to undermine the status of the office of the Chief Justice. This seems to us to be an over-reaction to the letter, and yet another example of a tendency on the part of the Chief Justice to overreact to what he perceived, but no objective observer would have perceived, as slights against the dignity and status of his office.

The Chief Justice's proceedings for judicial review

The application

- 6.15 On 30 August 2007, the Chief Justice filed an application for judicial review as Chief Justice in the Supreme Court of Gibraltar. He sought declarations that sections 6 and 37(3) of the Judicial Service Act were ultra vires the Constitution and therefore null and void. According to the grounds attached to the claim form section 6 "amounts to an unconstitutional re-definition and demotion of the office of Chief Justice, inconsistent with proper judicial independence". Advance publicity for these proceedings was given by an article in Gibraltar Chronicle of 24 August, entitled "Chief Justice Schofield set to challenge Gibraltar Parliament with local law suit", and incorporating a statement by the Chief Justice. An article to similar effect was carried in *Panorama* of the same date. Vox of that date had the headline "Bombshell as Chief Justice challenges Caruana". Two days later he submitted his response to the Signatories' memoranda. On 13 December he withdrew his application for judicial review after a hearing at which the court refused his application for adjournment until the conclusion of the proceedings before this Tribunal. In explanation of his decision to withdraw the Chief Justice said that he could not have coped with both proceedings at the same time, and that he could not have pursued an action as Chief Justice while he was suspended.
- 6.16 In evidence the Chief Justice said: "I was extremely reluctant. In fact, at one stage I had firmly decided that I was not going to take proceedings within my own court". He had sat on the opinion of Rabinder Singh QC (see paragraph 5.10), who had advised him that he could not seek judicial review until the Act came into force (which it did on 5 July). His immediate reaction had been that the Act was unconstitutional. He had waited to see how, "in the light of events", the Act was working in practice. After a month or so he had realised that section 6 was "a sham", and that its sole purpose must have been to remove him as head of the judiciary. It was, he said, unworkable. The President had no office in Gibraltar or in England. His telephone was often on answer-phone and

his only e-mail connection was through his daughter. There had been no indications about discussions as to the division of responsibilities. The practical aspect had not been thought through. Being head of the judiciary meant directing the way the courts operated, responding to complaints, deciding whether the Woolf reforms were to be adopted and so forth. On any major decision in relation to the courts of Gibraltar he, the Chief Justice, would have to have recourse to someone who was outside the jurisdiction, and was not as familiar with the jurisdiction as he was. Neither the President of the Court of Appeal nor the Minister for Justice had communicated with him. He expected the President to contact him to discuss how the Act would work. He had decided that he owed it to the Gibraltar jurisdiction to take these proceedings: he was primarily concerned with the administration of justice. He consulted Geoffrey Robertson QC and Kirsty Brimelow to whom he passed Mr Singh's advice. He instructed David Hughes (mentioned at paragraph 4.8 et seq). He realised that he would have to take action within a fairly short timeframe. However, the only specific event which occurred before he raised proceedings arose out of his decision, announced on 11 July, to cancel the Opening of the Legal Year, which (as we have noted in para 6.5) he accepted was "not (his) finest administrative action".

Comparison with orders sought by Mrs Schofield in her libel action

- 6.17 A striking feature of the application by the Chief Justice was the strong similarity between it and the orders for which Mrs Schofield applied in her libel action on 17 July, which had attracted publicity. The *Gibraltar Chronicle* of 21 August carried an article entitled "Chief Justice's wife mounts legal challenge to Judicial Service Act".
- 6.18 In each case the claimant sought a declaration that sections 6 and section 37(3) were ultra vires the Constitution and therefore null and void. In each case it was claimed that the Act undermined, diminished, or was inconsistent with, judicial independence; demoted the office of the Chief Justice, which had been set up in 1877; and questioned the independence of the JSC. Each referred to an assurance given to the Chief Justice at the meeting in the FCO in November 2006 that there was no intention to

reduce his office. The Chief Justice said in evidence that he had no foreknowledge of his wife's action. He had done nothing to approve it. He accepted that there were similarities between the orders sought, but denied that there had been any collusion, claiming that the similarities were because the issues were the same and he and his wife, both lawyers, were likely to articulate matters in the same way. He also accepted that some exhibits were common to both actions, but this was, he said, because there was a "family file" which contained copies of formal documents. According to his second witness statement his wife had had no hand in the filing or drafting of his application for judicial review. This was, in our view, misleading. Papers disclosed by his solicitors included an e-mail from his wife to them dated 18 August in which she expressed her views on some legal questions with which his action was concerned. He commented in cross-examination: "I can imagine that she would find it very hard to resist putting her finger in the pie, but it was not to be encouraged".

The Chief Justice's statement in support of the application

- 6.19 In support of his application for judicial review his solicitor filed a statement by the Chief Justice, in which he made a number of remarkable and disturbing claims. One of the headings of the statement was entitled "Personal attempts to remove me from office". Under this heading there appeared the following:
 - "14 In the course of late in 1998 and early 1999 I was called to a number of meetings with the then Governor Sir Richard Luce. Sir Richard asked me to consider my position as Chief Justice, and ultimately suggested that I accept a six month warrant of appointment. When I pressed him as to the reason for his suggestions he told me that the Chief Minister had made representations that my "contract" be not renewed. I made it clear to the Governor that I had security of tenure under the Constitution and would stay in post as per the Constitution.
 - 15 In 1999, at the ceremonial opening of the legal year, I spoke publicly of instances in which the delay or denial of funds by the

Government had had the potential adversely to affect the administration of justice....

- I referred to two particular incidents. Firstly, I had sought funding to attend a Judicial Studies Board seminar on the Woolf reforms, which were to be introduced into Gibraltar. Gibraltar has no equivalent of the Judicial Studies Board. Funding for this was refused by the Government, although the UK government eventually provided funding.
- Secondly, I had initiated discussions with the Government to institute a system of part time acting Stipendiary Magistrates, similar to recorders or deputy district judges in England & Wales. The benefit of this would have been twofold. It would have assisted in reducing backlog in the magistrates' court. It would also have provided a pool of local practitioners with judicial experience, which would have been valuable when making future judicial appointments. The Government were willing to agree to this, provided that they had some say in who was appointed. I considered this to be unacceptable.
- In February 2002, whilst I was sitting in a criminal trial, I was presented by the Governor with a warrant purporting to appoint me Chief Justice for 1 year. I asked counsel in the criminal trial to consider the position. My view was that, if the warrant did limit my appointment to 1 year, I would not be an independent tribunal as required by the 1969 Constitution then in force.
- 19 Counsel for two of the defendants (who is also one of my counsel in this matter) made representations that, pursuant to the 1969 Constitution, a Chief Justice appointed held office until the age of 67, or until removed by the constitutionally established procedure. Counsel for the other defendants (a member of the same firm) adopted these representations. Counsel for the prosecution (a member of the Attorney General's chambers) did not address the court in any meaningful way.
- I received no explanation from the Governor as to the powers he considered he had to circumvent my tenure of office.

- I had expressly asked the Attorney General to attend the Court and address me on the warrant. He did not do so, nor did he provide any explanation for why he did not do so.
- On 04.09.2002, Sir Desmond de Silva QC wrote to me, to record that he had received an approach from the Attorney General, indicating that I would be assisted in finding judicial employment elsewhere if I were to leave Gibraltar....
- I considered and consider that I would be untrue to my oath if I were to accept such an inducement or bow to such pressure."
- 6.20 Next, under the heading "Legislative attempts to remove me from office/diminish the office of Chief Justice", he stated:
 - The pressure to leave office moved into legislative format. When the draft of the 2006 Constitution was published I attempted to raise objections in relation to the judiciary provisions. I sought to convey these concerns privately, but found an unwillingness to address their substance. I obtained an opinion from three distinguished jurists, Sir Sydney Kentridge QC, Keir Starmer QC and Richard Tur, Benn Fellow, Oriel College Oxford, who opined that the (then draft) Constitution was defective in certain respects
 - Although comment was made to the media about these aspects of the draft Constitution, at no point was their substance addressed.
 - I spoke publicly of these issues in my address at the opening of the Legal Year in October 2006.
 - In November 2006, I attended a meeting at the Foreign and Commonwealth Office, with Sir Sydney Kentridge QC and Richard Tur, at which my concerns regarding the Constitution were discussed. Although this did not result in any amendment to the text of the Constitution, an addition was made to the explanatory note to the Constitution.
 - At the said meeting, one of the concerns expressed by the jurists was the apparent diminution of the office of Chief Justice, by appointing the President of the Court of Appeal Chairman of the

- JSC. I was assured that this was in no way intended to diminish the office of Chief Justice, who was still the most senior judge in Gibraltar.
- On the 20th of February 2007, three days before I was due to go on a scheduled vacation to Argentina with my daughter, I received a consultation paper in relation to the JSA bill. My holiday date was well-known to the Chief Minister.
- However, I did respond to the draft, albeit briefly because of the timing of the consultation, and more fully upon my return...
- Further, the General Council of the Bar in Gibraltar made representations that certain provisions of the JSA may be inconsistent with the Constitution
- The executive appeared determined to press ahead with the Bill. I believe that this determination was bourn (sic) out of a desire to diminish my powers as Chief Justice because I would not leave. Further, downgrading the office of Chief Justice ensures that future Chief Justices will retain independence but little power to use it should the necessity arise."
- 6.21 At paragraph 41 of his statement the Chief Justice, referring to the membership of the JSC, alleged: "The Attorney General has, at the very least, allowed himself to be the conduit for improper inducements to be made to myself". The basis for this claim appeared to be the conversation between the Attorney General and Sir Desmond de Silva in August 2000. For the reasons given in paragraph 3.57 we preferred the evidence of the Attorney General as to this conversation. The same applied to the conversation between him and Mr Picardo.
- 6.22 In evidence the Chief Justice said that he had signed the statement after taking legal advice and in good faith. He considered that there was an evidential basis for everything he had stated. He confirmed that it was his belief that he had been the victim of a series of personal and legislative attempts to remove him from office. In his fifth witness statement he maintained that section 6 was "a legislative device to demote me and that the guiding force was the Chief Minister, who had recently appointed

himself Minister for Justice"; he had a duty to discuss with the Chief Justice the proposed amendments prior to the publication of the consultation paper but had failed to do so and had displayed hostility in defence of the Act. The Chief Justice also believed that the Attorney General was knowingly involved in an improper attempt to remove him from office.

- 6.23 Mr Hughes said that these claims were included in order to show that the Chief Justice's concerns were not theoretical, and that there had been moves that were consistent with an insufficient respect for judicial independence. The Chief Justice said that he had been reluctant to include these claims, because he knew that this would be criticised, and initially he had not been persuaded that they were necessary for a constitutional challenge. He said that there had been a series of telephone calls, ending with his accepting the advice of Ms Brimelow that "we ought to tell it like it is". We do not accept the evidence of the Chief Justice. It is plain, and we find, that he was far from reluctant to include these claims. An e-mail of 17 August (from the e-mail address which the Chief Justice shared with his wife), showed that the Chief Justice had sent to Mr Hughes an "original draft" containing a series of allegations to the same effect as those in his statement in the judicial review proceedings. The Chief Justice accepted that he must have prepared this draft. An attendance note made two days before the issue of proceedings recorded that the Chief Justice "wanted ... the whole story to be told".
- 6.24 In addition to these claims counsel for the Chief Justice put to the Chief Minister in his cross-examination "the various matters which the Chief Justice says supports his reasonable suspicion that he is being or has been encouraged" to leave office. The first was "whether you in the run-up to the renewal of the contract in February 1999 made representations to Lord Luce that you didn't want him to continue as Chief Justice?" The second was "the simple denial of funding in 1999 when it's just simply said... you can't have funding to go, it is not appropriate and not relevant to [Gibraltar]?". The third was the Government press release of 9 October 2000 in connection with the maids issue. The fourth was "the 2002 one

- year appointment". The fifth was "the failure to consult with the Chief Justice over the 2006 Constitution, that is to say to get back to him". The sixth was "the Judicial Service Act designed to reduce his status and his role in the judicial system in Gibraltar".
- 6.25 Were these allegations well founded and justified?
- 6.26 In regard to the alleged remarks of Sir Richard Luce in 1998 or 1999 we preferred his evidence and that of the Chief Secretary and the Chief Minister to that of the Chief Justice (see paragraph 4.6).
- 6.27 As to the second matter, we concluded in paragraphs 2.17 and 2.18 that the Chief Justice's references to funding at the Opening of the Legal Year in October 1999 were seriously inaccurate and misleading, and that there was no basis for the suggestion that a delay of denial of funds by the Government had had the potential to adversely affect the administration of justice. In these circumstances paragraphs 15-17 of the Chief Justice's statement were grossly misleading.
- 6.28 As to the third matter, despite the fact that counsel for the Chief Justice founded on the press release of 9 October 2000 in his cross-examination of the Chief Minister, it is evident that, by the time he came to be cross-examined, the Chief Justice had had second thoughts on the matter. As we noted in paragraph 3.19, he did not dispute that the Government had been acting in good faith.
- 6.29 As to the fourth matter, as we remarked in paragraph 4.21, we do not know why the Governor chose to give the Chief Justice a warrant for one year in February 2002. However, the Governor told him that the practice of issuing time limited warrants could not affect his security of tenure under the Constitution. We stated in paragraph 4.25 that the hearing in chambers was contrived to enable him to give vent to his outrage at the Governor's action. We should add that we see no connection between the action of the Governor and the Government.
- 6.30 As regards the 2006 Constitution, we note that the Chief Justice's position was expressed in more than one form. In his statement in the judicial

review proceedings (paragraph 6.20) he complained about a failure to address his objections to the text which was published in July 2006. As we noted in paragraph 6.24, the point which was put on his behalf in cross-examination of the Chief Minister was his failure to get back to the Chief Justice after his letter of 9 January 2006 (see paragraph 4.47). However, in cross-examination the Chief Justice accepted that the 2006 Constitution did not represent a legislative attempt to remove him from office. The claim by the Chief Justice in paragraph 25 of his statement that at no point was the substance of his objections addressed was untrue. The Chief Justice claimed in his oral evidence that his written statement in his proceedings for judicial review had been drafted very quickly. He had been about to go to the Commonwealth Law Conference in Nairobi. He remembered discussing his statement with his counsel who said it might have to be amended later. He took as much care as he could, adding that it "may have been more happily phrased".

- 6.31 Even making allowance for the fact, as he saw it, that there was a need to take proceedings without delay, we find it extraordinary that the Chief Justice chose to include among a set of very serious and controversial allegations a complaint in regard to the 2006 Constitution from which he so readily retreated in cross-examination. It is unclear what points, if any, were not covered by this concession. However, we are certain, in the light of our observations in paragraphs 4.52 4.57 and 4.61 4.62, that, far from the Chief Minister or the Government seeking to remove him or encourage him to leave, it was the Chief Justice who behaved improperly and inappropriately.
- 6.32 Whatever may have been the merits of the arguments against section 6 in the context of Gibraltar, the Chief Justice's claim that the section was enacted at the behest of the Government as a means of removing him from office, or, in the less ambitious language used by his counsel, for the purpose of encouraging him to leave, does not bear examination.
- 6.33 In his evidence the Chief Minister flatly denied there had been any longstanding intention or attempt to remove the Chief Justice or encourage him to leave. He said that, while the Government had been very unhappy

with some of the conduct of the Chief Justice over the years, it had exercised considerable restraint. He also pointed to a series of decisions over the years when the Government had acceded to requests by the Chief Justice, for example to increase his salary, to waive rent on his official residence and to pay a proportion of his terminal gratuity in advance, all of which were inconsistent with the alleged intention or attempt. What the Chief Minister described as "the beginning of the terminal process" was, he said, probably Mrs Schofield's statements about the Chief Justice being hounded out of office, when the boundary of what society and the Government ought to be expected to tolerate was beginning to be crossed. That process was concluded by the Chief Justice's witness statement in his proceedings for judicial review when he openly accused the Government of using executive and legislative means to remove him from office. "At that point", said the Chief Minister, "I think the Government decided that this is it. One of us has to go". He went on to say that the system of government in Gibraltar, and in the United Kingdom, was not built to withstand this degree of antagonism in the public domain. However, it had not been the position of the Government that it would seek to bring about the removal of the Chief Justice – until, that is, the Governor decided to convene the Tribunal and suspend the Chief Justice. Over the years there had been long periods of complete normality of relations. That was why, the Chief Minister said, he could not understand the idea that going back to 1998 there had been some sort of machination for his removal, trying to create the impression that the Government was "a sort of serial attempted remover".

- 6.34 The claim that the Government had been seeking the removal of the Chief Justice since 1998 or 1999 is, in our view, without foundation. We should add that we would have reached the same conclusion even without regard to the evidence of the Chief Minister.
- 6.35 Once again we draw attention to the effect of the conduct of the Chief Justice. It is plain that these allegations were also bound to cause public controversy, and to have a profound effect on the well-being of the relationship between the Chief Justice and the Government, the Chief

Minister and the Attorney General. The Chief Justice accepted that he would have "problems" in hearing cases in which the Chief Minister was personally involved. The same we find would be likely to apply in the case of the Attorney General in view of the Chief Justice's remarks about him.

6.36 There is another aspect. On his own evidence the Chief Justice never gave thought to whether, in taking the stance which he had done, he might be seen as aligning himself with the Opposition which had opposed the passage of the Act and had pledged to repeal it. According to his evidence he was disinterested, concerned only for the constitution and the administration of justice. However, the Leader of the Opposition was quick to take the benefit of the challenge which the Chief Justice had mounted. On 3 September 2007 the Gibraltar Chronicle carried an article entitled "Schofield 'well placed' to question legal standing of Judicial Service Act – Bossano".

CHAPTER 7

CONCLUSIONS

This chapter contains-

- our general conclusions as to the conduct of the Chief Justice on the basis of our findings in the previous chapters (paragraphs 7.1 7.27)
- the grounds for the removal of a Chief Justice (paragraphs 7.28-7.33)
- our assessment (paragraphs 7.34 7.42)
- our advice (paragraph 7.43).

The conduct of the Chief Justice

- 7.1 In the course of giving our conclusions we will have occasion to have regard to the *Bangalore Principles of Judicial Conduct* (2002) and the 2008 edition of *Guide to Judicial Conduct*, published by the Judges' Council of England and Wales, to which we were referred by Counsel to the Tribunal and counsel for the Government. We will do so, not for the purpose of deriving from them any definitive test for the application of "inability" or "misbehaviour" in terms of the Gibraltar Constitution, but in order to give us some assistance in seeing whether, and to what extent, the conduct of the Chief Justice fell below the standard expected of a judge. The standards with which we will be mostly concerned are those relating to a judge's integrity and the propriety of a judge's conduct.
- 7.2 As we narrated in chapter 3, the conduct of the Chief Justice in regard to the maids issue was previously considered by the Governor in 2000, with a view to the possible establishment of a Tribunal of Inquiry under the then Constitution. In evidence the Chief Justice acknowledged that he

- had made grave errors in regard to the payment of PAYE and social insurance contributions in respect of Ms Danvers.
- 7.3 The decision of the Governor on 4 October 2000 caused considerable controversy in Gibraltar and legal practitioners in particular (paragraph 3.23). So far as concerned Ms Williams, the Governor relied on the Chief Justice's assurance that he had no outstanding liabilities. In the light of the discussion at the directions hearing on 8 April 2008 we were satisfied that we should revisit this issue (see the Second Schedule). This led to investigation of the circumstances in which Ms Williams was employed, the results of which were the subject of evidence before us. This demonstrated that the Chief Justice had been in breach of the law relating to work permits, taxation and social insurance contributions in her case. This was in addition to his having irresponsibly signed work permit forms for her in blank, which he accepted showed "reckless disregard" (paragraphs 3.5). Furthermore we found that his communications to the Governor were less than full and frank. They were positively misleading in regard to the legal advice he purported to receive and pass on (paragraphs 3.13 - 3.15). We have no doubt that the damage to the reputation of the Chief Justice which was caused by the maids issue in 2000 would have been all the greater if the full facts in regard to Ms Williams had been known.
- 7.4 It is also plain that when he made his fifth witness statement to this Tribunal the Chief Justice continued to take an attitude towards compliance with the law which was nothing less than cavalier, in that he asserted that he had had no obligation to make contributions in respect of Ms Williams and accused the Government of having acted in bad faith. When he gave evidence before us he conceded that he was in error in both respects (paragraph 3.19).
- 7.5 It hardly needs to be pointed out that integrity is essential to the proper discharge of the judicial office. A judge must maintain high standards in private as well as public life. His or her conduct has to be "above reproach in the view of a reasonable observer" (*Bangalore Principles* paragraph 3.1). "Judges have to accept that the nature of their office

exposes them to considerable scrutiny and puts constraints on their behaviour which other people may not experience... Behaviour which might be regarded as merely unfortunate if engaged in by someone who is not a judge might be seen as unacceptable if engaged in by a person who is a judge and who, by reason of that office, has to pass judgment on their behaviour of others" (*Guide to Judicial Conduct*, paragraph 4.1).

- 7.6 We now turn to the evidence about other matters. It soon became evident during the course of the hearing that the Chief Justice's conduct repeatedly took a number of inappropriate forms. While it is convenient for the purpose of description to set them out under separate headings, it is plain that many of them are inter-related.
- 7.7 First, his conduct repeatedly fell far short of what befitted the dignity of his office. This included a tendency to over-react to perceived slights. He did not, however, appear to be alive to that. We refer to the following examples:
 - (i) His behaviour in regard to the attendance of the Chief Minister at the Registry of the Supreme Court and Magistrates Court in December 1999 showed unwarranted hostility and petty discourtesy (paragraph 2.35).
 - (ii) His remark to Robert Vasquez in October 2000, in the aftermath of the maids issue, that he should not "rape the Constitution", was inconsistent with the dignity and status of his office (paragraph 3.23).
 - (iii) The conduct of his defence when he was prosecuted in 2001 betrayed a remarkable lack of judgment and sense of proportion and a disregard for the damage done to the administration of justice in Gibraltar (paragraph 3.52).
 - (iv) He behaved petulantly towards the Acting Governor and without regard to the constitutional consequences of his action because he took the view that the Acting Governor had wrongly given the

- Chief Minister precedence over him on the departure of the former Governor in July 2006 (paragraph 4.37).
- (v) In February 2007 he flagrantly disregarded the Chief Minister's stipulation when providing him with a draft copy of the Judicial Service Bill that he should treat the draft confidentially and for his eyes only by showing the draft to his wife and leaving it available to her with the consequence that it was leaked to the press, and he took provocative action in bypassing the Bar Council by circulating copies of the draft to members of the Bar, without reference to the Government (paragraph 5.7).
- (vi) In allowing his wife and her counsel to attend on him formally in his chambers on the afternoon of 16 April 2007, and in acceding to their request for guidance in relation to a possible application for injunctive relief against the Bar Council, he acted in flagrant disregard of accepted standards of judicial behaviour; yet even before us he appeared unable to appreciate that (paragraphs 5.51 5.62).
- (vii) He cancelled of the Opening of the Legal Year 2007 2008 without attempting to contact the President of the Courts of Gibraltar (paragraph 6.5 and 6.8)
- (viii) In his proceedings for judicial review of provisions of the Judicial Service Act he made very serious allegations against the Government and the Chief Minister which he either abandoned or which we found to be without foundation (paragraphs 6.15 ff).

We should add that the tendency of the Chief Justice to resort to combat with the Government by press release or a letter to the press, not to mention legal proceedings or the threat of them (to which we refer below), was hardly becoming the impartial dispenser of justice.

7.8 Secondly, the confrontational manner in which the Chief Justice addressed what he claimed were threats to judicial independence, in reality the status of his office and his continuance in that office, was in our view

improper in nature and in any event inappropriate and disproportionate. As we have seen, this included his courting publicity for serious accusations, and that without having adequately discussed or checked matters with the Government. His statements were frequently inaccurate and liable to lead to public speculation, yet he showed no inclination to explain them even when aware that they had been misunderstood or distorted, let alone any understanding of their effect on the respect in which the judiciary should be held.

- 7.9 The Guide to Judicial Conduct states at paragraph 5.1(6), in words derived from the Bangalore Principles, that "a judge, like any other citizen, is entitled to freedom of expression, belief, association and assembly, but in exercising such rights, a judge shall always conduct himself or herself in such a manner as to preserve the dignity of the judicial office and the impartiality and independence of the judiciary". The Guide goes on in paragraph 8.2.2 to counsel that, in participating in public debate, a judge should take care to avoid causing the public to associate the judge with a particular organisation, group or cause. We should add that the reason for such guidance is obvious. It is essential that the judge has, and is seen to have, the detached, unbiased, unprejudiced, impartial, open-minded and even-handed approach which is the hallmark of a judge. If a judge enters the political arena and participates in public debate by expressing opinions on controversial subjects which do not directly affect the operation of the courts, the independence of the judiciary or aspects of the administration of justice, by entering into disputes with public figures or by publicly criticising the government, he or she may well be perceived as not impartial when presiding in court in cases relating to such controversies or involving such figures or the government. For that reason, it is of the utmost importance that the judge's public statements are accurate, and that they are suitably restrained, and not framed in inflammatory or tendentious terms.
- 7.10 These public statements by the Chief Justice, which were highly critical of the executive, betrayed a complete lack of grasp of the need for restraint.

 The fact that the subject matter was said by him to be the independence or

position of the judiciary or the administration of justice does not, in our view, significantly affect the position since what mattered was the manner in which, and the context in which, his interventions were made.

7.11 We refer to the following examples:

- (i) The remarks which he made in his public address at the Opening of the Legal Year in October 1999 about the Government delaying or withholding funding suggested that the independent functioning of the judiciary had been at risk of improper restraint by the Government. They were inaccurate and without foundation. They were in language which was so unspecific as to create an obvious risk of speculation. The Chief Justice appeared to fail to understand the concern which his remarks had created. His claim that for him to have explained his remarks would have run the risk of misinterpretation was unconvincing. His claim that the matters to which he alluded were or might be confidential and could not be disclosed without risk of contravention of the Official Secrets Act was without substance (see paragraphs 2.17 2.24).
- (ii) His letter to the *Gibraltar Chronicle* dated 11 November 1999 about consultation in regard to judicial appointments courted controversy with the Government and was inaccurate (see paragraphs 2.30 2.31).
- (iii) In February 2002 after receiving only a one year warrant of appointment he readily accused the Governor of a calculated attempt to undermine him (paragraph 4.7). He made angry remarks at a contrived hearing to enable him to give vent to his outrage at the Governor's action and threatened the suspension of sittings. Although the hearing took place in chambers, the substance of his complaint was leaked to the press by a source close to him (paragraphs 4.18 4.25)
- (iv) The remarks which he made in his public address at the Opening of the Legal Year in October 2006 on the subject of the draft Constitution followed his threat of legal proceedings against the

executive. He did not have the full backing of the local judiciary for these remarks. They were misleading and unbalanced (he claimed in evidence that he was not responsible for how the press interpreted them). They represented his opening of a public campaign with the referendum in mind. This was improper. His involvement of solicitors and other legal advisers and his threat of legal proceedings in support of that campaign were in any event unnecessary and grossly disproportionate. That campaign became part of Gibraltar politics (see paragraphs 4.53 - 4.63).

- (v) His engagement of at least three teams of legal advisers in response to the Judicial Service Bill showed again an over-reaction to what he claimed was a threat to judicial independence (paragraphs 5.5, 5.10 and 6.16).
- 7.12 Thirdly, we refer to his hostility towards the Government and the Chief Minister in particular, against whom he made serious and unfounded accusations. The *Guide to Judicial Conduct* observes in paragraph 2.1, in connection with the subject of judicial independence, that the relationship between the judiciary and the legislative and executive arms of government "should be one of mutual respect, each recognising the proper role of the others". It is plain, particularly in the case of a Chief Justice, that along with mutual respect there needs to be a positive and constructive relationship with the executive arm. There were failures in both respects, for which the Chief Justice was directly responsible.
- 7.13 The Chief Justice's proceedings for judicial review of the Judicial Service Act represented a constitutional challenge by the head of the resident judiciary to the legislative and executive arms of government. In support of his application he not only questioned the power of the legislature to enact certain provisions of the Judicial Service Act, but also alleged that the Government had had for a long time attempted to have him removed from office. In the course of the hearing before us these allegations were either departed from or shown to our satisfaction to be unfounded (paragraphs 6.26 6.34). As we observed in paragraph 6.35, these allegations were bound to cause public controversy, and to have a

- profound effect on the well-being of the relationship between the Chief Justice and the Government, the Chief Minister and the Attorney General.
- 7.14 They would moreover affect the ability of the Chief Justice to discharge the functions of his office. In evidence the Chief Justice accepted that he would have "problems" in presiding over any case in which the Chief Minister was involved personally, either as a witness or because a policy of Government with which he was concerned was in issue (paragraph 6.35). According to the evidence of Mr Mendez, the Government is involved in only about 10% of civil cases, and less than 10 judicial review cases were heard each year. However, we do not think it correct to confine attention to Government cases with which the Chief Minister is directly connected. Mr Catania gave evidence that any attempt to draw a clear distinction between them and Government cases which were unconnected to the Chief Minister might be problematic in a jurisdiction as small as Gibraltar. "In Gibraltar", he said, "people see Government involvement in everything that happens, and obviously with the Government comes the head of Government". In any event it is clear that cases involving the Chief Minister would be expected to be among the most important in their impact on the public. In addition one should not underestimate the potential effect of the Chief Justice's allegations on his reputation for impartiality in cases at large.
- 7.15 Counsel for the Tribunal also pointed to further implications which could arise from these allegations. He submitted that it was difficult to see how the concession which the Chief Justice had made in respect of the Chief Minister would not also extend to cases involving the Attorney General, in the light of the Chief Justice's evidence that the Attorney General had been knowingly involved in improper attempts to remove him from office. Further, Counsel to the Tribunal argued, there was a risk that any interventions which the Chief Justice made in regard to the administration of justice, such as making representations in accordance with section 5 of the Judicial Service Act, could be undermined by the perception of the fair-minded observer that the Chief Justice might be motivated by animosity towards the Chief Minister, rather than by a desire to address

- matters of substantial concern. In his evidence the Chief Justice accepted that in "certain situations" this might "perhaps" be the case. That was a considerable understatement. We accept these submissions. We would add that the Chief Justice's known antipathy towards certain members of the Bar Council, in particular Mr Neish, would be likely to affect the ability of the Chief Justice to be seen to do justice in individual cases.
- 7.16 We now turn to the question of whether the Chief Justice supported, or was associated with, the statements and actions of Mrs Schofield.
- 7.17 Before coming to the events in 2007 relating to the draft Judicial Service Bill, we should point out that all along she made no secret of her determination to fight what she claimed were attempts to remove him from office. As for the Chief Justice, we are in no doubt that he was content that this should be known, even when he knew that there was no foundation for the claim.
- 7.18 He was aware in 2000 that she had said that if the Chief Minister fought him in the gutter she would meet him in the gutter. The Attorney General gave evidence, which we accepted, that the Chief Justice had remarked to him that he could not be involved in a public controversy but she would get into the gutter and fight (paragraph 3.22).
- 7.19 The Chief Justice was aware, in connection with alleged phone tapping, that she had given a press interview in which she had claimed that "they are trying to hound him out of office", and that it was also reported in the press that he believed that he was under surveillance because of a clash with the Chief Minister over claims of political interference in the judiciary's independence. He said in evidence that he had not considered making a public statement to contradict either statement. As we observed in paragraph 3.36, his unwillingness to correct what had been attributed to him and to dissociate himself from her comments was not satisfactorily explained by him. In paragraph 2.31 we drew attention to the completely different approach of the Chief Justice when concerned about a Government statement with which he disagreed but which he was anxious not to be perceived to approve by staying silent.

- 7.20 He was aware that, at the invitation of his wife, observers from the IBA and the ICJ were present at his trial in 2001, apparently briefed by her that the prosecution was politically motivated, a claim which formed no part of his defence and was in any event without substance. Yet he was quite content with the attendance of these observers (see paragraph 3.52).
- 7.21 As we stated in paragraph 4.26, the perception given publicly in February 2002 was of concerted action by the Chief Justice and his wife in response to the action of the Governor in presenting him with a one year warrant of appointment.
- 7.22 It is thus clear that, before the issuing of the draft Judicial Service Bill in February 2007, Mrs Schofield was well known for campaigning against what she claimed were attempts to have the Chief Justice removed from office. On certain occasions she and the Chief Justice gave the impression that their actions were co-ordinated. However that may be and we return to that aspect below we have no doubt that in the absence of his distancing himself from her statements, a fair minded and well informed observer would have considered that she had spoken with his approval.
- 7.23 As regards the draft Judicial Service Bill, we have found that the Chief Justice expected that his wife would "go public"; that he was content that she should do so and supported her in that respect (paragraph 5.17). He was aware that she intended to pursue the matter of her correspondence with the Bar Council as vigorously as possible (paragraph 5.22). While he accepted that it would have been improper for him to have sent the emails which she sent to the Bar Council, he was reckless as to the effect which they had on his position, in the absence of any attempt to distance him from them (paragraph 5.29). There was, in our view, abundant evidence that the Chief Justice would have been perceived as supporting, or at any rate approving, her statements, hence improperly associating himself with her communications with the Bar Council.
- 7.24 The events that followed provided confirmation for that perception.

 Showing reckless disregard for the requirements and reputation of his office the Chief Justice allowed his wife and her solicitor to attend on him

- formally in connection with her libel action and then offered guidance for the protection of her position (paragraph 5.74). He thereafter treated his wife as an interested party in regard to an application for him to recuse himself in the light of her conduct (paragraphs 5.93 and 5.97). These actions were in themselves improper.
- 7.25 A further example of grounds for the perception that the Chief Justice and his wife were acting in concert was provided by the strong similarity between the orders sought by her in her libel action, and those sought by him, not long afterwards, in his proceedings for judicial review (paragraphs 6.17 6.18)
- 7.26 To this we add evidence given by the Attorney General which is relevant to Mrs Schofield's involvement in her husband's concerns. In his witness statement and orally he described an occasion when both the Chief Justice and his wife came to see him "quite early on in my time as Attorney General" (he was appointed to that office in 1997). They complained that the circulation list for the Revision of Laws Bill had read: "Governor. Chief Minister. Chief Justice". The Chief Justice was, he said, very affronted and wanted to know why he had not intervened to get the order changed. He described this as a threat to the independence of the judiciary. The Attorney General had replied that this had nothing to do with the independence of the judiciary, but was to do with the standing order of precedence. He said he had wondered why Mrs Schofield was "quite so involved in something that I would have thought really only concerned her husband". He regarded it as an example of them acting in concert. He also recalled an evening when, for at least an hour, he was harangued by them about the alteration of the order of precedence and how this interfered with the independence of the judiciary, which struck him as "a completely disproportionate response".
- 7.27 Finally, we have noted that the Chief Justice gave evidence that he was concerned by a press report in April 2007 of an e-mail in which his wife listed the maids issue as one of the reasons why she claimed that the Chief Minister wanted to get rid of him (paragraph 5.35). He said that he did not share that view but had not considered dissociating himself from that

suggestion "because once I embark on that, I then have to embark upon qualifications, explanations, which may lead us to further problems" (paragraph 3.26). We again find that explanation unconvincing and his inaction inappropriate.

The grounds for the removal of a Chief Justice

7.28 Section 64 (2) of the Gibraltar Constitution provides:

"The Chief Justice, a Puisne Judge, the President of the Court of Appeal or a Justice of Appeal may be removed from office only for inability to discharge the functions of his office (whether arising from infirmity of body or mind or any other cause) or for misbehaviour...".

These grounds for the removal of a judge, which are similar to statutory provisions in other jurisdictions, are in harmony with protocol 18 of the UN Statement on the Basic Principles of the Independence of the judiciary, which states:

"Judges shall be subject to suspension or removal only for reasons of inability or misbehaviour that renders them unfit to discharge their duties".

The requirements of section 64(2) are clearly stringent, and rightly so, since a judge's security of tenure is essential for securing the independence of the judiciary, which is fundamental to the proper functioning of a democratic society.

7.29 Counsel to the Tribunal submitted, in our view correctly, that inability is a broader concept than disability and embraces more than physical or mental infirmity. It might also extend to an accumulation of matters which taken together, were indicative of a lack of ability to perform fundamental duties. He referred to *Stewart v Secretary of State for Scotland* 1998 SC (HL) 81. In that case it was found, in the light of what happened in eighteen cases, that the conduct of a sheriff had been improper and that it stemmed from a defect of character which rendered him unable to

- perform the judicial functions of a sheriff. It was held that the removal of the sheriff from office by reason of "inability" was justified.
- 7.30 Counsel for the Chief Justice placed reliance on a passage in the speech of Lord Jauncey of Tullichettle in *Stewart* at page 85 where he adopted a statement by Lord Coulsfield in the Inner House: "What has to be shown is that [the sheriff] is not really capable of performing the proper function of a judge at all". He also submitted that, leaving aside cases of infirmity, this ground for removal of a judge required some inherent or acquired failing, some personality defect or some other similar cause.
- 7.31 We make two comments on the submissions by counsel for the Chief Justice. First, we note that when adopting the statement of Lord Coulsfield Lord Jauncey was seeking to contrast "mere lack of efficiency or competence *per se*", which was "very unlikely to measure up to inability". The implication of Lord Jauncey's remarks is, in our view, that "inability" is concerned with incapacity in one or more fundamental respects. Secondly, it seems to us that the submissions of counsel for the Chief Justice tended to treat the facts of *Stewart* as if they provided a definition of "inability", that is to say, where there was no question of infirmity. However, the words "or any other cause" do not restrict the scope of what may give rise to "inability".
- 7.32 As regards "misbehaviour", counsel for the Chief Justice submitted that the threshold was a high one. In the light of the history of that ground, it imported serious misconduct, such as corruption, partisanship or gross neglect in office, or some crime of moral turpitude outside office. He pointed out that almost all of the recent cases in which "misbehaviour" had been given serious consideration involved conduct amounting to a crime. He referred to a number of examples, including *Clark v Vanstone* [2004] FCA 1105. He emphasised, by reference to a passage in the judgement of Gray J at para 85 of that case a passage approved by the Privy Council in *Lawrence v Attorney General of Grenada* [2007] 1 WLR 1474 that the content of its meaning was to be determined by reference to the effect of the conduct on the capacity of the person to hold office. Gray J continued:

"In turn, the capacity to continue to hold office has two aspects. The conduct of the person concerned might be such that it affects directly the person's ability to carry out the office. Alternatively, in addition, it might affect the perceptions of others in relation to the office, so that any purported performance of the duties of the office will be perceived widely as corrupt, improper or inimical to the interests of the persons, or the organisation, for whose benefit the functions of the office are performed. In either case, the danger is that the office itself will be brought in to disrepute as a result of the conduct of its holder. If that is likely to be the case, then the conduct is properly characterised as misbehaviour for the purposes of relevant legislation".

7.33 This quotation demonstrates that the significance of both inability and misbehaviour as a well-founded ground for the removal of a judge is ultimately the same, namely that the judge is shown to be unfit to hold office. In each case that unfitness must lead to removal since it undermines public confidence in the judiciary and the administration of justice. However, the route by which inability or misbehaviour comes to be a ground for removal is somewhat different. Inability, in whatever respect it applies, goes to the root of the judge's ability to discharge the functions of his office. It is of a fundamental character. By contrast a judge may be guilty of misbehaviour without it necessarily constituting a ground for his removal. That will depend on whether it is such as to show him to be unfit for his office (cf *Therrien v Canada (Minister of Justice)* [2001] 2RCS 3 at para 147). Only then will it be misbehaviour within the meaning of the statutory provision.

Our assessment

7.34 Counsel for the Tribunal and counsel for the parties represented before us were agreed that it was for us to reach our own conclusion as to whether the facts found by us showed any inability on the part of the Chief Justice to discharge the functions of his office or any misbehaviour on his part; and if so, to reach our own conclusion as to whether such inability or misbehaviour should warrant the removal of the Chief Justice. We are

- satisfied that this would be consistent with section 64 of the Gibraltar Constitution.
- 7.35 We are in no doubt that in a number of instances there was misbehaviour on the part of the Chief Justice, without our going so far as to say that any single instance amounted to such misbehaviour as to show that he was thereby unfit to hold office. We give as examples:
 - (i) his unfounded complaints about funding in 1999, which he sought to exploit in order to embarrass the Government;
 - (ii) the maids issue, where he persisted in a reckless disregard for compliance with the law and made a less than frank disclosure to the Governor;
 - (iii) the way in which he assisted his wife to make attacks on the Bar Council which it would have been improper for him to make; and heard her counsel in chambers of on the afternoon of 16 April 2007; and
 - (iv) his serious and unfounded allegations against the Government in 2007.
- 7.36 The conduct of the Chief Justice which we have summarised earlier in this chapter directly affected the way in which he discharged part of the responsibilities of his office, such as his relations with the Governor and the Government (covering such matters as funding, proposed legislation and appointments), and his relations with the representatives of the Bar. The conduct stemmed, in our view, from a number of characteristics of his personality and attitude, as follows.
- 7.37 First, the Chief Justice did not seem to be alive to the boundary between what was and what was not proper for someone in his position to do or say. He repeatedly showed a lack of judgment in this respect. He also showed the lack of a sense of proportion, and tended to over-react to perceived slights. He did not observe appropriate restraint or respect for accuracy in his public pronouncements.

- 7.38 Secondly, he showed a pre-occupation, bordering on an obsession, with judicial independence. He claimed that it was under threat when this was not the case. This led to his responding in an improper or excessive manner to executive action of which he disapproved. Allied to this was his pre-occupation with the status of his office and his continuance in office. This showed itself in a number of ways ranging from petty discourtesy to the Chief Minister to the unfounded accusation that the Government had long sought to have him removed from office.
- 7.39 Fourthly, he showed himself to be unable to restrain himself from supporting his wife in her attack of the members of the Bar Council or her libel action against its Chairman. Although he affected a lack of interest in her communications with the Bar Council he was more than content that his silence should be interpreted as support for her communications. He knew that it would have been improper for him to have sent them. He was unable to grasp that his association with them would have been seen by a fair minded and well informed observer as improper.
- 7.40 Fifthly, the perceptions arising from the conduct of the Chief Justice inevitably rendered it impossible for the Chief Justice to sit in a significant number of cases.
- 7.41 At the same time the Chief Justice showed himself to be indifferent as to the effect, or the perceived effect, of his conduct on his relations with the Government and the Bar, the standing of the judiciary and the administration of justice in Gibraltar. This would inevitably affect the reputation of his office. In the particular context of Gibraltar, which is a small jurisdiction as a number of witnesses reminded us, the significance of public perception is inevitably magnified. In his witness statement Mr Neish observed: "The public in Gibraltar is much closer to public figures than in the case of say, England. Their scrutiny is more intense and their actions more directly felt". While it is true that public opinion in Gibraltar is not unanimous in its disapproval of the conduct of the Chief Justice we are in no doubt that its effect has been to polarise public opinion in a way which is damaging to the reputation of the office, and hence to the

- interests of good governance of Gibraltar. By his conduct he has antagonised a large number of those who practise before him.
- 7.42 In these circumstances we conclude that the Chief Justice is unable to discharge the functions of his office. We are satisfied that this inability warrants the removal of the Chief Justice from office.

Our advice

7.43 Accordingly, in terms of section 64(4)(c) we advise the Governor that he should request that the question of the removal of the Chief Justice should be referred by Her Majesty to the Judicial Committee of the Privy Council.

Dated 1.2. November 2008

Cullen of Whiteking

The Rt Hon Lord Cullen of Whitekirk KT

The Rt Hon Sir Peter Gibson

The Rt Hon Sir Jonathan Parker

Touri. Ins.

First Schedule

Procedural history

- 1. At the first directions hearing, which was held in public in London, with a video-link to Gibraltar, on 31 January 2008, we ruled that the Signatories and the Government should be entitled to be represented before the Tribunal in accordance with section 11 of the Commissions of Inquiry Act 1888 (the 1888 Act) which applies to the Tribunal by virtue of subsection (5) of section 64 of the Constitution. We also invited various named individuals and entities to make representations relevant to the Memoranda, the Chief Justice's Preliminary Response and our terms of reference. Directions were also given for the preparation and service of a draft Statement of Issues, with opportunity to the Chief Justice to identify any issues which he contended fell outside the our proper remit. Other directions were given for production of documents and the service of witness statements collated by the Solicitors to the Tribunal.
- 2. Representations were thereafter received from the Signatories, the Government, Mrs Schofield, the Attorney General of Gibraltar, the Registrar and the Deputy Registrar of the Supreme Court of Gibraltar, a number of members of the legal profession in Gibraltar, the Legal Adviser to the FCO, the ICJ and the IBA. The Gibraltar Government supported the view of the Signatories that the Chief Justice should be removed from office.
- 3. At the second directions hearing, which was held in public in Gibraltar on 8 April 2008, we heard and decided two applications by the Chief Justice. The first application was that we should not inquire into certain fresh matters introduced in representations made by the Signatories and the Government or into other matters which were said to have been already

determined. The second was that, in relation to all or some of the allegations, there was no case to answer. In the result, save in respect of certain of the issues in the draft Statement of Issues, these applications were dismissed. The reasons for our decisions were issued on 23 April 2008, and are reproduced in the Second Schedule. In the light of that decision a revised Statement of Issues was prepared and issued to the participants. Its headings in respect of factual issues in chronological order are set out in the Third Schedule. The evidence arising out of the investigation of the matters referred to in the revised Statement of Issues is the subject of the chapters of this Report.

- 4. At this hearing we also made a recommendation, as agreed by the Chief Justice and the Government, as to the costs of the representation of the Chief Justice, in accordance with section 13 of the 1888 Act.
- 5. We granted an application by Mrs Schofield that she should be entitled to be represented by counsel before the Tribunal. We were not persuaded that it was appropriate that her representation should be publicly funded. The text of our rulings in respect of Mrs Schofield, issued on 23 April 2008, is reproduced in the first part of the Fourth Schedule.
- 6. By letter dated 16 June the Solicitors to the Tribunal informed the solicitors for the Chief Justice of the Tribunal's decision in relation to applications which they had made on his behalf. It was satisfied that certain witnesses should be added to the current list of proposed witnesses. It was also satisfied that his counsel should be permitted to cross-examine each of the witnesses giving oral evidence, subject to the qualifications that such cross-examination was limited to matters of factual dispute of which the witness had personal knowledge; that the issues on which cross-examination was to be pursued, and time estimates for the same, were identified in advance; and that the time permitted for such cross-examination was subject to such reasonable limits as might be directed by the Tribunal, and consistent with the time frame allocated for the hearing. Further, all cross-examination permitted by the Tribunal was subject to its overriding power to intervene at any time in order to regulate its own procedures and to control the proceedings, in the

- interests of justice, fairness and proportionality. Cross-examination proceeded on that basis.
- 7. The full hearing was held in public in Gibraltar. It commenced on 7 July with opening submissions of counsel supplementing written submissions which had been provided earlier. We refused a written application by Mrs Schofield for permission to cross-examine certain witnesses in regard to certain matters. We did so for the reasons advanced by Counsel to the Tribunal, namely that there was no reason to believe that counsel for the Chief Justice could not pursue such lines of questioning as were relevant to the issues before the Tribunal.
- 8. The full hearing concluded on its twelfth day (28 July) with closing submissions by counsel orally and in writing.
- 9. The following was the representation before us:
 - The Chief Justice by Mr Edward Fitzgerald QC and Ms Caoilfhionn Gallagher, instructed by Messrs Bindmans LLP of London
 - The Government of Gibraltar by Mr James Eadie QC, instructed by the Government.
 - The Signatories by Mr Antony White QC and Mr Robert Vasquez, of Triay & Triay, Gibraltar
 - Mrs Schofield was not represented before the Tribunal
- 10. The witnesses who gave evidence before the Tribunal did so on oath or affirmation. Their written statements were supplied to the participants in advance and spoken to. With the exception of the Chief Justice, each of the witnesses was examined by Counsel to the Tribunal, and cross-examined by counsel for the Chief Justice. The Chief Justice was examined by his counsel and cross-examined by Counsel for the Tribunal, and, in respect of one area of inquiry, by counsel for the Government. The written statement of each witness and his or her oral evidence were directed to the matters covered by the revised Statement of Issues so far as relevant to that witness.

- 11. In the order in which they commenced their oral evidence the witnesses were as follows:
 - Mrs Katherine M Dawson, formerly Registrar of the Supreme Court and the Court of Appeal of Gibraltar
 - Mr Clive J Mendez, Deputy Registrar of the Supreme Court and the Court of Appeal of Gibraltar
 - Mr Robert M Vasquez, Barrister, Gibraltar
 - Mr Alfred Vasquez QC, Gibraltar
 - Mr Francis J Triay, Barrister, Gibraltar
 - Ms Annabelle Desoiza, Registrar of the Supreme Court and the Court of Appeal of Gibraltar
 - Mr Charles Gomez, Barrister, Gibraltar
 - Mr David Hughes, Barrister, Gibraltar (now practising as a solicitor in the United Kingdom)
 - Mr James Neish QC, Chairman of the Bar Council
 - Mr Richard Garcia MBE, Chief Secretary to the Gibraltar Government
 - Mr Peter Caruana QC, Chief Minister of Gibraltar
 - Mr Reginald Rhoda QC, Attorney General of Gibraltar
 - The Hon Philip R Barton, CMG OBE, Deputy Governor of Gibraltar
 - Mr Joe Bossano, Leader of the Opposition, formerly Chief Minister
 - Mr Stephen Catania, Barrister, Gibraltar
 - Mr J E Triay QC, Gibraltar
 - The Chief Justice
 - Mr Charles Pitto, the Stipendiary Magistrate, Gibraltar

- The Tribunal was also provided with written statements of a number of persons who did not give oral evidence, but whose evidence was considered by the Tribunal. Their statements were likewise directed to the matters covered by the revised Statement of Issues so far as relevant to them. They were:
 - Mr Felix Alvarez, Chairman of the Equality Rights Group, Gibraltar
 - Bishop Charles Caruana, Gibraltar
 - Mr James Dingemans QC, London
 - Mr David J V Dumas QC, Gibraltar
 - Mr Byron Georgiadis, Barrister, Nairobi
 - Ms Gillian M Guzman, Barrister, Gibraltar
 - The Rt Hon Lord Luce KG
 - Mr Fabian Picardo, Barrister, Gibraltar
 - Mr Peter Schirmer, Associate Editor, *Vox* newspaper
 - Miss Amanda Schofield
 - Mrs Anne W Schofield

The second part of the Fourth Schedule sets out the circumstances in which Mrs Schofield did not give evidence.

- 13. Following the conclusion of the hearing on 4 August a further written statement of Mr David Hughes was tendered to us. Having invited and considered submissions from the other participants we decided that it should be taken into account along with the rest of the evidence.
- 14. On 11 August 2008 the Signatories made an application to us that that we should recommend to the Governor that our Report should be disclosed to the Chief Justice, the Government and the Signatories and that it be made public. Having invited and considered the submissions of the participants on this subject we refused the application for the reason

advanced by Counsel to the Tribunal in his submissions, namely, in short, that it was not part of the function of the Tribunal to make such a recommendation. We stated that any submissions of the interested parties for the publication of the Report should in due course be addressed to the Governor.

Second Schedule

Rulings on applications by the Chief Justice at the directions hearing on 8 April 2008

1. At a directions hearing in Gibraltar on 8 April 2008 we heard and decided two applications by the Chief Justice, notice of which had been served on 26 March 2008. We indicated at the end of the hearing that we would give the reasons for our decisions later. The following are our reasons.

The first application

2. It was for a direction or order in the following terms:

"Remit: The Tribunal may not inquire into fresh matters not originally referred to the Tribunal in September 2007 or into matters that have already been determined"

3. The members of the Tribunal were appointed by Governor of Gibraltar by letters dated 14 September 2007, and, in the case of its Chairman, dated 11 December 2007, under section 64 (4) (a) of the Gibraltar Constitution Order 2006 and in accordance with the advice of the Judicial Service Commission (JSC), to advise him on the matter of the Chief Justice. The letters of appointment stated:

"The Tribunal should inquire into and report on whether the Chief Justice is unable to discharge the functions of his office by reason of inability or for misbehaviour having regard to the Memorandum and Supplementary Memorandum with two Appendices submitted by 13 senior representatives of the legal profession on 17 April 2007 and any other

submissions and evidence which may be placed before it and report on the facts thereof to me and advise whether I should request that the removal of that judge should be referred by Her Majesty to the Judicial Committee".

- For The Chief Justice Mr Fitzgerald explained that the "fresh matters" mentioned in the application extended, not only to entirely new material contained in the representations of the Signatories and the Gibraltar Government in response to the Tribunal's Directions Ruling No 1 of 31 January 2008, but also to the inclusion in such representations of new detail of allegations that were previously wholly unparticularised or vague.
- 5. Mr Fitzgerald maintained in the first place that consideration of the fresh matters would be unconstitutional. Removal of a judge for inability or misbehaviour was subject to two key safeguards. In compliance with subsection (8) of section 64 the Governor had to act in accordance with the advice of the JSC. As a matter of fairness, he also had to afford the judge an opportunity to be heard (cf Rees v Crane [1994] AC 173). The fresh allegations had not been subjected to the "constitutional filter", namely the identification by the Governor, acting on the advice of the JSC, of specific allegations when he decided, in terms subsection (4), that the "question of removing the Chief Justice... ought to be investigated". The latter allegations were "the matter" into which the Tribunal had to inquire in accordance with subsection (4)(b). The filter was intended to prevent vexatious or frivolous complaints from reaching the Tribunal. The lack of such a filter would remove the Chief Justice's ability to make representations before matters proceeded beyond the stage of the JSC. Mr Fitzgerald recognised that the inclusion in the Tribunal's terms of reference of the phrase "and any other submissions and evidence which may be placed before it" supported what he referred to as "the uncertain, ever-growing approach to the Tribunal's remit favoured by the Government and the Signatories". However, the terms of reference fell to be read restrictively as subject to section 64, and, where there was a conflict, the Constitution was to be preferred. Mr Fitzgerald sought to draw a parallel with the direction of the Tribunal, communicated by the

letter from Clifford Chance dated 21 December 2007, that insofar as it might be suggested that there was any discrepancy between the terms of reference and section 64, the text in the Constitution was to be preferred. To the extent that the terms of reference supported the "roving brief" approach, they were unlawful and did not bind the Tribunal.

6. Mr Fitzgerald enlisted in support of this submission the Report of the Tribunal chaired by Lord Mustill, dated 14 December 2007, in regard to the Chief Justice of Trinidad and Tobago. It was concerned with a similar constitutional framework. At paragraph 6 of its Report the Tribunal stated:

"The task must therefore be to ascertain what...is 'the matter' which the Tribunal is authorised and required to investigate. Plainly, this cannot be a question completely at large whether the Chief Justice is fit to continue in office. The Tribunal cannot have been intended to trawl through his past behaviour, to see whether faults of any description could be brought to light which might qualify as 'misbehaviour'".

Mr Fitzgerald pointed out that the Mustill Tribunal determined that its remit was set by the terms of the reference letter, and, more importantly, the documentation which had led the Governor to determine that there was sufficient prima facie evidence justifying its appointment.

7. We were satisfied that the "fresh matters" fell within the scope of the words "and any other submissions and evidence which may be placed before it" in our terms of reference, and that there was no ground for restricting those words in the way contended for by Mr Fitzgerald. Section 64 does not, expressly or impliedly, prohibit a tribunal considering additional information which was not before the JSC or the Governor. We accordingly rejected the notion that section 64 contains some kind of "constitutional filter", as submitted by Mr Fitzgerald. Mr Otty, Counsel for the Tribunal, was, in our opinion, correct in submitting that "the matter" referred to in subsection (4)(b) is "the question of removing the Chief Justice...from office for inability... or for misbehaviour". It is on the facts of that "matter" that a tribunal is required to report. Section 64 nowhere envisages a set of allegations which is to define the scope of

inquiry. We found no merit in Mr Fitzgerald's argument that, if the terms of reference were not given a restricted meaning, there would be a failure to observe safeguards and a "constitutional filter". By means of the arrangements which were made for the making of representations and a draft Statement of Issues at an early stage in the proceedings the Chief Justice was given an opportunity to object to the inclusion of the "fresh matters". We should add that we did not consider that the observations of the Mustill Tribunal on which Mr Fitzgerald relied were in point in the present case. That Tribunal was essentially concerned with a single allegation of a very serious character, and it is not surprising that it chose to focus on that allegation, on which the question of possible removal turned. In the present case it is clear that from the outset there was seen to be a need to investigate a number of allegations relating to the conduct of the Chief Justice over a number of years. Furthermore, there is no question in the present case of the Tribunal pursuing the question of the fitness of the Chief Justice "at large". The procedure which has been followed by the Tribunal is designed to identify the issues at an early stage, and to ensure that fairness to the Chief Justice is observed throughout. We would add that in any event the Tribunal is not competent to entertain a challenge to the lawfulness of our terms of reference. Such a challenge could only be made in proceedings for judicial review, and it is now too late for that. For these reasons we rejected the first submission.

8. Mr Fitzgerald submitted in the second place that inclusion of the "fresh matters" would be extremely oppressive and unfair. He pointed out that many of them were almost a decade old. The Signatories could have raised many of them in 2000, when a complaint about the Chief Justice was before the then Governor, but evidently chose not to do so. Neither they nor the Government had raised any of them between April 2007 and the meeting of the JSC, or at the time when the reference to the Tribunal was made. They raised them when the Chief Justice had been suspended from office and had no access to his office or relevant papers. The "fresh matters" were plainly the result of a trawl through his career – the kind of result-oriented, historical fishing expedition expressly ruled out by Lord

- Mustill. The Chief Justice had been denied the opportunity to be heard in response to these matters before the reference was made to the Tribunal. He could not be expected to meet "a continually moving target". Mr Fitzgerald went on to express sharp criticism of the way in which the "fresh matters" had been raised. He questioned the impartiality of the Chief Secretary and the Attorney General who were members of the JSC. He suggested that these ministers had taken advantage of the licence which the JSC had given, by reason of the width of the terms of reference, to introduce allegations which could have been made and disposed of at an earlier stage. This was an abuse of process.
- 9. We found much of these submissions to be misguided. The proceedings of the Tribunal are inquisitorial in nature. Under its terms of reference this Tribunal has a public duty to inquire into matters which come to its attention from whatever source where such matters have a potential bearing on the question of the inability or misbehaviour of the Chief Justice. This is subject, of course, to its duty throughout to ensure that he is fairly treated. There are no parties in the ordinary sense in which that term is used in litigation or prosecution. Persons may be represented only in accordance with section 11 of the Commissions of Inquiry Act. There are no pleadings and no indictment. It follows that the dilatoriness or motive with which such a person is alleged to have behaved in raising matters before the Tribunal does not provide a ground for excluding them. What does matter is whether, because of the stage at which they are raised or for some other reason, considerations of fairness would require their exclusion. There is no question of the Chief Justice being expected to meet "a continually moving target". As we have already explained, the procedure followed by the Tribunal has been designed to avoid that.
- 10. The submissions of Mr Fitzgerald were primarily directed to the wholesale exclusion of the "fresh matters". In the alternative, he invited us to exclude certain of them, as set out in a Schedule, on a number of grounds including that of fairness. We were not persuaded, at least at the present stage in the proceedings, that it would be inconsistent with fair treatment of the Chief Justice for us to entertain the generality of the

- "fresh matters". Accordingly we rejected Mr Fitzgerald's primary submission. Since the Schedule invoked several different grounds for exclusion we will deal with it below.
- 11. The last part of the first application sought the exclusion of "matters that have already been determined". This referred to the outcome in October 2000 of the assessment by the then Governor of allegations made about the employment by the Chief Justice of Ms Danvers as a domestic employee. It had been alleged that the Chief Justice had failed to make punctual payment of PAYE and social security contributions relating to that employment. According to a news release dated 5 October 2000 (6/2551) the Governor announced:

"The information I have received shows that Ms Danvers' employment was registered with the Employment Training Board and that all outstanding PAYE payments and social security contributions have now been met. It is regrettable that matters were not regularised at an earlier stage but I have accepted that the Chief Justice did not deliberately seek to avoid his obligations. The Chief Justice has also assured me, in relation to another former employee of his, Ms Williams, that he has no outstanding liabilities. I have concluded, in view of the information and assurances which I have received, that it would not be appropriate to me to take any formal action in exercise of my constitutional powers."

12. Mr Fitzgerald submitted that we might derive assistance from Regulation 14 (1) (g) of The Judicial Discipline (Prescribed Procedures) Regulations 2006 of England and Wales under which a complaint about a judicial office holder fell to be dismissed if "it raises a matter which has already been dealt with ... and does not present any material new evidence". It would be an abuse of process and contrary to natural justice to allow this matter to be re-opened, particularly where, he said, it was not claimed that there was any new evidence relating to it. It had not been rekindled by the Chief Justice or indeed his wife. There was no evidence that in 2000 the Governor had been misled. The Governor had accepted the Chief Justice's explanation in mitigation and decided that investigation of the question of his removal was not warranted. That disposal of the matter

- gave rise to a legitimate expectation on the part of the Chief Justice that it would not be revisited.
- 13. We were not persuaded that there was sufficient to give rise to such an expectation. In the first place the Governor did not go so far as to say, or imply, that the conduct of the Chief Justice could not be considered along with other matters in connection with a later investigation of the question of his removal, let alone that his conduct in regard to the employment of Ms Danvers had been beyond reproach. This Tribunal is required to consider a considerable number of complaints about the conduct of the Chief Justice along with that relating to her employment. In the second place, as was pointed out by Mr White for the Signatories, the Tribunal had before it a complaint relating to the employment of Ms Williams, which arguably constituted material new evidence. As Mr Otty submitted, the regulations to which Mr Fitzgerald referred to are no more than a guide and do not preclude consideration of other matters along with an earlier complaint. In these circumstances we also rejected this submission by Mr Fitzgerald.

The second application

14. It was for a direction or ruling in the following terms:

"In relation to all or some of the allegations, there is no case to answer as a matter of law, and/or as a matter of fact"

15. Mr Fitzgerald submitted that the Tribunal should consider whether the allegations against the Chief Justice, even if true, could conceivably reach the constitutional threshold of "inability to discharge the functions of (his) office" or "misbehaviour", which he referred to as the "8 April test". "Inability" suggested, he said, a high threshold: a personal incapacity to perform as a judge at all due to some inherent or acquired failing, some personality defect or some similar cause, as in *Stewart v Secretary of State for Scotland* 1998 SC (HL) 81. It was plainly not enough for the removal of a judge that he had a difference of opinion with the executive, or that he made controversial statements in public. Nor was it enough that there was a lack of confidence in him among society or some practitioners. A

- perception that the Chief Justice was biased, as was alleged, could only affect his sitting in some cases. There was no evidence that he had made it impossible for himself to sit in a substantial proportion of the cases which came before the court.
- 16. It was important, he said, to keep the concept of "misbehaviour" distinct from that of "inability". "Misbehaviour" imported serious misconduct, typically of a criminal, or morally abhorrent or grossly unprofessional nature, which constituted an affront to the standing of the high office of judge. It originated from the common law crime of misbehaviour in a public office (Boulanger v the Queen [2006] 2 SCR 49). Any lower test would create an unwarranted threat to a judge's security of tenure. All the recent cases on "misbehaviour" were concerned with criminal conduct. He accepted, however, that "misbehaviour" did not always have to be criminal, as was observed by the Mustill Tribunal at paragraph 87 of its Report. In the case of the most senior judge in a given jurisdiction the meaning of "misbehaviour" must be different from its meaning in the case of a civil servant or office-holder as in *Lawrence v Attorney General of* Grenada [2007] 1 WLR 1474, which should be distinguished from the present case. Clark v Vanstone [2004] FCA 1105 and Vanstone v Clark [2005] FCAFC 189 supported the view that, in order to suffice for removal from office, misbehaviour, even if it was criminal, must also render the judge unfit to hold office. None of the allegations made by the Signatories and the Government could amount to "misbehaviour" on the part of the Chief Justice. There was no significant criminality or suggestion of wrongdoing, apart from some contrived allegation of misrepresentation. Some of the allegations were manifestly insufficient. The Signatories and the Government had wrongly gone on to the question of perception of unfitness without laying a basis for a finding of "misbehaviour".
- 17. Mr Fitzgerald accepted that "inability" might be inferred, as it was in *Stewart*, from a number of incidents. However, "misbehaviour" could not be constituted by the cumulative effect of a number of instances of conduct, where none of them was in itself "misbehaviour".

- 18. Some of the matters alleged were, he said, irrelevant. For example, the conduct of the Chief Justice's wife could not be a foundation of the case against him. It might justify an application for the Chief Justice to recuse himself in a particular case, but that could not justify a finding of "inability" or "misbehaviour".
- 19. Mr Fitzgerald also invited us to consider in addition or alternatively whether there was a realistic prospect of our reaching the conclusion that the allegations were proved beyond reasonable doubt, which was the standard of proof applied in the Mustill Report (see paragraph 82). In regard to all or some of the allegations there was no case to answer as a matter of fact. Many had too weak an evidential basis to go forward to full hearing; some were wholly unsupported by evidence. As regards the conduct of his wife, the Chief Justice denied any involvement with her activities. She had not acted as his proxy. There was no evidence to the contrary. Without any evidence that she had acted as proxy for him that could realistically reach the appropriate standard applied in the Mustill Report, this would leave only the extraordinary assertion that he was responsible for her actions.
- 20. It may be noted that in arguing that there was "no case to answer" an expression redolent of adversarial proceedings Mr Fitzgerald addressed not only the "fresh matters" introduced by the Signatories and the Government, but also the original allegations in the Supplementary Memorandum. His submissions, in effect, challenged the decision of the Governor, on the advice of the JSC, that the question of removing the Chief Justice ought to be investigated. This entailed that the Tribunal should not do what it was appointed to do, namely to "inquire into the matter and report on the facts of to the Governor...".
- 21. We did not, however, require to rely on these considerations in dealing with this application, since we were not persuaded that Mr Fitzgerald's approach to the application of the law to the present case was correct. It is reasonably clear that "inability" and "misbehaviour" have no fixed meaning. In particular the content of "misbehaviour" is to be determined by reference to the effect of the conduct on the capacity of the person to

- continue to hold the office, including what is perceived (*Clark v Vanstone*, Gray J, paragraph 85, approved by the Privy Council in *Lawrence v Att Gen of Grenada*). In *Vanstone* Weinberg J said at paragraph 164, with reference to the meaning of "misbehaviour" that "the conduct in question must have the potential to undermine the standing of the courts, or destroy public confidence in the judge's ability to continue to perform his or her functions".
- 22. Mr Otty was, in our opinion, correct when he submitted that the concepts of "inability" and "misbehaviour" were closely related, and that the interpretation of one may assist in the interpretation of the other (Vanstone, Black CJ at paragraph 20). We also accepted his submission that there was no reason in principle why "misbehaviour" justifying removal may not be based on the cumulative effect of a number of incidents, at least where some of them can themselves be properly characterised as "misbehaviour". That is consistent with the approach taken by the Privy Council in *Lawrence v Att Gen of Grenada*. While the facts of that case and the office with which it was concerned were plainly different from the present we found no good reason to distinguish the principles on which the judgments were based. Mr Fitzgerald was right to emphasise the high test that had to be satisfied before the removal of a judge was warranted. However, it also has to be borne in mind that the performance of a judicial function calls for virtually irreproachable conduct (Therrien v Canada (Minister of Justice) [2001] 2 RCS 3, paragraph 111).
- 23. In the present case it is clear that from the outset the task of the Tribunal was to consider the allegation of a systemic problem affecting the Chief Justice, arising out of a number of allegations relating to incidents over a period of years. We considered that in general it is not feasible at this stage, before the full hearing when the whole of the evidence is before us, to pick out which of those allegations, if any, are of no materiality or relevance. We might add that there was force in the observation of Mr White for the Signatories that paragraph 2.21 of the written submissions for the Chief Justice appeared to accept that at least some of the allegations did meet the test for "misbehaviour".

- 24. As regards the allegations about Mrs Schofield, we noted that in the Supplementary Memorandum the Signatories assert that the Chief Justice not disassociate himself from certain of her statements and actions, and in their representations they say that the only proper inference to draw is that her statements and actions were part of a coordinated joint approach adopted by her and the Chief Justice; and that in any event any fairminded observer would have believed that he shared and endorsed the views that she expressed. We were not persuaded that we should exclude any of the allegations relating to the conduct of Mrs Schofield, or any other allegations, on the ground that they are irrelevant. However, we considered that it would be going too far and too fast to affirm their relevance at this stage on the basis of what has been asserted about them. We reserved our judgment as to their relevance until we have the whole evidence at the full hearing. In the meantime we noted that the conduct of a relative may be relevant, for example, to a real danger or suspicion of bias (R v Bow Street Metropolitan Magistrate ex parte Pinochet (No 2) [2000] 1 AC 119, Lord Browne-Wilkinson at page 135).
- 25. We next refer again to the Schedule for the Chief Justice. Having regard to considerations of fairness and proportionality and the fact that they relate to the conduct of the Chief Justice when he was serving in another jurisdiction we were satisfied that it is appropriate to exclude the allegations to which paragraphs 13 and 14 of the Draft Statement of Issues relate. For the Signatories Mr White argued that the allegations about the behaviour of the Chief Justice in court to which paragraphs 15-19, 31-32, 46-48 and 63-64 relate should be regarded as falling into a pattern. While there are a number of similarities between these allegations, they are not likely to make any material difference when considered in conjunction with the other allegations. Accordingly we excluded these paragraphs. We did not take the same course with paragraph 49 as it raises different issues. We also agreed that in the interests of the efficiency of the inquiry it was right to exclude paragraph 45 which relates to the conduct of the Chief Justice at the land frontier with Spain. Lastly, paragraph 37(d) raises an issue as to whether the Chief Justice acted improperly when prosecuted under the Traffic Act in arguing that particular account should be taken of

his position as Chief Justice in the exercise of discretion to stay the proceedings. In the light of the evidence which has now been filed we accepted the suggestion of Mr Otty that that paragraph should be excluded.

- As regards Mr Fitzgerald's submission that the Tribunal should find that there was no case to answer as a matter of fact, we considered, subject to what we say in the next paragraph, that it was premature to reach any view as to sufficiency of evidence. As Mr Otty pointed out, evidential inquiries on behalf of the Tribunal have only recently begun, and the Government have indicated that they intend to offer further evidence. In the meantime we should not be understood as accepting the proposition, in reliance on the Mustill Report, that the allegations against the Chief Justice will require to be proved beyond reasonable doubt. In view of the nature of the conduct with which that tribunal was concerned, its use of that standard is understandable. In the present case the more flexible approach to which it referred at paragraph 81 may be more appropriate.
- 27. We excepted from the above the allegations concerning Hassans (Statement of Isssues paragraphs 41 and 53-54) since it was plain from the response of that firm that these allegations should be excluded.

Disposal

28. In the result, save in respect of the issues raised in paragraphs 13-14, 15-19, 31-32, 37 (d), 41, 45-48, 53-54 and 63-64, of the draft Statement of Issues, the applications of the Chief Justice were dismissed.

Third Schedule

The headings of the revised Statement of Issues

General issues (general criticisms of the conduct of the Chief Justice)

Factual issues in chronological order:

- 1. The comments made by the Chief Justice in respect of the funding of the judiciary at the opening of the legal year 1999-2000.
- 2. The comments made by the Chief Justice in November 1999 in respect of judicial appointments.
- 3. The failure of the Chief Justice in 1999 to make punctual PAYE and social security payments in respect of domestic staff.
- 4. Instructions from the Chief Justice relating to the attendance by the Chief Minister at the Registry of the Supreme Court in December 1999.
- 5. Instructions to the Registrar of the Supreme Court in respect of expenditure in May 2000.
- 6. The alleged statement by the Chief Justice in October 2000 to the then Chairman of the Bar, Mr Robert Vasquez, that he should not "rape the constitution".
- 7. The allegation of interception of the Chief Justice's telephone communications in 1999 and 2000.

- 8. The alleged nature of defence arguments advanced on behalf of the Chief Justice on his prosecution in 2001 for an offence under the Traffic Act.
- 9. The comments made by the Chief Justice in open court in February 2002 following, and relating to, his re-appointment as Chief Justice for 1 year.
- 10. The conduct of the Chief Justice in respect of certain proceedings in November 2004.
- 11. The conduct of the Chief Justice on the departure of the former Governor Sir Francis Richards in July 2006.
- 12. The involvement of the Chief Justice in debate over the 2006 Constitution.
- 13. The conduct of the Chief Justice in relation to the Judicial Service Bill 2007.
- 14. The nature and course of a hearing before the Chief Justice on the morning of 16 April 2007.
- 15. The nature and course of a hearing in chambers before the Chief Justice in the afternoon of 16 April 2007.
- 16. The nature of the Chief Justice's reply when asked for information in respect of the hearing in chambers on 16 April 2007 referred to above.
- 17. The conduct of Mrs Schofield's libel proceedings against Mr James Neish QC from April 2007 onwards.
- 18. The nature and course of a hearing on 24 April 2007.
- 19. The recusal applications of May 2007.
- 20. The complaint by Mrs Schofield against Mr Freddie Vasquez QC dated 21 August 2007.
- 21. The pursuit of judicial review proceedings in relation to the Judicial Service Act 2007 between August and December 2007.
- 22. The purported cancellation of the Ceremonial Opening of the Legal Year in October 2007

23. Alleged inappropriate criticism of Registry staff in October 2007.

Other undated instances of alleged misconduct on the part of the Chief Justice raised by the Deputy Registrar of the Supreme Court

Fourth Schedule

First Part

Rulings on applications by Mrs Schofield at the directions hearing on 8 April 2008

- 1. On behalf of Mrs Schofield Mr Stanley invited us to recognise that she should be accorded representation by counsel at the Inquiry, in accordance with section 11 of the Commissions of Inquiry Act, albeit, as he accepted, on a limited basis. He argued that she was "implicated... in the matter under inquiry", failing which we should exercise our discretion in favour of her being so represented. He also sought a recommendation that the legal expenses of her representation should be paid in accordance with section 13 of the Act. These applications were supported by written submissions by Mr Michael J Beloff QC.
- 2. Mr Stanley pointed to two respects in which her interests were at stake, as meriting her separate representation. First, the Tribunal was being asked to endorse criticisms of her conduct, perhaps regardless of whether it came to the view that her conduct should be attributed to the Chief Justice. Her personal and professional reputation might be affected by the conduct and reporting of the proceedings. Secondly, if there were a rule that the Chief Justice was obliged to control his wife or to distance himself publicly from anything that she said, she would be affected in a different way from him, being always in the position of seeing cold water poured over her views by a person who enjoyed a position of some responsibility in the community.

- 3. Turning to the issue of funding, Mr Stanley emphasised the inquisitorial nature of the inquiry. Whatever the outcome, the Tribunal would be required to determine the facts and give advice to the Governor. It was in the public interest that confidence in the judicial system was maintained. The representation of persons who had an interest in the subject matter of an inquiry served the public interest as well as private interests. He referred to passages in the Report of the Royal Commission on Tribunals of Inquiry under the Chairmanship of Lord Justice Salmon in 1966 (Cmnd 3121) in which it considered the circumstances in which a witness should be given opportunity to prepare his case, be assisted by legal advisers and have his legal expenses met out of public funds.
- 4. Mr Stanley submitted while Mrs Schofield should be entitled to separate representation, it needed to be limited and confined to those matters where she had an interest which was or might be different from that of the Chief Justice. Funding for representation might cover advice about the preparation of the witness statements, written submissions perhaps briefly amplified orally and possibly participation when witnesses who had something to say specifically about her were being cross-examined.
- 5. We consider in the first place whether there is a basis for Mrs Schofield being accorded representation by counsel. In his written submissions Mr Beloff maintained that allegations made by the Signatories raised important issues. These included, he said, whether, because of her status as a judge's wife, Mrs Schofield's freedom of expression was curtailed, and whether the wife of a serving judge was obliged to maintain any particular standards of behaviour in her capacity as such. We note that Mrs Schofield put forward a similar argument in support of her application for representation dated 15 February 2008. In the light of the discussion at the directions hearing on 8 April we are by no means satisfied that these are live issues, let alone issues which are appropriate for resolution by this Tribunal. As Mr Otty pointed out, whatever the outcome, the Tribunal process will not involve any determination of any of Mrs Schofield's own civil rights. Counsel for the Signatories and the Gibraltar Government accepted that there was no challenge to her right of

free speech. Mr Beloff also maintained that that there were issues as to whether a judge was obliged to disassociate himself from any behaviour (including statements) of his wife, and, more fundamentally, whether there was any basis for not treating wife and husband as separate persons with separate rights and obligations. However, as was pointed out by Mr Otty, these issues were addressed by the Chief

- 6. Justice in his Preliminary Response to the Supplementary Memorandum and Messrs Bindman's letter to the Governor dated 26 July 2007 enclosing it. If it be the case that the resolution of the first of these issues would have different implications for the Chief Justice and Mrs Schofield, this does not, in our view, support her being separately represented, since their respective interests would not be in conflict but in alignment. Both the Chief Justice and Mrs Schofield are at one in asserting that he was under no duty to disassociate himself from her activities.
- 7. When we reach our conclusions at the end of this inquiry we will be concerned with statements and actions of Mrs Schofield only in so far as they bear on the question of inability or misbehaviour of the Chief Justice for the purposes of section 64. Thus we will not be concerned with criticisms of Mrs Schofield in so far as they affect her only. Whether statements or actions of Mrs Schofield bear on that question will depend, in the first place, on whether the Chief Justice was associated, or would reasonably have been perceived to be associated, with them, arising from his having taken a joint approach with her or from having failed to disassociate himself, as is alleged. In the second place it will depend on whether, if he was so associated, the nature of such statements or actions point to his inability or misbehaviour. The Chief Justice has an interest to maintain that neither of these requirements will be satisfied. At present he clearly disputes the first of them. Mr Fitzgerald stated that the Chief Justice would certainly seek to justify what his wife had said in some cases. Where her expressed views coincided with the views expressed by him, he would obviously want to defend them. However, he might not agree with everything his wife had done, and there were some things in which he had no interest.

- 8. It thus appears that, to a significant but not clearly defined extent, it is likely that the Chief Justice will seek to defend what Mrs Schofield said and did in the exercise of her right of free speech. In these circumstances we are persuaded that it is appropriate that Mrs Schofield should be represented by counsel. We emphasise that it is on the limited basis that such representation is for countering criticisms of her statements and actions which are not covered by the representatives of the Chief Justice despite the claim by the Signatories and the Gibraltar Government that they affect him. We are content to accept that Mrs Schofield should be regarded as "implicated or concerned in the matter under inquiry" for the purpose of section 11.
- 9. As regards the separate question of funding, this plainly depends on the circumstances. We see no good reason why Mrs Schofield's witness statement should not be taken by her husband's solicitors, as was suggested by Mr Otty. There is no conflict of interest which would make this inappropriate, and it is not inconsistent with his position that he did not have to disassociate himself from what she said or did. We have already pointed out the limited basis for the representation of Mrs Schofield. We note the extent of her representation in the proceedings which was envisaged by Mr Stanley. For the Gibraltar Government Mr Pannick QC, who opposed funding, pointed out that Mrs Schofield would not be eligible for civil legal aid. In all the circumstances we are not persuaded that it is appropriate for us to recommend that the representation of Mrs Schofield should be publicly funded.
- 10. Mr Stanley indicated at the hearing that he left it to us to deal with the substantive applications by Mrs Schofield dated 25 March 2008. It is abundantly plain that we have no jurisdiction to make the rulings or directions sought in paragraphs 1, 4, 6, 7 8, and 9 of the application notice. We agree with Mr Otty that paragraphs 2 and 5 add nothing to similar applications by the Chief Justice and fall to be rejected for the same reasons. As regards paragraph 3, since the proceedings are inquisitorial in nature there is no question of the Signatories bearing a burden of proof. Statements of truth are not required at this stage. We are content to grant

Mrs Schofield permission, as sought in paragraph 11, to disclose the statements of Mr Neish, Mr Vasquez, Mr Mendez and the Attorney General in her libel action.

Second Part

The circumstances in which Mrs Schofield did not give evidence before the Tribunal

- 1. Mrs Schofield was from the outset listed as an intended witness. She was named on the provisional list of witnesses which was circulated on 3 March 2008 along with the draft Statement of Issues.
- 2. On the first day of the full hearing (7 July) we refused a written application by her for permission to cross-examine particular witnesses in regard to certain matters. We did so for the reasons advanced by Counsel to the Tribunal, namely that there was no reason to believe that counsel for the Chief Justice could not pursue such lines of questioning as were relevant to the issues before the Tribunal. Counsel to the Tribunal also pointed out that there were real practical difficulties with her proposal that Mr Charles Gomez should take responsibility for cross-examination, in that he would be appearing as a witness, and had made a number of criticisms of the Signatories in his witness statement. On day 6 (14 July) we dismissed as lacking any justification an application by Mrs Schofield for a direction that Counsel to the Tribunal and the Solicitors to the Tribunal should disclose information relating to their appointment and the negotiation of their fees, and the briefing of Counsel to the Tribunal; and that there should be disclosure as to an alleged briefing of Mr James Neish QC by the Secretary to the Tribunal.
- 3. On day 3 (9 July) Mr Charles Gomez, who had acted for Mrs Schofield in 2007, gave evidence about Mrs Schofield consulting him in regard to a possible challenge to the Judicial Service Bill (see paragraphs 5.14 5.15). In the course of his evidence the Chairman of the Tribunal asked him whether, for the purpose of its having an accurate and clear account of the

sequence of events, it would be possible for him to come with his file of papers on the following day when he was due to continue giving evidence. Mr Gomez said that it would. Thereafter the Solicitors to the Tribunal wrote to Mr Gomez asking him to confirm with Mrs Schofield that she had no objection to his providing such papers. Mrs Schofield responded to them that at that stage she did not consent, and that she wished to consider the papers herself and take legal advice, in particular on the question of legal professional privilege. She asked that the reasonable costs of her obtaining such advice be the subject of a recommendation by the Tribunal. On day 4 (10 July) Counsel to the Tribunal stated that this investigation could be important both for the purpose of achieving an accurate chronology and for consideration of the allegation that Mrs Schofield and the Chief Justice had acted in concert. As then proposed by Counsel to the Tribunal, we made a formal recommendation that the reasonable expenses of Mrs Schofield in obtaining legal advice on certain matters should be included in the expenses of the Tribunal and paid to her. These matters were the extent of professional privilege over any documentation in the possession of Mr Gomez relevant to certain specified issues; whether any such documentation could be redacted so as to allow disclosure without infringing such legal professional privilege; and whether such privilege should be waived in respect of any such documentation so as to allow production before the Tribunal.

4. On day 7 (15 July) we considered an application by Mrs Schofield, made by e-mail to the Solicitors to the Tribunal on that date, in which she claimed that her right to privilege had been breached during the examination of Mr Gomez by Counsel to the Tribunal. She sought a recommendation by us that she should be funded for the expense of obtaining the advice of counsel "as to the impact of the breach of privilege on my further participation in the Tribunal". She went on to state that "I shall be unable to appear on the 21st (being the day on which she was then due to give evidence) unless I have received independent advice on the impact of the breach of privilege on my participation in the Tribunal". Mrs Schofield also sought a recommendation of funding to cover the

expense of her solicitor travelling to London for the purpose of her obtaining advice from counsel about the alleged breach of privilege, on the ground that "I have no confidence in the privacy of phones and emails in Gibraltar". We ruled that, whether or not the advice fell strictly within our previous recommendation, we were satisfied that our recommendation should include the cost of taking such advice, but declined to recommend the expense of her solicitor travelling to London in that it was not justified by the reason stated for it.

- 5. By e-mail to the Solicitors to the Tribunal on 18 July Mrs Schofield sought confirmation from us that, for the purpose of obtaining such advice, fees of counsel estimated at £6,400 (16 hours at £400 per hour) and solicitors at £1,900 (10 hours at £190) would be met. We authorised the Solicitors to the Tribunal to respond, as they did by e-mail on 20 July, that we did not consider it appropriate to comment on these specific figures; that our recommendation had made it clear that her expenses must be reasonable; and we did not consider that we were in a position to go beyond that recommendation.
- 6. On day 9 (22 July) Counsel to the Tribunal stated that the Government had indicated through its counsel that it was content to authorise whatever fees we considered were reasonable for the purpose of Mrs Schofield obtaining legal advice in respect of the subject of the recommendation made by the Tribunal. Counsel for the Government recorded its concern as to the fee levels which were being sought, commenting that they seemed extremely large for the amount of work and given the nature of the issues. We reiterated that expenditure on obtaining legal advice about the matter of confidentiality required to be reasonable and it was not appropriate for us to state a specific level of fees for counsel and solicitors. That was for the reason it was not practical for us to express a view on figures or indeed a range of figures. However, we felt bound to say that, at least on the face of it, the number of hours for counsel and solicitors which had been stated struck us as excessive for the limited purpose of getting advice on the matters mentioned in our recommendation and in accordance with our rulings. We also

emphasised that, for the proper running of the proceedings before the Tribunal within the period which had been arranged for them, it was essential that Mrs Schofield attend as a witness to give evidence on Friday 25 July and no later, observing that we were not aware of any good reason why she should not so attend.

7. Mrs Schofield responded by e-mail to the Solicitors to Tribunal later on 22 July, stating:

"In the light of the ruling, I am unable to seek legal advice.

In the absence of funding for my participation, I am unable to proceed in any form before the Tribunal before receiving advice on different aspects of my participation and legal issues.

I shall therefore not be appearing before the Tribunal on Friday. I reserve my rights before the Privy Council".

- 8. On Day 10 (23 July), when this message was referred to before us, we observed that there was no question of Mrs Schofield being unable to obtain legal advice. It remained the position that we had recommended that she was entitled to the reasonable costs of taking legal advice in relation to privilege, including the question of an alleged breach and what the consequences might or might not be. We reiterated that it was not appropriate or practicable for us to sanction a particular number of hours or a particular figure or range of figures. We also observed that the absence of legal advice provided no justification for Mrs Schofield not appearing as a witness as we expected on Friday. We regarded the absence of legal advice as nothing more than a pretext for her non-appearance. We accepted the proposal of Counsel to the Tribunal that we should not summon Mrs Schofield under section 8 of the 1888 Act, applied by section 64 (5) of the Constitution, since:
 - (i) the Tribunal had made its expectation and requirements for the efficient running of the inquiry abundantly clear;
 - (ii) it was not clear that the issue of a summons would achieve any greater effect than that clear statement;

- (iii) it was not clear whether Mrs Schofield was in the United Kingdom or Gibraltar, and so difficulties arising out of service or its effectiveness might arise; and
- (iv) the expectation that Mrs Schofield should give evidence had been clear from the outset, and it would not be in the interests of either the Chief Justice or the people of Gibraltar that there be any delay to the further conduct of this inquiry.
- 9. On Day 11 (24 July) we were referred to a further e-mail from Mrs Schofield to the Solicitors to the Tribunal of that date, in which she requested that Tribunal to respond to the figures which had been quoted. We stated that we had nothing to add to what we had previously stated.
- 10. Following the conclusion of the hearing Mrs Schofield made an application on 1 August that a Foreign Affairs Committee Report on Overseas Territories should be placed before us, as bearing on the conversation between Lord Luce and the Chief Justice referred to in chapter 2 of this Report. Having considered the application we declined it as being of insufficient relevance.
- 11. Copies of the relevant written and e-mail communications between Mrs Schofield and the Solicitors to the Tribunal were included in the papers of the Tribunal.

Fifth Schedule

The standard of proof

Counsel for the Chief Justice submitted that we should apply the criminal standard of proof beyond reasonable doubt "given the gravity of the allegations and the magnitude of what is at stake". He referred in particular to allegations that the Chief Justice had committed criminal offences, and, in his words, had "colluded with his wife in a dishonourable and dishonest way", and "made complaints about the Government which he either knew or ought to have known were false". He submitted that, in the alternative, we should apply a standard, intermediate between the criminal and the civil, requiring "clear and convincing evidence", as in certain other jurisdictions. Finally, he said, we should at least apply the "flexible approach", described in *In Re H* [1996] AC 563 by Lord Nicholls of Birkenhead at page 586 where he said: "The more improbable the event, the stronger must be the evidence that it did occur".

It is clear that, apart from certain exceptions, the general standard of proof in civil proceedings is proof on the balance of probabilities (*In Re D* [2008] 1 WLR 1499, Lord Carswell at paragraph 23). As he pointed out, the exceptions include disciplinary proceedings brought against members of a profession, where proof beyond reasonable doubt is required (cf *Campbell v Hamlet*, Privy Council, 25 April 2005). However, we do not consider that the proceedings before us are analogous to disciplinary proceedings. Although they arise out of complaints about the conduct of the Chief Justice, he is not being 'prosecuted' by an accuser. The proceedings before us are inquisitorial in nature. The Chief Justice is entitled to be represented before us since his conduct is the subject of inquiry. The proceedings as a whole are concerned with the public interest - on the one hand,

to protect a judge against unfounded or illegitimate interference with his tenure of office, and on the other, to remove from office a judge who is unfit to hold office. That unfitness does not need to be based on misbehaviour. It may be based on his inability.

As Lord Hoffmann said in *In re B* [2008] 3 WLR 1 at paragraph 13: "...there is only one civil standard of proof and that is proof that the fact in issue more probably occurred than not". Accordingly we reject the idea of an intermediate standard. It is also inappropriate, in our view, to modify the civil standard because of the gravity of the allegations or the magnitude of the consequences for the Chief Justice (see *In Re D*, Lord Brown of Eaton-under-Heywood at paragraph 47; *In Re B*, Baroness Hale of Richmond at paragraph 70).

In these circumstances we apply the civil standard of proof on the balance of probabilities. In doing so we adopt the flexible approach that "the more improbable the event, the stronger must be the evidence that it did occur" (see *In Re D*). There are no matters in dispute which call for the application of the criminal standard of proof beyond reasonable doubt.